THE PSYCHOLOGICAL ASSESSMENT OF REFUGEES IN THE CONTEXT OF WORK-RELATED INTEGRATION

DAN ASFAR



The Psychological Assessment of Refugees in the Context of Work-Related Integration

Dan Asfar

VRIJE UNIVERSITEIT

THE PSYCHOLOGICAL ASSESSMENT OF REFUGEES IN THE CONTEXT OF WORK-RELATED INTEGRATION

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad Doctor aan de Vrije Universiteit Amsterdam, op gezag van de rector magnificus prof.dr. J.J.G. Geurts, in het openbaar te verdedigen ten overstaan van de promotiecommissie van de Faculteit der Gedrags- en Bewegingswetenschappen op donderdag 11 mei 2023 om 11.45 uur in een bijeenkomst van de universiteit, De Boelelaan 1105

Cover: Els Lorié Lay-out: Ilse Modder (www.ilsemodder.nl) Printing: Gildeprint Enschede (www.gildeprint.nl)

© Copyright, 2023, Dan Asfar, The Netherlands.

All rights reserved. No part of this thesis may be reproduced or transmitted in any form or by any means, electronically or mechanically by photocopying, recording, or otherwise, without written permission of the author. The copyright of previously published chapters of this thesis remains with the publisher or journal. door

Dan Asfar

geboren te Bagdad, Irak

promotor:	prof.dr. M.Ph. Born
copromotor:	prof.dr. J.K. Oostrom
promotiecommissie:	dr. B.G. Adams
	prof.dr. J.M. Dagevos
	prof.dr. F. Fleischmann
	prof.dr. E.M. Sijbrandij
	prof.dr. R.E. de Vries

Financial Support

Stichting NOA financially supported the present Ph.D. project.

TABLE OF CONTENTS

Chapter 1	General Introduction	9
Chapter 2	Psychological Clusters and Integration Outcomes of Recently Arrived Refugees in the Netherlands	19
Chapter 3	Psychological Individual Differences as Predictors of Refugees' Local Language Proficiency	43
Chapter 4	A Hindering and Facilitating Individual-Difference Factors Framework for Predicting Refugees' Workforce Participation	67
Chapter 5	The Normative Judgment Test of Honesty-Humility: An Implicit Instrument for the Organizational Context	111
Chapter 6	General Discussion	141
Appendices	Summary Dankwoord References Supplementary Material	156 160 164 192

CHAPTER 1

General Introduction

Dan Asfar Vrije Universiteit Amsterdam, the Netherlands NOA B.V., Amsterdam, the Netherlands

GENERAL INTRODUCTION

The share of foreign-born (non-European) population in Europe was 3.5% at the end of the 1990s, and this number has increased to 12.8% by 2021 (Eurostat, 2021a). Almost two-thirds (64.2%) of the foreign-born population in 2020 in Europe were non-European Union (EU) nationals (Eurostat, 2021a), which is primarily the consequence of forced replacement of large groups of refugees due to war and persecution in the MENA (Middle East and North Africa) region. Between 2014 and 2021, in total 6,204,095 asylum seekers requested humanitarian protection in EU countries, and 199,620 of them did so in the Netherlands (Eurostat, 2021a). The influx of these recently arrived refugees has increased the cultural diversity in the Netherlands but has also raised the question of how our society can effectively help refugees successfully integrate into their new host country. In this dissertation, I address this question from an individual-level psychological approach. I demonstrate that psychological individual characteristics can influence refugees' integration outcomes and must therefore be examined to design effective customized integration trajectories. Additionally, with the aim to improve the psychological assessment of refugees, I have developed and validated a novel implicit instrument with potential utility for refugee assessments.

REFUGEE INTEGRATION

In order to study the factors that influence the successful integration of refugees, it is necessary to first define and describe the term *integration*, and to subsequently clarify what constitutes successful. Yet, after several decades of research, there is still no clear consensus on how to define and assess migrants' integration (e.g., Castles et al., 2002; Saharso, 2019; Schinkel, 2018). Several authors have even argued that integration is a "chaotic concept" (e.g., Ager & Strang, 2008; Penninx, 2019; Schneider & Crul, 2010). This unclarity is reflected in studies on migrants' and refugees' integration that usually either avoid using or proposing a definition of the term, or that stress its complex nature (Harder et al., 2018; Puma et al., 2018). In research, until now two definitions have been most influential. In one definition, integration is described as a two-way process of becoming an accepted part of society (Bijl & Verweij, 2012; Council of Europe, 1997). In another definition, integration is referred to as the process through which immigrants come to a similar socioeconomic position as native-born people (OECD/EU, 2015; Schneider & Crul, 2010). However, the following issues have been raised about these definitions: they are ambiguous and hence multi-interpretable, do not specify the dimensions through which migrants should be compared with the native population, are normative and assume that immigrants should assimilate into the host country, and are influenced by policies (Abdou, 2019; Faist, 2000; Loch, 2014; Puma et al., 2018; Saharso, 2019; Schneider & Crul, 2010).

In an attempt to address the above-mentioned issues, Harder et al. (2018) recently proposed a framework that has received much scientific interest. The authors define integration as the

"degree to which immigrants have the knowledge and capacity to build a successful, fulfilling life in the host society" (Harder et al., 2018, p. 11484). In this framework, six dimensions of integration have been conceptualized and operationalized: The psychological dimension (one's sense of belonging in the host society), the economic dimension (employment outcomes and income), the political dimension (political and civic participation), the social dimension (social ties with natives), the linguistic dimension (the ability to use the local language), and the navigational dimension (managing basic needs in the host country).

This dissertation primarily concentrates on the economic and linguistic integration dimensions for three main reasons. First, this Ph.D. project has been conducted in the domain of industrial and organizational psychology, where the economic and the linguistic dimensions are highly relevant. Second, there is robust evidence for a positive link between local language proficiency (the linguistic dimension) and employment (the economic dimension), and social well-being (e.g., Beiser & Hou, 2001; Paul & Moser, 2009), stressing the importance of investigating these integration dimensions. Third, a higher workforce participation (economic integration) and its prerequisite local language proficiency (linguistic integration) are important during the current labor market shortages, as refugees can help to diminish the employment gaps (UNHCR, 2013).

THE PREDICTORS OF REFUGEES' INTEGRATION SUCCESS

To explain and predict differences in refugees' integration success, several factors have been investigated until now. These can be organized on three different levels: The institutional (or macro) level, the organizational (or meso) level, and the individual (or micro) level (e.g., Al Ariss et al., 2012; Lee et al., 2020). The institutional level consists of factors that pertain to international, national, and local policies, and to regulations and legislation. Most empirical work has focused on migrants' and refugees' employment, indicating integration success. Examples of factors and procedures that have shown to affect integration success include the length of the asylum process (e.g., Bakker et al., 2014; Hainmueller et al., 2016), housing allocation policies (e.g., Beckers & Borghans, 2011, Fasani et al., 2022), and the accreditation of foreign credentials (Krahn et al., 2000).¹

The organizational level pertains to factors that support refugees and employers toward societal integration. In the Netherlands, for example, governmental, non-governmental, and non-profit organizations such as the Dutch council for refugees (Vluchtelingenwerk) and the Foundation for Refugee Students (UAF) assist refugees in local language learning, education, and employment. These organizations provide resources, information, and a social network to refugees that help them with access to facilities, learning, education, and employment, and as such, help them towards integration (Garkisch et al., 2017; Lacroix

¹ For a recent review on integration policies in the Netherlands, readers are referred to Dagevos et al. (2020).

et al., 2015). Additionally, employers can exert influence on the economic integration of refugees through inclusive and fair personnel recruitment and selection, and training and development opportunities (Szkudlarek, 2019). In fact, several studies have shown that employers who had hired refugees were mostly satisfied with their performance and were likely to recruit refugees in the future and recommend refugee job seekers to other organizations (Lundborg & Skedinger, 2016; Szkudlarek, 2019).

The individual level - the last of the three levels - pertains to refugees' individual differences characteristics that are associated with integration outcomes. The most frequently studied individual difference predictors of linguistic and economic integration are sociodemographic and human capital variables such as age, gender, educational attainment, and work experience, and local language proficiency has been studied as a predictor of economic integration, see Chapter 2 to Chapter 4. However, this dissertation draws upon the literature on applied psychology to study refugees' integration outcomes from the individual-level perspective. A long tradition of research in this literature has identified and studied numerous psychological traits and categories of traits in which people differ from each other (e.g., Sackett et al., 2017). The most frequently studied traits are cognitive abilities (general intelligence or specific abilities; Carroll, 2005), personality (typically through five or six dimensions; Ashton et al., 2004; Costa & McCrae, 1992), vocational interests (such as through six typological domains; Holland, 1959), and values (e.g., the ten basic human values; Schwartz, 1992). These psychological individual differences, particularly cognitive ability and personality, have frequently been associated with performance and behavior at work and in education (e.g., Connelly & Ones, 2010; Poropat, 2009; Sackett et al., 2022).

Given the importance of psychological individual differences in predicting outcomes in these domains, it is surprising that only a handful of studies has investigated psychological traits among refugees as predictors of education- and work-related integration outcomes (for exceptions, see Hahn et al., 2019; Kosyakova & Laible, 2021; Thum, 2014). There are two probable explanations for the limited applied psychological work on this issue. One explanation is that the domain of applied personality is relatively small in comparison to other research domains, and hence, comparatively few scholars take an applied psychological approach to the literature on refugees' integration success. Another explanation is that psychological traits are relatively difficult and time-consuming to study. Whereas sociodemographic variables are available in governmental databases, the study of psychological traits requires special assessments. This dissertation aims to address this research gap by studying psychological individual differences, among which general mental ability (GMA; or intelligence) and personality traits, with respect to refugees' integration outcomes.

BEYOND TRADITIONAL (PERSONALITY) SELF-REPORT MEASURES

Many psychological individual differences, including personality traits, have typically been assessed using self-report measures, where participants indicate the extent to which they agree with items presented on a Likert-scale. Concentrating on the assessment of personality, decades of research, primarily in Western samples, revealed robust evidence for the validity of personality inventories such as the (five-dimensional) NEO PI-R (Costa & McCrae, 1992) and the (six-dimensional) HEXACO-PI-R (Ashton et al., 2004). There is some support for the cross-cultural validity of such inventories (cf. lon et al., 2017; Schmitt et al., 2007), although such studies have typically examined only few non-Western samples (Heine & Buchtel, 2009). In fact, there is empirical work that has raised several issues about the cross-cultural validity of traditional personality inventories (cf. Gurven et al., 2013; Laajaj et al., 2019; Smaldino et al., 2019).

One of these issues pertains to response styles, which are a systematic tendency to select particular scale options, regardless of the target construct (Paulhus, 1991). As such, response styles introduce error and therefore interfere with the measurement of the target construct (Podsakoff et al., 2012). Commonly studied response styles are the acquiescent response style (the tendency to agree regardless of item content), the extreme response style (the tendency to overuse the endpoints of a scale), the midpoint response style (the tendency to overuse the middle point of a scale), and socially desirable responding (the tendency to answer questions in a way that makes oneself look good). Abundant research has shown that response styles vary between cultural groups (cf. Danner et al., 2015). For example, higher rates of acquiescent and extreme response styles have been observed among Arab versus Jewish Israelis (Baron-Epel et al., 2010) and among first-generation versus secondgeneration migrants (Morren et al., 2012). Given the variations in response styles across cultures, some researchers have tried to "correct" for response style. Yet, despite such attempts, correcting for response styles has yielded no meaningful positive effects on the structural validity and the national-level criterion-related validity of personality inventories (e.g., Diamantopoulos et al., 2006; Dudley et al., 2005; Hoffmann et al., 2013).

Another issue with traditional personality measures in the cross-cultural context pertains to self-presentation, which concerns having unrealistic positive views of the self (Taylor & Brown, 1988). Self-presentation on self-report measures occurs cross-culturally, although differences have been observed in the traits in which cultures tend to self-enhance (Sedikides et al., 2003). For example, in two studies, Japanese people rated themselves as more positive than the midpoint on collectivistic attributes such as "cooperative" and "respectful", but did not self-enhance on individualistic attributes such as "self-reliant" and "unique", whereas the reverse pattern was found among American participants (Sedikides et al., 2003). As such, cultural self-presentation tendencies pose a threat to the validity of cross-cultural personality assessments.

Finally, cross-cultural personality assessments with traditional inventories may be problematic due to the reference-group effect (RGE; Heine et al., 2002). This effect refers to "the tendency for people to respond to subjective self-report items by comparing themselves with implicit standards from their culture" (Heine et al., 2008, p. 309). Several studies have shown that using different references based on age, gender, relative, and nationality affects mean-level personality trait test scores (Credé et al., 2010; De Vries et al., 2014; Wood et al., 2012). Thus, differences in mean-level test scores of different cultural or social groups might be masked and hence have suboptimal validity (Allik & Realo, 2016).

Altogether, the issues described above pose a threat to the validity of cross-cultural personality assessments. One potential viable solution for these issues is measuring personality through implicit instruments (Uhlmann et al., 2012). Implicit instruments assess individual attributes that people might not be willing to disclose or are unaware of (Moors et al., 2010), and thus might be less prone to biases of self-report measures (e.g., Vianello et al., 2013). One of the aims of this dissertation therefore was to develop and validate a new implicit instrument based on an existing paradigm (i.e., a method of assessing implicit cognition, such as the implicit association test [IAT; Greenwald et al., 1998] and the conditional reasoning test [CRT; James, 1998]), with the ultimate goal to make this instrument a valid and useful tool in the assessment of refugees. To this end, the existing implicit paradigms were reviewed based on the following criteria: The evidence for their (1) content- and (2) criterion-related validity, their (3) cross-cultural applicability, (4) applicant reactions, (5) susceptibility to faking, and (6) whether automatic scoring is possible.² Based on this review, an implicit instrument was developed based on the partially-structured attitude paradigm, where individuals judge trait levels of hypothetical persons who are described in vignettes (Vargas et al., 2004). As explained in Chapter 5, these judgments indicate something about the trait level of the respondents.

The implicit instrument which was developed aims to measure the personality trait Honesty-Humility, which has been defined as "the tendency to be fair and genuine in dealing with others, in the sense of cooperating with others even when one might exploit them without suffering retaliation" (Ashton & Lee, 2007, p. 156). Honesty-Humility is a relevant trait to measure as it is an important predictor of several employee behaviors and work outcomes (e.g., see Lee et al., 2019). Additionally, Honesty-Humility has been shown to be associated with the cultural adaptation of sojourners (Geeraert et al., 2019). The instrument was labeled as the Normative Judgment Test of Honesty-Humility (NJT-H). This dissertation provides some initial evidence for the validity and usefulness of the NJT-H, albeit among native Dutch samples.

THE CONTEXT OF THIS DISSERTATION

In an influential article, Henrich et al. (2010) introduced the acronym WEIRD, which stands for Western, Educated, Industrialized, Rich, and Democratic. The authors showed that the majority of the participants in psychological studies are characterized as being WEIRD, and argued that research should focus on human psychology and behavior that is more reflective of the global human diversity. However, this is particularly a challenging task with the so-called "hard-toreach" populations, such as people with disabilities, elderly people, and traveler families, but also migrants and refugees (e.g., Jones & Newburn, 2001). Nonetheless, as a consequence of the great influx of refugees in the Netherlands since 2014, in this dissertation, it was possible to collect data on thousands of refugees. The great influx of refugees induced the need for an assessment that portrays the profile of a refugee - their motivations, capabilities, personal barriers, and values - as this profile could be used to offer optimal and custom integration trajectories. Such an assessment could have several benefits: it is systematic as it screens refugees in a similar way, objective because it provides normative responses and test scores, and practical as it offers a lot of information about the refugee, which is less prone to cultural and linguistic communicative issues like in interviews. Such an assessment was developed by NOA BV, a consultancy agency that focuses on the development and the assessment of psychological instruments for work- and education-related purposes. The implementation of the refugee assessment started in June 2016. By June 2022, 88 Dutch municipalities and organizations have used the refugee assessment to assist refugees in their integration, which resulted in approximately 27.000 individual assessments – and thus a unique (psychological) dataset of refugees became available. However, by the time of launching this assessment, little was known about the psychometric properties (such as reliability and validity) of the assessment components among the understudied group of refugees. To provide such insights, the present Ph.D. project was initiated.

OUTLINE OF THIS DISSERTATION

This dissertation is organized as follows. The first and present chapter (Chapter 1) is the General Introduction. Next, this dissertation includes four empirical chapters (Chapter 2 to Chapter 5). These chapters all related to individual and psychological differences and the integration of refugees and work outcomes. Finally, the last chapter (Chapter 6) presents the General Discussion. Each chapter can be read independently from the other chapters.

Chapter 2 identifies four statistically derived psychological clusters of refugees based on their GMA, achievement motivation, and psychological distress. These psychological clusters are compared in terms of their level of work search intention and local (Dutch) language proficiency. This chapter explores to what extent the psychological clusters differentiate refugees in terms of these integration outcomes. Also, it examines whether the clusters differ in refugees' demographic characteristics, namely nationality, gender, age, and educational level.

² Given that the applied psychological literature on implicit instruments has only recently been started to emerge, there is limited robust evidence for each criteria mentioned above for each implicit method (for a review, see Uhlmann et al., 2012).

Chapter 3 focuses on the importance of psychological individual difference characteristics as predictors of refugees' local language proficiency. In a cross-sectional study among Syrian and Eritrean refugees in the Netherlands, this chapter examines the effects of several sociodemographic factors and psychological distress, next to the potential effects of GMA, work search intention, and the personality traits Conscientiousness and Openness to Experience on refugees' level of local language proficiency.

Chapter 4 concentrates on the individual-difference factors of refugees' workforce participation. This chapter introduces and investigates a novel integrative framework for predicting refugees' workforce participation, which organizes individual-difference factors into hindering (impeding demographics and health- and family-related challenges) and facilitating (acquired human and social capital and work-relevant traits) characteristics. This framework was examined among Syrian and Eritrean refugees in the Netherlands through a time-lagged data design.

Chapter 5 describes the validation of an implicit instrument – the Normative Judgment Test of Honesty-Humility (NJT-H) – to measure the personality trait Honesty-Humility in the organizational context. In this chapter, the NJT-H is examined with respect to employees' work performance and behaviors in two studies among employees living in the Netherlands who do not necessarily have a foreign cultural background.

Chapter 6 presents a general discussion of the four empirical chapters of this dissertation. I summarize and integrate the main findings which emanated from the four studies, describe their theoretical implications and contributions to the literature, offer practical implications to support flourishing refugees' integration trajectories, and shed light on future research opportunities.

Figure 1. A Visual Representation of the Content of this Dissertation



Chapter/title	Studies	Independent Variables	Dependent Variables	Methodology	Purpose	Status
2. Psychological Clusters and Integration Outcomes of Recently Arrived Refugees in the Netherlands	~	 Demographics: nationality, gender, age, educational level Psychological clusters based on GMA, achievement motivation, and psychological distress 	 Work search intention Local language proficiency 	Sample • Refugees from Syria, Eritrea, Iran, Irad, and Afghanistan (N = 4648) Design • Cross-sectional	To identify refugee clusters based on psychological traits and examine how these clusters relate to integration outcomes	Manuscript in preparation
3. Psychological Individual Differences as Predictors of Refugees' Local Language Proficiency	-	 Demographics: gender, age of arrival, educational level Psychological distress GMA Work search intention Conscientiousness Openness 	 Local language proficiency 	Sample • Refugees from Syria (<i>n</i> = 1054) and Eritrea (<i>n</i> = 500) Design • Cross-sectional	To investigate the influence of psychological traits on refugees' local language learning, and their incremental validity over demographics	Published: Asfar, D., Born, M. P., Oostrom, J. K., & Vugt, (2019). Psychological m. (2019). Psychological individual differences as predictors of refugees' local language proficiency. <i>European Journal of European Journal of</i> 7385–1400. <u>https://doi.</u> 03/10.1002/ejsp.2592.
4. A Hindering and Facilitating Individual- Difference Factors Framework for Predicting Refugees' Workforce Participation	-	 Impeding demographics: age sex, nationality Health- and family-related challenges: physical health problems, PTSD symptoms, spouse/children in country of origin Acquired human and social capital: PM educational level, PM work experience, local language proficiency, frequency of contact with natives Work-related traits: Agreeableness, Extraversion, Emotional Stability, GMA, Conscientiousness, work centrality 	 Workforce participation (employment and longest employment duration) Highest hourly wage 	Sample • Refugees from Syria (<i>n</i> = 1867) and Eritrea (<i>n</i> = 844) Design • Time-lagged (assessment (assessment data matched to governmental data with a time span of five years)	To introduce and test a new framework of the individual-difference predictors of refugees' workforce participation	Submitted for publication
5. The Normative Judgment Test of Honesty- Humility: An Implicit Instrument for the Organizational Context	0	 HEXACO personality traits NJT-H 	 Study 1: self-ratings of CWB and OCB Study 2: CWB (self- and supervisory ratings), OCB and task performance (supervisory ratings) 	Sample • Dutch and foreign participants living in the Netherlands (Study 1, N = 230; Study 2, N = 124) Design • Cross-sectional (matched data of employee-supervisor in	To develop and validate an integrity-related implicit instrument of the personality trait Honesty-Humility for the organizational context – with the ultitmate aim to make this instrument a valid tool for the assessment of refugees	Submitted for publication

CHAPTER

Psychological Clusters and Integration Outcomes of Recently Arrived Refugees in the Netherlands

> Dan Asfar^{1,2} Marise Ph. Born^{1,3,4} and Janneke K. Oostrom¹

 \mathcal{D}

¹ Vrije Universiteit Amsterdam, the Netherlands
 ² NOA B.V., Amsterdam, the Netherlands
 ³ Erasmus University Rotterdam, the Netherlands
 ⁴ North-West University, South Africa

ABSTRACT

When examining refugees, researchers and practitioners tend to focus on the demographic subgroups to which they belong. The present study investigated whether it is useful and beneficial to examine refugees as part of psychological clusters (groups of individuals that share similar profiles) through means of cluster analysis. The study was conducted among recently arrived refugees from Syria (n = 2881), Eritrea (n = 1183), Iran (n = 270), Iraq (n = 188), and Afghanistan (n = 126) living in the Netherlands. Drawing upon the ability-motivation-opportunity (AMO-)framework, we assessed refugees' general mental ability (GMA), achievement motivation, and psychological distress to identify and explore psychological clusters. The cluster analysis revealed four clusters, namely a bright, incapable, distressed, and undistinctive-ordinary cluster. We related these clusters and refugee demographics (i.e., nationality, gender, age, and educational level) to refugees' level of work search intention and local language proficiency. The results showed that the psychological clusters differ particularly in the latter outcome, and did so more than the demographic subgroups. These results provide some evidence for the utility of using psychological clusters in the domain of refugee integration.

KEYWORDS

2

refugees, integration, psychological clusters, demographic subgroups

PSYCHOLOGICAL CLUSTERS AND INTEGRATION OUTCOMES OF RECENTLY ARRIVED REFUGEES IN THE NETHERLANDS

War, persecution, and unsettled social conditions in many world regions, in particular Africa and the Middle East, have resulted in large groups of refugees coming to Europe from 2014 onwards. In the Netherlands alone, a total of 230,686 asylum applications were submitted from 2014 to 2020, of which 61.8% were granted (Eurostat, 2022a). The influx of these refugees has underscored the importance of understanding the factors that influence refugees' integration success into their new host country. Examples of such factors include governmental integration policies (e.g., Jongen et al., 2020; Koopmans, 2010), support from NGOs and non-profit organizations (Garkisch et al., 2017; McIntosh & Cockburn-Wootten, 2019), and the length of the asylum process in a country (e.g., Bakker et al., 2014). Besides such situational factors, demographic variables, such as refugees' gender and educational attainment, may also affect integration success (e.g., Bloch, 2002).

In the Dutch context, several studies and reports have been published on the link between demographic variables and integration success, both among the earlier waves (i.e., the 1990s; Dourleiin & Dagevos, 2011; Jennissen, 2011; Maliepaard et al., 2017), and more recent waves of refugees (i.e., 2014 and later; CBS, 2022; Dagevos et al., 2018; Jongen et al., 2020; Odé & Dagevos, 2017). The insights from these studies and reports are important for at least two reasons. First, differences between demographic subgroups help to identify the barriers that certain subgroups face, and as such, serve to identify solutions. Examples of refugee subgroup barriers include the large cultural distance between Eritrean refugees' home country and their European country of residence (Sterckx et al., 2018), and women's difficulties to enter the labor market in Western societies due to the traditional gender roles in their home countries according to which men are the income providers (Baranik, 2020; Bear & Glick, 2017; Razenberg et al., 2018). Second, a better understanding of demographic variables in relation to integration success is useful in practice, as these will help practitioners to examine which refugees will and which refugees will not need individual assistance (Miltenburg & Dagevos, 2021). For example, the finding that pre-migration educational attainment is associated with faster local language acquisition suggests that refugees with lower levels of pre-migration education might generally benefit more from intensive language courses (e.g., Asfar et al., 2019; Van Tubergen, 2010). Such insight can be taken into consideration by constructing custom integration trajectories.

Despite these benefits, a focus on demographic variables could result in an irrelevant salience and stereotyping of demographic subgroups, which have been associated with several negative effects. The salience of negative stereotypes can trigger the activation of stereotype threat, which is the psychological discomfort that is thought to arise when

individuals are confronted with a negative stereotype about their own group in a situation in which the negative stereotype could be confirmed (Steele & Aronson, 1995; Steele et al., 2002). Stereotype threat has for instance been found to negatively influence cognitive ability test scores, including those of migrants (Appel et al., 2015). Negative stereotypes of refugee subgroups can also result in discrimination, prejudice, and unethical practices. For example, several meta-analyses on fictitious job applications have shown that ethnic minorities, particularly female ethnic minorities, with a comparable resume and application letter, face substantial hiring discrimination compared to natives (Lippens et al., 2022; Thijssen et al., 2022; Zschirnt & Ruedin, 2016). Another recent example of unethical practices related to demographic data is the "Dutch childcare benefits scandal" (in Dutch: "Toeslagenaffaire"), where between 2013 and 2019, the authorities used data on citizens' second nationality to identify individuals who made fraudulent benefit claims for day-care (Huisman, 2020). Consequently, 26,000 parents were wrongly accused of fraud and were required to pay back the allowances they had received, which drove many families into severe financial hardship.

These issues motivated us to investigate an alternative way of describing and examining refugees. One potentially effective way is to array refugees into clusters based on relevant psychological traits. As we explain below, psychological traits related to one's ability, motivation, and opportunity are important and could hence be used for this purpose. Numerous studies have shown that arraying individuals into groups (clusters) based on psychological traits can help identify individuals at "risk" and serve to apply customized interventions. For example, Cerda-Navarro et al. (2019) used engagement factors to identify the profiles of students that were at risk of dropouts, which could, in turn, be used to prevent such occurrences. Similarly, in the present research, we use measures of general mental ability (GMA), achievement motivation, and psychological distress to identify psychological clusters and to investigate whether and to what extent these clusters are associated with integration outcomes, namely work search intention and local language proficiency. As such, we can establish whether it is useful to examine refugees by their psychological profile. Additionally, it allows comparing the usefulness of psychological clusters in comparison to demographic subgroups. In this study, these are nationality (Syria, Eritrea, Iran, Iraq, and Afghanistan), gender (men and women), age (three age groups), and educational level (five levels). Taken together, in this study, we address three research questions:

Research Question 1 (RQ1): What clusters among refugees can we identify based on psychological traits?

Research Question 2 (RQ2): To what extent do psychological clusters differ in the studied integration outcomes?

Research Question 3 (RQ3): How large are the differences in integration outcomes of the psychological clusters compared to demographic subgroups?

INTEGRATION OUTCOMES

There has been a long debate on the definition and the assessment of migrants' integration (e.g., Castles et al., 2002; Saharso, 2019; Schinkel, 2018). Recently, however, Harder et al. (2018) proposed a framework for refugees' integration that has received considerable scientific interest. In this framework, the authors define integration as the "degree to which immigrants have the knowledge and capacity to build a successful, fulfilling life in the host society" (Harder et al., 2018, p. 11484). This framework differentiates six dimensions of integration: The psychological dimension (the sense of belonging in the host society), the economic dimension (employment outcomes and income), the political dimension (political and civic participation), the social dimension (social ties with natives), the linguistic dimension (the ability to use the local language), and the navigational dimension (managing basic needs in the host country). In the present research, we focus on the economic and linguistic dimensions.

Most research on the economic dimension of integration has focused on refugees' quick and successful employment, as employment is regarded as a key aspect of integration (e.g., Ager & Strang, 2008; Esser, 2004), and as many refugees end up unemployed (Fasani et al., 2022). To illustrate, in the Netherlands, only 19% of the refugees who had received a residence permit in 2014 found a job within three years (CBS, 2021). This high unemployment rate among refugees is worrisome because it is associated with negative consequences, both on the macroeconomic (community) level (such as high economic costs for governments; Aiyar et al., 2016) and the individual level (such as poor mental health; McKee-Ryan et al., 2005; Paul & Moser, 2009). In the present study, we do not examine actual employment, but instead, focus on an important predictor of actual employment, namely, work search intention (Van Hooft & Noordzij, 2009; Van Hooft et al., 2021). Work search intention has been defined as the extent to which individuals are willing to try hard to perform necessary job-search behaviors, or the effort they are planning to exert engaging in job-search behavior (Van Hooft et al., 2004). Work search intention is a relevant alternative to actual employment in the present study, as the majority of participants are still unemployed by the time of the research due to their relatively short length of stay in the Netherlands. Thus, a measure of their employment intentions can, in comparison to actual employment, provide more power to detect significant differences between the psychological clusters and demographic subgroups.

Furthermore, we examine local language proficiency as a measure of linguistic integration of refugees. Local language proficiency has been related to refugees' higher well-being

(Beiser & Hou, 2001), lower psychological distress (Asfar et al., 2019), faster employment (Aldashev et al., 2009; Bloch, 2002; also, see Chapter 4), higher income (Chiswick & Miller, 2007; Dustmann, 1994; Shields & Price, 2002; also, see Chapter 4), and lower odds of marginalization (Bloch, 2002; Schellekens, 2001). Many previous studies have assessed refugees' local language proficiency through self-reports (e.g., Carliner, 2000; Chiswick & Miller, 2002; Van Tubergen, 2010) or interviews (e.g., Van Tubergen & Kalmijn, 2005; Van Tubergen & Wierenga, 2011). However, in the present research, we measured local language proficiency through an objective and validated standardized instrument (Edele et al., 2015).

THE PRESENT RESEARCH

The present research aimed to (1) identify clusters of refugees based on psychological traits, (2) examine the extent to which de clusters differ in integration outcomes, and (3) compare these differences with differences between demographic subgroups. With regard to demographic characteristics, a recent report of the Dutch Central Bureau of Statistics (CBS) has revealed that, three years after receiving a residence permit in the Netherlands, refugees from Afghanistan have the highest workforce participation (37%), whilst the other four nationalities show lower but about similar workforce participation rates, varying from 15% (Eritrea) to 20% (Iraq) (CBS, 2022). This CBS report also revealed that refugees from Iran have the highest local language proficiency, followed by refugees from Syria, Afghanistan, Iraq, and Eritrea. Furthermore, empirical work among recently arrived refugees in the Netherlands has shown that gender is associated with workforce participation (higher among male refugees; see Chapter 4), but not associated with local language proficiency (Asfar et al., 2019). These two studies have also shown that age is negatively associated with workforce participation and local language proficiency, and that educational level is not significantly associated with workforce participation but is positively associated with local language proficiency.

In this study work search intention and local language proficiency between the demographic subgroups were examined, but also examine psychological clusters. Through the statistical technique of cluster analysis, refugees were classified into clusters: groups of individuals who are similar in terms of the variables entered into the analysis. To our knowledge, only a few studies have employed cluster analysis among refugee samples (e.g., to identify rehousing trajectories, participation profiles, and socio-cultural starting positions; Damen et al., 2022; De Hoon et al., 2021; Miltenburg & Dagevos, 2021). Here, we apply cluster analysis using refugees' psychological individual differences by drawing upon the ability-motivation-opportunity (AMO) framework (Appelbaum et al., 2000). The AMO framework posits that employee performance – and in turn organizational performance – depends on three fundamental individual factors: ability (which represents a person's ability to perform the task), motivation (which represents an individual's willingness to act), and opportunity

(which refers to the extent to which circumstances facilitate or hinder behavior) (Appelbaum et al., 2000). The three factors in the AMO framework are essential features of several influential psychological theories, such as expectancy theory (Vroom, 1964) and social exchange theory (Blau, 1964).

The extensive literature from applied psychology was used to select and to justify the constructs for the three AMO factors (Sackett et al., 2017). In applied psychology, ability has generally been studied as general mental ability (GMA, or intelligence), which is "the ability to understand complex ideas, to adapt effectively to the environment, to learn from experience, to engage in various forms of reasoning, to overcome obstacles by taking thought" (Neisser et al., 1996, p. 77). As several authors have contended, the measurement of skills is intimately connected to the measurement of GMA (Kuncel & Beatty, 2013; Lubinski & Dawis, 1992). Across industries and jobs, GMA affects one's capability to learn and perform well on tasks (e.g., Nye et al., 2022; Schmidt & Hunter, 2004). Moreover, in research, GMA is usually positioned as the standard against which other predictors are compared (Ree et al., 1994). Among refugees, GMA has also been positively associated with workforce participation (see Chapter 4) and local language proficiency (Asfar et al., 2019; Edele et al., 2015). Altogether, GMA is an important psychological trait that pertains to individual differences in ability, and is therefore used in the present research.

Achievement motivation has been considered the trait that best reflects individual differences in motivation (e.g., Kanfer & Heggestad, 1997). Individuals high in achievement motivation set high standards and aspire to accomplish difficult tasks (Jackson, 1974). Correspondingly, achievement motivation is a positive predictor of academic performance (Spence et al., 1989) and job performance (Dudley et al., 2005). To the best of our knowledge, there are no studies on the relationship between achievement motivation and the integration outcomes examined in this research. Altogether, in this research, we use a measure of achievement motivation to assess AMO's motivation factor.

Finally, opportunity in the AMO framework has been more broadly interpreted and used in the literature. Opportunity has typically been studied as a situational variable (e.g., job autonomy, involvement in policymaking; Boselie, 2010). However, individual-level variables can also facilitate or hinder behavior, and could therefore be considered under the umbrella of the opportunity factor in the AMO framework. For example, factors that negatively influence (re)employment include disabilities and illness (Wanberg et al., 2002), larger households (Van Hooft et al., 2005), and psychological distress (McKee-Ryan et al., 2005; Paul & Moser, 2009). However, psychological distress is highly relevant among refugee samples, as poor psychological health is an important issue among such groups given the traumatic experiences they had to go through and may still encounter (e.g., Blackmore et al., 2020; Bogic et al., 2015; Charlson et al., 2019; Fazel et al., 2005). Therefore, the present

26 | CHAPTER 2

arriving in the Netherlands and completing the assessment was 22.99 months (SD = 15.75). The descriptive statistics for the study variables are reported in Table 2 to Table 5 for the

2

MATERIALS

All measures had been developed in the Dutch language, but a certified translation agency had translated the measures into different languages, including English, Modern Standard Arabic (for Arabic refugees, including refugees from Syria and Iraq), Tigrinya (for Eritrean refugees), and Farsi (for refugees from Iran and Afghanistan). The refugees completed the assessment in their native language (92.3%), in English (7.2%), or in Dutch (0.6%), according to their own preference.

the candidates had attained secondary education, 17.1% of the candidates had attained

secondary vocational education, 18.1% of the candidates had attained higher education (cf. bachelor's degree), and 10.0% of the candidates had attained university master's level. In

total, 64.7% of the refugees reported having pre-migration work experience. The median

year in which refugees completed the assessment was 2017. The average duration between

nationalities, genders, age groups, and educational levels, respectively.

General Mental Ability

Two non-verbal subtests of the multicultural capacities test were used to assess GMA, which had been developed to measure fluid intelligence and to minimize any potential bias that might be imposed by cultural background (e.g., due to language or knowledge domain differences) (MCT-M; Van Breemen et al., 2018; Van den Berg, 2001).³ The subtest Components consists of 30 items with a time limit of 9 minutes, in which participants must select two out of six spatial parts that can make up one displayed figure. The subtest Exclusion consists of 30 items with a time limit of 7 minutes, in which participants must select the figure that does not match the other four presented figures. Research has supported the cross-cultural validity and applicability of the MCT-M (Asfar et al., 2019) and provided evidence for its predictive validity in the domains of social functioning and academic achievement among native Dutch candidates and several Dutch migrant groups (Van den Berg, 2001). The correlation between the two MCT-M subtests in the present research was r = .57, p < .001. Earlier research revealed support for the aggregation of the two subtests into one total score of GMA (Asfar et al., 2019). The raw total score on the two subtests is M= 33.00 (SD = 11.00). In a representative sample of the Dutch population, a raw score of M =44.44 (SD = 9.19) corresponds to an IQ score of 100. Correspondingly, GMA was computed as the average of the standardized scores on both subtests. To provide more insights into the reliability of the MCT-M subtests, a sample of 35 refugees completed Components twice and 29 refugees completed Exclusion twice with average time intervals between the

In sum, in the present research, GMA, achievement motivation, and psychological distress were used to identify psychological clusters among refugees. Since cluster analysis is an exploratory method, no predictions were formalized about the actual clusters or their relations with dependent variables (Maxwell et al., 2002). The psychological clusters are compared for the two integration outcomes to identify which psychological cluster(s) secure more or less favorable integration success. Furthermore, the magnitude of the integration outcome differences between the refugee demographic subgroups and psychological clusters are also examined.

METHOD

2

PARTICIPANTS AND PROCEDURE

In total, 34 municipalities and organizations that are involved with refugees (such as employment organizations and educational institutions) invited refugees within a few months to a few years after arriving in the Netherlands to complete a digital psychological assessment to support them in their integration into the Dutch society. Participating in the assessment was voluntary, and although exact numbers about the response rate are unavailable, assessment administrators estimated it to be at least 95%. The assessment was offered by a consultancy agency that focuses on the development and the assessment of psychological instruments for work- and education-related purposes. The assessments were usually administered in distraction-free rooms in the municipality or the organization, and took about two hours to complete. During the assessments, one or more staff members were present, and no communication with other candidates was allowed. The assessment consists of sociodemographic questions and several psychological measures (for details, see Chapter 3 and Chapter 4 of this dissertation). The criteria for participating in the assessment were being at least 18 years old, having a residence permit, and being literate. Participating in the assessment was not monetarily compensated.

The data were collected from June 2016 to November 2019. In this period, 7609 refugees completed the assessment. However, we only used the data of the five largest nationalities in our dataset (N = 4648), which were Syria (n = 2881; 62.0%), Eritrea (n = 1183; 25.5%), Iran (n = 270; 5.8%), Iraq (n = 188; 4.0%) and Afghanistan (n = 126; 2.7%). Most of the refugees were male (3141; 67.6%). The mean age was 30.50 years (SD = 9.44; Mdn = 28 years). The educational levels of the refugees in the country of origin had been converted to the Dutch education system using the Nuffic classification (Nuffic, 2022). In terms of this system, 15.2% of the candidates had no or basic education (though they were literate), 39.6% of

³ To see the test environment and some sample items of the two GMA subscales, readers can visit the webpage https://www.noa-online.net/practicequestions/mct-m. After entering one's email address, the login instructions are sent.

assessments of respectively 6.5 and 6.2 months. These data revealed test-retest reliabilities of r = .88 for Components and r = .93 for Exclusion.

Achievement Motivation

To assess achievement motivation, the Competentietest was used (NOA, 2016). The scale includes ten items, but two items were excluded because of low item-rest correlations (r = .13 and r = .03). The items were rated on a 5-point Likert scale, ranging from 1 = (Almost) never to 5 = (Almost) always. An example item is "I make high demands on myself in my work". The raw mean score of the scale is M = 3.73 (SD = 0.66). In a representative sample of the Dutch population, a mean raw score of M = 4.08 (SD = 0.51) has been observed. In the current sample, the alpha coefficient of this scale equaled .77 (respectively .77, .72, .82, .83, and .81 for refugees from Syria, Eritrea, Iran, Iraq, and Afghanistan).

Psychological Distress

To assess psychological distress, the 10-item Kessler Psychological Distress Scale was used (K10; Kessler et al., 2002). In the K10, participants are asked to indicate on a 5-point Likert scale how often they experienced or felt something during the last 30 days. An example item is "About how often did you feel hopeless?". The raw total score on the K10 is M = 17.81 (SD = 7.74; range = 10-50). Several studies have shown that the K10 is a reliable and valid instrument to assess anxiety and depressive disorders in clinical and in non-clinical populations (e.g., Cairney et al., 2007; Furukawa et al., 2003; Kessler et al., 2003). Furthermore, the K10 also has shown good psychometric qualities among non-Western samples, supporting the cross-cultural validity of the instrument (Fassaert et al., 2009). In the current sample, the alpha coefficient of this scale equaled .89 (respectively .90, .83, .95, .93, and .94 for refugees from Syria, Eritrea, Iran, Iraq, and Afghanistan).

Work Search Intention

Work search intention was assessed with a 10-item subscale of a work motivation questionnaire (AWV; NOA, 2005). Research has revealed evidence for the convergent validity of this instrument with other work motivation instruments (Dusseldorp et al., 2018). An example item is "How much time do you spend on searching for vacancies on the internet?". Participants indicated how much time they spend on such activities using a 5-point Likert scale, ranging from 1 = Not at all to 5 = Very frequently. The raw mean score of the scale is M = 2.50 (SD = 0.94). The alpha coefficient of the work search intention scale in the current sample equaled .91 (respectively .91, .89, .93, .92, and .94 for refugees from Syria, Eritrea, Iran, Iraq, and Afghanistan).

Local Language Proficiency

A test developed by the psychological consultancy agency that gathered the data assessed local (Dutch) language proficiency (NOA, 2006). The Dutch language proficiency

test contains two short stories written in Dutch with 80 incomplete words which need to be completed to make meaningful words within the context. Scores on this test can range from 0 (no word fragment completed correctly) to 80 (all word fragments completed correctly). The raw total score on the test is M = 22.41 (SD = 18.60). Several studies have provided support for the construct- and criterion-related validity of the Dutch language proficiency test among migrant and refugee groups (Asfar et al., 2019; CINOP et al., 2002; NOA, 2006). The alpha coefficient of the Dutch language proficiency test in the current sample equaled .97 (respectively .98, .96, .98, .97, and .97 for refugees from Syria, Eritrea, Iran, Iraq, and Afghanistan). Furthermore, a random sample of 119 refugees who had conducted the assessment, completed the Dutch language proficiency test twice with an average time interval between the assessments of 12 weeks and revealed a test-retest reliability of r =.84 (NOA, 2021).

STATISTICAL ANALYSIS

To identify psychological clusters, a cluster analysis was conducted with GMA, achievement motivation, and psychological distress according to a two-step approach. First, a hierarchical cluster analysis was performed to determine the number of clusters. Second, we conducted a non-hierarchical cluster analysis to assign participants to the clusters (Gore, 2000; Hair et al., 2010; Tan et al., 2016). The hierarchical cluster analysis was conducted using Ward's method based on squared Euclidian distances. Ward's method is one of the most robust methods in hierarchical cluster analyses (Gore, 2000), and minimizes the within-cluster differences (Brusco et al., 2017). The non-hierarchical cluster analysis was done with the *k*-means technique to assign each participant to one of the identified clusters.

One-way analysis of variance (ANOVA) was used to test whether demographic subgroups and psychological clusters differ in the studied variables. By means of the conservative posthoc Tukey's Honest Significant Difference test, it was examined whether specific subgroups or clusters show significant differences in the integration outcomes. The magnitude of the differences in the study variables between refugee subgroups and psychological clusters were reported in partial eta squared values (η_p^2). Values of $\eta_p^2 < .035$, $.035 \le \eta_p^2 \le .10$, and $\eta_p^2 > .10$ are respectively considered small, medium, and large effects (Cohen, 1988). For ease of interpretability, the scores on the psychological traits (GMA, achievement motivation, and psychological distress) and the integration outcomes (work search intention and local language proficiency) were transformed to *z*-scores. The level of the mean scores and mean score differences were interpreted according to the Cohen's *d* guidelines, where values of *d* = 0.20, *d* = 0.50, and *d* = 0.80 are respectively considered small, medium, and large (Cohen, 1988). Based on these guidelines, we described mean differences of *z* < 0.10 as average, 0.10 $\le z \le 0.35$ as slightly above (or below) average, 0.35 $\le z \le 0.65$ as moderately high (or low), and *z* > 0.65 as very high (or low). The analyses were conducted in SPSS (version 26).

RESULTS

PRELIMINARY ANALYSES

The correlations between the study variables are reported in Table 1. Work search intention significantly correlated with gender (r = -.14, p < .001; males higher), age (r = -.03, p = .042), pre-migration educational level (r = .11, p < .001), pre-migration work experience (r = .14, p < .001), achievement motivation (r = .21, p < .001), and psychological distress (r = -.06, p < .001). Furthermore, local language proficiency significantly correlated with age (r = -.08, p < .001), pre-migration educational level (r = .26, p < .001), GMA (r = .32, p < .001), achievement motivation (r = .06, p < .001), and psychological distress (r = -.08, p < .001), pre-migration educational level (r = .26, p < .001), GMA (r = .32, p < .001), achievement motivation (r = .06, p < .001), and psychological distress (r = -.10, p < .001). Finally, work search intention and local language proficiency were positively correlated (r = .15, p < .001). Most significant correlations were small to moderate.

Table 1. Bivariate Correlations between the Study Variables

	1	2	3	4	5	6	7	8
1. Gender (Male = 0, Female = 1)	-							
2. Age	.03	-						
3. PM educational level	00	.20**	-					
4. PM work experience (yes)	31**	.34**	.21**	-				
5. GMA	02	.04*	.33**	.10**	-			
6. Achievement motivation	12**	.07**	.16**	.16**	.15**	-		
7. Psychological distress	.05**	.08**	07**	.04*	07**	14**	-	
8. Work search intention	14**	03*	.11**	.14**	.01	.21**	06**	-
9. Local language proficiency	.02	08**	.26**	.01	.32**	.06**	10**	.15**

Note. PM = pre-migration, GMA = general mental ability. * p < .05, ** p < .01.

DEMOGRAPHIC SUBGROUPS

The descriptive statistics of and differences in study variables for nationality, gender, age group, and educational level are reported in Table 2 to Table 5. The nationalities revealed medium differences in work search intention, F(4, 4643) = 81.34, p < .001, $\eta_p^2 = .065$, where refugees from Eritrea (M = 0.38, SD = 0.97) and Iran (M = 0.36, SD = 1.02) scored higher than refugees from the other nationalities (respectively *p*-values \leq .001 and *p*-values \leq .011), but did not differ from each other (p = .997). Furthermore, the nationalities revealed small differences in local language proficiency, F(4, 4643) = 13.65, p < .001, $\eta_p^2 = .012$, where Iran scored significantly higher than the other nationalities (*p*-values \leq .001), except Afghanistan (p = .059).

Men and women revealed small differences in work search intention, F(1, 4646) = 89.37, p < .001, $\eta_p^2 = .019$, where men (M = 0.10, SD = 0.96) scored higher than women (M = -0.20, SD = 1.04). Furthermore, men (M = -0.01, SD = 0.98) and women (M = 0.03, SD = 1.04) showed no significant differences in local language proficiency, F(1, 4646) = 1.87, p = .171, $\eta_n^2 = .000$.

The age groups revealed small differences in work search intention, F(2, 4645) = 7.15, p = .001, $\eta_p^2 = .003$, where the age groups of 18-30 (M = 0.03, SD = 1.00) and 30-45 (M = 0.00, SD = 1.01) years old scored significantly higher than the 45-65 (M = -0.16, SD = 0.96) years old age group (respectively p < .001 and p = .006), but the former two age groups did not differ significantly from each other (p = .674). Similarly, the age groups revealed small differences in local language proficiency, F(2, 4645) = 17.71, p < .001, $\eta_p^2 = .008$, where the age groups of 18-30 (M = 0.03, SD = 1.02) and 30-45 (M = 0.02, SD = 0.98) years old scored significantly higher than the 45-65 (M = -0.26, SD = 0.87) years old age group (p-values < .001), but the former two age groups did not differ significantly from each other (p = .911).

The educational levels revealed small differences in work search intention, F(4, 4643) = 40.02, p < .001, $\eta_p^2 = .033$, where refugees with secondary education and master's level scored respectively significantly lower and higher than the other educational levels (respectively all *p*-values < .001 and all *p*-values < .011). Furthermore, the educational levels revealed medium differences in local language proficiency, F(4, 4643) = 83.08, p < .001, $\eta_p^2 = .067$, where refugees with no or basic education scored significantly lower than refugees from the other educational levels (all *p*-values < .001).

	Syria (<i>n</i> = 2881)	Eritrea (<i>n</i> = 1183)	lran (<i>n</i> = 270)	Iraq (<i>n</i> = 188)	Afghanistan (<i>n</i> = 126)	F (p)	η,²
Gender (% male; ref = 67.6%)	66.9%	70.9%	61.9%	63.3%	69.8%	3.13 (.014)	.003
Age (ref = 30.50; 9.44)	31.87 (10.10)	26.67 (6.32)	32.91 (8.34)	31.91 (10.39)	27.90 (8.44)	76.14 (< .001)	.062
PM educational level (0-4; ref = 1.68; 1.22)	1.93 (1.11)	0.99 (1.21)	2.31 (1.06)	1.78 (1.14)	1.05 (1.20)	174.27 (< .001)	.131
No or basic education (ref = 15.2%)	76 (2.6%)	551 (46.6%)	8 (3.0%)	17 (9.0%)	54 (42.9%)		
Secondary education (ref = 39.6%)	1315 (45.6%)	343 (29.0%)	61 (22.6%)	83 (44.1%)	39 (31.0%)		
Secondary vocational education (ref = 17.1%)	577 (20.0%)	104 (8.8%)	76 (28.1%)	24 (12.8%)	12 (9.5%)		
Higher education (ref = 18.1%)	564 (19.6%)	120 (10.1%)	89 (33.0%)	52 (27.7%)	15 (11.9%)		
Master's level (ref = 10.0%)	349 (12.1%)	65 (5.5%)	36 (13.3%)	12 (6.4%)	6 (4.8%)		
PM work experience (% yes; ref = 64.7%)	68.1%	53.9%	80.4%	61.2%	57.9%	27.72 (< .001)	.023
GMA	0.24 (0.92)	-0.73 (0.82)	0.64 (0.86)	-0.01 (1.00)	-0.23 (1.07)	266.58 (< .001)	.195
Achievement motivation	0.10 (0.98)	-0.19 (0.94)	0.01 (1.04)	-0.18 (1.20)	-0.44 (1.12)	24.20 (< .001)	.022
Psychological distress	-0.05 (0.98)	-0.10 (0.85)	0.30 (1.23)	0.36 (1.23)	0.77 (1.29)	36.12 (< .001)	.031
Work search intention	-0.18 (0.96) ^{2.3}	0.38 (0.97) ^{1,4,5}	0.36 (1.02) ^{1.4.5}	-0.22 (0.99) ^{2.3}	0.02 (1.02) ^{2.3}	81.34 (< .001)	.065
Local language proficiency	0.01 (1.05) ^{2,3}	-0.10 (0.83) ^{1,3}	0.38 (1.13) ^{1,2,4}	-0.06 (0.97) ³	0.10 (0.98)	13.65 (< .001)	.012
Note. PM = pre-migration, GMA = general menti	al ability. For work s	earch intention and	local language profi	ciency, the numbers	in superscript indi	cate that the subgrou	up scored

left. the from ÷ pue , D £ Znd, of the g ഹ pue 4 m Ń _ he Φ ıp(s), subgre e (an)oth from ently f diffe

Table 3. Descriptive Statistics of and Differences in Study Variables between the Genders

	Men (<i>n</i> = 3141)	Women (<i>n</i> = 1507)	F (p)	η_p^2
Age (ref = 30.50; 9.44)	30.32 (9.42)	30.89 (9.48)	3.74 (.053)	.001
PM educational level (0-4; ref = 1.68; 1.22) No or basic education (ref = 15.2%) Secondary education (ref = 39.6%) Secondary vocational education (ref = 17.1%) Higher professional education (ref = 18.1%) Master's level (ref = 10.0%)	1.69 (1.23) 479 (15.2%) 1235 (39.3%) 556 (17.7%) 536 (17.1%) 335 (10.7%)	1.67 (1.21) 227 (15.1%) 606 (40.2%) 237 (15.7%) 304 (20.2%) 133 (8.8%)	0.08 (.775)	.000
PM work experience (% yes; ref = 64.7%)	74.8%	43.5%	488.13 (< .001)	.095
GMA	0.02 (1.00)	-0.03 (0.99)	2.43 (.119)	.001
Achievement motivation	0.08 (0.96)	-0.17 (1.07)	57.62 (< .001)	.013
Psychological distress	-0.03 (0.99)	0.06 (1.01)	9.27 (.002)	.002
Work search intention	0.10 (0.96)	-0.20 (1.04)	89.37 (< .001)	.019
Local language proficiency	-0.01 (0.98)	0.03 (1.04)	1.87 (.171)	.000

Note. PM = pre-migration, GMA = general mental ability.

Table 4. Descriptive Statistics of and Differences in Study Variables between Age Groups

	18-30 y/o (n = 2682)	30-45 y/o (<i>n</i> = 1515)	45-65 y/o (n = 452)	F (p)	η_p^2
Gender (% male; ref = 67.6%)	68.6%	66.9%	63.9%	2.17 (.114)	.001
Age (ref = 30.50; 9.44)	23.95 (3.36)	36.00 (4.13)	50.94 (4.38)		
PM educational level (0-4; ref = 1.68; 1.22) No or basic education (ref = 15.2%) Secondary education (ref = 39.6%) Secondary vocational education (ref = 17.1%) Higher professional education (ref = 18.1%) Master's level (ref = 10.0%)	1.50 (1.19) 535 (20.0%) 1108 (41.3%) 405 (15.1%) 431 (16.1%) 202 (7.5%)	1.89 (1.21) 153 (10.1%) 558 (36.8%) 294 (19.4%) 320 (21.1%) 190 (12.5%)	2.07 (1.19) 18 (4.0%) 175 (38.7%) 94 (20.8%) 89 (19.7%) 76 (16.8%)	77.53 (< .001)	.032
PM work experience (% yes; ref = 64.7%)	51.9%	80.3%	80.4%	257.38 (< .001)	.100
GMA	-0.05 (103)	0.09 (0.97)	-0.02 (0.87)	8.06 (< .001)	.004
Achievement motivation	-0.05 (1.01)	0.06 (0.97)	0.08 (1.01)	7.35 (.001)	.003
Psychological distress	-0.05 (0.95)	0.05 (1.06)	0.15 (1.08)	10.41 (< .001)	.005
Work search intention	0.03 (1.00) ³	0.00 (1.01) ³	-0.16 (0.96) ^{1,2}	7.15 (.001)	.003
Local language proficiency	0.03 (1.02) ³	0.02 (0.98) ³	-0.26 (0.87) ^{1,2}	17.71 (< .001)	.008

Note. PM = pre-migration, GMA = general mental ability. For work search intention and local language proficiency, the numbers in superscript indicate that the subgroup scored differently from (an)other subgroup(s), where the 1, 2, and 3 respectively represent the groups of the 2^{nd} , 3^{rd} , and 4^{th} column from the left.

2

Table 2. Descriptive Statistics of and Differences in Study Variables between the Nationalities

able 5. Descriptive Statistics of c	nd Differences in Stu	dy Variables bet	ween the Educati	onal Levels			
	No or basic (<i>n</i> = 1867)	Secondary (<i>n</i> = 844)	Secondary vocational	Higher professional	Master's level (<i>n</i> = 719)	F (p)	η ²
			(766L = U)	(RL / = N)			
Gondor (% malo: rof = 67 6%)	G7 9%	G7 1%	70.1%	62 2%	71 6%	7 26 10771	

			(n = 1992)	(n = 719)			
3ender (% male; ref = 67.6%)	67.8%	67.1%	70.1%	63.8%	71.6%	2.86 (.022)	.002
Age (ref = 30.50; 9.44)	26.77 (7.57)	29.96 (9.74)	31.60 (9.61)	31.92 (8.63)	33.87 (9.88)	53.73 (< .001)	.044
PM work experience (% yes; ref = 64.7%)	48.7%	57.7%	75.7%	77.3%	74.8%	63.02 (< .001)	.052
GMA	-0.79 (0.91)	-0.07 (0.94)	0.21 (0.93)	0.33 (0.92)	0.41 (0.91)	176.46 (< .001)	.138
Achievement motivation	-0.32 (1.05)	-0.07 (1.01)	0.11 (0.97)	0.14 (0.95)	0.25 (0.90)	32.73 (< .001)	.029
² sychological distress	0.08 (0.98)	0.04 (1.02)	0.01 (1.04)	-0.03 (0.98)	-0.22 (0.87)	7.23 (< .001)	.006
Work search intention	0.14 (1.00) ^{2,3,5}	-0.20 (0.97) ^{1,3,4,5}	-0.00 (0.98) ^{1,2,4,5}	0.14 (1.00) ^{2.3,5}	0.33 (1.01) ^{1,2,3,4}	40.02 (< .001)	.033
-ocal language proficiency	-0.37 (0.79) ^{2,3,4,5}	-0.15 (0.96) ^{1,3,4,5}	0.09 (1.00) ^{1,2,4,5}	0.32 (1.02) ^{1.2,3}	0.42 (1.07) ^{1,2,3}	83.08 (< .001)	.067
ote. PM = pre-migration. GMA = general r	mental ability. For y	work search intentic	on and local language	te proficiency. the n	iumbers in superscrip	ot indicate that the sub	aroup scored

dno the hat cate script he .⊆ oers ÷. nun bi the prot 2nd language ips of the 2 9 σ ЧO 다 work 5 resp For and ability. | 2, 3, 4, a the 1, 2 eral gei, GMA e-migration, ((an)other sul pre Ρ Note.

PSYCHOLOGICAL CLUSTERS

The results from the hierarchical cluster analysis with GMA, achievement motivation, and psychological distress indicated four clusters to provide the most robust solution. Next, the *k*-means cluster analysis arrayed the participants into the four clusters, yielding n = 1377 in Cluster 1, n = 876 in Cluster 2, n = 576 in Cluster 3, and n = 1357 in Cluster 4 (Table 6). The clusters differed significantly in GMA (*F*[3, 4182] = 5290.85, p < .001, $\eta_p^2 = .791$), achievement motivation (*F*[3, 4182] = 35.29, p < .001, $\eta_p^2 = .025$), and psychological distress (*F*[3, 4182] = 5290.85, p < .001, $\eta_p^2 = .791$), see Figure 1.

In comparison to the other refugee clusters, Cluster 1 is characterized by a very high score on GMA (M = 1.04, SD = 0.41), a slightly above-average score on achievement motivation (M = 0.16, SD = 0.90), and a moderately low score on psychological distress (M = -0.37, SD = 0.55). Cluster 2 is characterized by a very low score on GMA (M = -1.41, SD = 0.45), a slightly below-average score on achievement motivation (M = -0.22, SD = 0.90), and an average score on psychological distress (M = 0.06, SD = 0.87). Cluster 3 is characterized by a slightly above-average score on GMA (M = -0.20, SD = 0.69), a slightly below-average score on achievement motivation (M = -0.18, SD = 1.07), and a very high score on psychological distress (M = -0.18, SD = 1.07), and a very high score on psychological distress (M = -0.17, SD = 0.36), an average score on achievement motivation (M = -0.17, SD = 0.36), an average score on achievement motivation (M = -0.17, SD = 0.36), an average score on achievement motivation (M = -0.45, SD = 0.48). Based on these scores, we labeled Cluster 1 to Cluster 4 respectively as the *bright, incapable, distressed,* and *undistinctive-ordinary* cluster (RQ1).⁴

The bright cluster includes relatively many refugees from Syria (78.8%) and Iran (10.2%), and relatively few refugees from Eritrea (5.4%). Additionally, this cluster is characterized by a high mean level of pre-migration educational level (M = 2.17, SD = 1.16). The incapable cluster includes relatively many refugees from Eritrea (51.7%), and relatively few refugees from Syria (38.6%) and Iran (1.7%). Additionally, this cluster is characterized by a low mean level of pre-migration educational level (M = 1.15, SD = 1.13). The distressed cluster includes relatively many refugees from Iran (10.6%), Iraq (7.8%), and Afghanistan (6.1%), and relatively few refugees (M = 32.14, SD = 9.86). Finally, the undistinctive-ordinary cluster includes refugees that are distributed in proportion to the whole sample in terms of their demographic characteristics (e.g., a similar share of the nationalities and genders).

The clusters revealed small differences in work search intention F(3, 4182) = 4.24, p = .005, $\eta_n^2 = .003$, where the bright and undistinctive-ordinary cluster scored higher

⁴ These labels depict primarily one of the three psychological traits that characterize the cluster. However, one should recognize that these labels are simplified representations of the clusters, and that the clusters also denote different mean levels of the other psychological traits.

than the distressed cluster (respectively p = .005 and p < .009). However, the incapable cluster did not differ significantly from the other clusters (all *p*-values > .179). Furthermore, the clusters revealed medium differences in local language proficiency F(3, 4182) = 123.47, ρ < .001, η_{p}^{2} = .081, where the bright and incapable clusters scored respectively higher and lower than the other clusters (all p-values < .001) (RQ2). Overall, the effect sizes revealed that the demographic subgroups of nationality (η_{p}^{2} = .065), gender (η_{p}^{2} = .019), and educational level ($\eta_n^2 = .033$) differ more in work search intention than the psychological clusters (η_n^2 = .003), but the psychological clusters differ more in local language proficiency (η_n^2 = .081) than the demographic subgroups ($\eta_p^2 = .000$ to $\eta_p^2 = .067$) (RQ3).

Figure 1. The Psychological Profiles of the Four Refugee Clusters



Table 6. Descriptive Statistics of and Differences in Study Variables between the Clusters

	Bright (<i>n</i> = 1377)	Incapable (<i>n</i> = 876)	Distressed (<i>n</i> = 576)	Undistinctive-ordinary (<i>n</i> = 1357)	F (p)	η_{p}^{2}
Nationality Syria (ref = 62.0%) E-itro- (rof - 75.5%)	1085 (78.8%) 74 /6 1%)	338 (38.6%) 462761 7%)	351 (60.9%) 04744 6%)	852 (62.8%) 206 /20 2%)	42.03 (< .001)	.029
Enitrea (ref = 20.3%) Iran (ref = 5.8%)	141 (10.2%) 141 (10.2%)	15 (1.7%)	61 (10.6%) 61 (10.6%)	46 (3.4%)		
Iraq (ref = 4.0%)	47 (3.4%)	41 (4.7%)	45 (7.8%)	37 (2.7%)		
Afghanistan (ref = 2.7%)	30 (2.2%)	29 (3.3%)	35 (6.1%)	26 (1.9%)		
Gender (% male; ref = 67.6%)	69.1%	69.2%	65.1%	67.7%	1.18 (.318)	.001
Age (ref = 30.50; 9.44)	30.26 (8.67)	29.05 (9.10)	32.14 (9.86)	30.83 (10.02)	13.66 (< .001)	.010
PM educational level (0-4; ref = 1.68; 1.22)	2.17 (1.16)	1.15 (1.13)	1.67 (1.13)	1.65 (1.20)	142.91 (< .001)	.093
No or basic education (ref = 15.2%)	46 (3.3%)	289 (33.0%)	74 (12.8%)	189 (13.9%)		
Secondary education (ref = 39.6%)	465 (33.8%)	345 (39.4%)	236 (41.0%)	593 (43.7%)		
Secondary vocational education (ref = 17.1%)	290 (21.1%)	104 (11.9%)	114 (19.8%)	216 (15.9%)		
Higher professional education (ref = 18.1%)	356 (25.9%)	100 (11.4%)	112 (19.4%)	223 (16.4%)		
Master's level (ref = 10.0%)	220 (16.0%)	38 (4.3%)	40 (6.9%)	136 (10.0%)		
PM work experience (% yes; ref = 64.7%)	69.5%	58.3%	68.8%	63.3%	11.55 (< .001)	.008
GMA	1.04 (0.41)	-1.41 (0.45)	0.20 (0.69)	-0.17 (0.36)	5290.85 (< .001)	.791
Achievement motivation	0.16 (0.90)	-0.22 (1.10)	-0.18 (1.07)	0.05 (0.96)	35.29 (< .001)	.025
Psychological distress	-0.37 (0.55)	0.06 (0.87)	1.80 (0.85)	-0.45 (0.48)	1801.71 (< .001)	.564
Work search intention	0.04 (0.97) ³	-0.01 (1.02)	-0.12 (0.96) ^{1,4}	0.04 (1.02) ³	4.24 (.005)	.003
Local language proficiency	0.40 (1.11) ^{2,3,4}	-0.35 (0.79) ^{1,3,4}	-0.11 (0.96) ^{1,2}	-0.07 (0.92) ^{1,2}	123.47 (< .001)	.081
Vote. PM = pre-migration, GMA = general mental a differently from (an)other subgroup(s), where the 1, 2, and educational level represent their percentages w	ability. For work searcl , 3, and 4 respectively i vithin the whole sampl	n intention and local l epresent the groups o e.	language proficiency, of the 2 nd , 3 rd , 4 th , and 5	the numbers in superscript indi th column from the left. The "ref="	cate that the subgrou of the categories of n	p scored ationality

DISCUSSION

The present study investigated profiles of recently arrived refugees living in the Netherlands by applying cluster analysis, which is a statistical technique that arrays refugees into clusters: groups of individuals who are similar in terms of the variables entered into the analysis. Based on the ability-motivation-opportunity (AMO) framework (Appelbaum et al., 2000), GMA (ability), achievement motivation (motivation), and psychological distress (opportunity) were assessed to identify psychological clusters among refugees. We studied the integration outcomes work search intention (as a proxy of employment) and local language proficiency. Employment and local language proficiency are considered important aspects of integration (e.g., Ager & Strang, 2008; Esser, 2004), and have been linked to refugees' well-being (Beiser & Hou, 2001; McKee-Ryan et al., 2005; Paul & Moser, 2009). We aimed to identify clusters based on three psychological traits (RQ1), examine to what extent these clusters are related to the studied integration outcomes (RQ2), and compare the magnitude of these effects with those of demographic subgroups based on nationality, gender, age, and educational levels (RQ3).

With respect to RQ1, we found four clusters based on GMA, achievement motivation, and psychological distress, which we labeled the bright, incapable, distressed, and undistinctiveordinary cluster (Figure 1). In their respective order, these clusters are mainly characterized by a high score on GMA, a low score on GMA, a high score on psychological distress, and a low score on psychological distress, although they also differ in the other psychological traits. The differences in achievement motivation between the clusters were smaller than the differences in GMA and psychological distress. With respect to RQ2 and RQ3, small differences in work search intentions were found between the clusters: the bright and undistinctive-ordinary cluster scored somewhat higher on work search intention than the distressed cluster. However, the difference in work search intention between the refugee subgroups based on nationality, gender, and educational level was larger than those based on the psychological clusters. With regard to local language proficiency, mediumsized differences were found between the clusters, where the bright and incapable clusters scored respectively higher and lower than the other clusters. These cluster-based differences were larger than the differences between the subgroups in nationality, gender, age, or educational level.

CONTRIBUTIONS AND IMPLICATIONS

The present study offers several contributions and theoretical and practical implications to the literature on refugee profiles (Damen et al., 2022). We contributed to this literature by showing that the application of cluster analysis of psychological traits among refugees provides meaningful distinguishable groups (RQ1). This finding signifies that refugees can be investigated and identified by their psychological profiles instead of or in addition to

the demographic subgroup they belong to. Currently, differences between refugees are, both in research and in practice, commonly examined based on demographic subgroups, but investigating refugees and their differences based on their psychological profiles provides a novel – and, as we contend, an advantageous – approach. To clarify, given that psychological profiles are not directly visible or identifiable such as demographic characteristics, we anticipate that examining refugees by psychological profiles prevents or restrains issues such as stereotype threat (Appel et al., 2015; Steele & Aronson, 1995) or unjust practices of discrimination, such as those in job applications (Lippens et al., 2022; Thijssen et al., 2022; Zschirnt & Ruedin, 2016). Additionally, in contrast to demographic analyses, psychological profiles inherently reveal the personal strengths and weaknesses of an individual, and hence could help navigate one's self-development opportunities. Thus, practitioners could benefit from identifying refugees based on their psychological profiles by offering more effective support. For example, the incapable cluster can benefit from education and additional language training to compensate for their low levels of GMA, and the distressed cluster can benefit from therapeutic health support to alleviate their high levels of psychological distress.

Furthermore, we contributed to the literature on refugees' integration (Lee et al., 2020) by examining how the psychological clusters differ in work search intention and local language proficiency (RQ2). Previous research on psychological risk factors of poor outcomes using cluster analysis has focused on other (non-refugee) samples, such as students and their chance of dropouts (e.g., Cerda-Navarro et al., 2019) The significance of examining and using psychological clusters is ultimately revealed by their ability to differentiate refugees in terms of their integration outcomes. Indeed, we provided evidence that the psychological clusters differ in work search intention, and more strongly in local language proficiency. This is likely because local language proficiency is relatively strongly related to GMA. The current research findings indicate that counselors need to pay special attention to the incapable and distressed clusters, as refugees of these clusters are more likely to have lower levels of work search intention and/or local language proficiency.

Finally, we contribute to the literature on refugee individual differences and integration success (Huijnk et al., 2015) by comparing psychological clusters with demographic subgroups on the integration outcomes (RQ3). For work search intention, we found that the differences between the demographic subgroups based on nationality, gender, and educational level were larger than those of the psychological clusters. However, for local language proficiency, the differences between the psychological clusters were larger than the differences between the demographic subgroups of nationality, gender, age, or educational level. These findings indicate that one can predict more accurately the mastering of the local language of refugees by examining to which psychological cluster they belong compared to using their demographic subgroup. The practical implication is

that, when designing integration trajectories for language learning, counselors can benefit from assessing the psychological traits to allocate refugees into a cluster. More specifically, the incapable cluster requires intensive language courses, whereas the bright cluster requires less language training.

LIMITATIONS AND FUTURE DIRECTIONS

2

The present research has several limitations and corresponding future research opportunities. First, this study focused on economic and linguistic integration (through measures of work search intention and local language proficiency). Although these outcomes are important, we did not address other relevant integration dimensions (e.g., the psychological, political, social, and navigational dimensions; Harder et al., 2018). To provide more evidence for the utility of identifying and using psychological clusters, it is essential to show that they differentiate refugees substantially on different sorts of integration outcomes – and preferably do so more than the demographic subgroups.

Second, work search intention formed a proxy for refugees' economic integration (as a substitute for actual employment, because of the relatively short length of stay of the present sample). Yet, the results show that the means on this scale of refugee nationality subgroups do not correspond well to their employment statistics in the Netherlands. Specifically, the employment rates (and between the parentheses the means of work search intention) of the 2016 refugee cohort in the Netherlands after 36 months of receiving a residence permit are 25% for Syrian refugees (M = -0.18), 28% for Eritrean refugees (M = 0.38), 35% for Iranian refugees (M = 0.36), 31% for Iragi refugees (M = -0.06), and 38% for Afghan refugees (M= 0.10).⁵ Thus, the employment rates and the work search intention mean levels are not greatly aligned, and as such, the work search intention scale appears to be a suboptimal proxy of the economic integration (employment) of refugees for cross-national comparisons (CBS, 2022). One possible reason for the unalignment is that the intention-behavior gap (Sheeran & Webb, 2016) differs between cultures, alike to gap differences that have been observed between socioeconomic groups (Conner et al., 2013). Therefore, we suggest that future research investigates the intention-behavior gap in the cross-cultural context to understand the magnitude of and the explanations for this gap between nationalities and cultures. Besides, future research could also study actual employment among refugees with longer residence lengths.

A third suggestion is that future research could focus on a broader spectrum of traits. In this study, we used the ability-motivation-opportunity (AMO) framework (Appelbaum et al., 2000) to use GMA (ability), achievement motivation (motivation), and psychological distress

(opportunity) for examining the psychological clusters of refugees. Although this framework allows to cover important and different aspects of the psychological profile of individuals, one can investigate whether more differentiated psychological profiles can be revealed by using different or supplemental traits, such as emotional intelligence (e.g., Joseph & Newman, 2010) or cultural intelligence (e.g., Ward et al., 2009).

⁵ The correlation between the country-level employment rates and the means of work search intention is r = .35 (p = .559). However, this analysis with N = 5 countries severely lacks statistical power and needs to be interpreted carefully.

3

CHAPTER

Psychological Individual Differences as Predictors of Refugees' Local Language Proficiency

> Dan Asfar^{1,2} Marise Ph. Born^{1,3,4} Janneke K. Oostrom¹ Mark van Vugt¹

¹ Vrije Universiteit Amsterdam ² NOA B.V. ³ Erasmus University Rotterdam ⁴ North-West University

This chapter has been published as: Asfar, D., Born, M. P., Oostrom, J. K., & Van Vugt, M. (2019). Psychological individual differences as predictors of refugees' local language proficiency. *European Journal of Social Psychology, 49*(7), 1385–1400. https://doi.org/10.1002/ ejsp.2592

Acknowledgments We thank Remko van den Berg, Jacek Buczny, and Reinout E. de Vries for their assistance with the data collection and the statistical analyses.

ABSTRACT

Learning the local language is important for the successful integration of immigrants. Previous research has identified a number of sociodemographic factors that are associated with the effectiveness of local language acquisition among immigrants, but little is known about the influence of psychological differences on immigrants' local language acquisition. In the present research, individual differences in general mental ability (GMA), work search intention, and personality traits Conscientiousness and Openness were studied among recently arrived Syrian (n = 1054) and Eritrean (n = 500) refugees in the Netherlands. The results revealed that in addition to the effects of age of arrival, local length of stay, premigration educational attainment, and psychological distress, GMA and work search intention were positively associated with refugees' local language proficiency. Additionally, work search intention was found to strengthen the effect of GMA on local language proficiency. No positive linear effects were observed for Conscientiousness and Openness. Some evidence was found for curvilinear relationships between psychological predictors and local language proficiency. Implications are discussed.

KEYWORDS

refugees, local language proficiency, GMA, work search intention, Conscientiousness, Openness

PSYCHOLOGICAL INDIVIDUAL DIFFERENCES AS PREDICTORS OF REFUGEES' LOCAL LANGUAGE PROFICIENCY

From 2014 until the end of 2017, about four million refugees entered Europe, and many of them have applied for asylum (UNHCR, 2017). Given the differences between their home culture and the culture of residence, refugees face challenges with adjustment (Berry, 1997; Rudmin, 2003). Earlier research showed that successful integration and adjustment require learning the local language, which is associated with positive outcomes in social well-being, work, and education (e.g., Joly, 1996). Research among earlier waves of immigrants showed that host country language proficiency is positively related to psychological well-being (Beiser & Hou, 2001), employment (Aldashev et al., 2009; Bloch, 2002), and higher earnings (Chiswick & Miller, 2007; Dustmann, 1994; Shields & Price, 2002). In contrast, immigrants with poor local language skills were shown to have a higher chance to be marginalized from the community, are more dependent on social networks, and have less access to the labor market (Bloch, 2002; Schellekens, 2001; Valtonen, 1994).

Several earlier studies have investigated predictors of local language acquisition of immigrants and refugees (see Chiswick & Miller, 2007; Esser, 2006). However, this research suffers from several shortcomings. First, most of these studies have focused on sociodemographic factors as predictors of local language acquisition and have ignored psychological differences. Second, these studies focused predominantly on labor and family immigrants or made no distinction between such voluntary immigrants and refugees (Fennelly & Palasz, 2003). In comparison to voluntary immigrants, refugees are generally more highly educated (Liebau & Salikutluk, 2016), show lower employment rates (Salikutluk et al., 2016), are at higher risk for a variety of psychiatric disorders (Fazel et al., 2005; Hollifield et al., 2002; Schock et al., 2016), and are worse at acquiring the local language (Chiswick & Miller, 2001, 2007; Van Tubergen & Kalmijn, 2005). For these reasons, findings from voluntary immigrant samples cannot be generalized to refugee samples. Lastly, previous research has typically used self-reports (Beenstock et al., 2001; Carliner, 2000; Chiswick & Miller, 2002; Van Tubergen, 2010) or interviews (e.g., Van Tubergen & Kalmijn, 2005; Van Tubergen & Wierenga, 2011) to assess local language proficiency. Although the interview-based measure is arguably a more valid criterion than self-reports, objective measures of local language proficiency, such as standardized tests, would further increase the validity of research findings (Edele et al., 2015).

To address these concerns, the present research draws on the psychological literature on personnel selection and academic performance (e.g., Judge & Zapata, 2015; Roberts et al., 2007; Schmidt & Hunter, 1998; Von Stumm et al., 2011), investigating the predictive validities of individual differences for local language acquisition. Specifically, we test the

impact of general mental ability (GMA), work search intention, and the personality traits Conscientiousness and Openness on local language proficiency among a recent wave of Syrian and Eritrean refugees residing in the Netherlands.

LOCAL LANGUAGE PROFICIENCY

The majority of studies on immigrants' local language proficiency have been conducted in the field of sociology and economics and they typically draw on the standard theoretical model (Chiswick & Miller, 2001, 2007; Esser, 2006; Hwang & Xi, 2008; Mesch, 2003). This model contends that there are three general determinants of immigrants' local language acquisition: (1) exposure, that is, the extent to which immigrants hear and read the local language, (2) incentives, that is, the advantages one can obtain by mastering the local language weighted against the costs of learning the local language, and (3) efficiency, that is, the innate abilities to learn and acquire a new language (Chiswick & Miller, 2007). Based on this model, a number of sociodemographic predictors of local language acquisition have been identified, including gender (e.g., Beiser & Hou, 2000; Kristen et al., 2016; Van der Slik et al., 2015), length of stay in the country of residence (e.g., Carliner, 2000; Van Tubergen & Kalmijn, 2005), age of arrival (e.g., Kristen et al., 2016), and premigration education level (Beiser & Hou, 2000; Hayfron, 2001; Hou & Beiser, 2006; Van Tubergen, 2010). Mental health has also been studied often among immigrants and refugees (e.g., see Fazel et al., 2005; Porter & Haslam, 2005), and good mental health has frequently been associated with better local language acquisition (Beiser & Hou, 2001; Chiswick & Miller, 2001; Van Tubergen & Kalmijn, 2005; for exceptions, see Van Niejenhuis et al., 2015; Van Tubergen, 2010). Although these findings are informative, we argue that this literature fails to recognize that psychological and individual differences - in addition to situational and external factors - could also influence local language acquisition among migrants (Dörnyei, 2005).

Little is known about psychological predictors of local language acquisition among refugees. In the psychological literature on individual differences and personnel selection (e.g., Judge & Zapata, 2015; Roberts et al., 2007; Schmidt & Hunter, 1998), researchers have identified several important psychological predictors of performance in the domains of work and education, such as cognitive ability and personality traits. We expect that these psychological individual differences show similar or stronger relationships with local language acquisition compared to the effects that are observed in studies on academic performance. There is evidence that transition periods in life function as a catalyst for personality differences to be magnified (Caspi & Moffitt, 1993). That is, under conditions of change – such as refugees' forced resettlement into a new country that brings with it an unfamiliar culture – personality traits become accentuated and have a stronger effect on behavior, relative to under ordinary and undisrupted life conditions. This phenomenon might strengthen the effect of individual traits on the local language acquisition of refugees. In the following section, the theoretical and empirical basis of psychological predictors of

learning performance is reviewed with a particular focus on local language acquisition.

PSYCHOLOGICAL PREDICTORS General Mental Ability

GMA (or intelligence) has been defined as "the ability to understand complex ideas, to adapt effectively to the environment, to learn from experience, to engage in various forms of reasoning, to overcome obstacles by taking thought" (Neisser et al., 1996, p. 77). GMA determines an individual's capability to learn and perform well on tasks. GMA is a positive predictor for a number of performance indicators, including academic performance and achievement (Deary et al., 2007; Duckworth & Seligman, 2005; Sternberg et al., 2001), attained occupational level (Schmidt & Hunter, 2004), training success (Salgado et al., 2003), and work performance (Sackett et al., 2022; Schmidt & Hunter, 1998, 2004). GMA is also a predictor of local language proficiency in non-immigrant samples (e.g., Dörnyei, 2005; Pishghadam & Khajavy, 2013; Skehan, 1991). To the authors' best knowledge, only one study on local language proficiency among immigrants (in this study, voluntary immigrants) assessed GMA, and these researchers found a positive effect of GMA on local language proficiency (Edele et al., 2015). In line with these findings, we predict that GMA is positively associated with local language proficiency among refugees (H1).

Work Search Intention

Immigrants' local language proficiency is positively associated with employment and voluntary work (Bloch, 2002; Dustmann & Fabbri, 2003; Potocky-Tripodi, 2004). A possible explanation for this is that refugees who seek a job are more motivated to learn the local language. Correspondingly, recent findings on Syrian refugees in the Netherlands and in Greece showed that job search self-efficacy – the belief that one is competent in searching for and finding employment (Saks & Ashforth, 1999) – is positively correlated with local language proficiency (Pajic et al., 2018). Research has also shown that immigrants' employment history is a predictor of local language proficiency (Beiser & Hou, 2000). Drawing on these findings, we predict that work search intention is positively related to local language proficiency among refugees (H2).

Personality Traits

Two widely accepted personality taxonomies are the five- (the Big Five; Costa & McCrae, 1992) and the six-dimensional model (Ashton et al., 2004). One of the traits in these models is the personality factor that is known as Conscientiousness (Costa & McCrae, 1992). Individuals high in Conscientiousness are organized, responsible, and industrious (Lee & Ashton, 2004). Conscientiousness is one of the strongest non-cognitive predictors of academic achievement (Noftle & Robins, 2007; O'Connor & Paunonen, 2007; Robbins et al., 2004), and it even predicts academic success when it is assessed in childhood (Shiner et al., 2003). Alongside education, local language acquisition also requires discipline to

learn, and it can therefore be anticipated that refugees high in trait Conscientiousness do better in local language acquisition. Although we are not aware of direct evidence for this link, there is one relevant study among Dutch sixth-grade children which showed that Conscientiousness and Openness are positively related to foreign language vocabulary, grammar, and reading test scores (Verhoeven & Vermeer, 2002). Correspondingly, we predict that Conscientiousness is positively associated with local language proficiency among refugees (H3).

Another relevant personality trait is labeled Openness (or Openness to Experience; Costa & McCrae, 1992; Lee & Ashton, 2004). Individuals high in Openness are aesthetically sensitive and intellectual (Lee & Ashton, 2004), and they are hence expected to have a higher proclivity for learning a foreign language. Openness is associated with academic success (but see Busato et al., 2000), SAT scores (Noftle & Robins, 2007), and final grades (Farsides & Woodfield, 2003). In addition to the previously mentioned study by Verhoeven and Vermeer (2002) that showed a positive association between Openness and indices of foreign language skills, another study among students found that HEXACO Openness is correlated with subjective self-reported local language fluency (r = .20; Gargalianou et al., 2015). Lastly, a recent study among international students in the Netherlands found a small positive effect of Openness on local language proficiency (Van Niejenhuis et al., 2018). In line with these findings, we predict that Openness is positively associated with local language proficiency among refugees (H4).

Interaction Effects

Multiplicative models in industrial/organizational (I/O) psychology explicate that performance is a function of ability (typically operationalized as GMA) times motivation, suggesting that the positive effect of GMA on performance is stronger at higher levels of motivation (Klehe & Anderson, 2007; Mitchell & Nebeker, 1973). Although a recent meta-analysis on this issue concluded that the interaction effect between GMA and motivation explains little additional variance in job performance (Van Iddekinge et al., 2018), this meta-analysis only included measures of motivation, but not the general personality trait Conscientiousness. Studies that examined Conscientiousness as a moderator of GMA for predicting job performance reveal inconsistent findings: Some studies found no support for an interaction (Mount et al., 1999; Sackett et al., 1998), whereas other studies revealed support for interaction effects between GMA and achievement motivation related facets of Conscientiousness (Perry et al., 2010), and between GMA and contextualized measures of achievement motivation (Hirschfeld et al., 2004). In educational psychology, the effect of GMA on grade point average was found to be stronger at higher levels of (the facets of) Conscientiousness (Bergold & Steinmayr, 2018; Di Domenico & Fournier, 2015; Ziegler et al., 2009; but see Zhang & Ziegler, 2015). Openness has also been found to moderate the effect of GMA on academic performance, such that the effect of GMA on academic performance is stronger at lower levels of Openness (Bergold & Steinmayr, 2018; Di Domenico & Fournier, 2015; Zhang & Ziegler, 2015). GMA and Openness have a compensatory effect on performance, and hence, the lower one's level of Openness, the larger the relative contribution of GMA on performance (see Ziegler et al., 2012).

Although work search behavior and Conscientiousness are moderately related (r=.30; Kanfer et al., 2001), their shared variance is small enough to predict additive interaction effects of GMA with both work search intention and Conscientiousness. Whereas Conscientiousness covers one's general level of industriousness and orderliness (DeYoung et al., 2007), work search intention can be considered a more specific, contextualized measure of motivation – and contextualized measures show higher validities than non-contextualized measures (e.g., see Shaffer & Postlethwaite, 2012). In line with these findings, we anticipate that GMA interacts with work search intention, Conscientiousness, and Openness for predicting local language proficiency among refugees. Thus, we predict that the effect of GMA on local language proficiency among refugees is stronger at higher levels of work search intention (H5), at higher levels of Conscientiousness (H6), and at lower levels of Openness (H7).

EXPLORATORY CURVILINEAR ANALYSES

In addition to testing the hypotheses, nonlinear relationships between the predictors and local language proficiency are explored. To our knowledge, no previous literature has reported curvilinear effects of predictors of second language learning. Nonetheless, previous studies have revealed inverted U-shaped relationships between Conscientiousness and task and contextual performance (Janssen, 2001; LaHuis et al., 2005; Le et al., 2011; Whetzel et al., 2010; Wihler et al., 2017; but see Robie & Ryan, 1999), training performance (Vasilopoulos et al., 2007), and grade point average (Cucina & Vasilopoulos, 2005). We are only aware of one study that revealed a nonlinear – in this case, a U-shaped – relationship between Openness and grade point average (Cucina & Vasilopoulos, 2005). Regarding GMA, some scholars have theorized that the positive effect of GMA on performance weakens at higher levels of the construct (Jensen, 1998; Robertson et al., 2010; Te Nijenhuis & Hartmann, 2006). Empirical work, however, has failed to find support for this proposition in the context of work (Coward & Sackett, 1990) and education (Coyle, 2015; Lubinski, 2009; Park et al., 2008; Ziegler & Peikert, 2018). To our knowledge, there is no literature on curvilinear relationships between work search intention and performance indicators.

METHOD

PARTICIPANTS

3

We obtained data of refugees from 81 countries (mostly from the Middle East and Africa), but we only report the findings of refugees from Syria (n = 1054) and refugees from the East-African country Eritrea (n = 500), as they comprise the largest refugee groups in the current data set (M_{22} = 29.14 years, SD = 8.76; 72.5% males). Previous research has shown that refugees from these countries flee from harsh societal conditions and they typically report to have undergone traumatic events and to have suffered from mental health problems (Dagevos et al., 2018; Sterckx et al., 2018). The average age of arrival was M = 27.53 (SD = 8.60). The sample sizes were not a priori determined, as the principal aim of the assessments initially concerned refugee consultation instead of research. Also, the seven hypotheses in this study are presented as confirmatory hypotheses, but for the sake of transparency, we indicate that this study was not pre-registered. From the analyses, we have excluded five participants because of suspicious response patterns (i.e., identical responses on all items, or only extreme responses), and we have excluded two participants because they have reported a length of stay in the Netherlands that is substantially higher than the length of stay reported by the other participants (i.e., 87 and 214 months; 8.48 and 24.33 SD above the mean).

The average local length of stay in the Netherlands was 16.95 months (SD = 8.26), and the Syrian refugee group (M = 16.19, SD = 8.27) had a shorter local length of stay than the Eritrean refugee group (M = 18.61, SD = 8.01), t(1516) = -5.33, p < .001. The refugee participants resided in one of four large Dutch municipalities, and some refugees conducted the assessment through an invitation from the Foundation for Refugee Students (UAF), which is a foundation that supports refugees by providing education and finding work. Less than half of the participants were married (43.2%), and 47.0% of the participants reported having no family in the Netherlands at the time of assessment. Attained education levels at the country of origin were converted by Nuffic, a Dutch certificated agency, to match the standards of the European Qualifications Framework (EQF). Defined by the EQF standards, 18.0% of the participants attained level 2 (cf. basic education), 22.5% participants attained level 3 (cf. vocational secondary education), 29.5% participants attained level 4 (cf. associate's degree), 27.9% participants attained level 6 (cf. bachelor's degree), and 2.1% of participants attained level 7 (cf. master's degree). The majority of the refugees had worked in their country of origin (62.8%), reporting work experience of less than 1 year (6.6%), 1 to 2 years (15.4%), 2 to 5 years (17.3%), 5 to 10 years (12.6%), 10 to 20 years (6.8%), and more than 20 years (4.1%).

PROCEDURE

50 | CHAPTER 3

A Dutch psychological consultancy agency that focuses on the development of psychological tests and the assessment of individuals for work and educational-related purposes has

gathered data among a large group of refugees through assessments for municipalities and NGOs since the year 2016. The inclusion criteria of the consultancy agency for assessing refugees were being at least 18 years old, having a residency permit, and being literate. Through a formal letter, refugees were requested to take part in an online assessment. In this letter, it was emphasized that participation was voluntary and that the assessment could help with finding a job or an education, and therefore stimulate integration into the Dutch society. A large proportion of the invited refugees agreed to participate. Although the exact value is unknown, employees at the municipalities estimated the response rate to be at least 95%. The assessments were typically administered in a distraction-free room of the municipality's town hall. One or more counselors were present to help respondents with questions if necessary, and no communication with others was allowed during the assessments. No compensation was offered in return for participation.

MATERIALS

The measures were developed in Dutch and had subsequently been translated into Modern Standard Arabic (for Arabic refugees, including Syrians) and into Tigrinya (for Eritrean refugees) by a certified translation agency. Thus, all assessment components (i.e., the instructions, the tests, and the questionnaires) were administered in the refugees' mother tongue. The full assessment consists of sociodemographic questions, two GMA subtests, scales for self-reported competencies, personality traits, and work motivation, a posttraumatic stress disorder checklist (the PCL-5; Blevins et al., 2015), a measure of psychological distress (the K10; Kessler et al., 2002), and a Dutch and an English language proficiency test. Here, we only describe the instruments that are relevant to the present study. Readers who are interested in the test environment or who would like to see sample items of the two GMA subtests are referred to the footnote.⁶

Measurement Invariance and Refugee Group Differences

To investigate whether the assessed instruments have similar validities among the two refugee groups, we tested for metric measurement invariance – that is, whether item and factor loadings are equivalent across the Syrian and the Eritrean refugee group. Scholars have recommended considering support for metric invariance when Δ CFI \leq 0.01 (Cheung & Rensvold, 2002). Comparing latent variable models for Syrian and Eritrean refugees, based on the Δ CFI \leq 0.01 cut off, we found support for metric invariance for the measures of work search intention (Δ CFI = .002), Conscientiousness (Δ CFI = .006), Openness (Δ CFI = .008), and local language proficiency (Δ CFI = .008), but not for GMA (Δ CFI = .016) and psychological distress (Δ CFI = .013). Given the small deviations of the latter two instruments and the unequal sample ratio in this research, we considered it unnecessary to remove

⁶ To see some sample items of the two GMA subscales that were completed by the refugee participants, readers can visit the webpage: https://www.noa-online.net/practicequestions/mct-m

items from the GMA test and the psychological distress scale. We have also conducted Confirmatory Factor Analyses of the instruments in our study, and the fit indices are reported in the Supplementary Material (Table S1).

General Mental Ability

GMA was assessed by two non-verbal subtests of the multicultural capacities test (MCT-M; Van den Berg, 2001). The subtests were developed to measure fluid intelligence (Cattell, 1971), and aimed to reduce or eliminate a potential bias that might be imposed by cultural background when using tests that contain cultural elements such as language or knowledge (Van de Vijver & Tanzer, 1997). One subtest is labeled *Components*, in which candidates have to select two out of the six spatial parts that can make up one displayed figure. The other subtest is labeled *Exclusion*, in which candidates have to select the figure that does not match the other four presented figures. The Components subtest has a time limit of 9 minutes, and the Exclusion subtest has a time limit of 7 minutes. Earlier empirical work showed evidence for the cross-cultural applicability of the MCT-M and its predictive validity in the domains of social functioning and academic achievement among native Dutch candidates and among several Dutch migrant groups (Van den Berg, 2001). In the present study, a total score for GMA was computed by adding up the number of correctly answered items of the two subtests. The correlation between the two subtests was r = .55, p < .001. Coefficient alpha of the two subtests in the current sample was .92.

To support the aggregation of the two subtests into one total score of GMA, we conducted confirmatory factor analyses using the R package Lavaan (version 0.05-23.1097; Rosseel, 2012). We compared two latent variable models. One model is the hypothesized hierarchical bi-factor model that includes a general factor of cognitive ability in addition to two factors that represent the two subtests. The other model is a correlated-factors model with two latent variables that represent the two subtests. The models were analyzed using the maximum likelihood estimation method, and missing values were dealt with by using full information maximum likelihood (El-Sheikh et al., 2017). The analyses indicated that the hierarchical bi-factor model (χ^2 [1650] = 2747.02, p < .001, CFI = .937, TLI = .932, RMSEA = .021, SRMR = .032), has a better fit than the correlated two-factor model (χ^2 [1719] = 3703.18, p < .001, CFI = .885, TLI = .881, RMSEA = .027, SRMR = .046), χ^2 (59) = 956.16, p < .001, justifying the aggregation of the two subtests of the MCT-M into one total score.

Work Search Intention

We assessed work search intention with a 10-item subscale of a work motivation questionnaire (AWV; NOA, 2005). Research has shown convergent validity for this instrument with other work motivation instruments (Dusseldorp et al., 2018). Example items are "How much time do you spend on searching for vacancies on the internet?", and "How frequently do you approach employers for job opportunities?". Participants were instructed to rate on a 5-point

Likert scale how much time they spend on such activities, ranging from 1 = Not at all to 5 = Very frequently. In the current sample, coefficient alpha of this scale was .91.

Conscientiousness and Openness

To assess the personality traits Conscientiousness and Openness, we used the MPT-BS-QS Basic (NOA, 2009; Holtrop et al., 2014), which is a short (60-item) version of a personality inventory that consists of six factors, and that corresponds to the HEXACO model of personality (Lee & Ashton, 2004). In the present article, we focus exclusively on Conscientiousness and Openness, as we consider these traits to be the most relevant for predicting local language acquisition. The correlations between the other four personality dimensions and the study variables are reported in the Supplementary Material (Table S2). Ten items were used to assess each personality dimension. Example items of Conscientiousness are "I do things very precisely", and "I think carefully before I act", and example items of Openness are "I often come up with plans to do new things", and "I have often more than one idea on how to do something". Coefficient alpha of Conscientiousness and Openness in the current sample were respectively .69 and .83. Participants were instructed to rate on a 5-point Likert scale how much they agree or disagree with each statement, ranging from 1 = Totally disagree to 5 = Totally agree. The two personality scales correlated quite highly, r = .66, p < .001, so we conducted a confirmatory factor analysis to verify the underlying two-factor structure. Results showed that the two-factor structure fits the data reasonably well (χ^2 [151] = 859.63, p < .001, CFI = .908, TLI = .895, RMSEA = .055, SRMR = .042), and significantly better than a one-factor solution (χ^2 [152] = 983.97, p < .001, CFI = .891, TLI = .878, RMSEA = .059, SRMR = .044), Δx^2 (1) = 124.34, p < .001.

Local Language Proficiency

Local (Dutch) language proficiency was assessed using an instrument that had been developed by the psychological consultancy agency that gathered the data (NOA, 2006). In this test, participants read two short stories written in Dutch (in total, 198 words), which include sentences with 80 incomplete words. Participants were instructed to complete the word fragments such that they make meaningful words in their context, within a 15-minute time limit. Scores on this test could range from 0 (no word fragment completed correctly) to 80 (all word fragments completed correctly). Previous unpublished research has shown that the scores on this test correlate strongly with scores on a Dutch language test that is used nationwide (NT2; CINOP, Citogroep, Bureau ICE, & BVE Raad, 2002), supporting the construct validity of the test. Specifically, the Dutch language proficiency test that was used in the current study correlated with the NT2 subtests of reading (r = .65), writing (r = .78), vocal understanding (r = .48), and speaking (r = .50) (NOA, 2006). In the current sample, coefficient alpha of the Dutch proficiency test was .97.

Control Variables

Demographic Variables. To establish the relative importance of psychological differences in local language acquisition above and beyond the effects of sociodemographic predictors, the variables gender, age of arrival in the Netherlands, length of stay in the Netherlands, and premigration educational attainment were included in the analyses. Previous research has revealed mixed findings regarding the relationship between gender and local language acquisition among immigrants. Some studies reported greater local language proficiency among female immigrants (e.g., Van der Slik et al., 2015), whereas other studies reported greater levels of local language proficiency among male immigrants (e.g., Beiser & Hou, 2001; Dustmann & Fabbri, 2003; Fennelly & Palasz, 2003; Van Tubergen, 2010), or showing no gender differences (e.g., Kristen et al., 2015). Previous research has shown that local language proficiency among immigrants is negatively associated with the age of arrival in the country of residence (e.g., Kristen et al., 2016), and positively associated with local length of stay (Carliner, 2000; Van Tubergen & Kalmijn, 2005) and premigration education level (Beiser & Hou, 2000; Hayfron, 2001; Hou & Beiser, 2006; Van Tubergen, 2010).

Psychological Distress. Despite some exceptions (Van Niejenhuis et al., 2015; Van Tubergen, 2010), most previous research findings revealed a negative effect of psychological distress on local language proficiency (Beiser & Hou, 2001; Chiswick & Miller, 2001; Van Tubergen, & Kalmijn, 2005). Psychological distress was measured with the 10-item Kessler Psychological Distress Scale (K10; Kessler et al., 2002). Participants were asked to indicate on a 5-point Likert scale how often they experienced or felt something during the last 30 days. Example items are "About how often did you feel nervous?", and "About how often did you feel hopeless?". Previous research has demonstrated that the K10 is a reliable and valid tool to assess anxiety and depressive disorders in clinical and in non-clinical populations (e.g., Cairney et al., 2007; Furukawa et al., 2003; Kessler et al., 2003), and it has shown predictive validity for several psychiatric disorders (Donker et al., 2010). Good psychometric qualities of the K10 were also demonstrated among non-Western samples, supporting the cross-cultural validity of the instrument (Fassaert et al., 2009). In the current sample, coefficient alpha of this scale was .87.

RESULTS

PRELIMINARY ANALYSES

Independent sample *t*-tests were conducted to exploratory investigate differences in test and scale scores between Syrian and Eritrean refugees. The analyses showed that Syrian refugees scored higher on GMA, t(1058.47) = 21.46, p < .001, Conscientiousness, t(1541) =13.57, p < .001, and Openness, t(1541) = 11.26, p < .001, whereas the Eritrean refugees scored higher on work search intention, t(1545) = -13.25, p < .001. No refugee group differences were observed for psychological distress, t(1451) = 1.11, p = .27, and local language proficiency, t(1147.38) = 1.82, p = .07. Table 1 shows the means, standard deviations, effect sizes, and correlations between study variables of the total sample and of the Syrian and Eritrean refugees separately.

GMA showed the strongest correlation with local language proficiency (r = .29, p < .001), followed by local length of stay (r = .26, p < .001) and educational attainment (r = .26, p < .001), age of arrival (r = .15, p < .001), psychological distress (r = .13, p < .001), and work search intention (r = .09, p < .001). No significant correlations were observed between local language proficiency and Conscientiousness (r = .03, p = .24) or Openness (r = .02, p = .37). There was also no significant difference between men (M = 18.92, SD = 17.38) and women (M = 17.28, SD = 18.28) on local language proficiency, t(1545) = 1.70, p = .09.

HYPOTHESIS TESTING

In order to investigate the unique predictive validity of the variables of interest on local language proficiency, we conducted a hierarchical regression analysis with three steps (Table 2). The first step (Model 1) included the control variables gender, age of arrival, local length of stay, premigration educational attainment, and psychological distress. In the second step (Model 2), GMA, work search intention, Conscientiousness, and Openness were added, testing H1, H2, H3, and H4, respectively. In the third step (Model 3), the interaction terms of GMA with work search intention (H5), GMA with Conscientiousness (H6), and GMA with Openness (H7) were included. We expected that every predictor and interaction term would explain unique variance in local language proficiency. In Table 2, beta-coefficients and their confidence intervals were reported to present the effects of the predictors of local language proficiency. The R^2 and Cohen's f^2 statistic in Table 2 respectively indicate the total explained variance of the model and the corresponding effect size, where $f^2 \ge 0.02$ is a small effect, $f^2 \ge 0.15$ is a medium effect, and $f^2 \ge 0.35$ is a large effect (Cohen, 1988).

No differential effects of the predictors of local language proficiency were observed between the two refugee groups. Specifically, refugee group did not moderate the effect of GMA (β = .06, *t* = 1.50, *p* = .13), work search intention (β = .05, *t* = -1.48, *p* = .14), Conscientiousness (β = .05, *t* = 0.82, *p* = .41), Openness (β = .05, *t* = 0.87, *p* = .39), and the interactions between GMA and work search intention (β = -.02, *t* = -0.18, *p* = .75), Conscientiousness (β = .06, *t* = 0.89, *p* = .37), and Openness (β = -.01, *t* = -0.32, *p* = .85). Therefore, the hierarchical regression analysis was conducted for the entire sample.

Variables	
s of Study	
Correlation	
l Bivariate (
Sizes, anc	
ics, Effect	
tive Statist	
ic Descrip	
oup Speci	
lefugee Gr	
erall and F	
Table 1. <i>OV</i>	

3

Variable	M	200	M	su	MF	20,	D	_	V	ŋ	4	ŋ	D	~	0	ע	2
1. Gender	0.28	0.45	0.28	0.45	0.27	0.44	0.02	1	/*00. 07	29**/ 11*	.10**/ .10	01/ .07	05/ 08	18**/ 05	03/ 10*	16*/ 13**	
2. Age of arrival	27.53	8.60	29.05	9.78	24.69	5.71	0.54**	*90.	ı	13**/ 20**	.11*/ 07	.03/ .03	09*/ 07	.04/ .12**	/**80. .08	/**80. *60.	18**/ 12*
3. Local length of stay	16.95	8.26	16.19	8.27	18.61	8.01	0.30**	24**	17**	ı	10**/ 02	.05/ .02	.04/ .02	.05/ 11*	03/ .02	02/ .00	.29**/ .35**
 Educational attainment 	2.43	1.50	2.32	1.58	2.89	1.00	0.43**	**60.	*90.	07*		10**/ 03	.21**/ 12	.11**/ 11	.12**/ 07	.13**/ 10	.28**/ .12
5. Psychological distress	1.74	0.72	1.75	0.74	1.70	0.67	0.06	.02	.04	.04	10**	(.89/ .84)	11**/ 10*	07**/ 06	05/ 13**	02/ 07	15**/ 07
6. GMA	32.29	10.83	35.75	9.87	24.94	8.93	1.15**	04	.05*	03	.10**	08**	(.91/ .81)	.11**/ .12**	.07*/ .06	.16**/ .14**	.33**/
7. Work search intention	2.44	0.93	2.23	0.88	2.87	0.91	0.72**	 13**	02	.04	.12**	07**	05*	(.91/ (.88)	.16**/ .23**	.27**/ .25**	.15**/ 01
8. Conscientiousness	3.86	0.45	3.97	0.42	3.65	0.45	0.74**	05	.15**	06*	.05	07*	.21**	.06**	(.68/ .63)	.63**/ .62**	01/ .08
9. Openness	3.90	0.53	4.00	0.50	3.69	0.53	0.60**	-,14**	.14**	05*	.06*	03	.26**	.15**	.66**	(.84/ .74)	01/ .06
10. Local language proficiency	18.46	17.62	18.98	18.57	17.34	15.37	0.10	04	15**	.26**	.26**	13**	.29**	**60.	03	.02	/76.) (76.
Note. The subscript $_{o, s'}$ effect size, where 0.10 = below the diagonal. Corr	and _E resp small, 0.3C elations fo	ectively r) = mediu r respecti	epresent m, 0.50 = vely Syria	the overa large (Co in and Eri	all sample bhen, 1988 trean refu	$(N_o = 15)$ 3). Gende gees are	47), the S er, Male = : shown a	yrian refu 0, Female bove the c	gee grou e = 1. GM/ diagonal,	Ip $(n_s = 10)$ A = generic separate	53), and al mental d by a sla	the Eritrea ability. Co ish. Simila	an refuge orrelatior Irly, the v	ee group i is for the alues betv	$(n_E = 494)$ entire sal ween bra	. <i>d</i> = Coh nple are : ckets rep	en's <i>d</i> shown resent

Conscientiousness, ention, 1.70-5.00 for intention, Conscie search intention, 1. search i work s stress, ¹ distres: distre al di r psychological of for psychologica 1.00-5.00 for 1 is 1.00-5.00 fo is is le of scores is 1ge of scores Possible range c The actual range ocal language pro refugees, respectively. The local language proficiency. I-57 for GMA, and 0-78 for lo ean for ss, 1 alpha coefficients for Syrian and Eritre Openness, 0-60 for GMA, and 0-80 scientiousness, 1.60-5.00 for Openne. < .05, ** *p* < .01 (two-tailed). and -Cons Q

Model 1 (Table 2, Model 1) of the hierarchical linear regression analysis showed that local language proficiency is not associated with gender ($\beta = .01, t = 0.47, p = .64$), it is negatively associated with age of arrival ($\beta = -12$, t = -4.49, p < .001), positively associated with local length of stay ($\beta = .31$, t = 11.03, p < .001) and premigration educational attainment ($\beta = .28$, t = 10.34, p < .001), and negatively associated with psychological distress ($\beta = -.11$, t = -3.91, p < .001). Model 1 explained 19.7% variance in local language proficiency, which corresponds to Cohen's f^2 = .25, indicating a medium effect size.

In Model 2 (Table 2, Model 2), GMA (β = .29, t = 10.82, p < .001) and work search intention $(\beta = .06, t = 2.32, p = .02)$ showed a positive relationship with local language proficiency above and beyond the effects of the control variables, supporting H1 and H2. In contrast to the predictions of H3 and H4, no significant relationships were observed between refugees' local language acquisition and Conscientiousness ($\beta = .03, t = 0.75, p = .45$) or Openness (β = -.06, t = -1.67, p = .10). Overall, Model 2 explained 7.9% additional variance over and above Model 1 (i.e., in total 27.6% explained variance), F(4, 1116) = 30.42, p < .001, which corresponds to Cohen's f^2 = .39, indicating a large effect size.

Model 3 showed an interaction effect between GMA and work search intention on local language proficiency ($\beta = .09, t = 3.37, p = .001$), such that the effect of GMA on local language proficiency was stronger at higher levels of work search intention, supporting H5 (Figure 1). A simple slope analysis showed that the positive relationship between GMA and local language proficiency was stronger at +1 SD of work search intention (β = .36, t = 10.28, p < .001) compared to -1 SD of work search intention (B = .23, t = 6.78, p < .001). In contrast to the predictions of H6 and H7, no evidence was found for interaction effects between GMA and Conscientiousness ($\beta = -.04$, t = -1.10, p = .27) and between GMA and Openness $(\beta = -.05, t = -1.39, p = .17)$ on local language proficiency. Model 3 explained 1.1% additional variance in local language proficiency over and above Model 2, F(3, 1113) = 5.90, p = .001. The total amount of explained variance in local language proficiency of Model 3 is 28.7%, corresponding to Cohen's $f^2 = .40$, indicating a large effect size.⁷

⁷ The multiple regression analysis was also conducted without the control variables. The results are comparable to the results with the control variables, except for Openness. In the model without control variables, Openness (β = -.09, p = .003) as well as the interaction term of GMA and Openness ($\beta = .07$, p = .04) are significantly negatively associated with local language proficiency.

Table 2. Hierarchical Regression Analysis with Predictors of Local Language Pre-	roficiency
--	------------

	Local Language Proficiency					
-	Model 1		Model 2		Model 3	
Variable	ß	95% CI	ß	95% CI	ß	95% CI
Gender (male = 0, female = 1)	.01	[04, .07]	.03	[02, .09]	.03	[05, .05]
Age of arrival	12***	[18,07]	13***	[18,08]	13***	[18,08]
Local length of stay (months)	.31***	[.25, .36]	.31***	[.26, .37]	.31***	[.26, .37]
Educational attainment	.28***	[.23, .33]	.25***	[.20, .30]	.25***	[.20, .31]
Psychological distress	11***	[16,05]	08**	[13,03]	08**	[13,03]
GMA			.29***	[.24, .34]	.28***	[.23, .34]
Work search intention			.06*	[.01, .12]	.07**	[.02, .12]
Conscientiousness			.03	[04, .09]	.02	[05, .08]
Openness			06	[13, .01]	07	[14, .00]
GMA x Work search intention					.09**	[.04, .14]
GMA x Conscientiousness					04	[11, .03]
GMA x Openness					05	[12, .02]
R^2		.20		.28		.29
F		55.04***		47.21***		37.30***
ΔR^2				.08		.01
ΔF				30.23***		5.75*
f ²		25		39		40

Note. GMA = general mental ability; 95% CI = 95% confidence interval. f' = Cohen's f', where $f' \ge 0.02$, $f' \ge 0.15$, and $f' \ge 0.35$ represent small, medium, and large effect sizes, respectively (Cohen, 1988). N = 1547.

* p < .05, ** p < .01, *** p < .001.

Figure 1. The Relation between General Mental Ability and Refugees' Dutch Language Proficiency at +1 and -1 SD of Work Search Intention



Note. The values on the x-axis represent z-scores, and the grey area around the lines illustrates the 95% standard error confidence interval.

EXPLORATORY CURVILINEAR ANALYSES

To test for curvilinear relationships of the predictors with local language proficiency, we compared a linear regression model, $y = b_0 + b_1x$, with a quadratic regression model, $y = b_0 + b_1x + b_2x^2$, where y is the dependent variable local language proficiency, and where x represents the total score on one of the four predictors. Unique variance in local language proficiency was explained by the quadratic terms of Conscientiousness (*F*[1, 1540] = 6.83, p = .001, $\Delta R^2 = .01$), Openness (*F*[1, 1540] = 3.90, p = .02, $\Delta R^2 = .01$), and GMA (*F*[1, 1542] = 10.36, p = .001, $\Delta R^2 = .01$), but not work search intention (*F*[1, 1544] = 0.25, p = .62, $\Delta R^2 < .001$). Conscientiousness and Openness revealed inverted U-shaped relationships with refugees' local language proficiency (see Appendix, Figure A1 and Figure A2), whereas the relationship between GMA and local language proficiency showed an exponential trend (see Appendix, Figure A3).

DISCUSSION

Drawing on the literature on psychological individual differences and personnel selection (e.g., Judge & Zapata, 2015; Roberts et al., 2007; Schmidt & Hunter, 1998), the present study examined the utility of psychological traits to explain differences in local (Dutch) language acquisition among Syrian and Eritrean refugees in the Netherlands, above and beyond factors that have been studied in prior empirical work. Specifically, we examined the incremental validity of GMA, work search intention, and the personality traits Conscientiousness and Openness, above and beyond sociodemographic variables and psychological distress in the prediction of refugees' local language proficiency.

THEORETICAL IMPLICATIONS

This study contributes to the literature on refugees' local language acquisition in several important ways. First, most previous research on local language acquisition investigated voluntary immigrants, and only a few studies have investigated refugees exclusively (Fennelly & Palasz, 2003). We replicated some earlier findings among a sample of Syrian and Eritrean refugees residing in the Netherlands. The findings showed that local language proficiency levels were about similar for men and women, but higher among refugees who were younger, who had a longer length of stay in the Netherlands, who had a higher premigration education level, and who experienced lower levels of psychological distress. The latter finding is especially important, as some earlier studies did not find a negative link between psychological distress and local language acquisition among immigrants (Van Niejenhuis et al., 2015; Van Tubergen, 2010). This could possibly be explained by our use of more reliable and valid instruments of psychological distress and language proficiency.

Second, to our knowledge, we are the first to simultaneously test the effects of individual differences in GMA, work search intention, and personality traits as predictors of immigrants' local language acquisition. The results revealed that, above and beyond the effects of the sociodemographic variables and psychological distress, refugees' level of local language proficiency is most strongly and positively associated with GMA, and to a lesser extent, yet still significantly and positively associated with work search intention. Unexpectedly, no positive relationships were found between Conscientiousness (i.e., being organized and industrious) and Openness (i.e., being aesthetically sensitive and intellectual), and refugees' local language proficiency. Exploratory analyses revealed inverted U-shaped relationships of these latter two personality traits with local language proficiency. That is, higher local language proficiency scores were observed around the mean of the trait continuum, whilst both extreme ends of the continuum tend to display lower scores. Additionally, evidence was found for a curvilinear relationship between GMA and local language proficiency, such that this link is stronger at higher levels of GMA. Altogether, the findings indicate that cognitive ability and work motivation are positively related to refugees' local language proficiency, but that the personality traits Conscientiousness and Openness are not linearly positively associated with local language proficiency among refugees.

We propose two explanations for the lack of support for linear positive relations of Conscientiousness and Openness with refugees' local language proficiency. One possible explanation is that the effect of personality on behavior - as some theorists have argued - is limited in collectivistic cultures, due to individuals' higher responsibility to social roles and relationships (Heine, 2001; Markus & Kitayama, 1998; Shweder, 1991; for a discussion, see Church & Katigbak, 2017). According to this view, refugees' embeddedness in their social role is of more relative importance than their personality traits in predicting behavior. Another possible explanation is self-selection in personality - that is, a phenomenon where individuals with certain personality profiles are more or less inclined to migrate, resulting in a smaller variance of migrants' personality traits (Boneva & Frieze, 2001). Earlier research among non-refugee samples has found that immigration was predicted by low Conscientiousness and high Openness (Ciani & Capiluppi, 2011; Jokela, 2009; Tabor et al., 2015). Self-selection among refugees would manifest itself in a smaller range of test scores, decreasing the statistical power to detect relationships between psychological characteristics and local language acquisition. Unfortunately, it was impossible to test this hypothesis in the current study, as we have no assessment data of a representative sample of Syrian or Eritrean citizens residing in their home country.

The third contribution of this study is the investigation of interaction effects of GMA with work search intention, Conscientiousness, and Openness. Earlier research had shown that the positive effect of GMA on performance is stronger at higher levels of achievement motivation and at lower levels of Openness (e.g., Bergold & Steinmayr, 2018; Di Domenico &

Fournier, 2015; Ziegler et al., 2009). This study showed that the effect of GMA on refugees' local language proficiency was stronger at higher levels of work search intention. We found no support that Conscientiousness or Openness moderated the relationship between GMA and refugees' local language proficiency.

STRENGTHS, LIMITATIONS, AND FUTURE DIRECTIONS

The present study has several strengths. First, we studied participants from two samples of refugees that have been underexplored in the literature. Psychological traits and their predictive validity in the domain of work and education have been rarely studied in Middle-Eastern samples (Henrich et al., 2010). We showed that the relationships between the assessed psychological predictors and local language proficiency were comparable for Syrian and Eritrean refugees, who originate from different ecological environments. Another strength of this study is the large sample sizes of the refugee groups, which allow for reliable effect estimates. The use of an objective measure of local language proficiency is also a strength of this study. Previous research has shown that objective measures of language proficiency reveal substantially different findings from self-report measures of language proficiency (Edele et al., 2015). Furthermore, the subtests that were used in this study measure fluid (non-verbal) intelligence, which indicates that there is no contamination with the dependent variable. Lastly, as there were potential incentives associated with assessment results (i.e., opportunities in work and education), we believe that the participants were motivated to respond honestly to the questionnaires and perform well on the ability tests. This supports the validity of the research findings (Duckworth et al., 2011).

Despite the strengths of the current research, there are also some limitations. One limitation is that there has been no substantial empirical validation so far of the cross-cultural personality inventory that was used in the present study. It is therefore somewhat uncertain whether Conscientiousness and Openness were measured adequately, although the operationalizations of these constructs show similarities to the well-validated HEXACO model of personality. Second, given that our study has a cross-sectional design, considerable caution is necessary when deriving conclusions about the causal nature of the findings. Although a causal relationship with local language acquisition is rather indisputable for some predictors (e.g., age of arrival, local length of stay, premigration educational attainment, and GMA), other predictors such as psychological distress, work search intention, and personality traits could as well be influenced by mastering the local language, or these relationships could also be bidirectional.

An important avenue for future research is to replicate our findings among other refugee and non-refugee migrant groups and in other countries. For instance, cross-country differences have been found in local language acquisition (e.g., Fennelly & Palasz, 2003; Van der Slik et al., 2015), and immigrants who originate from countries with a larger linguistic distance (i.e., a

larger dissimilarity between languages) show a slower local language acquisition (Chiswick & Miller, 2001; Kristen et al., 2016). We also advocate for the use of personality inventories that have been validated across cultures, such as the HEXACO personality inventory (Lee & Ashton, 2004) or the International Personality Item Pool (IPIP; Goldberg, 1999). Alternatively, researchers could use emic (i.e., local-specific) personality inventories, such as The South African Personality Inventory (Fetvadjiev et al., 2015) or the Arab Personality Inventory (Zeinoun, et al., 2017). Additionally, researchers could examine the effect of personality traits at the level of their facets. Research has shown that personality facets could display substantially different correlations with a criterion compared to their overarching personality dimensions (e.g., Moon et al., 2008). In the present research, we examined the effects of refugees' individual differences after a relatively short duration in the country of residence – that is, on average about 17 months. Future research could investigate the predictive validity of psychological individual differences in local language proficiency after a longer local length of stay.

Beyond the future directions considered so far, we advocate for longitudinal research investigating refugees' individual differences in personality and cognitive abilities as predictors of local language acquisition. Such research could also be extended to other relevant outcomes for refugees, such as employment and social adjustment. Another suggestion for future research is to examine psychological differences between refugees and people from the same country who did not migrate under harsh societal circumstances. This would offer insights into the potential psychological precursors of migration. In this respect, research so far has only focused on voluntary migrants and students, mostly from Western countries (Boneva & Frieze, 2001). Finally, research could explore the potential influences of local resettlement policies and practices on local language acquisition (see Koopmans, 2010).

CONCLUSION

The present study investigated the relationship between refugees' psychological individual differences and local language acquisition. We revealed that in addition to the effects of sociodemographic variables and psychological distress, refugees' local language proficiency is positively associated with GMA and work search intention, and that the effect of GMA on local language proficiency is stronger at higher levels of work search intention. No linear positive relationships were found between personality traits Conscientiousness and Openness and local language proficiency, but some evidence was found for curvilinear relationships between these personality traits and local language proficiency. The findings suggest that among refugees, psychological individual differences in cognitive ability and work motivation are important for learning the local language.

APPENDIX

Figure A1, A2, and A3. (a) The inverted U-shaped relationship between Conscientiousness (z-scores) and Dutch language proficiency scores. (b) The inverted U-shaped relationship between Openness (z-scores) and Dutch language proficiency scores. (c) The curvilinear relationship between general mental ability (z-scores) and Dutch language proficiency scores





CHAPTER

A Hindering and Facilitating Individual-Difference Factors Framework for Predicting Refugees' Workforce Participation

> Dan Asfar^{1,2} Marise Ph. Born^{1,3,4} Janneke K. Oostrom¹ Remko H. van den Berg²

¹ Vrije Universiteit Amsterdam
 ² NOA B.V.
 ³ Erasmus University Rotterdam
 ⁴ North-West University

Paper submitted for publication

ABSTRACT

Finding employment is an important aspect of refugees' integration into their receiving society. Previous research has identified several individual-difference factors that form obstacles towards refugees' quick and successful workforce participation. In the present study, we organize these individual-difference factors into an integrative framework containing a wide range of factors that may either hinder or facilitate refugees' path to finding work. Using time-lagged data of recently arrived Syrian (n = 1867) and Eritrean (n = 844) refugees in the Netherlands, we examined the validity of refugees' individualdifference factors for predicting their workforce participation (i.e., employment and longest employment duration). We also exploratorily examined relations between individualdifference factors and refugees' highest hourly wage. The results showed that several hindering factors (i.e., older age, being a woman, and physical health problems) and several facilitating factors (i.e., pre-migration educational level, pre-migration work experience, local language proficiency, frequency of contact with natives, Extraversion, GMA, and work centrality) contributed to predicting one or more of these employment outcomes. There were some differences in the predictive validity of the individual-difference factors between refugee groups (i.e., based on age, sex, and nationality). These findings offer several important insights into refugees' workforce integration and provide implications for improving refugees' integration trajectories.

KEYWORDS

4

refugees, individual-difference factors, workforce participation, employment, highest hourly wage

A HINDERING AND FACILITATING INDIVIDUAL-DIFFERENCE FACTORS FRAMEWORK FOR PREDICTING REFUGEES' WORKFORCE PARTICIPATION

According to the United Nations Refugee Agency (UNHCR, 2022a), the number of refugees worldwide has reached the highest level on record. In the first few months of 2022, 26.6 million people fled their home country as a result of armed conflict, violence, persecution, or human rights violations (UNHCR, 2022b). This year's number is expected to exceed the total number of refugees in 2021 (27.1 million) and is more than double the annual number of refugees (10.5 million) the world saw a decade ago (UNHCR, 2022b). In contrast to other migrants, who leave their home countries because of, for example, economic considerations, refugees are those migrants who are forced to leave their home countries due to unforeseen, life-threatening circumstances (Dustmann et al., 2017). Furthermore, despite initially strong return intentions (UNHCR, 2022a), refugees often cannot return to their home countries due to well-founded fears of the prevailing circumstances (Lee et al., 2020). Thus, the massive global increase in refugees, and the enduring instabilities in the countries they have fled, caused an acute and growing need to assist refugees in rebuilding their lives in the countries where they sought protection and security.

Early employment is a key aspect – if not the most important aspect (Bloch, 2002; Colic-Peisker, 2005) – for refugees to rebuild their lives (Feeney, 2000). However, across the globe many refugees face long-term unemployment (Donato & Ferris, 2020; Fasani et al., 2022). For example, in the Netherlands, where the present research was conducted, only 19% of the refugees who received a residence permit in 2014 found a job within three years (CBS, 2021). A low employment rate is associated with several negative consequences. For the receiving country, refugees' unemployment results in high economic costs (Aiyar et al., 2016; Stenberg & Westerlund, 2008), higher levels of anti-immigrant attitudes (Chandler & Tsai, 2001), and less readiness to take up refugees in the future (Seidelsohn et al., 2020). For refugees, unemployment results in stress, poor physical health, bad health-related habits (Bambra & Eikemo, 2009; Cohen et al., 2007; Dettenborn et al., 2010), and poor mental health (McKee-Ryan et al., 2005; Paul & Moser, 2009). To prevent these negative consequences, it is essential to gain insights into the factors that influence refugees' quick and successful workforce participation.

In the present research, we concentrate on refugees' individual-difference factors because the personnel psychology literature has shown that these factors play a crucial role in workforce participation (e.g., Van Hooft et al., 2021). Furthermore, insights into how such factors influence workforce participation will allow receiving countries to identify specific assistance needs, develop customized and evidence-based integration trajectories, and identify opportunities for refugees to cultivate and train relevant traits and skills to promote their workforce participation. Although individual-difference factors are relatively stable attributes that individuals bring to the job (Sackett et al., 2017), we emphasize that these factors are not unchangeable; one's educational level and (language) skills are examples of individual-difference factors that are relatively stable but can alter by investing considerable time and effort.

The personnel psychology literature has offered various theoretical frameworks and metaanalyses of individual-difference factors (e.g., demographics, human and social capital variables, personality traits, and attitudinal factors) that successfully predict native-born job seekers' or economic migrants' workforce participation (e.g., Kanfer et al., 2001; Van Hooft et al., 2021), employability (Harari et al., 2021), and underemployment (Guerrero & Rothstein, 2012). However, these frameworks are unlikely to generalize to refugee populations for two reasons. First, refugees are forced to leave their home countries and they often do not have a choice in determining their receiving countries (Koser, 2007). Therefore, compared to native-born job seekers or economic migrants, the educational level, work experience, and (language) skills of refugees are less likely to match the needs of the job market (Lee et al., 2020). Second, refugees face a number of unique barriers compared to native-born job seekers or economic migrants, including health- and family-related challenges (e.g., Agbényiga et al., 2012). For example, due to dire circumstances in their home countries and perilous journeys to their receiving countries, refugees have a relatively high prevalence of post-traumatic stress disorder (PTSD) symptoms (Blackmore et al., 2020) and have often been forced to leave family members behind (e.g., Keita & Schewe, 2021). Although PTSD symptoms (Sienkiewicz et al., 2020) and family-related challenges (Beddoes & Pawley, 2014) are known to affect employment-related outcomes, such characteristics are not included in current frameworks within the personnel psychology literature.

To the best of our knowledge, there have been only two prior attempts at developing a theoretical framework of individual-difference factors for understanding workforce participation of refugees specifically (Boss et al., 2021; Lee et al., 2020; both will be explained in the theory section). However, the predictive value of these frameworks is unknown as neither of them has been quantitatively evaluated. More importantly, both frameworks comprise different individual-difference factors, suggesting that both lack important predictors of workforce participation. For example, the framework of Boss et al. (2021) does not include demographics (e.g., age, sex) and the framework of Lee et al. (2020) does not include any factors related to refugees' acquired human capital in their home countries (e.g., work experience, education), while prior research indicates that these factors play a significant role in refugees' workforce participation (De Vroome & Van Tubergen, 2010; Hunkler et al., 2021; Khawaja & Hebbani, 2018). Furthermore, although both frameworks identify PTSD symptoms, neither framework includes family-related challenges or psychological characteristics such as personality and general mental ability (GMA). The omission of psychological characteristics is especially surprising, as these characteristics are important predictors of workforce participation among native-born job seekers (Hogan et al., 2013; Kanfer et al., 2001), and of refugees' broader integration in their host societies (Echterhoff et al., 2020). Hence, there is a clear need for the development and actual testing of an integrated and more comprehensive theoretical framework.

In this study, we aim to advance our understanding of refugees' workforce participation by developing an integrative theoretical framework of individual-difference factors to predict workforce participation (see Figure 1). Importantly, we test the predictive validity of (the individual-difference factors within) our framework among a large group of recently arrived refugees living in the Netherlands. To this end, we relate these factors to two criteria: (1) employment (i.e., the actual occurrence and speed of finding a job), and (2) longest employment duration (i.e., the longest consecutive employment duration). These criteria are in concert with the definition of employability – that is "the ability to gain and the ability to maintain a job" (Fugate et al., 2004; Hogan et al., 2013). Furthermore, research has shown that many refugees have temporary jobs in the early years after resettlement (CBS, 2021), which makes longest employment duration a crucial indicator of refugees' workforce participation.

We aim to make three contributions. First, we add to the literature on refugees' workforce participation by integrating the sociology and economics literature on refugees' workforce participation with the psychology (i.e., personnel psychology) literature on job search and employability. Current refugee research is fragmented; different individual-difference factors are examined within different disciplines (Lee et al., 2020), and these disciplines have rigid boundaries in terms of which characteristics are included or excluded. While the discipline of sociology has focused on socio-demographics (e.g., Bakker et al., 2017; Khawaja & Hebbani, 2018), the discipline of economics has focused on factors related to acquired human and social capital (e.g., Chiswick & Miller, 2009; Friedberg, 2000). Complementary to these disciplines, the psychology literature indicates the importance of studying psychological characteristics in relation to workforce participation. As this latter discipline has exclusively focused on predicting employment of new labor market entrants, job-to-job seekers, and job losers (for meta-analyses, see Kanfer et al., 2001; Van Hooft et al., 2021), established frameworks within this discipline lack important individual-difference factors that are specifically relevant for refugees (e.g., language, health- and family-related challenges). Our study will break these disciplinary boundaries by integrating insights from sociology, economics, and psychology into one comprehensive theoretical framework containing a wide range of individual-difference factors that may either hinder or facilitate refugees' path to finding work. This framework will allow scholars to adequately build on existing knowledge and to propose more effective interventions to increase refugees' workforce participation.

70 | CHAPTER 4
Second, we contribute to the literature on refugees' workforce participation by actually testing our integrative theoretical framework among a large group of recently arrived Syrian and Eritrean refugees living in the Netherlands. The Syrian refugees had mostly fled their country because of the civil war, and the Eritrean refugees had mostly fled their country because of the obligatory military service, oppression, violence, and poverty (for more information about the sociodemographic and cultural differences between Syria and Eritrea and the background of these refugee groups, we refer readers to the Appendix, and to Dagevos et al. [2018] and Sterckx et al. [2018]). Given the current tight labor market, speeding up the workforce participation of these two relatively large (UNHCR, 2021) yet understudied refugee groups is not only important for the wellbeing of individuals, but also for organizations and societies at large.

Finally, our study offers an important methodological contribution. Many studies on refugees' workforce participation have adopted cross-sectional research designs with self-reported employment data (e.g., De Vroome & Van Tubergen, 2010), which limits causal inferences. The present study addresses this issue by adopting a time-lagged research design over a five-year time span with employment data from the Central Bureau of Statistics (CBS), which are objective, rich (recorded monthly), and unbiased (i.e., there is no selective participant dropout in this study that could bias the results; see AkI et al., 2012; Asendorpf et al., 2014).

4





A HINDERING AND FACILITATING INDIVIDUAL-DIFFERENCE FACTORS FRAMEWORK

In the domain of refugee employment and workforce integration, two earlier frameworks have been developed, namely by Lee et al. (2020) and by Boss et al. (2021). Both frameworks aim to include all conceivably relevant factors potentially affecting refugees' workforce participation. Lee et al.'s framework is based on a literature review of earlier publications in this domain and groups relevant factors into three categories: 1) institutional factors, focusing on country-level issues such as cross-country differences in immigrant regulations, asylum policies, socio-political climate, and public sentiment towards immigrants, 2) organizational factors, pertaining to problems of exclusion and isolation in the workplace, discrimination, but also whether an organization offers volunteering opportunities to refugees with the aim of providing them with local workplace experiences, and 3) individual-level factors, encompassing refugee motivation to integrate, their local language competency, age and sex, their social networks, and psychological responses (e.g., PTSD symptoms and anxiety).

Whereas Lee et al.'s framework is extensive, ranging from country-wide to individualdifference factors, Boss et al. (2021) proposed a framework that specifically focuses on refugees' job search process. They distinguish about 50 factors related to this process, including diversity cues and available job openings within organizations, local language proficiency, mental health and financial needs of refugees, person-organization and person-job fit, self-promotion behaviors and quality of answers during job interviews, but also perceptions of discrimination by the job seekers, and employer exploitation. From a small qualitative study in Canada among seven Syrian refugees and six service providers preparing Syrian refugees for the workforce, Boss et al. derived that local language proficiency, credential recognition, having work experience in the receiving country (a "catch 22"), cultural incongruencies in the job application process, employer exploitation in unethical work situations, and mental health were regarded as relevant for successful employment.

Both frameworks include a vast number of factors, of which PTSD symptoms (or mental health) form a distinctive issue for refugees. Neither framework includes refugee-specific family-related challenges such as having left behind family members in one's country of origin. Furthermore, an important group of variables from the perspective of personnel psychology has been neglected in both frameworks, namely psychological characteristics such as personality and GMA, which are related to workforce participation (Van Hooft et al., 2021; Vélez-Coto et al., 2021) and success in the job (Sackett et al., 2022). Finally, Lee et al.'s framework does not include any factors related to refugees' acquired human capital in their home countries (e.g., work experience, education), which play an important role in refugees' workforce participation (e.g., Chiswick & Miller, 2009; Friedberg, 2000), whereas

4

72 | CHAPTER 4

the framework of Boss et al. (2021) does not include refugee demographics (e.g., age, sex). As both frameworks are tremendously elaborate, it is extremely difficult to empirically put them to the test.

Given the importance of individual-difference factors within personnel psychology (e.g., Van Hooft et al., 2021), the present study's aim is to develop and test an integrative framework of individual-difference factors to predict refugees' workforce participation. Within the category of individual-level factors, Lee et al.'s (2020) framework distinguishes between demographics (age, sex), local language proficiency, social networks, psychological responses (PTSD symptoms), and motivation to find work and integrate. Both Lee et al.'s (2020) and Boss et al.'s (2021) frameworks focus on obstacles and challenges faced by refugees, with Lee et al. seeing a systemic barrier to refugee workforce integration ('the canvas ceiling') and Boss et al. perceiving broad barriers to employment. However, from the domain of positive psychology (cf. Seligman & Csikszentmihalyi, 2000; Titova et al., 2018), it is valuable to not only recognize hindering factors, but to also consider resources that refugees may have and employ towards thriving and reaching success.

To structure and extend Lee et al.'s (2020) and Boss et al.'s (2021) thinking on relevant individual-difference factors, we explicitly distinguish between a group of potentially hindering and a group of potentially facilitating factors. One's impeding demographics and health- and family-related challenges belong to the group of hindering characteristics. The group of facilitating characteristics comprises an individual's acquired human and social capital and work-relevant traits. Both groups of factors are elaborated upon in the following.

HINDERING INDIVIDUAL-DIFFERENCE FACTORS Impeding Demographics

Demographic variables have mostly been studied in the sociology literature (Bakker et al., 2017; Khawaja & Hebbani, 2018), and are associated with particular challenges due to discrimination and stereotyping, which can affect one's employment prospects (e.g., Colic-Peisker & Tilbury, 2006; Kofman, 2014). One relevant impeding demographic variable is age. Reemployment after job loss is particularly difficult for older workers (Klehe et al., 2012). Some challenges for older people include the lack of modern job skills (Fossum et al., 1986), low familiarity with modern job-search methods (Gibson et al., 1993; Westaby & Braithwaite, 2003), and employer hiring preferences for younger workers (Ahmed et al., 2012; Derous & Decoster, 2017). A recent meta-analysis indeed has shown that age is negatively related to reemployment success and positively related to unemployment duration (Wanberg et al., 2016). Furthermore, a UK study found that refugees' age is positively related to the number of barriers they face when searching for employment (Shiferaw & Hagos, 2002). Similarly, a Swedish study showed a positive correlation between refugees' age and their unemployment duration (Lundborg, 2013). Altogether, there is robust evidence that age

is negatively related to (re)employment success. Therefore, we propose the following hypothesis:

Hypothesis 1a (H1a): Age is negatively related to refugees' workforce participation.⁸

The next relevant hindering demographic variable is being a woman. Cross-national statistics show a gender gap in refugees' workforce participation, with men showing higher rates of employment than women (Bisello & Mascherini, 2017; Eurostat, 2021b). For example, in the Netherlands, the gender employment gap was 15.1% in 2020 among non-EU-born migrants, compared to 7.8% among native citizens (Eurostat, 2021b). Research has shown that this gender gap is still present 15 years after migration (Bakker & Dagevos, 2017; Maliepaard et al., 2017). Several factors have been offered as explanations for refugees' gender gap in workforce participation, including traditional gender roles according to which men are the income providers (Baranik, 2020; Razenberg et al., 2018), as well as women having lower education levels (Barslund et al., 2017), less work experience, and a confined social network (Razenberg et al., 2018; for a review of the barriers, see Howes et al., 2018). In line with these empirical findings, we propose the following hypothesis:

Hypothesis 1b (H1b): Workforce participation is lower among female refugees than male refugees.

The third demographic variable which may cause challenges for refugees is the cultural incongruence between one's home country and the receiving country. A lack of familiarity with many aspects of the new culture may hinder the chances of getting employment. As expressed in the per capita GDP ranking which ranges from 1 to 192, the living standard in Eritrea (rank 179) is considerably lower than in Syria (rank 137) and thus more incongruent than Syria with the prosperous country of the Netherlands (rank 13). The legatum prosperity index (Lind, 2014) reflects these country differences in the 2021-indicators such as living conditions, economy, health, and education. Having a nationality that reflects a more different cultural background may further slow down one's labor participation, leading us to the expectation that:

Hypothesis 1c (H1c): Workforce participation is lower among Eritrean refugees than Syrian refugees.

⁸ In the hypotheses, we use the generic term *workforce participation* to refer to two dependent variables (DVs): employment and the longest employment duration. Since empirical work on the latter criterion is scarce, we have not made any a priori distinctions in our hypotheses.

Health- and Family-Related Challenges

Distinctive for the refugee context are two demanding themes, namely health-related challenges (physical health problems and PTSD symptoms) and family-related challenges. There is ample evidence that refugees tend to have worse physical and mental health than other migrants and natives of a country (Donato & Ferris, 2020; Ruiz & Vargas-Silva, 2018). Using a dataset from the UK, Ruiz and Vargas-Silva (2018) for instance describe that refugees are more likely than both other groups to report that the type of work they can do and the amount of work they can engage in are reduced by health problems, which may contain physical health problems such as diabetes, rheumatism and heart problems, but also mental health problems. Related to mental health, several reviews and meta-analyses on the prevalence and risk factors of mental wellbeing among refugees reveal that PTSD is more common among refugees than among native populations (Blackmore et al., 2020; Charlson et al., 2019). PTSD symptoms such as intrusive memories and heightened arousal levels are relevant to investigate in the context of employment, as these might make someone less able to work. PTSD symptoms among refugees have been linked to substance abuse (Harris et al., 2019; Horyniak et al., 2016) and violent crime (Couttenier et al., 2019), and some authors have argued that trauma can affect refugees' workforce participation (e.g., Brell et al., 2020). In fact, the average duration of PTSD has been found to exceed seven years in US samples (Kessler, 2000), and has therefore the potential to impede the socio-economic integration of refugees in the early years after resettlement. Donato and Ferris (2020) state that most data sources about refugees do not include measures relating physical and mental health with employment. Nonetheless, to our knowledge, to date three studies have empirically examined the link between PTSD symptoms and refugees' workforce participation. One study reported a weak negative effect (Cheng et al., 2021b), while two other studies revealed no direct effect. Yet, these latter studies have several methodological limitations which could question their findings, such as a cross-sectional research design, a small sample size, and a short residence duration (Hunkler & Khourshed, 2020; Wright et al., 2016). Given the high prevalence of physical health problems and PTSD symptoms among refugees and some evidence of a negative effect of PTSD symptoms on refugees' workforce participation, we propose the following hypothesis:

Hypothesis 2a (H2a): Physical health problems are negatively related to refugees' workforce participation.

Hypothesis 2b (H2b): PTSD symptoms are negatively related to refugees' workforce participation.

2018), refugees with family members still in the home country from which they had fled, will ruminate and worry about their loved ones. They may be wondering whether their family still is alive, what their whereabouts are, and what they are doing. As such family-related challenges are understood to affect employment-related outcomes (Beddoes & Pawley, 2014), we anticipate that:

Hypothesis 2c (H2c): Workforce participation is lower among refugees who have versus those who have not left a spouse or children in their country of origin.

FACILITATING INDIVIDUAL-DIFFERENCE FACTORS Acquired Human and Social Capital

Generally, human and social capital constitute differences in individuals' investment decisions (i.e., choices that provide resources for the labor market), and these differences are the main predictors of employment outcomes (Becker, 1964). Human capital refers to individuals' educational, personal, and professional experiences that can enhance career attainment (Becker, 1964), and social capital refers to social relationships that provide access to various types of resources (Caspi et al., 1998; Portes, 2000).⁹ Individual differences in human and social capital determine one's occupational success (e.g., Borjas & Chiswick, 2019; Lancee, 2012). In line with earlier empirical work (e.g., De Vroome & Van Tubergen, 2010), we next describe the human and social capital variables that are relevant to refugees' workforce participation.

Educational attainment has been considered an important form of human capital (e.g., McArdle et al., 2007; Tharmaseelan et al., 2010), also for refugees (Chiswick & Miller, 2001; Hunkler et al., 2021). However, refugees' pre-migration education is often not easily transferable across national contexts (Chiswick & Miller, 2008; Friedberg, 2000). For instance, refugees who were highly educated in their country-of-origin face challenges with quickly finding employment that corresponds to their educational level (Cheng et al., 2021a). Nevertheless, refugees' pre-migration educational attainment level has been found to positively influence workforce participation (e.g., Hartog & Zorlu, 2009; Hunkler et al., 2021), although the effects are stronger for post-migration compared to pre-migration educational attainment (De Vroome & Van Tubergen, 2010; Kanas & Van Tubergen, 2009). Given the relatively short local length of stay of the refugees in our sample, we only focus on pre-migration educational attainment. We therefore propose the following hypothesis:

A highly important issue for refugees is family-related challenges. Refugees have frequently been compelled to leave family members behind (e.g., Keita & Schewe, 2021). In contrast to refugees who live with their nuclear family in the receiving country (Gambaro et al.,

⁹ Some authors have argued that human capital includes psychological traits (e.g., Caspi et al., 1998; Fugate et al., 2004), whereas other authors place psychological traits in a separate category (e.g., Judge et al., 1995; Ng et al., 2005). In line with the principles of the human capital theory (Becker, 1964), we argue that human capital constitutes investment decision characteristics, and hence, psychological traits (including GMA) are not included in the capital perspective.

Hypothesis 3a (H3a): Pre-migration educational attainment is positively related to refugees' workforce participation.

Another important form of human capital is work experience. Skills acquired in the workplace can positively affect future employment, as work experience is valued by employers (Chiswick & Miller, 2009). Although work experience in the country of origin could be less compatible with the host society's labor market (e.g., Friedberg, 2000), we argue that premigration work experience still is valuable. Indeed, there is evidence that pre-migration work experience positively predicts refugees' workforce participation (e.g., Cheng et al., 2021a, 2021b). One study reported a negative effect, but this was probably found because the authors did not control for age (Chiswick & Wang, 2016). Altogether, we propose the following hypothesis:

Hypothesis 3b (H3b): Workforce participation is higher among refugees with versus without pre-migration work experience.

The last form of human capital that is highly relevant to refugees is local language proficiency. Local language proficiency is an essential resource for fostering refugees' incorporation into society (Portes & Rumbaut, 2006). Language skills are essential in most professions (Chiswick & Miller, 2001), and employers therefore are more likely to hire employees with proper language skills (Bertone, 2004). Moreover, better local language skills increase refugees' range of jobs in the labor market for which they are qualified (Chiswick & Miller, 2001). Correspondingly, refugees with better local language proficiency have higher levels of job-search self-efficacy (Pajic et al., 2018). Given these findings, it is unsurprising that local language proficiency has been positively associated with migrants' and refugees' employment success in several countries, such as the US (e.g., Chiswick & Wenz, 2006), Canada (e.g., Frank, 2013), Australia (e.g., Cheng et al., 2021b; Guven & Islam, 2015; Waxman, 2001), the UK (Bloch, 2002; Dustmann & Fabbri, 2003), and also in non-English speaking countries such as Israel (Chiswick & Repetto, 2001; Chiswick et al., 2020), Germany (Aldashev et al., 2009), and the Netherlands (Chiswick & Wang, 2016; Zorlu & Hartog, 2018). Given the well-documented importance of local language proficiency for refugees' socioeconomic integration, we propose the following hypothesis:

Hypothesis 3c (H3c): Local language proficiency is positively related to refugees' workforce participation.

Next, we argue that refugees' social capital – or bridging social capital, referring to the amount of refugees' contact with native individuals (see Eisnecker & Schacht, 2016; Kanas et al., 2012) – positively affects their workforce participation (Portes, 2000). This notion is referred to by Lee et al. (2020) as one's social network. Refugees' social contact with

natives is likely to be beneficial in terms of employment prospects, as natives have more knowledge about the local labor market (Gericke et al., 2018) and as they more often are employed and have prestigious jobs (Kanas & Van Tubergen, 2009). Indeed, a Dutch study has shown that the odds of employment are higher among refugees who maintain more relationships with Dutch natives (De Vroome & Van Tubergen, 2010), which effect has been replicated in several countries (e.g., Drever & Hoffmeister, 2008). In line with the theory and findings described, we propose the following hypothesis:

Hypothesis 3d (H3d): Frequency of contact with natives is positively related to refugees' workforce participation.

Work-Relevant Traits

Finally, we investigate a group of facilitating factors that have not received much attention until now in refugee-related research, namely one's work-relevant traits. Work-relevant traits enabling finding and keeping a job can be perceived through the lens of the rewarding-ablewilling (RAW) model (Hogan et al., 2013). This perspective explicates that being *rewarding* to work with, being *able* to work, and being *willing* to work constitute three (interpersonal, cognitive, and intrapersonal) individual core attributes for finding and keeping a job, as these attributes are valued by employers and promote employees' desirable work behaviors (Hogan & Chamorro-Premuzic, 2015; Hogan et al., 2013). Based on the premise that the predictor space of occupational performance should contain measures of personality, cognitive ability, and values (Ackerman & Heggestad, 1997; Hogan & Chamorro-Premuzic, 2015), we describe seven psychological traits that fit under the umbrella of the three core attributes and that have shown positive effects in organizational contexts. As only a few studies have examined the predictive validity of psychological traits for the employment outcomes of refugees, most studies reviewed below have been conducted on native and Western samples.

Rewarding. In most jobs, individuals work with others or as part of a team (Levy & Cannon, 2016). Therefore, being rewarding to work with and getting along and working well with others is regarded as an important attribute in the workplace (also, see Sackett & Walmsley, 2014). In fact, when making employment decisions, employers often focus more on applicants' interpersonal skills than on their academic credentials (Taylor, 2006). The strongest personality correlates of the rewarding attribute are Agreeableness, Extraversion, and Emotional Stability (Boudreaux et al., 2022).

Agreeableness reflects individuals' selflessness, cooperativeness, helpfulness, and flexibility (Digman, 1990; McCrae & Costa, 2008). Agreeableness is one of two interpersonal traits (next to Extraversion; Trapnell & Wiggins, 1990) and can lead to positive interpersonal affect (Niven et al., 2012). Correspondingly, several studies have shown that Agreeableness relates positively to the likeability ratings of peers (De Vries et al., 2020; Wortman &

4

Wood, 2011). Agreeableness has also been found to be important in team settings where collaboration is required (Hogan & Holland, 2003; Witt et al., 2002). These research findings demonstrate Agreeableness' significance to getting along and working well with others, and thus fit the rewarding attribute. There is also some empirical evidence that Agreeableness is positively related to preparatory job-search behavior (i.e., the gathering of job-search information) and job-search intensity (Baay et al., 2014; Van Hooft et al., 2021). Altogether, in line with the RAW model, Agreeableness is anticipated to be important for refugees' workforce participation:

Hypothesis 4a (H4a): Agreeableness is positively related to refugees' workforce participation.

Extraversion refers to being sociable, open to others, assertive, active, and to like excitement (Costa & McCrae, 1992). One study found that, of the Big Five personality traits, Extraversion is the strongest predictor of social skills (Ferris et al., 2001). Correspondingly, Extraversion is the best personality predictor of interpersonal and organizational citizenship behavior (Pletzer et al., 2021). Moreover, Extraversion is positively related to organizational commitment (Wilmot et al., 2019) and job performance – particularly in jobs that require interpersonal skills (Barrick & Mount, 1991). These research findings indicate that extraverted people are rewarding to deal with. Furthermore, a meta-analysis showed that Extraversion is related to employment status (Van Hooft et al., 2021). Altogether, in line with the RAW model, we anticipate that Extraversion has a positive effect on refugees' workforce participation:

Hypothesis 4b (H4b): Extraversion is positively related to refugees' workforce participation.

The last trait relevant to the rewarding attribute is Emotional Stability (or its opposite: Neuroticism), which reflects the tendency to be confident, secure, and steady (Judge & Bono, 2001). Individuals high in Emotional Stability are more cooperative and have high-quality interactions with others in the work setting (LePine & Van Dyne, 2001). Emotional stability is a predictor of job satisfaction (Judge & Bono, 2001), job performance (Connelly & Ones, 2010; Hogan & Holland, 2003), task performance and contextual performance (Judge et al., 2013), and overall career success (Judge et al., 1999). Based on these research findings, Emotional Stability is likely to fit the rewarding attribute. Furthermore, a meta-analysis showed that Emotional Stability is positively related to employment status and the number of job offers (Kanfer et al., 2001). In line with the RAW model, we thus anticipate that Emotional Stability has a positive effect on refugees' workforce participation:

Hypothesis 4c (H4c): Emotional Stability is positively related to refugees' workforce participation.

Able. The second attribute of the RAW model is being able to do the job. This attribute is related to one's GMA (Boudreaux et al., 2022), which is defined as "the ability to understand complex ideas, to adapt effectively to the environment, to learn from experience, to engage in various forms of reasoning, to overcome obstacles by taking thought" (Neisser et al., 1996, p. 77), and determines an individual's capability to learn and perform well on tasks (Schmidt & Hunter, 1998, 2004). In fact, GMA is a positive predictor of several occupational success indicators, including attained occupational level (Schmidt & Hunter, 2004), training success (Salgado et al., 2003), and work performance (Schmidt & Hunter, 1998, 2004). Additionally, several prospective studies in the US (Caspi et al., 1998; Fergusson et al., 2005; Herrnstein & Murray, 1994), Sweden (Lindqvist & Vestman, 2011), and the UK (Egan et al., 2017) have found that GMA, as assessed in childhood or adolescence, is a positive predictor of the probability and duration of employment in adulthood. A recent meta-analysis has also shown a positive link between GMA and employment (Vélez-Coto et al., 2021). Furthermore, in a German sample, GMA was found to positively predict reemployment success (Gnambs, 2017). Altogether, GMA is a relevant trait for being able to do the job. Hence, in line with the RAW model, we anticipate that GMA has a positive effect on refugees' workforce participation:

Hypothesis 4d (H4d): GMA is positively related to refugees' workforce participation.

Willing. The last attribute of the RAW model is being willing (i.e., motivated) to work, and refers to one's proclivity to work hard and to produce high-quality results in a timely fashion (Boudreaux et al., 2022). Conscientiousness and work centrality map onto the willing attribute (Boudreaux et al., 2022). Conscientiousness pertains to being organized, responsible, and industrious (Lee & Ashton, 2004), and is related to different forms of performance motivation (Judge & Ilies, 2002). Furthermore, Conscientiousness is the personality trait with the highest predictive validity for occupational success (Roberts et al., 2007). For example, conscientious employees perform better at work (Barrick & Mount, 1991), engage more frequently in organizational citizenship behaviors (Lee et al., 2019; Pletzer et al., 2020). In one study in the UK, adolescent Conscientiousness was a predictor of lower lifetime unemployment (Egan et al., 2017). Hence, in line with the RAW model, we anticipate that Conscientiousness has a positive effect on refugees' workforce participation:

Hypothesis 4e (H4e): Conscientiousness is positively related to refugees' workforce participation.

Work centrality is considered to be a universal predictor of occupational success (Smith, 1994). It refers to the general importance that work has in an individual's life at any given time (MOW, 1987). High work centrality has been associated with better job performance

80 | CHAPTER 4

4

(Diefendorff et al., 2002; Sharabi & Harpaz, 2010), more job involvement (Diefendorff et al., 2002), more commitment to the organization (Kalleberg & Mastekaasa, 2001), less absenteeism and turnover (Kostek, 2012), and longer working hours (Sharabi & Harpaz, 2007; Snir & Harpaz, 2002). Furthermore, unemployment causes distress to people who have high levels of work centrality (Paul & Moser, 2009). Correspondingly, meta-analytic work has also shown that work centrality is a robust positive predictor of job-search behaviors and employment (Kanfer et al., 2001; Van Hooft et al., 2021). Taken together, work centrality is relevant to the willing attribute. Hence, in line with the RAW model, we anticipate that work centrality has a positive effect on refugees' workforce participation:

Hypothesis 4f (H4f): Work centrality is positively related to refugees' workforce participation.

MODEL INTEGRATION

Each of the four above groups of individual-difference factors (i.e., impeding demographics, health- and family-related challenges, acquired human and social capital, and work-relevant traits) contains a separate set of predictors. Although each perspective draws on different underlying mechanisms, the predictors will share some mutual variance. For instance, psychological traits might affect one's personal investment decisions (e.g., GMA could influence local language learning and Extraversion could influence the frequency of contact with natives) or human capital might be related to demographics (e.g., work experience is positively associated with age). Therefore, to test the hypotheses presented above, we not only analyze each group of predictors in independent statistical models, but also include the variables of each group in one model to reveal those individual-difference factors that explain the most unique variance in refugees' workforce participation. Given that this kind of research has not previously been conducted with psychological traits, we test this model exploratorily:

Research Question 1 (RQ1): Which individual-difference factors are the strongest predictors of refugees' workforce participation?

Highest Hourly Wage

Finally, wages have frequently been studied as an employment outcome among migrants and refugees (e.g., Chiswick & Wang, 2016) as they are important for establishing one's financial independence (Avrić et al., 2019) and provide an objective indicator of employment quality (Vinokur & Schul, 2002). Hence, we will also examine the predictive validity of the individual-difference factors for the highest hourly wage. However, in the early years of resettlement, employed refugees usually have low-quality jobs with low wages (e.g., Kosny et al., 2020). Given that the current study is conducted over a relatively short time span, we expected that there is relatively little variance in the wages of the refugees in our data. Hence, we formulated a research question rather than hypotheses about the predictive validity of individual-difference factors for highest hourly wage.

Research Question 2 (RQ2): Which individual-difference factors are the strongest predictors of refugees' highest hourly wage?

METHOD

PARTICIPANTS AND PROCEDURE

Within a few months to a few years after arriving in the Netherlands, refugees were invited by municipalities to complete an online psychological assessment that is aimed to support their integration into the Dutch society. The assessment consists of sociodemographic questions and several psychological and aptitude measures.¹⁰ The criteria for participation in the assessment were being at least 18 years old, having a residence permit, and being literate. The assessment is offered by a consultancy agency that focuses on the development and the assessment of psychological instruments for work- and educational-related purposes. The consultancy agency trained the counselors of the municipalities on how to interpret the assessment results and on how to use these results to support refugees in their socioeconomic integration. Participation in the assessment was voluntary, and although exact numbers about the response rate are unavailable, assessment administrators estimated it to be at least 95%. The assessments were usually administered in distraction-free rooms in the municipality's town hall and took about two hours to complete. During the assessments, one or more staff members were present, and no communication with other candidates was allowed. Participating in the assessment was not monetarily compensated. This research was approved by the university's ethics committee.

The assessments started in June 2016 and were still being carried out by the time of this research. However, in October 2019, all available assessment data were retrieved. In February 2021, the variables zip code (at the time of the assessment), date of birth, and sex were used to match the assessment data to the data from the CBS. The CBS data were available from January 2014 to July 2020 and were registered monthly. From the CBS dataset, we used the data on employment, education, physical health problems (general practitioner consultation costs), urbanization of the municipality (place of residence), and the dates of arrival in the Netherlands and receiving a residence permit.

¹⁰ The assessment consists of sociodemographic questions, two GMA subtests, scales for self-reported competencies and personality traits, a scale of work motivation and attitudes, a measure of psychological distress, a post-traumatic stress disorder checklist, and a Dutch and an English language proficiency test. In the present research, we only used the measures that are relevant for the research purpose. The complete instruments of the consultancy agency are not available for proprietary reasons.

We only used the data of refugees from Syria and Eritrea, as these groups comprise most subjects in the dataset (respectively 62.6% and 17.4%). Additionally, we excluded participants from the analyses who (a) passed away, (b) left the Netherlands, (c) did not receive a residence permit, (d) completed a different version of the assessment, or (e) who received their residence permit after July 2017 (since their employment data are not available far enough into the future for conducting the analyses). For details about the data exclusion, see Supplementary Material. After excluding these participants, the final dataset consisted of 2711 refugees (n_{syria} = 1867, $n_{Eritrea}$ = 844). The majority of the refugees was male (1992; 73.5%). Detailed descriptive statistics of the study variables for the whole sample and the two nationalities and sexes separately are presented in Table 1. A visual representation of the study events in time is presented in Figure 2.



edian obs

Table 1. Overall and Refugee Group Specific Desc	criptive Statistics of Study Vai	iables	Table 1. Continued.				
	Total (N = 2711)	Syrians (n = 1867)	Eritreans (n = 844)	Men (<i>n</i> = 1992)	Women (<i>n</i> = 719)	d or φ Nationality	d or φ Sex
Covariates							-
Year RP (<i>Mdn</i>) 2014 2015 2016 2017	2015 359 (13.2%) 1024 (37.8%) 1104 (40.7%) 224 (8.3%)	2016 180 (9.6%) 596 (31.9%) 907 (48.6%) 184 (9.9%)	2015 179 (21.2%) 428 (50.7%) 197 (23.3%) 40 (4.7%)	2015 289 (14.5%) 778 (39.1%) 795 (39.9%) 130 (6.5%)	2016 70 (9.7%) 246 (34.2%) 309 (43.0%) 94 (13.1%)	29*	.12*
Duration betw arrival and RP (months)	2.83 (3.27)	2.80 (3.63)	2.91 (2.30)	3.23 (2.86)	1.71 (3.99)	0.04	-0.44*
Duration betw RP and assessment (months)	19.13 (11.50)	18.53 (11.79)	20.44 (10.73)	19.60 (11.30)	17.81 (11.95)	0.17*	-0.15*
Urbanization 1 2 3 4 5	4.15 (1.20) 113 (4.2%) 283 (10.4%) 238 (8.8%) 517 (19.1%) 1560 (57.5%)	4.09 (1.21) 84 (4.5%) 203 (10.9%) 186 (10.0%) 382 (20.5%) 1012 (54.2%)	4.30 (1.15) 29 (3.4%) 80 (9.5%) 52 (6.2%) 135 (16.0%) 548 (64.9%)	4.16 (1.20) 83 (4.2%) 210 (10.5%) 182 (9.1%) 356 (17.9%) 1161 (58.3%)	4.15 (1.18) 30 (4.2%) 73 (10.2%) 56 (7.8%) 161 (22.4%) 399 (55.5%)	0.18*	-0.01
Followed education (yes)	494 (18.2%)	300 (16.1%)	194 (23.0%)	383 (19.2%)	111 (15.4%)	.08*	04*
Having social support (yes)	1766 (65.2%)	1375 (73.7%)	391 (46.4%)	1317 (66.2%)	449 (62.6%)	27*	03
Impeding Demographics ^a							
Age at receiving a RP	29.27 (9.18)	31.04 (9.77)	25.36 (6.14)	29.40 (9.18)	28.92 (9.19)	-0.70*	-0.05
Health- and Family-Related Challenges							
Physical health problems ^b	34.53 (43.6)	37.87 (48.15)	27.14 (29.83)	28.15 (38.83)	52.20 (50.58)	-0.27*	0.53*
PTSD symptoms (0-80)	14.89 (13.67)	16.40 (14.42)	11.55 (11.17)	15.06 (13.99)	14.41 (12.75)	-0.38*	-0.05
Spouse/children in country of origin (yes)	818 (30.9%)	503 (27.7%)	315 (37.8%)	611 (31.4%)	207 (29.5%)	.10*	02
Acquired Human and Social Capital							
Pre-migration educational level (0-4) No or basic education High school Associate's Bachelor's Master's	1.66 (1.21) 406 (15.0%) 1102 (40.6%) 475 (17.5%) 456 (16.8%) 272 (10.0%)	1.90 (1.11) 61 (3.3%) 849 (45.5%) 386 (20.7%) 350 (18.7%) 221 (11.8%)	1.13 (1.24) 345 (40.9%) 253 (30.0%) 89 (10.5%) 106 (12.6%) 51 (6.0%)	1.71 (1.22) 275 (13.8%) 799 (40.1%) 366 (18.4%) 329 (16.5%) 223 (11.2%)	1.53 (1.17) 131 (18.2%) 303 (42.1%) 109 (15.2%) 127 (17.7%) 49 (6.8%)	-0.65*	-0.15*
Pre-migration work experience (yes)	1792 (66.1%)	1341 (71.8%)	451 (53.4%)	1506 (75.6%)	286 (39.8%)	18*	33*
Local language proficiency (0-80)	24.31 (18.24)	25.27 (19.47)	22.37 (15.30)	24.39 (17.86)	24.05 (19.39)	-0.17*	-0.02
Frequency of contact with natives (0-4) No contact Once per month Once per two weeks Once per week Two or more times per week	2.22 (1.58) 657 (24.8%) 302 (11.4%) 311 (11.7%) 555 (21.0%) 823 (31.1%)	2.30 (1.56) 394 (21.7%) 236 (13.0%) 203 (11.2%) 386 (21.3%) 595 (32.8%)	2.04 (1.62) 263 (31.5%) 66 (7.9%) 108 (12.9%) 169 (20.3%) 228 (27.3%)	2.31 (1.58) 453 (23.3%) 204 (10.5%) 229 (11.8%) 403 (20.7%) 658 (33.8%)	1.97 (1.57) 204 (29.1%) 98 (14.0%) 82 (11.7%) 152 (21.7) 165 (23.5%)	-0.16*	-0.22*
Work-relevant traits							
Agreeableness (1-5)	3.98 (0.51)	4.07 (0.45)	3.78 (0.59)	4.00 (0.50)	3.92 (0.53)	-0.55*	-0.16*
Extraversion (1-5)	3.57 (0.42)	3.60 (0.43)	3.48 (0.40)	3.59 (0.42)	3.51 (0.43)	-0.29*	-0.19*
Emotional Stability (1-5)	3.44 (0.60)	3.35 (0.61)	3.66 (0.53)	3.50 (0.60)	3.29 (0.59)	0.54*	-0.35*
GMA (0-60)	32.28 (10.99)	35.53 (10.23)	25.07 (9.00)	32.75 (10.90)	30.97 (11.13)	-1.09*	-0.16*
Conscientiousness (1-5)	3.95 (0.53)	4.05 (0.47)	3.71 (0.57)	3.97 (0.52)	3.89 (0.55)	-0.65*	-0.15*
Work centrality (1-5)	3.89 (0.70)	3.83 (0.67)	4.02 (0.76)	3.97 (0.66)	3.68 (0.78)	0.27*	-0.40*
Employment							
Employment (yes) ^c	928 (34.2%)	660 (35.4%)	268 (31.8%)	824 (41.4%)	104 (14.5%)	04	25*
Longest employment duration (months) ^d	7.45 (6.27)	7.67 (6.40)	6.91 (5.90)	7.68 (6.35)	5.65 (5.29)	-0.12	-0.35*
Total employment duration (months) ^d	8.07 (6.47)	8.33 (6.60)	7.41 (6.10)	8.31 (6.55)	6.15 (5.48)	-0.14*	-0.36*
Duration until work (months) ^d	24.57 (8.82)	24.05 (8.82)	25.85 (8.70)	24.38 (8.83)	26.07 (8.64)	0.21*	0.19*
Highest hourly wage (EUR) ^d	12.88 (3.55)	12.93 (3.54)	12.77 (3.59)	12.97 (3.48)	12.20 (4.06)	-0.04	-0.20*

Highest hourly wage (EUR)' 12.88 (3.55) 12.93 (3.54) Note. RP = Residence permit, GMA = General mental ability. The values within brackets represent standard deviations or percentages. The differences between the nationalities (Syria and Eritrea) and sexes are presented in the last two columns with the Cohen's *d* (for continuous variables; identifiable by a 0 before the dot) or φ values (for categorical variables; identifiable by no 0 before the dot). Values of *d* = 0.20 and φ = .10 are small, *d* = 0.50 and φ = .30 are medium, and *d* = 0.80 and φ = .50 are large (Cohen, 1988). For urbanization, values ranged from 1 = weak urbanization to 5 = strong urbanization.

^a Sex and Nationality were omitted from the table as these statistics are reported in the title row.

* p < .05.

^b Operationalized as the average yearly general practitioner consultation costs (in the euro currency) over the three years after receiving a residence permit.

^c Having been employed for at least one month in the three years after receiving a residence permit.

^d These statistics have been retrieved by only analyzing the refugees who have been employed.

MEASURES

All measures were developed in the Dutch language and had also been translated into different languages, including English, Modern Standard Arabic (for Arabic refugees, including Syrians), and Tigrinya (for Eritrean refugees) by a certified Dutch translation agency. The refugees completed the assessment in their native language (93.8%), in English (5.8%), or in Dutch (0.5%), according to their own preference.

Age

4

Age was assessed as the age (in years) of the refugee at the time of receiving a residence permit.

Sex (Being a Woman)

Sex was assessed by asking refugees whether they are a man (= 0) or a woman (= 1).

Nationality (Cultural Incongruence)

Participants were asked to select their country of birth from a list of all the countries in the world. In this research, we only examined refugees from Syria (= 0) and Eritrea (= 1).

Physical Health Problems

The average yearly general practitioner consultation costs over the three years after receiving a residence permit were used as a proxy for physical health problems and varied between 0 EUR and 519.48 EUR.

PTSD Symptoms

Post-traumatic stress disorder (PTSD) symptoms were measured with the PTSD Checklist for the DSM-5 (PCL-5) (Blevins et al., 2015). The PCL-5 contains 20 items to which participants must indicate how frequently they have been bothered by a problem in the past month, using a five-point Likert scale ranging from 0 = *Not at all* to 4 = *Extremely*. An example item is "Repeated, disturbing dreams of the stressful experience". The scale has shown good psychometric properties in diverse samples, including Middle Eastern samples (Blevins et al., 2015; Ibrahim et al., 2018; Wortmann et al., 2016). In the current sample, the alpha coefficient of this scale equaled .93.

Having Left a Spouse or Children in the Country of Origin

To assess if refugees left a spouse or children in their country of origin, we asked two questions: "Is your wife/husband in your country of origin or in the Netherlands?" (0 = in *the Netherlands* and 1 = in *the country of origin*), and "Do you have minor children (below 18 years) in your country of origin?" (0 = no and 1 = yes). If either or both the spouse and children were in the country of origin, this factor was coded as 1 (yes), and otherwise 0 (no).

Pre-migration Educational Level

A Dutch certification agency, Nuffic, converted the refugees' educational attainment in their home country to match the Dutch educational system (Nuffic, 2022). Translated to the US educational system, these educational levels represent 0 = *no or basic education*, 1 = *high school degree*, 2 = *associate's degree*, 3 = *bachelor's degree*, and 4 = *master's degree*.

Pre-Migration Work Experience

To assess pre-migration work experience, we asked refugees whether or not they had a paid job in their country of origin (0 = no and 1 = yes).

Local Language Proficiency

Local (Dutch) language proficiency was assessed using a test developed by the psychological consultancy agency that gathered the data (NOA, 2006). The Dutch language proficiency test contains two short stories written in Dutch with 80 incomplete words, which candidates were instructed to complete such that they make meaningful words within the context. Scores on this test could range from 0 (no word fragment completed correctly) to 80 (all word fragments completed correctly). Unpublished research had shown that the scores on this test correlate strongly with scores on a Dutch language proficiency test that is used nationwide (NT2; CINOP et al., 2002), supporting its construct validity. Specifically, the Dutch language proficiency test used in the current study correlated with test scores of reading (r = .65), writing (r = .78), vocal understanding (r = .48), and speaking (r = .50) (NOA, 2006). In the current sample, the alpha coefficient of the Dutch language proficiency test who conducted the assessment, completed the Dutch language proficiency test twice with an average time interval between the assessments of 12 weeks and revealed a test-retest reliability of the test of r = .84 (NOA, 2021).

Frequency of Contact with Natives

The reported frequency of contact with natives pertained to informal contact, and not formal contact with personnel of municipalities, organizations, or shops, and had five response options: $1 = no \ contact$, $2 = once \ per \ month$, $3 = once \ per \ two \ weeks$, $4 = once \ per \ week$, and $5 = two \ or \ more \ times \ per \ week$.

Personality

In the domain of work-relevant traits, we assessed the personality traits Agreeableness, Extraversion, Emotional Stability, and Conscientiousness with the MPT-BS-QS Basic (Holtrop, 2016; NOA, 2009), which is a short (60-item) version of a personality inventory that consists of six factors, corresponding to the Big Five model of personality and the HEXACO Honesty-Humility dimension (Lee & Ashton, 2004). The full personality inventory has been investigated among diverse cultural samples and has shown good psychometric

properties among Dutch native and non-Western migrant samples (NOA, 2009). To provide evidence for the validity of the MPT-BS-QS Basic, we conducted an international pilot study in Prolific (N = 238) to examine the relationship between the MPT-BS-QS Basic dimensions and the Big Five dimensions using the IPIP-100 (Goldberg et al., 2006). After excluding a few dysfunctional items from the scales, the results revealed overall proper convergent validities (correlations with the corresponding Big Five dimensions) and discriminant validities (correlations with the average of the non-targeted Big Five dimensions). The convergent and discriminant validities were respectively r = .55 (p < .001) and r = .26 (p < .001) for Agreeableness, r = .71 (p < .001) and r = .29 (p < .001) for Extraversion, r = .81 (p < .001) and r = .25 (p < .001) for Emotional Stability, and r = .73 (p < .001) and r = .22 (p < .001) for Conscientiousness. The size of these convergences is in line with other personality validation studies (Muck et al., 2007; Pace & Brannick, 2010; Rammstedt & John, 2007). Details about the pilot study and the item exclusions are available in the Supplementary Material.

In the MPT-BS-QS Basic, participants were instructed to rate how much they agree or disagree with each statement on a 5-point Likert scale, ranging from 1 = *Totally disagree* to 5 = *Totally agree*. Seven items were used to measure Agreeableness. An example item is "I try not to offend other people". Thirteen items were used to measure Extraversion. An example item is "I like to talk to people who I don't know". Eight items were used to measure Emotional Stability. An example item is "I get nervous quickly". Finally, eight items were used to measure Conscientiousness. An example item is "I set high standards for myself". The alpha coefficients equaled .72 for Agreeableness, .66 for Extraversion, .73 for Emotional Stability, and .77 for Conscientiousness.

General Mental Ability

We assessed GMA by two non-verbal subtests of the multicultural capacities test - intermediate level (MCT-M; Van Breemen et al., 2018; Van den Berg, 2001).¹¹ The subtests were developed to measure fluid intelligence and to minimize any potential bias that might be imposed by cultural background (e.g., due to language or knowledge domain differences). The subtest *Components* consists of 30 items with a time limit of 9 minutes, in which candidates must select two out of six spatial parts that can make up one displayed figure. The subtest *Exclusion* consists of 30 items with a time limit of 7 minutes, in which candidates must select the figure that does not match the other four presented figures. Earlier empirical work showed evidence for the cross-cultural validity and applicability of the MCT-M (Asfar et al., 2019) and its predictive validity in the domains of social functioning and academic achievement among native Dutch candidates and several Dutch migrant groups (Van den Berg, 2001). The correlation between the two MCT-M subtests was r = .56,

p < .001. Earlier research revealed support for the aggregation of the two subtests into one total score of GMA (Asfar et al., 2019). Correspondingly, GMA was computed as the average of the standardized scores on both subtests. The alpha coefficient of the two subtests in the current sample equaled .93. However, we are aware that the alpha coefficient could be an overestimation for tests that have a speed character (Evers et al., 2015).¹²

Work Centrality

We assessed work centrality with a subscale of the work motivation and work search questionnaire (in Dutch: arbeidsmotivatie- en werkzoekvragenlijst [AWV]) (NOA, 2005). Research has shown convergent validity for this instrument with other work motivation instruments (Dusseldorp et al., 2018). The AWV has shown comparable reliabilities and validities among native and non-Western migrant samples (NOA, 2005). An example item of work centrality is "I consider not having paid work ...", with response alternatives ranging from 1 = *Very unpleasant* to 5 = *Very pleasant*. The scale consists of seven items, but we excluded one item due to a low item-rest correlation (see Supplementary Material). In the current sample, the alpha coefficient of the scale equaled .72.

Employment Outcomes

To assess workforce participation, we used two criteria: (1) employment (i.e., the actual occurrence and speed of finding a job), and (2) longest employment duration (i.e., the longest consecutive employment duration). The highest hourly wage (RQ2) is presented in the euro (\in ; EUR) currency. All dependent variables were assessed monthly from the moment a refugee receives a residence permit until three years later. This time interval had been adopted for three reasons. First, according to Dutch governmental regulations, refugees need to finish their integration obligations within three years after receiving their residence permit, making three years a legitimate time interval. Second, the base rate (the percentage of refugees who are employed, three years after receiving a residence permit) is likely high enough to conduct analyses that are not severely restrained by statistical power to detect effects.¹³ With a shorter time interval, the base rate will be lower, and therefore the ability to detect significant predictive validities decreases. Third, with the three-year time span, a relatively large proportion of the sample remains available for the analyses, and as such, making the results less robust.

¹¹ To see the test environment and some sample items of the two GMA subscales, readers can visit the webpage https://www.noa-online.net/practicequestions/mct-m. After entering one's email address, the login instructions will be sent.

¹² To provide more insights into the reliability of the MCT-M subtests, a sample of 35 refugees completed Components twice and 29 refugees completed Exclusion twice with average time intervals between the assessments of respectively 6.5 and 6.2 months. These data revealed test-retest reliabilities of r = .88 for Components and r = .93 for Exclusion.

¹³ Dutch statistics have shown that 19% of the recently arrived refugees in the Netherlands have a paid job within three years after receiving a residence permit (CBS, 2021). We expected that the percentage in our research sample is similar or higher, as these refugees have completed an assessment that promotes the workforce participation. Indeed, the results showed that three years after receiving a residence permit, 34.2% of the refugees in our sample have been employed.

Covariates

We followed the guidelines by Bernerth and Aguinis (2016) to consider and select the covariates in our analyses. Based on the literature, we identified six variables that are relevant to include as covariates in the analyses as they might contaminate the measurement of or cause spurious relations between our focal variables. As such, we included the following variables as covariates in the analyses: (1) the year of receiving a residence permit (e.g., CBS, 2021), (2) the duration between arrival in the Netherlands and receiving a residence permit (e.g., Bakker et al., 2017), (3) the duration between receiving a residence permit and completing the assessment (e.g., Due et al., 2021), (4) the level of urbanization of the municipality where the refugee was resettled (e.g., Bakker & Dagevos, 2017), (5) whether a refugee followed education in the Netherlands during the three years after receiving a residence permit (e.g., De Vroome & Van Tubergen, 2010), and (6) social support (e.g., Newman et al., 2022). The degree of urbanization of the municipality (1 = weak urbanization to 5 = strong urbanization) was determined by where the participant lived for the largest part during the three years after receiving a residence permit. Whether or not a refugee followed education in the Netherlands was retrieved from the CBS data. Social support was assessed by one question, namely "Do you have any friends or family nearby that you can turn to for help?" (0 = no and 1 = yes). The analyses were conducted in SPSS (version 25) and in R with the package survival (Therneau, 2021) and survminer (Kassambara et al., 2017).

STATISTICAL ANALYSIS

4

We conducted the analyses on the combined sample of Syrian and Eritrean refugees. The employment outcomes (i.e., employment, longest employment duration, and highest hourly wage) were predicted with the individual-difference factors that were organized into four variable groups: impeding demographics and health- and family-related challenges - which form hindering individual-difference factors - and acquired human and social capital, and work-relevant traits - forming facilitating individual-difference factors. To assess employment (probability), we used Cox regression analyses (also called survival or event history analysis; see Tierens et al., 2021). The results from these analyses are reported in hazard ratios (HRs), which indicate the increase (if HR > 1) or decrease (if HR < 1) in the probability that one becomes employed in a particular month, given that it has not occurred prior to that month, for every one-unit increase in the independent variable. The Schoenfeld residual plots of the predictors of employment revealed no time-variant pattern, maintaining the proportional hazards assumption of the Cox model. In line with the suggestions by Azuero (2016) for binary variables, we considered HR < 1.60 (or > 0.63) as a small effect, $1.60 (0.63) \le HR \le 2.35 (0.43)$ as a medium effect, and HR > 2.35 (or < 0.43) as a large effect. For continuous variables, we considered HR < 1.30 (or > 0.77) as a small effect, 1.30 (0.77) \leq $HR \le 1.68$ (0.60) as a medium effect, and HR > 1.68 (or < 0.60) as a large effect (cf. Burns et al., 2019). The HR within each parenthesis is the negative equivalent of the HR left to each parenthesis. Given that model fit indices such as the explained variance in Cox regression are sensitive to the proportion of censored observations (i.e., the employment base rate), we used Nagelkerke R^2 in logistic regression analysis with 'having been being employed (yes/no) in the three years after receiving a residence permit' as the dependent variable to determine the (incremental) explained variance.¹⁴

Longest employment duration and highest hourly wage were examined as dependent variables in multiple regression analyses.¹⁵ The collinearity statistics showed VIF values below 10 and tolerance statistics above 0.1, indicating that there is no threat of multicollinearity to the multiple regression models in our research (Bowerman & O'Connell, 1990). In line with the suggestion by Acock (2014) to adopt the conventional correlation coefficient effect sizes for standardized beta coefficients (β), we considered $\beta < .20$ as a small effect, $.20 \le \beta \le .40$ as a medium effect, and $\beta > .40$ as a large effect (Cohen, 1988). Analyses with these two dependent variables were conducted for refugees who became employed during the three years after receiving a residence permit. For all models in the present research, we used Cohen's f^2 statistic to indicate the effect size of the predictive ability of a model, considering $f^2 < .09$ as a small effect, $.09 \le f^2 \le .25$ as a medium effect, and $f^2 > .25$ as a large effect (Cohen, 1988).

RESULTS

Table 2 presents the correlations between all study variables for the whole sample and the Syrian and Eritrean refugees separately.

HYPOTHESIS TESTING

Below we describe the results of the hindering and facilitating individual-difference factors per variable group (Table 3). Results of the covariate tests are not included in the description of our findings and can be found in Table 3.

¹⁴ The results from the logistic regression analyses highly correspond to the results from the Cox regression analyses.

¹⁵ Employment durations with one month of unemployment in between were summed up, as job transitions can take a couple of weeks. Furthermore, we excluded 16 participants from the analyses with highest hourly wage because these participants had hourly wages below the minimum wage at the criterion year (e.g., 4.43 EUR in 2019).

Table 2. Overall and Refugee Group Specific Bivariate Correlations among the Study Variables

Table 2. Continued.

	noup o	000000	5 D. (G.)			0110 01	nong u	10 010	ay ran					00//////												
	1	2	3	4	5	6	7	8	9	10	11	12		13	14	15	16	17	18	19	20	21	22	23	24	25
1. Year RP	-	.21 / .02	55/ 51	.06 / 06	.06/ .16	01/ 11	.02/ .18	.13/ .11	-	.06/ .00	09 / 04	.03 0	3/ 02	05 / 04	08/ 08	18/ 11	12 / 04	.03/ .01	.03/ .03	02/ 05	04/ .03	.03/ 01	02/ .06	.08/ .29	.12/ .24	.14/ .23
2. Duration arrival – RP	.15	-	- .09 / .06	.06 / 04	.04/ .04	.02/ 08	08 / 01	27 / 02	-	07 / 06	.02/ 03	04 0	4/ 03	01/ 00	.06 / 05	02/ 02	.05 / 02	04/ .02	.02/ .01	.03/ .00	01/ 05	00/ .01	. 06 / .03	.17 / .05	.07/ .03	06/ 01
3. Duration RP – assessment	54	06	-	35/ 45	11/ 25	.03/ .17	. 09 / .01	09 / 02	-	- .06 / .03	.14 / .03	00 0	0/ 02	03/ .00	.02/ .05	.25/ .19	.15/ .26	.05/ .04	06 / 05	05/ 08	05/ 11	03/ .02	. 08 / .03	14/ 26	21/ 25	15 / 09
4. Urbanization	00	.04	37	-	.05/ .12	01/ 10	03/ 16	02/ .02	-	03 / 01	.01/ .03	05 0	5 / 00	.07 / 01	.04/ 07	10/ 12	13⁄ 27	01/ 11	.03/ 03	.02/ .06	.05/ .08	.02/ 06	08/ 13	.07/ .09	. 09 / .08	.01/ 00
5. Followed education (yes)	.07	.04	15	.08	-	.06 / 01	43/ 32	04/ 06	-	10/ 12	08/ 08	18 1	8/ I 0	00/ .07	19/ 08	.21/ .13	.10 / 02	04/ .04	.02/ 01	.00/ 02	.10/ .13	.02/ 02	03/ .02	.18/ .34	.01/ .01	11 / .00
6. Having social support	.03	01	.05	06	.01	-	07/ .07	.01/ 10	-	05/ 06	13 / 01	10 0	0 / D1	.08/ .14	02/ .05	.11/ .15	.16/ .18	.05/ .11	.09/ .10	.07 / .05	.07 / .04	.0 7 / .06	.06/ .11	.05 / 00	.00. .00	05/ .04
7. Older age	.12	07	.05	08	39	.05	-	.02/ 13	-	.17 / .06	.11/ .09	.29 .1	9/ 2	.07/ .16	.31/ .21	16 / 07	06/ .14	.13/ .16	.01/ .05	.07 / .08	11 / .00	.01/ .10	.10/ .16	26 / 03	06/ .11	.21 / .11
8. Being a woman	.11	21	07	00	04	03	02	-	-	.24/ .32	03/ .01	.04 1	4/ I 3	03/ 13	34/ 32	.05/ 14	09/ 10	03/ 13	07/ 10	16/ 18	05/ 11	05/ 08	24/ 07	26/ 23	09 / 12	05/ 10
9. Nationality (Eritrea)	27	.02	.08	.08	.08	27	29	.02	-	-	-	-		-	-	-	-	-	-	-	-	-	-	-	-	-
10. Physical health problems	.07	07	05	03	11	02	.18	.24	11	-	.13/ .11	.04 0	4/ 04	02/ .06	04/ 04	08/ 06	03/ 01	.04/ 05	.03/ 03	12/ 11	08 / 04	.06 / 01	07 / 03	12/ 19	01/ .07	01/ -/01
11. PTSD symptoms	03	.01	.10	.00	09	05	.14	02	16	.14	(.94/ .90)	.04 .0	4/ 05	04/ .01	.07/ .03	08 / 02	11 / 02	01/ 03	12/ 19	42/ 35	- .06 / .01	08 / 06	/ 09 .00	04/ 04	10 / .10	05/ .04
12. Spouse/children in country of origin (yes)	01	03	.00	03	14	09	.20	02	.10	.01	.03	-		09 / 02	.09/ .08	12 / 04	05/ 03	.05 / .02	01/ 01	01/ 01	07 / 07	01/ .04	.05 / .03	10 / .04	04/ .04	.05/ 03
13. Pre-migration educational level	.04	01	04	.02	00	.17	.16	07	30	.04	.02	0	9	-	.10/ .20	.27/ .26	.14/ .16	.03/ .21	.11/ .18	.12/ .10	.26/ .22	.14/ .16	.15/ .16	.03/ .14	. 09 / .03	.25 / 03
14. Pre-migration work experience (yes)	03	.03	.02	01	16	.05	.31	33	18	02	.09	.0	07	.18	-	04/ .10	.06/ .16	.10/ .12	.11/ .14	.12/ .10	.04/ .09	.06/ .10	.25/ .13	.08 / .06	.02/ .01	.15 / .11
15. Local language proficiency (0-80)	13	02	.22	11	.18	.14	12	01	08	07	06	1	10	.28	.02	(.98/ .96)	.28/ .19	04/ .15	.04/ .09	.07/ .05	.34/ .36	. 06 / .05	.11/ .09	.09/ .13	.06/ 04	.11 / 03
16. Frequency of contact with natives	07	.03	.17	18	.05	.18	.01	10	08	02	07	0	05	.17	.11	.26	-	.14/ .15	.18/ .18	.13/ .07	.13 / .04	.12/ .14	.19/ .11	.12 / .06	.05/ .03	.07/ .15
17. Agreeableness	.09	02	.02	06	03	.14	.20	07	26	.04	.03	.0	D1	.17	.15	.04	.16	(.69/ .73)	.42/ .46	.14 / .06	.04/ .16	.54/ .70	.29/ .36	.00/ .05	.00/ 03	.08 / 05
18. Extraversion	.07	.02	07	.00	00	.13	.06	08	13	.03	11	0	02	.16	.14	.07	.18	.44	(.70/ .53)	.18/ .30	.05/ .12	.52/ .54	.24/ .23	.08/ .10	.04/ .05	.07/ 08
19. Emotional Stability	09	.02	04	.05	.02	01	00	16	.24	14	43	.0	D1	.04	.07	.04	.09	.04	.17	(.75/ .62)	.11 / .07	.14/ .13	.23 / 02	.06/ .08	.09 / 11	.09 / 04
20. GMA	.10	02	09	.02	.06	.17	.05	07	44	01	.03	1	11	.34	.13	.34	.13	.18	.12	02	(.92/ .89)	.10/ .09	.11/ .11	.08/ .15	.05/ 00	.12 / .00
21. Conscientiousness	.10	01	04	03	02	.14	.11	06	30	.08	02	0	02	.22	.12	.08	.14	.64	.53	.05	.21	(.75/ .76)	.26/ .32	.03/ .03	.06/ 06	.11 / 03
22. Work centrality	03	.06	.07	08	00	.04	.07	18	.12	07	08	.0	05	.11	.18	.09	.15	.27	.22	.18	.04	.23	(.72/ .72)	.13 / .03	03/ .09	.16 / .00
23. Employment (yes) ^b	.15	.14	18	.07	.23	.04	19	25	04	13	04	0	06	.07	.08	.11	.11	.03	.09	.06	.11	.04	.09	-	-	-
24. Longest employment duration	.16	.06	22	.08	.00	.02	01	10	06	.01	05	0	03	.08	.03	.04	.05	.01	.05	.03	.05	.04	.01	-	-	.19/ .14
25. Highest hourly wage	.17	04	13	.00	08	01	.19	07	02	00	02	.0)2	.16	.14	.08	.10	.04	.03	.05	.09	.06	.10	-	.18	-

Note. Correlations for the whole sample are shown below the diagonal. Correlations for respectively Syrian and Eritrean refugees are shown above the diagonal, separated by a slash. The bolded values are significant (p < .05, two-tailed). RP = Residence permit, GMA = General mental ability. The values between the parentheses on the diagonal represent the alpha coefficients among respectively the Syrian and Eritrean refugees. For urbanization, values ranged from 1 = weak urbanization to 5 = strong urbanization.

^b Having been employed for at least one month in the three years after receiving a residence permit.

REFUGEES' WORKFORCE PARTICIPATION | 95

Table 3. Cox and Multiple Regression Models on Employment (Emp), Longest Employment Duration (LED), and Highest Hourly Wage (HHW)

Table 3. Continued.

	Impe	ding Demog	aphics	Health-	and Family-	Related	Acquired H	luman and S	ocial Capital	Wor	k-Relevant Tr	aits		Full Model	
	Emp	LED	HHW	Emp	LED	HHW	ama	LED	HHW	Emp	LED	HHW	Emp	LED	HHW
	HR	ß	ß	HR	ß	ß	HR	ß	ß	HR	ß	ß	HR	ß	ß
	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%CI]	[95%Cl]	[95%CI]
Covariates															
Year RP	1.23***	.04	.14**	1.09*	.01	.15***	1.13**	.01	.17***	1.08	.01	.15***	1.25***	.04	.17***
	[1.12, 1.34]	[05, .12]	[.06, .21]	[1.00, 1.19]	[07, .09]	[.07, .22]	[1.03, 1.24]	[07, .09]	[.09, .25]	[0.99, 1.18]	[07, .09]	[.07, .23]	[1.13, 1.38]	[05, .13]	[.08, .26]
Duration arrival – RP	1.07*	10.0	08*	1.1/****	.04	09*	1.15***	.04	08**	1.18	.04	08*	1.07	10.	07
Duration PP assossment	0.78***	21***	[14,0 1] 09*	0.76***	[02, .11] 21***	[13,02]	[1.00, 1.22]	25***	[13,01]	0.76***	21***	07	0.65***	2//***	10*
	[0.71. 0.85]	[29, .14]	[16,01]	[0.69, 0.83]	[29,14]	[1601]	[0.59, 0.73]	[33,16]	[1801]	[0.70, 0.84]	[29,13]	[15, .01]	[0.58, 0.73]	[3316]	[1901]
Urbanization	0.85***	13***	03	0.88***	14***	02	0.92*	13***	01	0.87***	13***	03	0.92*	13***	01
	[0.80, 0.91]	[20, .07]	[09, .04]	[0.82, 0.94]	[20,07]	[09, .05]	[0.85, 0.99]	[20,06]	[08, .06]	[0.82, 0.93]	[20,07]	[09, .04]	[0.85, 0.99]	[20,06]	[08, .06]
Followed education (yes)	1.37***	06	01	1.90***	04	10**	1.70***	06	08*	1.99***	04	10**	1.17	08*	03
	[1.17, 1.61]	[13, .01]	[08, .06]	[1.64, 2.20]	[11, .02]	[17,03]	[1.44, 2.00]	[13, .01]	[15,01]	[1.72, 2.30]	[10, .03]	[17,03]	[0.97, 1.39]	[16,01]	[12, .05]
Having social support (yes)	1.06	.02	.01	1.11	.02	01	0.93	00	05	1.08	.02	02	0.90	00	02
	[0.92, 1.23]	[05, .08]	[06, .07]	[0.96, 1.27]	[05, .08]	[07, .06]	[0.80, 1.09]	[07, .07]	[12, .02]	[0.93, 1.24]	[05, .08]	[08, .05]	[0.76, 1.06]	[07, .07]	[10, .05]
Impeding demographics															
Higher age	0.70***	04	.18***										0.72***	04	.15**
Deine europee	[0.64, 0.76]	[11,.03]	[.11,.25]										[0.64, 0.80]	[12,.05]	[.06, .23]
being a wonian	[0.27	12 [-18-05]	08										[0 29 0 47]	14	00
Nationality (Fritrea)	0.93	- 00	03										0.91	02	07
Nationality (Entrea)	[0.79, 1.09]	[07, .07]	[04, .10]										[0.74, 1.11]	[07, .11]	[02, .16]
Health- and family-related ch	allenges														
Physical health problems ^{E+}	0			0.78***	.02	02							0.93	.05	02
				[0.71, 0.85]	[05, .08]	[08, .05]							[0.84, 1.02]	[02, .12]	[10, .05]
PTSD symptoms ^y				1.02	04	01							1.08	03	.00
				[0.95, 1.09]	[10, .03]	[08, .06]							[0.99, 1.17]	[11, .05]	[08, .08]
Spouse/children in country of				0.89	02	.01							0.89	01	02
origin (yes) ³				[0.77, 1.04]	[08, .05]	[06, .08]							[0.74, 1.06]	[08, .06]	[10, .05]
Acquired human and social															
Pro migration oducational							0.07	02	1 1 **				0.00	04	00*
level ^{E,f}							[0 90 1 04]	[- 05 10]	[03 18]				[0 91 1 07]	.04 [- 04 11]	[01 17]
Pre-migration work							1 38***	01	12**				1 23*	- 02	09*
experience (yes)							[1.17, 1.62]	[06, .08]	[.05, .19]				[1.03, 1.47]	[10, .06]	[.01, .17]
Local language proficiency ^{f+,o+}							1.22***	.08*	.10**				1.19***	.10*	.11**
							[1.13, 1.32]	[.01, .16]	[.03, .18]				[1.09, 1.29]	[.02, .18]	[.03, .19]
Frequency of contact with							1.15***	.06	.07*				1.13**	.05	.07*
natives ¹⁺							[1.07, 1.24]	[01, .13]	[.00, .15]				[1.05, 1.23]	[02, .12]	[.00, .15]
Work-relevant traits															
Agreeableness										0.97	04	03	0.99	03	05
Future version										[0.88, 1.06]	[12,.05]	[11,.00]	[0.90, 1.09]	[12,.07]	[14,.04]
Extraversion										[1.05 1.23]	.03	03	1.06	.01	04 [_13_04]
Emotional Stability®										1.04	03	03	1 02	00	- 01
2otional otability										[0.97, 1.11]	[04, .09]	[03, .10]	[0.94, 1.11]	[08, .08]	[09, .07]
GMA ^{f+}										1.13**	.00	.07*	1.03	05	.03
										[1.05, 1.21]	[06, .07]	[.01,.14]	[0.95, 1.12]	[12, .03]	[05, .10]
Conscientiousness										0.95	.02	.05	0.95	.01	.05
										[0.87, 1.05]	[07, .11]	[04, .14]	[0.86, 1.06]	[09, .10]	[05, .15]
Work centrality ^s										1.17***	.02	.08*	1.08	01	.03
$D^2(f^2)$ model	240 (22)	002/00	079 (08)	147/17	070 (00)	049 (05)	475 / 24)	002 (00)	008 (1 1)	[I.UO, I.ZO]	[US, .U8]	[.01,.15]	[U.33, I.18]	[U0, .U7]	[US, . [1]
π (F) Houel	.240 (.32)	.002 (.09)	.070 (.08)	.14/(.1/)	.070 (.08)	.040 (.05)	.1/5(.21)	.002 (.09)	.030 (.11)	. 133 (. 18)	.071 (.00)	.003 (.07)	.207 (.30)	. 101 (.11)	.123 (.14)
Change in K* (f*)	(13)	.014~ (.01)	.030**** (.03)	.018 (.02)	.002 (.00)	.000 (.00)	.038* (.04)	.013* (.01)	.050 (.05)	.025 (.03)	.003 (.00)	.0101 (.02)	.127 (.15)	.032 (.03)	.077 (.08)

Note. RP = Residence permit, GMA = General mental ability, HR = Hazard ratio. The HRs indicate the increase (if HR > 1) or decrease (if HR < 1) in the probability that one becomes employed in a particular month, given that it has not occurred prior to that month, for every one-unit increase in the independent variable. The superscripts indicate that

96 | CHAPTER 4

the effect on employment probability is stronger for Syrian (^{S+}), Eritrean (^{E+}), male (^{m+}), female (⁺), young (^{y+}), or old (^{o+}) refugees, or is only found among Syrian (^S), Eritrean (^E), male (^m), female (¹), young (^y), or old (^o) refugees. * $\rho < .05$, ** $\rho < .01$, *** $\rho < .001$.

Hindering Individual-Difference Factors

Impeding Demographics. We predicted that age (H1a) and being a woman (H1b) are negatively associated with refugees' workforce participation, and that workforce participation is higher among Syrian compared to Eritrean refugees (H1c). Our results revealed a significant negative effect of age on employment (HR = 0.70, p < .001), indicating that refugees' odds of becoming employed decreased by 30% for every standard deviation increase in age. However, there was no significant effect of age on longest employment duration ($\beta = ..04$, p = .240). Thus, H1a was only supported for employment. Furthermore, there was a significant negative effect of being a woman on employment (HR = 0.27, p < .001), indicating that the odds for female refugees to become employed are 73% lower compared to the odds for male refugees to become employment duration ($\beta = .12$, p < .001). Thus, H1b was fully supported. Finally, in contrast to H1c, we found no significant effect of nationality on employment (HR = 0.93, p = .351) or longest employment duration ($\beta = .00$, p = .975). Figure 3 presents the monthly employment rates during the three years after receiving a residence permit, for the Syrian and Eritrean male and female refugees.

Figure 3. Refugee Subgroup Employment Statistics During the Three Years After Receiving a Residence Permit



Note. The semitransparent areas around the lines represent the 95% confidence intervals.

Health- and Family-Related Challenges. We predicted that workforce participation is lower among refugees with higher levels of physical health problems (H2a) and PTSD symptoms (H2b), and among those who left a spouse or children in their country of origin (H2c). The results showed a significant negative effect of physical health problems on employment (HR = 0.78, p < .001), indicating that refugees' odds of becoming employed decreased by 22% for every standard deviation increase in physical health problems. However, we found no significant effect of physical health problems on longest employment duration ($\beta = .02$, p = .610). Thus, H2a was only supported for employment. Furthermore, in contrast to H2b, there was no significant effect of PTSD symptoms on employment (HR = 1.02, p = .680) or longest employment duration ($\beta = -.04$, p = .578). Finally, in contrast to H2c, there was no significant effect of having left a spouse or children in the country of origin on employment (HR = 0.89, p = .134) or longest employment duration ($\beta = -.02$, p = .578).

Facilitating Individual-Difference Factors

Acquired Human and Social Capital. We predicted workforce participation to be higher for refugees with a higher pre-migration educational level (H3a), more pre-migration work experience (H3b), a higher level of local language proficiency (H3c), and a higher frequency of contact with natives (H3d). The results showed, contrary to H3a, no significant effect of pre-migration educational attainment on employment (HR = 0.97, p = .388) or longest employment duration ($\beta = .02$, p = .531). With respect to H3b, we found a significant positive effect of pre-migration work experience on employment (HR = 1.38, p < .001), indicating that the probability of becoming employed is 38% higher for refugees with (versus without) pre-migration work experience. However, there was no significant effect of pre-migration work experience on longest employment duration ($\beta = .01$, p = .870). Thus, H3b was only supported for employment. Furthermore, there was a significant positive effect of local language proficiency on employment (HR = 1.22, p < .001), indicating that the odds for refugees to become employed are 22% higher for every standard deviation increase in their local language proficiency. Additionally, we found a significant positive effect of local language proficiency on longest employment duration ($\beta = .08$, p = .033). Thus, H3c was fully supported. Finally, we found a significant positive effect of frequency of contact with natives on employment (HR = 1.15, p < .001), indicating that the probability for refugees to become employed is 15% higher for every standard deviation increase in the frequency of contact with natives. However, there was no significant effect of frequency of contact with natives on longest employment duration ($\beta = .06$, p = .108). Thus, H3d was only supported for employment.

Work-Relevant Traits. We predicted workforce participation to be higher for refugees with higher levels of Agreeableness (H4a), Extraversion (H4b), Emotional Stability (H4c), GMA (H4d), Conscientiousness (H4e), and work centrality (H4f). The results revealed, contrary to H4a, no significant effect of Agreeableness on employment (HR = 0.97, p = .458) or longest employment duration (β = -.04, p = .404). With respect to H4b, there was a significant

positive effect of Extraversion on employment (HR = 1.13, p = .002), indicating that the odds to become employed are 13% higher for every standard deviation increase in refugees' level of Extraversion. However, there was no significant effect of Extraversion on longest employment duration (β =.03, p = .476). Thus, H4b was only supported for employment. Furthermore, in contrast to H4c, we found no significant effect of Emotional Stability on employment (HR = 1.04, p = .300) or longest employment duration ($\beta = .03$, p = .411). With respect to H4d, we found a significant positive effect of GMA on employment (HR = 1.13, p = .001), indicating that the odds for refugees to become employed are 13% higher for every standard deviation increase in their level of GMA. However, there was no significant effect of GMA on longest employment duration ($\beta = .00$, p = .931). Therefore, H4d was only supported for employment. Furthermore, H4e could not be supported, as we found no significant effect of Conscientiousness on employment (HR = 0.95, p = .311) or longest employment duration $(\beta = .02, p = .647)$. Finally, we found a significant positive effect of work centrality on employment (HR = 1.17, p < .001), indicating that the odds for refugees to become employed are 17% higher for every standard deviation increase in their level of work centrality. However, we found no significant effect of work centrality on longest employment duration ($\beta = .02$, p = .638). Thus, H4f was only supported for employment.

RESEARCH QUESTIONS

4

Our first research question (RQ1) focused on which individual-difference factors are the strongest predictors of refugees' workforce participation. We examined this RQ by testing a regression model that included all study variables (Table 3). The strongest negative predictors of employment were older age (HR = 0.72, p < .001) and being a woman (HR = 0.37, p < .001), and the strongest positive predictors of employment were pre-migration work experience (HR = 1.23, p = .020), local language proficiency (HR = 1.19, p < .001), and frequency of contact with natives (HR = 1.13, p = .002). For longest employment duration, the significant predictors were being a woman ($\beta = .14$, p < .001) and local language proficiency ($\beta = .10$, p = .014).

Our second research question (RQ2) referred to which individual-difference factors are the strongest predictors of refugees' highest hourly wage. For the impeding demographics, we found a significant positive effect of age ($\beta = .18$, p < .001) and a significant negative effect of being a woman ($\beta = .08$, p = .021) on highest hourly wage. This model explained 3.0% (ΔF [3, 886] = 9.69, p < .001; $f^2 = .03$) unique variance above and beyond the covariates. None of the health- and family-related challenges predicted highest hourly wage. This model explained 0.0% (ΔF [3, 873] = 0.13, p = .945; $f^2 = .00$) unique variance above and beyond the covariates. None of the health- and family-related challenges predicted highest hourly wage. This model explained 0.0% (ΔF [3, 873] = 0.13, p = .945; $f^2 = .00$) unique variance above and beyond the covariates. For the acquired human and social capital factors, we found significant positive effects of pre-migration educational level ($\beta = .11$, p = .005), pre-migration work experience ($\beta = .12$, p = .001), local language proficiency ($\beta = .00$, p = .008), and frequency of contact with natives ($\beta = .07$, p = .046). This model explained 5.0% (ΔF [4, 738] = 10.33, p < .001; $f^2 = .05$)

unique variance above and beyond the covariates. Finally, for the work-relevant traits, we found a significant effect of GMA ($\beta = .07$, p = .034) and work centrality ($\beta = .08$, p = .025). This model explained 1.6% (ΔF [6, 862] = 2.39, p = .027; $f^2 = .02$) unique variance above and beyond the covariates.

EXPLORATORY ANALYSES

We also examined the potential moderating effects of nationality, sex, and age on the relation between the hypothesized predictors and employment.¹⁶ These analyses show the extent to which the relationships vary between Syrian and Eritrean refugees, male and female refugees, and refugees of different ages, which may help to identify the barriers that certain subgroups face in finding employment. The results showed that nationality (Syria = 0, Eritrea = 1) significantly moderated the effect of physical health problems on employment (β = -.15, *p* = .012), such that this effect was stronger among Eritrean refugees. Furthermore, having left a spouse or children in the country of origin (β = .11, *p* = .001) only significantly negatively influenced employment of Syrian refugees. Additionally, pre-migration educational level (β = .07, *p* = .028) only showed a significant positive effect on employment among Eritrean refugees. Finally, work centrality (β = -.09, *p* = .011) only significantly positively predicted employment of Syrian refugees.

With respect to sex (male = 0, female = 1), we found that pre-migration educational level ($\beta = .12$, p = .010) only significantly positively influenced employment among female refugees. Furthermore, the effect of local language proficiency on employment ($\beta = .17$, p < .001) was stronger among female refugees. Additionally, the effect of frequency of contact with natives on employment ($\beta = .10$, p = .031) was stronger among female refugees. Finally, the effect of GMA on employment ($\beta = .13$, p = .008) was stronger among female refugees.

With respect to age, we found only a negative effect of PTSD symptoms on employment (β = -.15, p < .001) among young refugees. Furthermore, the effect of local language proficiency on employment (β = .08, p = .044) was stronger among old refugees. Finally, Emotional Stability only showed a significant positive effect on employment among old refugees (β = .09, p = .015).

Δ

¹⁶ We only focused on employment as the effects on longest employment duration were mostly nonsignificant or weak.

DISCUSSION

The goal of the present study was to advance our understanding of refugees' workforce participation by developing a framework of hindering and facilitating individual-difference factors for predicting their participation in the workforce and to test the predictive validity of the individual-difference factors within our framework among recently arrived Syrian and Eritrean refugees in the Netherlands. We related the individual-difference factors to two criteria: employment and longest employment duration. Overall, the present study showed that several hindering factors (i.e., older age, being a woman, and physical health problems) and facilitating factors (i.e., pre-migration work experience, local language proficiency, frequency of contact with natives, Extraversion, GMA, and work centrality) predicted refugees' employment. In contrast, only one hindering factor (i.e., being a woman) and one facilitating factor (i.e., local language proficiency) predicted refugees' longest employment duration. We also exploratorily examined whether these same hindering and facilitating factors would be able to predict refugees' highest hourly wage. We found that two hindering factors (i.e., older age and being a woman) and several facilitating factors (i.e., pre-migration educational level, pre-migration work experience, local language proficiency, frequency of contact with natives, GMA, and work centrality) predicted this indicator of employment guality. Below, we describe our theoretical contributions, practical implications, limitations, and suggestions for future research.

4

THEORETICAL CONTRIBUTIONS

The present study offers several theoretical contributions. First, we add to the literature on refugees' workforce participation by integrating multidisciplinary insights into one integrative theoretical framework containing a wide range of individual-difference factors that may either hinder or facilitate refugees' path to finding work. Although the personnel psychology literature already offered various theoretical frameworks and meta-analyses of individual-difference factors that successfully predict native-born job seekers' workforce participation (e.g., Kanfer et al., 2001; Van Hooft et al., 2021) and employability (Harari et al., 2021), we show that these frameworks lack important individual-difference factors that are relevant for migrants in general, and refugees in particular. Indeed, we found several migrant-specific individual-difference factors (i.e., local language proficiency and frequency of contact with natives) and a refugee-specific factor (i.e., physical health problems) to predict one or more of our outcome variables. In fact, local language proficiency was among the strongest predictors of all three workforce participation outcomes, and frequency of contact with natives was among the strongest predictors of employment and highest hourly wage. Importantly, unlike GMA, Extraversion, and work centrality, other established individual-difference predictors of workforce participation among native-born job seekers, such as Agreeableness (Baay et al., 2014; Van Hooft et al., 2021), Emotional Stability (Kanfer et al., 2001), and Conscientiousness (Egan et al., 2017) showed no relation with refugees'

workforce participation. Furthermore, although we did find significant effects for GMA in predicting refugees' employment and highest hourly wage, the effect of GMA seems to be less strong for refugees than reported for native-born job seekers (Vélez-Coto et al., 2021). Together, these findings show that established individual-difference factors predicting workforce participation do not necessarily generalize to refugees, who face unique barriers compared to native-born job seekers or economic migrants (e.g., Agbényiga et al., 2012), and whose (language) skills are less likely to match the needs of the job market (Lee et al., 2020).

Second, the present study contributes to the literature on refugees' workforce participation by actually testing our integrative framework among a large group of recently arrived refugees living in the Netherlands. To the best of our knowledge, there have been only two prior attempts at developing a theoretical framework of individual-difference factors for understanding refugees' workforce participation (Boss et al., 2021; Lee et al., 2020). However, neither framework has been quantitatively examined. Furthermore, both frameworks lack important predictors of workforce participation: The framework of Boss et al. (2021) does not include refugee demographics (e.g., age, sex) and the framework of Lee et al. (2020) does not include any factors related to refugees' acquired human capital in their home country. Importantly, both frameworks focus on hindering factors only, and do not include refugeespecific family-related challenges or psychological characteristics such as personality and GMA. Our framework is thus distinctive in that it includes refugees' demographics and acquired human capital in their home country, family-related challenges, and psychological characteristics, and focuses on both hindering and facilitating factors. Many of these factors predicted refugees' employment and highest hourly wage (note that, of these specific factors, only being a woman was [negatively] related to longest employment duration). Furthermore, while these previous frameworks already highlighted the importance of certain hindering factors for understanding refugees' workforce participation, we theorized and empirically demonstrated that facilitating factors (i.e., acquired human and social capital and work-relevant traits) matter as well. In fact, the facilitating factors explained more variance in highest hourly wage (2-5%) than the hindering factors (0-3%).

Third, this study contributes to the literature on refugees' socioeconomic integration by demonstrating differential predictive validities for Syrian and Eritrean refugees, two groups that have been seldomly empirically studied (but see Hunkler & Khourshed, 2020), but also for male and female refugees and for older and younger refugees. While previous studies do not differentiate between refugee groups or focus on only one refugee group (e.g., Hahn et al., 2019; Renner & Senft, 2013), our study indicates that certain hypothesized individual-difference factors are only related to employment of specific refugee groups. For example, although the refugee-specific individual-difference factors of PTSD symptoms or having left a spouse or children in their country of origin did not show a relation with

102 | CHAPTER 4

employment for our overall refugee group, PTSD symptoms did predict employment among younger refugees while having left a spouse or children in their country of origin did predict employment among Syrian refugees. These findings provide further evidence for the importance of including refugee-specific individual-difference factors when predicting refugees' workforce participation. They also indicate that, when studying refugees (or citizens), it is essential to not only investigate the main effect but also the moderating role of nationality, sex, and age.

Fourth, we contribute to the literature on refugees' socioeconomic integration (Lee et al., 2020), by studying three workforce participation outcomes – employment, longest employment duration, and exploratively highest hourly wage – and showing that in the early years of resettlement, refugees' individual-difference factors are stronger predictors of employment and highest hourly wage than of longest employment duration. One potential explanation for the weak predictive validities for longest employment duration pertains to refugees' actual jobs in the early years of resettlement. These are usually jobs with temporary contracts (CBS, 2021), typically characterized by low wages, poor working conditions, and limited upward mobility opportunities (e.g., Kosny et al., 2020), and are therefore challenging to maintain. Another explanation for these results may be that employment duration may be more dependent on organizational rather than individual-difference factors. Examples are job characteristics such as job security and rewards (Rubenstein et al., 2018). Future research is needed to investigate such relationships.

PRACTICAL IMPLICATIONS

The present study also has a number of important practical implications. First, most research on refugees' socioeconomic integration stems from sociology and economics, and has typically focused on demographics and acquired human and social capital (e.g., Kofman, 2014; Lancee, 2012). However, as it is difficult to administer tests and inventories among refugees and to relate their scores to time-lagged, objective employment data, there is little knowledge on the influence of work-relevant traits on refugees' workforce participation. Through the integrative framework introduced in this study, counselors can yield a comprehensive picture of a refugee's ability to gain and maintain (higher-quality) employment through the assessment of fairly observable and easy-to-capture individualdifference factors (i.e., age, being a woman, pre-migration educational level and work experience, local language proficiency, and frequency of contact with natives), but also more hidden work-relevant traits (i.e., Extraversion, GMA, and work centrality). Furthermore, earlier studies provide robust evidence that person-environment fit - the match between the traits of a person and the characteristics of a job, team, or organization - is a predictor of several important work outcomes (for a review, see Van Vianen, 2018). As psychological individual-difference factors are relatively stable and difficult to change (e.g., see Roberts et al., 2017), it is therefore important to select integration trajectories, trainings, and jobs that

fit these specific factors.

Second, taking into consideration the diversity in refugees' individual-difference factors is important in practice. In fact, it is considered to be "one of the most important challenges for host societies ... in order to release the full potential of those seeking integration" (Hahn et al., 2019, p. 11). As such, we recommend counselors to focus on the significant individual-difference factors that are highly malleable and trainable (i.e., local language proficiency and frequency of contact with natives), by stressing refugees' local language training, and helping them with increasing their network and contacts with natives. Additionally, we recommend counselors to provide individualized support and special attention to refugees with risk factors of low employment that are fixed or stable (i.e., [older] age, sex [woman], and [no] pre-migration work experience), or relatively stable (i.e., [low] Extraversion, GMA, and work centrality). It is also paramount to focus on those individual-difference factors that have predictive validities for the group the specific refugee group belongs to. So far, for instance, Dutch policymakers tend to adopt a one-size-fits-all approach to integration programs as they regard refugees as a homogenous group with similar background characteristics and aspirations (Miltenburg & Dagevos, 2021).

Third, whilst the covariates were not the primary focus of this study, findings concerning these variables have some important practical implications. For example, we found that a short time interval between receiving a residence permit and completing the assessment is positively associated with the odds of finding a job. This effect might be explained by the active support and counseling of the refugee that corresponds to the completion of the assessment, although strong data to justify the following claim is not available. For instance, many Eritrean refugees rely on the agency of the government and the municipalities to construct their integration trajectories, as it is the government that primarily determines citizens' educational trajectories and job industry in their home country (e.g., Sterckx et al., 2018). Thus, active counseling may help these refugees deal with the perceived barriers and difficulties to enter the labor market. As counselors are more familiar with effective jobsearch procedures, their support can increase the odds that a refugee quickly finds a job (Gericke et al., 2018). Altogether, it seems important to quickly start by assisting refugees in designing their integration trajectories. An individual assessment could be a useful tool in this regard, as it portrays the refugee's strengths and potential barriers to enter the labor market.

STRENGTHS, LIMITATIONS, AND FUTURE DIRECTIONS

The present study has several strengths: we adopted a time-lagged research design, examined multiple employment outcomes, used high-quality employment data, and measured local language proficiency through an objective test. However, despite these methodological strengths, the present study also has some limitations. We describe these

limitations below and suggest future research ideas beyond the ones already mentioned, related to the role of refugees' individual-differences factors in their workforce participation.

First, one important limitation of the present research pertains to the context of the assessment and subsequent counseling. The assessment can be considered an intervention, and the counselors of the refugees in this study were trained to use the results of the assessment to provide customized support for the refugee to sustain a prosperous integration. As such, when supporting refugees in their socioeconomic integration, counselors may have paid special attention to refugees with disadvantaged profiles and higher levels of undesirable traits, compared to refugees with more favorable characteristics. For example, a refugee with a low (high) score on Conscientiousness and more (fewer) PTSD symptoms is more (less) likely to be supported in his or her route to employment. As such, actual effects of individual-difference factors on workforce participation might have been partly masked in this research. Therefore, we suspect that the actual effects of these factors are stronger and that some null findings might be type II errors. Future research should investigate the predictive validity of refugees' individual-difference factors for employment outcomes in a research context where tests are only assessed for scientific purposes, and not as tools for assisting refugees in their socioeconomic integration.

Second, the personality inventory in this research has not been validated among refugee samples, and some scales showed relatively low internal consistencies (e.g., alpha coefficients of respectively .53 and .62 for Extraversion and Emotional Stability among Eritrean refugees). Nonetheless, relatively little research has been conducted on personality scales among citizens or refugees from Middle Eastern and African countries (for some exceptions, see Ion et al., 2017; Zeinoun et al., 2017). Future research could advance the literature on personality assessments of people from the Middle East and Africa by investigating the reliability and validity of inventories among such populations.

Third, in the present study, no data were available on job-search behaviors, the number of job interviews, and job offers. Hence, it was impossible to identify whether a refugee's unemployment is due to a lack of job search and initiative-taking or due to rejections by employers. A recent meta-analysis has shown that job-search intensity is a positive predictor of the number of interviews, job offers, and actual (re)employment (Van Hooft et al., 2021). Future research could therefore investigate whether job-search behavior and job-search quality mediate the effect of individual differences on refugees' workforce participation. Such research provides insights into the extent to which refugees' unemployment can be attributed to their own initiatives and to the employers' rigor in accepting job candidates. Additionally, researchers could also concentrate on the effectiveness of different types of job-search behaviors, such as formal job search (e.g., through employment agencies and online vacancies) and informal job search (e.g., through friends or relatives).

Fourth, although the employment outcomes that are examined in the present research employment, longest employment duration, and (exploratively) highest hourly wage - are important, employment success is also reflected by outcomes such as the job's extrinsic rewards (e.g., pension plan, insurances, benefits, favorable working hours), intrinsic rewards (e.g., need satisfaction, safety, security), and the person-job demands fit (e.g., the match between a person's skills and the requirements) (Saks, 2005; Wanberg et al., 2016). In fact, most recent refugees in the Netherlands who have a job work part-time (73%) and have a temporary contract (84%) (CBS, 2021). Furthermore, underemployment (i.e., overgualification) is a widely documented issue for highly skilled refugees (Campion, 2018; Ortlieb & Weiss, 2020). Approximately 60% of the employed refugees in European countries with tertiary education are overgualified for their jobs, whereas such overgualification occurs in 30% of the other non-EU-born migrants and in 21% of the natives (OECD, 2016). One recent study has found that refugees' career planning predicted their future quality of employment (Zikic & Klehe, 2021). As there is limited other empirical work that has examined refugees' individual-difference factors as predictors of quality of employment, underemployment, or other important job characteristics, further study of this issue would be of interest.

CONCLUSION

The present research has examined Syrian and Eritrean refugees' workforce participation (i.e., employment and longest employment duration) and highest hourly wage through an integrative framework that organizes various individual-difference factors in hindering and facilitating factors. The results showed that both hindering factors (i.e., age, being a woman, and physical health problems) and facilitating factors (i.e., pre-migration educational level and work experience, local language proficiency, frequency of contact with natives, Extraversion, GMA, and work centrality) contributed to predicting these employment outcomes. However, some effects were only found for specific refugee groups or for a particular workforce participation outcome. The findings provide important implications for improving refugees' integration trajectories.

APPENDIX

The table below provides information and statistics on the geography and population, sociodemographic characteristics, labor and economic characteristics, basic needs and health, and cultural dimensions of Syria, Eritrea, and the Netherlands. Most data reported this table were retrieved from the CIA World Factbook in April 2022 (CIA, 2022), and the cultural dimensions were retrieved from Hofstede (2001). For more information about the cultural dimensions, we refer readers to Hofstede (2011).

Countr	y Variable	Information	and Statistics	of Syria,	, Eritrea,	and the	Netherlands
--------	------------	-------------	----------------	-----------	------------	---------	-------------

Comparison	Syria	Eritrea	the Netherlands
Geography and populat	ion		
Location	The Middle East, bordering the Mediterranean Sea, between Lebanon and Turkey.	Eastern Africa, bordering the Red Sea, between Djibouti and Sudan.	Western Europe, bordering the North Sea, between Belgium and Germany.
Population size (July 2021)	20,384,316	6,147,398	17,337,403
Population distribution	Large population density along the Mediterranean coast. The highest concentration of citizens is found in Damascus, Aleppo, and Hims (Homs). More than half of the population lives in the coastal plain, the province of Halab, and the Euphrates River valley. ^a	The population density is highest in the center of the country and around the cities of Asmara and Keren, and smaller settlements are found in the north and south of the country.	An area known as the Randstad, anchored by the cities of Amsterdam, Rotterdam, the Hague, and Utrecht, is the most densely populated region in the Netherlands. The north tends to be less dense, although sizeable communities are found throughout the whole country.
Urbanized population (% of the total population in 2021)	56.1%	42.0%	92.6%
Ethnic groups	Arab ~50%, Alawite ~15%, Kurd ~10%, Levantine ~10%, other ~15% (includes Druze, Ismaili, Imami, Nusairi, Assyrian, Turkoman, Armenian) (year N.A.)	Tigrinya 55%, Tigre 30%, Saho 4%, Kunama 2%, Rashaida 2%, Bilen 2%, other (Afar, Beni Amir, Nera) 5% (2010)	Dutch 76.9%, EU 6.4%, Turkish 2.4%, Moroccan 2.3%, Indonesian 2.1%, German 2.1%, Surinamese 2%, Polish 1%, other 4.8% (2018)

Comparison	Syria	Eritrea	the Netherlands
Sociodemographic char	acteristics		
Median age (2020)	Total: 23.5 years Male: 23 years Female: 24 years	Total: 20.3 years Male: 19.7 years Female: 20.8 years	Total: 42.8 years Male: 41.6 years Female: 44 years
Languages	Arabic (official), Kurdish, Armenian, Aramaic, Circassian, French, English	Tigrinya (official), Arabic (official), English (official), Tigre, Kunama, Afar, other Cushitic languages	Dutch (official), Frisian (official in the Fryslan province)
Literacy (% of age 15 and older who can read and write)	Total population: 86.4% Male: 91.7% Female: 81.0% (2015)	Total population: 76.6% Male: 84.4% Female: 68.9% (2018)	Total population: 99% Male: 99% Female: 99% (2003)
Religions	Islam 87%, Christianity 10%, Druze 3% ^b	Christianity 62.9%, Islam 36.6%, Others or None 0.5%	Roman Catholic 23.6%, Protestant 14.9%, Islam 5.1%, other 5.6% (includes Hindu, Buddhist, Jewish), none 50.7% (2017)
Labor and economic ch	aracteristics		
GDP - per capita (PPP)	\$2,900 (2015)	\$1,600 (2019)	\$55,200 (2020)
Population below the poverty line	82.5% (2014)	50.0% (2004)	13.6% (2019)
Unemployment rate	50.0% (2017)	5.8% (2017)	3.4% (2019)
Workforce participation by occupation	Agriculture: 17% Industry: 16% Services: 67% (2008)	Agriculture: 80% Industry: 20% (2004)	Agriculture: 1.2% Industry: 17.2% Services: 81.6% (2015)
Basic needs and health			
Drinking water source (% of the total population)	Urban: 99% Rural: 99.3% Total: 99.4%	Urban: 73.2% Rural: 53.3% Total: 57.8%	Urban: 100% Rural: 100% Total: 100%
Sanitation facility access (% of the total population)	Urban: 99.6% Rural: 98.6% Total: 99.1%	Urban: 44.5% Rural: 7.3% Total: 15.7%	Urban: 100% Rural: 100% Total: 100%
Physician density (physicians/1000 citizens)	1.29 (2016)	0.06 (2016)	3.61 (2017)
Hospital bed density (beds/1000 citizens)	1.4 (2017)	0.7 (2011)	3.3 (2017)
Legatum prosperity index (rank in 2021)	158	162	6
Cultural dimensions ^c			
Power distance	80	70	38
Individualism	35	20	80
Masculinity	52	65	14
Uncertainty avoidance	60	55	53
Long-term orientation	30	N.A.	67
Indulgence	N.A.	46	68

Note. The values between parentheses denote the years in which the information has been retrieved.

^a The ongoing civil war has altered the population distribution.

^b The Christian population might be considerably smaller as many Christians have fled the country during the recent civil war.

^c The scores on the cultural dimensions range from 1 (lowest) to 100 (highest). The scores of Ethiopia are used in the column of Eritrea, as no data on Eritrea is available, and Ethiopia and Eritrea have similar cultures.

CHAPTER

The Normative Judgment Test of Honesty-Humility: An Implicit Instrument for the Organizational Context

> Dan Asfar^{1,2} Janneke K. Oostrom¹ Mark Van Vugt¹ Marise Ph. Born^{1,3,4}

5

¹ Vrije Universiteit Amsterdam ² NOA B.V. ³ Erasmus University Rotterdam ⁴ North-West University

Paper submitted for publication

ABSTRACT

Implicit instruments to measure attitudes and personality have received increased attention from organizational scholars in recent years. One understudied implicit paradigm is the partially structured attitude measure, which assesses individuals' attributes through their judgments of hypothetical persons described in vignettes. Based on this paradigm, we developed the Normative Judgment Test to assess the personality trait of Honesty-Humility (the NJT-H). In two studies among employees (N = 230 and N = 124), we examined the NJT-H's construct- and criterion-related validity. In both studies, the NJT-H was significantly and positively related to Honesty-Humility, and not meaningfully related to the other five HEXACO traits. Furthermore, the NJT-H was negatively related to counterproductive work behavior and positively related to organizational citizenship behavior and task performance, as measured through self-ratings (Study 1) and supervisor ratings (Study 2). The NJT-H also explained unique variance in these outcomes above and beyond Honesty-Humility and the other five HEXACO traits. Altogether, these findings provide initial evidence of the practical value of the NJT-H in organizational contexts.

KEYWORDS

NJT-H, Honesty-Humility, personality measure, implicit, validity, work outcomes

THE NORMATIVE JUDGMENT TEST OF HONESTY-HUMILITY: AN IMPLICIT INSTRUMENT FOR THE ORGANIZATIONAL CONTEXT

There is a growing understanding of the notion that the greatest asset of any organization is its employees. Indeed, employee behaviors – specifically, behaviors that are aligned with organizational interests such as low counterproductive work behavior (CWB) and high organizational citizenship behavior (OCB) – have emerged as a key factor contributing to organizational performance (Bolino et al., 2012; Camara & Schneider, 1994; Vardi & Weitz, 2004). Personality traits are important predictors of these specific employee behaviors (e.g., Barrick & Mount, 1991; Ones et al., 1993; Tett et al., 1991). Hence, to select employees whose behaviors are aligned with organizational interests, that is, who act ethically and cooperatively, numerous organizations worldwide have included personality measures in their selection programs (Rothstein & Goffin, 2006). In recent years, there is growing evidence in the personality literature that the trait Honesty-Humility (H-H) within the HEXACO personality model is one of the strongest and most consistent trait predictors of CWB and OCB (Lee et al., 2019; Pletzer et al., 2019, 2020, 2021). Assessing trait H-H in selection and promotion contexts could therefore be beneficial for organizations.

Whilst personality traits are most typically assessed with self-report measures, scholars have called for more research on "innovative techniques that go beyond, without replacing, self-report measures..." (Funder, 2002, p. 639; see also Sackett et al., 2017). One of these innovative techniques are implicit instruments, which assess traits, motives, and attitudes that people might not be willing to disclose or are unaware of (Moors et al., 2010). Implicit instruments have been found to predict relevant work outcomes and explain variance in these outcomes above and beyond the variance explained by self-report measures of the same construct (see Uhlmann et al., 2012). The goal of the present research is to develop an implicit instrument of H-H that can be used in organizational contexts. This instrument, which we label the Normative Judgment Test of Honesty-Humility (the NJT-H), is based on the understudied partially structured attitude measure (PSAM; Vargas et al., 2004). The PSAM assesses individuals' trait levels through their judgments of ambiguous behaviors of hypothetical persons described in vignettes. Uhlmann et al. (2012) suggested that this paradigm is "... useful for predicting who is likely to engage in high levels of organizational citizenship or volunteering ... or for discerning an individual's standards for ethical behavior or corporate social responsibility" (p. 579). To address this suggestion, we conduct two studies to investigate the NJT-H's construct- and criterion-related validity for predicting CWB and OCB. In Study 2, we exploratorily examine the relationship between the NJT-H and task performance. We demonstrate that the NJT-H is a promising complement or alternative to self-report measures for predicting CWB, OCB, and task performance.

PREDICTORS OF CWB AND OCB

CWB and OCB are two important and commonly studied employee behavior (Dalal, 2005; Rotundo & Sackett, 2002; Sackett & DeVore, 2001). CWB has been defined as "voluntary behavior that violates significant organizational norms and in so doing threatens the wellbeing of an organization, its members, or both" (Robinson & Bennett, 1995, p. 556). CWBs have been linked to high profit loss (Camara & Schneider, 1994; Vardi & Weitz, 2004), poor team performance (Dunlop & Lee, 2004), and high levels of employee burnout (Mackey et al., 2019). Furthermore, OCB has been defined as "individual behavior that is discretionary, not directly or explicitly recognized by the formal reward system, and that in the aggregate promotes the effective functioning of the organization" (Organ, 1988, p. 4). OCBs have been linked to several organizational performance indices, such as turnover and customer satisfaction (Nielsen et al., 2009; Organ et al., 2006; Podsakoff et al., 2009) and individual level outcomes, such as managerial performance evaluations (e.g., Podsakoff et al., 2000), reward allocations (Podsakoff et al., 2009), and lower employee turnover (Mossholder et al., 2005; for a meta-analytic review of the consequences of OCB, see Podsakoff et al., 2009).

To be able to prevent harmful and promote desirable workplace behaviors, much research has been dedicated to the predictors of CWBs and OCBs (Berry et al., 2007; Harari & Viswesvaran, 2018). Several organizational characteristics have been associated with employees' levels of CWBs and OCBs, including organizational justice, leadership style, and team empowerment (Colquitt et al., 2013; Kirkman & Shapiro, 2001; Mitchell & Ambrose, 2007). Furthermore, individual differences as predictors of CWBs and OCBs – particularly those in personality – have received increased attention in recent years (Lee et al., 2019; Pletzer et al., 2020; Pletzer et al., 2021). Personality is most commonly described in terms of the Big Five (or FFM) dimensions: Openness, Conscientiousness, Extraversion, Agreeableness, and Neuroticism/Emotional Stability. However, in the last two decades, an increasing number of studies in the personality psychology literature have indicated that personality might be more optimally described in terms of six instead of five dimensions (Ashton et al., 2004; De Raad et al., 2014; Saucier, 2009). The HEXACO is the dominant six-dimensional personality framework, which consists of the dimensions Honesty-Humility (H-H), Emotionality (E), eXtraversion (X), Agreeableness (A), Conscientiousness (C), and Openness to Experience (O) (Ashton & Lee, 2007; Ashton et al., 2014). Whilst the HEXACO and the Big-Five model differ somewhat in the conceptualizations of the dimensions particularly the dimensions of Agreeableness and Emotional Stability/Emotionality - the main difference between the personality models is the addition of the Honesty-Humility (H-H) dimension in the HEXACO (e.g., see Ashton & Lee, 2020).

H-H is defined as "the tendency to be fair and genuine in dealing with others, in the sense of cooperating with others even when one might exploit them without suffering retaliation" (Ashton & Lee, 2007, p. 156). This trait captures an individual's tendency to refrain from

manipulation, fraud, and exploitation (Ashton & Lee, 2007; Lee & Ashton, 2004). In line with H-H's description, this trait correlates positively with overt and personality-based integrity measures, with typical correlations of above r = .50 (e.g., Lee et al., 2005; Lee et al., 2019; Marcus et al., 2007). The propensity of individuals low in H-H to deceive and exploit others makes them also "more likely to behave in their own interest at the expense of the best interest of their employer" (Oh et al., 2011, p. 500). Indeed, recent meta-analytic work has shown that H-H is a good predictor of CWB (r = .35 to .39) and explains unique variance in CWB, above and beyond the variance explained by the other five HEXACO traits (Lee et al., 2019; Marcus et al., 2007; Pletzer et al., 2019, 2020). Furthermore, H-H is positively related to ethical leadership (De Vries, 2012), and negatively to delinquent work behaviors (De Vries & Van Gelder, 2015; Lee et al., 2005) and unethical business decisions (Ashton & Lee, 2008; De Vries et al., 2017).

Individuals who score high on H-H also tend to feel responsible for behaving prosocially toward others (Hilbig & Zettler, 2009; Oh et al., 2014). At work, high H-H employees have a proclivity to behave cooperatively toward colleagues and to adhere to the organizational rules (Ashton & Lee, 2007; Bourdage et al., 2012), because they believe that it is their moral responsibility to behave as such (Marcus et al., 2007). Correspondingly, meta-analyses have revealed that H-H is also a positive predictor of OCB (r = .10 to .18; Lee et al., 2019; Pletzer et al., 2021).

IMPLICIT INSTRUMENTS

A considerable amount of empirical work has shown that people process information about themselves and their environment not only explicitly (i.e., controlled or conscious), but also implicitly (i.e., automatic or unconscious; Bargh & Chartrand, 1999; Dijksterhuis & Nordgren, 2006; Epstein, 1994; Fazio, 1990; Greenwald & Banaji, 1995). To assess such unconscious psychological attributes, several implicit instruments have been developed and empirically investigated (for an overview, see Uhlmann et al., 2012). Common implicit instruments are the picture story exercise (PSE; Schultheiss et al., 2008; similar to the thematic apperception test [TAT]; Morgan & Murray, 1935), the implicit association test (IAT; Greenwald et al., 1998), and the conditional reasoning test (CRT; James, 1998). There is substantial empirical evidence that these instruments predict employee behaviors and explain unique variance in these behaviors above and beyond the variance explained by self-report measures (Apers et al., 2019; Dietl & Meurs, 2019; Galić et al., 2014; Lang et al., 2012; Leavitt et al., 2011). Furthermore, due to their indirect nature, these implicit instruments have shown to be resistant to faking under specific conditions (e.g., LeBreton et al., 2007; Steffens, 2004; Vecchione et al., 2014; Wiita et al., 2020).

Despite the positive findings regarding the predictive validity and fakability of these commonly studied implicit instruments, they also have some important limitations. The PSE

(or TAT) is time-consuming to administer and score, requires extensive training to score the subjective interpretation of participants' responses, lacks face validity, leading to defensive responses by applicants, and its reliability and validity remain debated (Lilienfeld et al., 2000). The IAT has a poor test-retest reliability (Cunningham et al., 2001; Egloff et al., 2005; LeBel & Paunonen, 2011), and participants have trouble seeing its job relevance and feel like they have little opportunity to perform (Wright & Meade, 2011). Additionally, in one recent study, an IAT of H-H showed no evidence for the criterion-related validity (Van Rensburg et al., 2022). Finally, the CRT-A (the most frequently studied CRT, which assesses aggression) has a highly skewed distribution of test scores with a mean of 3.89 on a scale of 0 (no aggression) to 22 (extremely aggressive) (James & LeBreton, 2012), which makes it difficult to discriminate amongst individuals with low levels of aggression (DeSimone & James, 2015). The issues with these commonly studied implicit instruments encouraged us to address the call for the development of novel implicit psychological instruments (Funder, 2002; Sackett et al., 2017). As we will explain and show in the remainder of this manuscript, the implicit instrument that we introduce deals with most of the issues described above: The NJT-H is relatively easy to administer and quick to assess, is automatically scored and standardized, has a good reliability and validity, and has no skewed distribution of test scores.

THE PARTIALLY-STRUCTURED ATTITUDE MEASURE

In 2004, Vargas et al. introduced a new paradigm of implicit testing: The partiallystructured attitude measure (PSAM). In the PSAM paradigm, individuals judge trait levels of hypothetical persons who are described in vignettes. This paradigm is based on the phenomenon that the self serves as an anchor in social judgment (cf. Sherif & Hovland, 1961; for an overview, see Dunning, 2012). Hence, one's judgments are an indication of the trait level of the individual. Research has shown that people judge others in ways that promote their self-esteem (Beauregard & Dunning, 1998), such that people who are high (low) on a desirable attribute are more (less) judgmental of people who are low (high) on the desirable attribute (Beauregard & Dunning, 1998; Dunning & Cohen, 1992; Dunning & Hayes, 1996; Eidelman & Biernat, 2007; Protzko & Schooler, 2019). For example, Dunning and Cohen (1992) measured students' athleticism (hours per week on physical activity), math ability (SAT score), punctuality (number of times too late to class), studiousness (hours of studying), and how well-read they are (number of books read per month), and asked them to judge the levels of several hypothetical persons on these characteristics. The authors found that students with higher levels of (or scores on) these characteristics judged the hypothetical persons lower on the corresponding characteristics (except for punctuality). Correspondingly, the series of studies by Vargas et al. (2004) demonstrated that trait level estimations of hypothetical persons in vignettes that pertain to honesty, religiosity, and political orientation were inversely associated with participants' self-reported and actual behaviors relevant to these domains. For example, participants who perceived hypothetical persons who engaged in ambiguous dishonest behavior as very dishonest, were themselves less likely to cheat on anagrams. Importantly, these judgments explained unique variance in the corresponding outcomes above and beyond explicit measures of the same constructs.

In the present research, our goal was to develop an implicit instrument of H-H, the NJT-H, that is based on the PSAM paradigm (Vargas et al., 2004). The NJT-H is different from Vargas et al.'s (2004) PSAM of honesty, as the NJT-H assesses H-H, which is a broader construct than honesty, and as it contains 17 items (versus six items in Vargas et al. [2004]), increasing the reliability of the measure. The NJT-H is thought to predict CWB and OCB and explain unique variance in these employee behaviors, above and beyond the variance explained by personality self-reports (Uhlmann et al., 2012). We believe that H-H is a construct that can be adequately measured with the PSAM paradigm because it is closely related to the PSAM honesty measure in Vargas et al. (2004), who showed that this measure was negatively related to cheating on anagrams.

THE PRESENT RESEARCH

In the current research, we investigated the construct- and criterion-related validity of the NJT-H. The NJT-H's construct-related validity was established by examining its relationship with the HEXACO traits (Lee & Ashton, 2004). Implicit instruments generally show modest positive correlations with explicit measures of the same construct (e.g., Bosson et al., 2000; Hofmann et al., 2005; James et al., 2005; Vargas et al., 2004). Several explanations provided for the modest correlations between implicit and explicit instruments are the motivational bias in the report of consciously accessible representations (Fazio & Olson, 2003), a lack of introspective access to implicitly assessed representations (Greenwald & Banaji, 1995), and method-related characteristics of the two instruments (e.g., type of responses; see Payne et al., 2008). Based on the weak correlations found until now between implicit and explicit instruments, we posit the following hypothesis for the convergent validity of the NJT-H:

Hypothesis 1 (H1): The NJT-H is modestly and positively correlated with HEXACO H-H. 17

The NJT-H was developed to exclusively measure the trait H-H and no other personality traits. Thus, we posit the following hypothesis regarding the discriminant validity of the NJT-H:

Hypothesis 2 (H2): The NJT-H is not significantly correlated with any of the five other HEXACO traits.

¹⁷ Following Cohen (1988), we regard r = .10, r = .30, and r = .50 respectively as small, moderate, and large correlations.

People who score high on H-H tend to be honest, sincere, and greed avoidant. These are integrity-related traits and should be associated with the avoidance of undesirable and harmful behaviors. Indeed, H-H has been shown to be a strong negative predictor of CWB (Lee et al., 2019; Marcus et al., 2007; Pletzer et al., 2019, 2020). We are not aware of studies that have examined the relationship between an implicit assessment of H-H and employee behaviors. Therefore, the present study provides the first test of the criterion-related validity of an implicit H-H assessment for predicting CWB. Altogether, for the criterion-related validity, we posit the following hypothesis:

Hypothesis 3 (H3): The NJT-H is negatively correlated with CWB.

People who score high on H-H also tend to feel responsible for behaving prosocially toward others (Hilbig & Zettler, 2009; Oh et al., 2014). Correspondingly, meta-analyses have shown that H-H is a positive predictor of OCB (Lee et al., 2019; Pletzer et al., 2021). As described above, no implicit instrument of H-H has been studied in relation to OCB. The present study provides the first test of the criterion-related validity of an implicit H-H assessment for predicting OCB. Altogether, for the criterion-related validity, we posit the following hypothesis:

Hypothesis 4 (H4): The NJT-H is positively correlated with OCB.

As explained earlier, implicit instruments capture unique variance in attributes to explicit self-report measures of the same construct (e.g., Hofmann et al., 2005). Hence, implicit instruments can explain variance in work outcomes above and beyond the variance explained by traditional self-report measures (for an overview, see Uhlmann et al., 2012). In line with these findings, we propose the following two hypotheses:

Hypothesis 5 (H5): The NJT-H explains unique variance in CWB above the variance explained by HEXACO H-H.

Hypothesis 6 (H6): The NJT-H explains unique variance in OCB above the variance explained by HEXACO H-H.

STUDY 1

The goal of Study 1 was to examine the construct-related, criterion-related, and incremental validity of the NJT-H by examining its relationship with the HEXACO traits, CWB, and OCB.

METHOD STUDY 1

PARTICIPANTS AND PROCEDURE

The survey was distributed in the year 2019 within the personal networks of three international master's students at a Dutch university (mostly family members, friends, and colleagues, who all had a paid job), via an online link and through paper-and-pencil administration. Participants in the current sample typically had white-collar jobs. Participants signed an informed consent that included information about their rights, the data protection procedures, and the researcher's contact information. The inclusion criteria for the study were being at least eighteen years old, having a paid job for at least one year with two working days a week, and having sufficient English language proficiency (measured through self-evaluations). At the end of the survey, participants were debriefed about the purpose of the study and were thanked for their contribution. Participants received a small gadget in return for their participation.

A total of 246 participants completed the survey. Sixteen participants were excluded from the analyses because they did not meet the job contract inclusion criterion or because they had finished the survey in an unrealistically short time (i.e., less than 500 seconds; median survey response time = 19.15 minutes). The remaining 230 participants provided 99.7% power to detect a medium-sized effect for the relationships in this study (r = .30; Cohen, 1992), with α = .05. The sample (M_{are} = 34.99 years, SD = 13.30; 57.8% men) consisted of 152 employees and 78 students who had a part-time job. Most participants had obtained a bachelor's degree (n = 104; 45.2%), a master's degree (n = 48; 20.9%), or a high school degree (n = 35; 15.2%). The majority of the participants was born in the Netherlands (n =169, 73.5%), and the other participants were born in other European countries (n = 27; 11.7%), Asia (n = 27; 11.7%), or another continent (n = 10; 4.3%). The participants had 1 to 45 years of work experience (M = 14.81, SD = 11.75), and worked 8 to 60 hours per week (M = 32.75, SD = 11.75) 10.93). The participants who completed the online survey (n = 152) and the paper-and-pencil survey (n = 78) differed significantly in age (respectively M = 36.72, SD = 14.53 and M = 31.63, SD = 9.74; t[212.17] = 3.15, p = .002, d = 0.41) and working hours (respectively M = 30.28, SD= 11.71 and M = 37.50, SD = 7.20; t[219.66] = -5.75, p < .001, d = 0.74). However, controlling for these group differences in age and working hours did not affect the significance of any of the effects.

MATERIALS

Normative Judgment Test of Honesty-Humility (NJT-H)

The development of the NJT-H and the construction of its items was an iterative process. We followed test development recommendations by Lane et al. (2016), and developed 23 NJT-H-items based on the method of Vargas et al. (2004) to measure trait H-H. The items have the form of vignettes, followed by a question about the hypothetical person described in the vignette (for an example item, see Table 1). As Dunning and Cohen (1992) demonstrated that participants varied mostly in their evaluation of hypothetical persons who score low rather than high on an attribute, all vignettes described a hypothetical person who engaged in ambiguous behavior that can be considered low on H-H.

In the item development process, we first focused on the content-related validity (Colquitt et al., 2019). First, the authors of this manuscript, together with three master's students, designed 23 items in total. The items pertain to situations at work (11 items), but also other life domains (6 items). To capture all H-H aspects, each item relates to one of the four facets of H-H (Sincerity, Fairness, Modesty, and Greed avoidance). However, as our goal was to develop a relatively short instrument, we intended to measure H-H as a whole rather than its four underlying facets separately. Second, the authors and students had a group discussion about the clarity, face validity, and potential risk of bias of each item. This resulted in relatively small adjustment of most of the items. Third, the item pool was reviewed on these same aspects by fourteen I/O psychologists who work in a consultancy that develops and assesses psychological instruments. The I/O psychologists had 2 to 23 years (M = 9.0, SD = 7.73) of total work experience in psychological tests and assessments. Specifically, through a survey, we asked the reviewers to indicate for each item (a) whether it was clearly formulated and easy to understand in terms of word usage, complexity, and ambiguity (and if not, why this might be the case), (b) to what extent - using a 5-point Likert scale - it measures each of the six HEXACO personality dimensions, (c) if it might function differently for certain groups, such as men and women or different ethnic groups (and if so, why this might be the case), and (d) whether the reviewers had suggestions to improve the item. The review highlighted several issues in the items, and we adjusted these items accordingly. The formulations of fourteen items were adjusted to make them clearer. Formulations of thirteen items were adjusted because they could potentially measure other personality traits. Finally, five items were revised because these items might have been biased against particular social groups or might have been differently interpreted by different groups. In the Supplementary Material, an illustration is provided of how an item has been revised.

Next, we focused on the adjectives that we could use in the question that followed each of the vignettes (e.g., How [*adjective*, e.g., dishonest] do you consider Jane to be?). We favored custom adjectives for each item to default adjectives, as the hypothetical person could be more adequately judged when the adjective matches the vignette. To adopt the

most applicable adjectives, we first reviewed Lee and Ashton's (2008) list of adjectives that load on H-H, and then selected nine adjectives that are clear and easy to comprehend and that are applicable (e.g., dishonest, selfish).¹⁸ To determine which H-H-related adjective is the most applicable to each item, we presented 31 Amazon Mechanical Turk (MTurk) workers with the NJT-H items together with the nine adjectives. We asked them to indicate which adjective, according to them, is the most applicable to the hypothetical person in the vignette. For each item, we used the adjective that was selected the most often.

In the NJT-H, participants were asked to judge the hypothetical person's level of Honesty/ Humility described in the vignette using a 5-point Likert scale. The item response options varied from 1 = *not at all [adjective]* to 5 = *extremely [adjective]*. The adjectives were phrased negatively (e.g., dishonest, immodest). Higher scores were therefore anticipated to indicate higher levels of trait H-H. Due to practical constraints, only 12 of the 23 developed items were used in Study 1. However, Study 2 revealed that the complete NJT-H correlates highly with the 12-item NJT-H (r = .86, p < .001). We selected four items per H-H facet. We conducted an exploratory factor analysis with the 12 NJT-H-items using the principal axis factoring extraction method (e.g., Chuah et al., 2006). The Kaiser–Meyer–Olkin measure (Kaiser, 1970) verified the sampling adequacy for the analysis, KMO = .736 (see Hutcheson & Sofroniou, 1999). Bartlett's test of sphericity (χ^2 [66] = 330.62, p < .001) indicated that the correlations between items were sufficiently large for the principal axis factoring method. The scree plot showed an inflexion that would justify retaining one factor, which explained 23.5% variance. The alpha coefficient of the NJT-H in the current study was .69.

Table 1. Example Item of the Normative Judgment Test of Honesty-Humility (NJT-H)

Jane works in a lunchroom. In front of the lunchroom, there are usually two product promoters, distributing products like chocolate and drinks. Jane gives them a free lunch, and in return, they give her many of the products that they are supposed to distribute to the people outside. Jane takes these products home with her. How dishonest do you consider Jane to be? 1 = not at all dishonest 2 = a little dishonest 3 = moderately dishonest 4 = very dishonest 5 = completely dishonest

HEXACO

The HEXACO is a personality model that consists of six broad traits: Honesty-Humility (H-H), Emotionality (E), Extraversion (X), Agreeableness (A), Conscientiousness (C), and Openness to Experience (O). We used 32 items of the 100-item HEXACO-PI-R (Lee & Ashton, 2004)

120 | CHAPTER 5

¹⁸ The list of adjectives consisted of Dishonest, Insincere, Greedy, Arrogant, Self-centered, Selfish, Immodest, Egoistical, and Untruthful. The former four adjectives were most frequently selected and used in the items.

to assess Honesty-Humility ($\alpha = .81$) and Conscientiousness ($\alpha = .82$), because these are integrity-related traits (Marcus et al., 2007) that are usually negatively related to desirable and undesirable employee behaviors (e.g., De Vries et al., 2017; Lee et al., 2005; Lee et al., 2019). The other four personality dimensions (E, X, A, and O) were measured with the Brief HEXACO Inventory (BHI; De Vries, 2013). This inventory includes four items per dimension, where each item belongs to a unique facet of the dimension. The alpha coefficient in the current sample was .49 for E, .60 for X, .34 for A, and .38 for O. Despite these low alpha coefficients, research demonstrated a high temporal consistency of the BHI dimensions and little loss in construct validity compared to the full 200-item Dutch HEXACO-PI-R (De Vries, 2013). In Study 2, we have assessed a longer version of the HEXACO inventory. Example items are "I wouldn't use flattery to get a raise or promotion at work, even if I thought it would succeed" (H), "I have to cry during sad or romantic movies" (E), "I easily approach strangers" (X), "Even when I'm treated badly, I remain calm" (A), "I plan ahead and organize things, to avoid scrambling at the last minute" (C), "I like people with strange ideas" (O). In both the 100-item HEXACO-PI-R and the BHI, items were rated on a five-point Likert scale (1 = strongly disagree, 5 = strongly agree).

Counterproductive Work Behavior (CWB)

We used the short 10-item version of the Counterproductive Work Behavior Checklist (CWB-C; Spector et al., 2010) to measure CWB. Example items are "Made fun of someone's personal life" and "Started an argument with someone at work". Participants indicated on a 5-point Likert scale how frequently they engaged in each behavior, ranging from 1 = never to 5 = every day. Coefficient alpha was .78.

Organizational Citizenship Behavior (OCB)

We measured OCB with nine items from Smith et al. (1983), see Kelloway et al. (2002). Example items are "Helping other employees with their work when they have been absent" and "Volunteering to do things not formally required by the job". Participants indicated the extent to which each statement characterizes them, using a 5-point Likert scale that ranges from 1 = not at all to 5 = extremely. Coefficient alpha was .82.

RESULTS STUDY 1

We first compared the correlations of the NJT-H and HEXACO H-H with age, gender, and nationality. The correlation between HEXACO H-H and age (r = .42, p < .001) was significantly stronger (z = 3.73, p < .001) than the correlation between the NJT-H and age (r = .42, p < .001) was significantly women (M = 3.62, SD = 0.52) scored higher than men (M = 3.32, SD = 0.57) on HEXACO H-H (t[228] = -4.06, p < .001, d = 0.55), while the difference between women (M = 3.09, SD = 0.55) and men (M = 2.95, SD = 0.53) on the NJT-H did not reach below the significance threshold

(t[228]=-1.97, p = .050, d = 0.26). Dutch participants (M = 3.60, SD = 0.47) scored higher on HEXACO H-H than non-Dutch participants (M = 3.02, SD = 0.60; t[87.68] = 6.83, p < .001, d = 1.02). However, Dutch participants (M = 2.96, SD = 0.54) scored lower on the NJT-H than non-Dutch participants (M = 3.14, SD = 0.53; t[228] = -2.32, p = .021, d = -0.33). The correlation between the NJT-H and CWB was not significantly different (z = -0.07, p = .471) between Dutch (r = -.34, p < .001) and non-Dutch (r = -.33, p = .005) participants. Similarly, the correlation between the NJT-H and OCB was not significantly different (z = -0.33, p = .370) between Dutch (r = .21, p = .007) and non-Dutch (r = .26, p = .047) participants.

HYPOTHESIS TESTING

Means, standard deviations, and bivariate intercorrelations of the sociodemographic variables, the NJT-H, the HEXACO traits, CWB, and OCB are presented in Table 2. We hypothesized that the NJT-H correlates modestly and positively with HEXACO H-H (H1), but not with the other five HEXACO traits (H2). The results revealed a significant modest and positive correlation between the NJT-H and HEXACO H-H (r = .26, p < .001), supporting H1. No significant correlations were found between the NJT-H and any other HEXACO traits, supporting H2 (Table 2). Furthermore, we predicted that the NJT-H is negatively correlated with CWB (H3) and positively correlated with OCB (H4). Indeed, the results revealed a negative correlation between the NJT-H and CWB (r = ..32, p < .001), and a positive correlation between the NJT-H and CWB (r = ..32, p < .001), and a positive correlation between the NJT-H and OCB (r = ..21, p = .002), supporting H3 and H4.

We also formulated two hypotheses regarding the incremental validity of the NJT-H. We predicted that the NJT-H explains unique variance in CWB and OCB, above and beyond the variance explained by HEXACO H-H (H5 and H6, respectively). To test these hypotheses, hierarchical regression analyses were conducted with CWB and OCB as the dependent variables (Table 3). HEXACO H-H was included in the first step (Model 1), and the NJT-H was added in the second step (Model 2). In predicting CWB, H-H showed a significant negative beta weight in the first step ($\beta = -.41$, t = -6.84, p < .001), and the NJT-H additionally showed a significant negative beta weight in the first step ($\beta = -.41$, t = -6.84, p < .001), and the NJT-H additionally showed a significant negative beta weight in the second step ($\beta = -.23$, t = -3.78, p < .001), supporting H5. Model 2 explained 21.9% of the variance in CWB (F[2, 229] = 31.90, p < .001), that is, 4.9% more variance explained than the variance explained in Model 1. Furthermore, in predicting OCB, there was a nonsignificant beta weight of H-H in the first step ($\beta = .09$, t = 1.30, p = .195), and the NJT-H showed a significant positive beta weight in the second step ($\beta = .20$, t = 2.91, p = .004), supporting H6. Model 2 explained 4.3% of the variance in OCB (F[2, 227] = 5.10, p = .007), that is, 3.6% more variance explained than the variance explained 4.3% of the variance in OCB (F[2, 227] = 5.10, p = .007), that is, 3.6% more variance explained than the variance explained than the variance explained in Model 1.

We conducted additional analyses to test whether the NJT-H explains unique variance in CWB and OCB, above and beyond the variance explained by all the six HEXACO traits. The table with the results is included in the Supplementary Material (Table S2). In the hierarchical

regression analyses, the six HEXACO traits were included in the first step (Model 1), and the NJT-H was added in the second step (Model 2). In predicting CWB, H-H (β = -.34, t = -5.33, p < .001) and Conscientiousness (B = -.21, t = -3.17, p = .002) showed significant negative beta weights in the first step. This model explained 21.8% of the variance in CWB (FIG. 229) = 10.34, p < .001). Model 2 showed that the NJT-H is significantly negatively related to CWB $(\beta = -.23, t = -3.82, p < .001)$, and explained 4.8% unique variance in CWB, above and beyond the variance explained by the six HEXACO traits (ΔF [1, 222] = 14.55, p < .001). Furthermore, in a hierarchical regression analysis with OCB as the dependent variable, Extraversion $(\beta = .26, t = 3.99, p < .001)$ and Conscientiousness $(\beta = .33, t = 5.06, p < .001)$ were significantly positively related to OCB. This model explained 22.3% of the variance in OCB (F[6, 227] = 10.57, p < .001). Model 2 showed that the NJT-H is significantly positively related to OCB (B = .22, t = 3.59, p < .001), and explained 4.3% unique variance in OCB, above and beyond the variance explained by the six HEXACO traits (ΔF [1, 220] = 12.87, p < .001).

Table 2. Mec	ans, Stand	ard Deviati	ions, and E	3ivariate In	tercorrelat	tions of Var	iables in S	tudy 1						
Variable	Ν	SD	1	2	ω	4	5	9	7	00	6	10	11	12
1. Gender	0.42	0.50	(-)											
2. Age	34.99	13.30	80.	(-)										
3. WH	32.75	10.93	29**	.26**	(-)									
4. NJT-H	3.00	0.54	.13	.14*	60.	(69)								
5. H	3.45	0.57	.26**	.42*	01	.26**	(.81)							
6. E	2.89	0.68	.43**	03	20**	.12	03	(.49)						
7. X	3.80	0.65	.21**	.12	02	.03	.25**	09	(09.)					
8. A	2.97	0.57	04	05	04	.10	.07	08	07	(.34)				
9. C	3.56	0.51	.21**	02	.02	60 [.]	.34**	03	.37**	02	(.82)			
10. O	3.55	0.58	03	80.	.04	07	.12	05	.24**	08	.18**	(.38)		
11. OCB	3.39	0.67	:03	-00	.06	.21**	60.	12	.35**	.02	.40**	80.	(.82)	
12. CWB	1.70	0.49	16*	22**	.01	32**	41**	90.	.13	.13	31**	06	06	(.78)
Note. Gender: Agreeablenes. scores for the N = 230. * $p < .05$; ** $p <$	male = 0, s, C = Cons scales is 1.0 : .01 (two-tail	female = 1, cientiousnes 0-5.00. lec).	WH = Worki ss, O = Opei	ing hours, Nnness to Exi	UT-H = Nor perience, O	mative Judg CB = Organi	ment Test c zational citiz	of Honesty-F zenship beh	avior, CWB -	= Counterpri	umility, E = oductive wo	Emotionality, ork behavior.	, X = Extrav The possib	le range of
Table 3. Hier	rarchical R	egression	Analyses I	vith HEXA	CO H-H an	d the NJT-,	H as Prediv	ctors of CV	VB and OC	CB in Study	1			
				CWB							OCB			
		Moc	del 1			Model 2			Model	-			Model 2	

R^2	.170	.219	.007	
F	46.81***	31.90***	1.69	
ΔR^2		.049		
ΔF		14.26***		

[-.10; .17] [.06; .33]

β 20**

22]

[-.04;

60

[-.47; -.23] [-.35; -.11]

35***

6

 \overline{O} 95% 53;

9 4

H NJT-H R²

Model 1

.043 5.10** .036

±0**

ive work behav

Note. H = Honesty-Humility, UCB = Urgain N = 230. * p < .05; ** p < .01; *** p < .001 (two-tailed).

5

DISCUSSION STUDY 1

The goal of Study 1 was to examine the construct-related, criterion-related, and incremental validity of the NJT-H. In line with the hypotheses, the results showed that the NJT-H (a) is modestly and positively associated with HEXACO H-H and not significantly associated with the other five HEXACO traits, (b) is negatively associated with CWB and positively associated with OCB, and (c) explains unique variance in CWB and OCB, above and beyond the variance explained by HEXACO H-H. Furthermore, the additional analyses showed that the NJT-H explained unique variance in CWB and OCB, above and beyond the variance by the six HEXACO traits. We also found that, with respect to age, gender, and nationality, score differences were significantly smaller on the NJT-H than on the HEXACO H-H scale. Altogether, these findings provide initial support for the validity of the NJT-H.

However, the findings of Study 1 are subject to at least three limitations. First, the employee behaviors were measured using self-reports, which may have inflated the relationship between the predictors and the criteria due to common source bias (Meier & O'Toole, 2013; Podsakoff et al., 2003). However, as research shows that other-reports add little to the assessment of CWB to self-reports (Berry et al., 2012), this limitation pertains primarily to the measurement of OCB. Nonetheless, it is important to also examine the NJT-H's criterion-related validity using other-reports of work outcomes (e.g., Jaramillo et al., 2005). Second, only a selection of NJT-H items was used in this study, which may have reduced the reliability and, hence, the validity of the test. Therefore, we expect to find a higher validity with more NJT-H items. Third, four of the HEXACO traits were measured with the BHI. Although the validity of the BHI has been demonstrated (De Vries, 2013), it is important to test the convergent and discriminant validity and the incremental validity of the NJT-H with a longer version of the HEXACO inventory.

In Study 1, we focused on CWB and OCB. Another important employee behavior is task performance, which has been defined as "activities that contribute to the organization's technical core either directly by implementing a part of its technological process, or indirectly by providing it with needed materials or services" (Borman & Motowidlo, 1997, p. 99). Whereas a conceptual and empirical link could be made between H-H and CWB and OCB, there is no such evident link between H-H and task performance. So far, only a few studies have examined H-H as a predictor of task performance. In a recent metaanalysis with seven studies, a weak positive relationship was found between H-H and task performance (Lee et al., 2019). However, some studies revealed no effect (e.g., Oh et al., 2014). Furthermore, there is no consistent evidence that dark triad traits - traits that are associated with callous, selfish, and malevolent interpersonal behaviors (Paulhus & Williams, 2002), and that have high conceptual similarities with the opposite end of H-H (e.g., Hodson et al., 2018) – are related to task performance (O'Boyle et al., 2012). Altogether, there is currently only indirect and inconsistent support for a relationship between H-H and task performance. Therefore, we propose two research questions for the relationship between the NJT-H and task performance:

Research Question 1 (RQ1): To what extent is the NJT-H correlated with task performance?

Research Question 2 (RQ2): To what extent does the NJT-H explain unique variance task performance, above and beyond the variance explained by HEXACO H-H?

In addition to exploratorily examining the relationship between the NJT-H and task performance, another goal of Study 2 was to provide another test of our hypotheses based on self-ratings of CWB and supervisor ratings of CWB and OCB. In the analyses with supervisor ratings, we controlled for the supervisor-subordinate interaction frequency, because this could influence the supervisor ratings (Kacmar et al., 2003). Furthermore, we administered the 23 NJT-H items and the HEXACO-60 inventory (Ashton & Lee, 2009). Finally, we used longer and more common scales of CWB and OCB.

METHOD STUDY 2

PARTICIPANTS

The participants in the present study are employees and their supervisors. Employees had to meet four criteria to participate in this study: Being between the age of 18 and 65 years

old, being born in the Netherlands, having worked for the current company for at least six months with a minimum of 16 hours per week, and having sufficient English proficiency to understand the survey. Supervisors also had to meet four criteria to participate in this study: Being between the age of 18 and 65 years old, having supervised the employee for a minimum of three months, having interacted with the employee at least once a week, and having sufficient English proficiency to understand the survey (measured through self-evaluations). Participants in the current sample typically had white-collar jobs.

The final dataset consisted of 123 employees (78 male; 63.4%) and 93 employee-supervisor dyads (59 male employees and 55 male supervisors). The power analysis indicated that we obtained 93.3% power with 123 participants and 85.1% power with 93 participants to detect a medium-sized effect for the relationships in this study (r = .30; Cohen, 1992), with $\alpha = .05$. On average, the employees were 32.89 years old (SD = 10.49). Most employees had obtained a bachelor's degree (n = 72; 58.5%), a master's degree (n = 22; 17.9%), or an associate degree (n = 21; 17.1%). Most of the employees (100; 81.3%) had a permanent contract. The employees had 2 to 45 years of work experience (M = 13.90, SD = 10.81), and had an average organizational tenure of 3.80 years (SD = 5.28) with working hours ranging from 16 to 60 hours per week (M = 38.11, SD = 8.16). The supervisors were on average 38.52 years old (SD = 8.62). Most supervisors had obtained a bachelor's degree (n = 45; 48.4%) or a master's degree (n = 37; 39.8%). The average supervisors' organizational tenure was 6.84 years (SD = 6.99), and their average managerial experience was 7.55 years (SD = 6.92). The supervisors worked 43.26 hours per week on average (SD = 9.44), varying from 24 to 70 hours. The employee-supervisor dyads worked together for 1.88 years on average (SD = 2.35).

PROCEDURE

The data were collected in the Netherlands in the year 2020. The employee-supervisor dyads were recruited by three master's students who worked collaboratively on a research project. Participants were recruited through the students' internship providers and through a few other organizations located in the Netherlands. First, the students contacted potential supervisors, explained the aim of the research, and asked them to participate in this study. If supervisors were willing to participate in this study, they were asked to provide the names of their subordinates who met the participation criteria. Second, the students contacted these employees, explained the aim of the research, and asked them to participate in this study. The employees were asked to complete the survey that includes the NJT-H, the HEXACO, and a scale for CWB. At the end of the survey, the employees had to enter a random 6-digit code. The employees were asked to share this code with their supervisor, together with the supervisor's survey link that was provided at the end of their survey. Subsequently, at the beginning of the supervisor's survey, the supervisors entered the 6-digit code. This survey includes a scale for CWB, OCB, and task performance. This procedure enabled us to match the surveys of the employees and the supervisors while maintaining participants' anonymity

and the confidentiality of their responses. In the consent form, participants were informed about their rights and the anonymous data processing and the confidential treatment of their data for scientific research. In the debriefing of the surveys, the goal of the study was explained, and the participants were thanked for their participation. The employee survey took about 15-20 minutes to complete, and the supervisor survey took about 10-15 minutes to complete. The present research has been approved by the faculty's ethics committee.

MATERIALS

Normative Judgment Test of Honesty-Humility (NJT-H)

All 23 NJT-H items were administered. However, based on the psychometric properties of the items, we selected 17 items to form the final version of the NJT-H. However, we also conducted the analyses with the 23 NJT-H items (see Supplementary Material). Overall, the 17-item NJT-H showed comparable or slightly stronger effects than the 23-item NJT-H. The alpha coefficient of the NJT-H in the current sample was .77. In Study 1, we conducted an exploratory factor analysis, and found a single factor explaining most of the variance in test scores. To confirm the unidimensional factor structure of the NJT-H, we conducted a confirmatory factor analysis where we compared a single-factor model with a four-factor model (representing the four H-H facets). The models were analyzed using the maximum likelihood estimation method, and missing values were dealt with using full information maximum likelihood (El-Sheikh et al., 2017). The analyses indicated that the single-factor model (x² [119] = 206.77, p < .001, CFI = .703, TLI = .660, RMSEA = .077, SRMR = .086) showed a comparable fit to the four-factor model (χ^2 [113] = 194.25, p < .001, CFI = .725, TLI = .669, RMSEA = .076, SRMR = .082), χ^{2} (6) = 12.51, p = .052. Given these results, we treat the NJT-H as a unidimensional construct. The item-level factor loadings and the zero-order correlations between the NJT-H items and the HEXACO dimensions are reported in Table S1 (see Supplementary Material).

To provide more evidence for the reliability of the NJT-H, we conducted a small online study in Prolific among 140 participants who completed the 17 NJT-H items. Two and a half weeks later, we asked the participants to complete the 17 NJT-H items again. In total, 104 participants (53.8% female; M_{age} = 35.35, SD = 12.88) completed the NJT-H twice (i.e., a retention rate of 74.3%). The average time interval between the two test administrations was 15.71 days (SD = 1.00, min = 15 days, max = 19 days). The test-retest reliability of the NJT-H was r = .71. This test-retest reliability is comparable to the test-retest reliability of the CRT-A (James & LeBreton, 2012), and higher than several other implicit instruments such as the IAT (Cunningham et al., 2001; Egloff et al., 2005; LeBel & Paunonen, 2011) and the TAT (Lilienfeld et al., 2000).

HEXACO

We used the HEXACO-60 (Ashton & Lee, 2009; De Vries et al., 2009) to measure the six

HEXACO traits among employees. Items were rated on a 5-point Likert scale (1 = *strongly disagree*; 5 = *strongly agree*). Example items are provided in Study 1. The alpha coefficients in the current study were .71 for H-H, .69 for E, .80 for X, .67 for A, .79 for C, and .73 for O.

Counterproductive Work Behavior (CWB)

The 19-item CWB scale by Bennett and Robinson (2000) was administered to both the employees and their supervisors. Example items are "Made fun of someone at work" and "Taken property without permission". Items were rated on a 5-point Likert scale, ranging from 1 = never to $5 = every \, day$. Coefficient alpha was .83 for both the employee self-reports and the supervisory ratings.

Organizational Citizenship Behavior (OCB)

We measured supervisor ratings of OCB with the 16-item OCB scale by Lee and Allen (2002). Supervisors were asked to indicate how often their subordinate(s) engaged in certain behaviors, ranging from 1 = never to 5 = always. Example items are "Defend the organization when other employees criticize it" and "Willingly give their time to help others who have work-related problems". Coefficient alpha was .89.

Task Performance

To measure supervisory ratings of task performance, we used nine items from Goodman and Svyantek (1999). Example items are "Achieves the objective of the job" and "Demonstrates expertise in all job-related tasks". Items were rated on a 5-point Likert scale, ranging from 1 = *strongly disagree* to 5 = *strongly agree*. Coefficient alpha was .86.

Interaction Frequency

Supervisor ratings are affected by the frequency of interaction between the supervisor and their subordinate (Kacmar et al., 2003). To take into consideration this potential confound, supervisors completed a 4-item scale by McAllister (1995) that measures the employee-supervisor interaction frequency. An example item of this scale is "How frequently do you interact with this person at work informally or socially?". Response alternatives ranged from 1 = *once or twice in the last 6 months* to 7 = *many times daily*. Higher scores on this scale indicate perceptions of a higher communication frequency. Coefficient alpha was .87.

RESULTS STUDY 2

In the current study, there were no significant differences between men and women on the NJT-H (respectively M = 3.64, SD = 0.46 and M = 3.59, SD = 0.45; t[119] = 0.63, p = .530, d = 0.11), HEXACO H-H (respectively M = 3.51, SD = 0.58 and M = 3.59, SD = 0.56; t[118] = -0.67, p = .506, d = 0.14), or any other personality trait. There were only significant differences between men (M = 40.39, SD = 7.07) and women (M = 34.00, SD = 8.64) on their working hours per week, t(73.26) = 4.14, p < .001, d = 0.81. Furthermore, employees' age was positively correlated with the NJT-H (r = .32, p < .001) and HEXACO H-H (r = .36, p < .001), and these correlations did not differ significantly from each other (z = 0.39, p = .349). Age was negatively correlated with supervisory ratings of CWB (r = .24, p = .024) and positively correlated with their ratings of OCB (r = .30, p = .004). Age was not significantly correlated with supervisory ratings of task performance (r = .06, p = .593). Finally, interaction frequency was negatively correlated with Conscientiousness (r = .21, p = .041), and positively correlated with supervisory ratings of CWB (r = .25, p = .015).

HYPOTHESIS TESTING

Means, standard deviations, and bivariate intercorrelations of sociodemographic variables, the NJT-H, the HEXACO traits, and the employee behaviors are presented in Table 4. The results revealed a significant modest positive correlation between the NJT-H and HEXACO H-H (r = .43, p < .001), supporting H1. Furthermore, the NJT-H showed no significant correlations with the other HEXACO traits, except a weak correlation with Conscientiousness (r = .19, p = .041), largely supporting H2. Furthermore, we predicted that the NJT-H is negatively correlated with CWB (H3). We found a correlation of r = ..31 (p = .001) between the NJT-H and self-ratings of CWB, and a correlation of r = ..46 (p < .001) between the NJT-H and supervisory ratings of CWB, supporting H3. We also predicted that the NJT-H is positively correlated with OCB (H4). Indeed, there was a positive correlation between the NJT-H and supervisory ratings of OCB (r = .48, p < .001), supporting H4.

We expected that the NJT-H explains unique variance in CWB (H5) and OCB (H6), above and beyond the variance explained by HEXACO H-H. To test these hypotheses, hierarchical regression analyses were conducted with self-reported CWB and supervisory ratings of CWB and OCB as the dependent variables (Table 5). For self-ratings of CWB, HEXACO H-H was included in the first step (Model 1), and the NJT-H was added in the second step (Model 2). For supervisory ratings of CWB and OCB, interaction frequency was included as a control variable in the first step (Model 1), the HEXACO H-H scale was included in the second step (Model 2), and the NJT-H was added in the third step (Model 3). In the Supplementary Material, we have included the results of the hierarchical regression without the control variable interaction frequency (Table S5). The results without interaction frequency are comparable to the results of the analyses that are reported in the main text.

Table 4. Means, Sto	andard D	eviation	is, and B	ivariate i	Intercorr	elations	of Varic	ibles in S	study 2								
Variable	Μ	SD	1	2	ŝ	4	5	6	7	8	6	10	11	12	13	14	15
1. Gender	0.36	0.48	T														
2. Age	32.89	10.49	02	ı													
3. WH	38.11	8.16	37**	11	,												
4. IF	5.21	1.10	05	16	.16	(88)											
5. NJT-H	3.45	0.43	06	.32**	.15	01	(.77)										
6. H	3.55	0.57	90.	.36**	.08	, - - -	.43**	(.71)									
7. E	2.96	0.52	.50**	02	10	.03 0	01	.01	(69)								
8. X	3.61	0.56	-00	05	.19*	.19	.18	.07	-11	(.80)							
9. A	3.13	0.49	04	.06	.04	0.	14	.33**	02	.01	(.67)						
10. C	3.63	0.54	.13	:03	00	22*	.19*	.35**	60.	60 [.]	.05	(.79)					
11. 0	3.41	0.57	07	08	11	.10	03	02	-00	.05	.19*	03	(.73)				
12. OCB	3.53	0.57	06	.30**	.25*	11	.48**	.37**	60.	.24*	.15	.17*	01	(8.9)			
13. CWB-E	1.51	0.36	07	17	.08	.1	31**	41**	00.	15	-20*	28**	.12	12	(.83)		
14. CWB-S	1.28	0.29	.04	24*	.03	.25*	46**	45**	.05	21*	-11	30**	24*	42**	.46**	(.83)	
15. TP	3.87	0.50	-11	06	.07	03	.23*	.04	.10	.10	.05	.18	60.	.43**	.03	33**	(98.)
Note. Gender: male =	0, female	= 1, WH =	= Working	hours, IF	= Interac	tion Freq	uency, N.	JT-H = No	rmative J	udgment	Test of H	Honesty-H	umility, H	= Hones	ty-Humilit	y, E = Emo	otionality,

behavior ept for IF, work b Counterproductive w the scales is 1.00-5.00, behavior, CWB-E = nge of scores for th citizenship t possible rang OCB = Organizational e. 0el ience, = d1 Exper ō. Q. Openness O = Oess, beha Conscientiousn uctive . U U ersion, A = Agreeableness, C = self-reports, CWB-S = Counterr a possible range of 1.00-7.00. self-r self-r s a pr io N = ployee s ch has a 93 to N < .05; ** s ∼ s

a possible יייין 1 = 123. * מ < .01 (two-tailed).

In the hierarchical regression analysis with self-ratings of CWB as the dependent variable, H-H showed a negative beta weight in Model 1 ($\beta = -.41$, t = -4.91, p < .001), explaining 16.8% of the variance in CWB self-ratings (F_{1} , 121] = 24,15, p < .001). Model 2 showed that the NJT-H $(\beta = -16, t = -1.80, p = .075)$ did not significantly explain unique variance in self-ratings of CWB above and beyond the variance explained by HEXACO H-H (F[2, 121] = 13.92, p < .001). In the hierarchical regression analysis with supervisory ratings of CWB as the dependent variable, interaction frequency ($\beta = .25$, t = 2.47, p = .015) explained 6.3% variance (F[1, 92]) = 6.11, p = .015). In Model 2, H-H (β = -.43, t = -4.60, p < .001) showed a significant negative beta weight, explaining 24.1% of the variance in supervisory ratings of CWB (F[2, 92] = 14.30, p < .001). Model 3 showed that the NJT-H is significantly negatively related to supervisory ratings of CWB (β = -.34, t = -3.49, p = .001), and explained 33.2% variance in supervisory ratings of CWB (F[3, 92] = 14.77, p < .001), that is, 9.1% unique variance above and beyond the variance explained in Model 2. The hierarchical regression analysis without the control variable interaction frequency showed that H-H (β = -.45, t = -4.77, p < .001) and the NJT-H $(\beta = -.32, t = -3.28, p = .001)$ maintained significant negative beta weights. Altogether, these results provide partial support for H5.

In the hierarchical regression analysis with supervisory ratings of OCB as the dependent variable, interaction frequency ($\beta = .11$, t = 1.06, p = .290) showed no significant beta weight. In Model 2, H-H (β = .39, t = 3.99, p < .001) showed a significant and positive beta weight, explaining 16.0% of the variance in supervisory ratings of OCB (F[2, 92] = 8.60, p < .001). Model 3 showed that the NJT-H is significantly and positively related to supervisory ratings of OCB (β = .39, t = 3.89, p < .001), and explained 28.6% variance in supervisory ratings of OCB (F[3, 92] = 11.60, p < .001), that is, 12.1% additional variance explained above and beyond the variance explained in Model 2. The hierarchical regression analysis without the control variable interaction frequency showed that H-H (β = .20, t = 2.02, p = .047) and the NJT-H ($\beta = .39$, t = 3.91, p < .001) maintained significant negative beta-weights. Altogether, these results provide support for H6.

We conducted additional analyses to test whether the NJT-H explains unique variance in CWB and OCB, above and beyond the variance explained by all the six HEXACO traits. The table of the results is included in the Supplementary Material (Table S6). In the hierarchical regression analyses, interaction frequency was included in the first step (Model 1), the six HEXACO traits were included in the second step (Model 2), and the NJT-H was added in the third step (Model 3). Step 1 was not applied in the analyses with CWB self-reports. The results showed that in predicting CWB self-reports, the NJT-H did not explain unique variance in CWB self-reports above and beyond the variance explained by the six HEXACO traits ($\beta = -.07$, t = -0.74, p = .461). However, in predicting supervisory ratings of CWB, the NJT-H (β = -.31, t = -3.27, p = .002) explained 7.4% unique variance above and beyond the variance explained by the six HEXACO traits (ΔF [1, 84] = 10.71, p = .002). Finally, in

132 | CHAPTER 5

predicting supervisory ratings of OCB, the NJT-H (β = .36, *t* = 3.51, *p* = .001) explained 10.1% unique variance in supervisory ratings of OCB (ΔF [1, 84] = 11.67, *p* = .001). These additional analyses showed that the NJT-H also explained unique variance in work outcomes above and beyond the complete HEXACO.

Table 5. Hierarchical Regression Ana	yses with Predictors of	^c CWB and OCB in Stud	y 2
--------------------------------------	-------------------------	----------------------------------	-----

		Model 1			Model 2	
	β		95% CI	β		95% CI
CWB-E		$R^2 = .16$	3		$R^2 = .226 (\Delta R^2 = .$	022)
Н	41	***	[57;24]	34*	*** [52;16]
NJT-H				16	1	35; .02]
	Μ	odel 1	Ν	lodel 2	M	odel 3
	β	95% CI	β	95% CI	β	95% CI
CWB-S	R^2	= .063	$R^2 = .24$	1 (<i>△R</i> ² = .178)	R ² = .332	2 (<i>∆R</i> ² = .091)
IF	.25*	[.05; .45]	.21	[.02; .39]	.22*	[.05; .39]
Н			43**	* [61;24]	28**	[47;09]
NJT-H					34**	[53;14]
	Μ	odel 1	N	lodel 2	M	odel 3
	β	95% CI	β	95% CI	β	95% CI
OCB	R^2	= .012	R ² = .16	O (<i>∆R</i> ² = .148)	$R^2 = .281$	(<i>∆R</i> ² = .121)
IF	.11	[10; .32]	.15	[04; .35]	.10	[09; .30]
Н			.39**	* [.19; .58]	.22*	[.02; .42]
NJT-H					.39***	[.19; .58]

Note. IF = Interaction Frequency, H = Honesty-Humility, NJT-H = Normative Judgment Test of Honesty-Humility, CWB-E = Counterproductive work behavior employee self-reports, CWB-S = Counterproductive work behavior supervisory rating, OCB = Organizational citizenship behavior.

N = 123 and N = 93.

* p < .05; ** p < .01; *** p < .001 (two-tailed).

EXPLORATORY RESEARCH QUESTIONS

We posited two exploratory research questions for the relations between the NJT-H and task performance. In RQ1, we proposed examining to what extent the NJT-H is correlated with task performance. The results showed that the NJT-H is significantly and positively correlated with task performance, r = .23 (p = .025). In RQ2, we proposed examining to what extent the NJT-H explains unique variance in task performance, above and beyond the variance explained by HEXACO H-H (Table 6). After controlling for interaction frequency ($\beta = .03, t = .025, p = .805$), H-H showed no significant beta weight for supervisory ratings of task performance ($\beta = .04, t = 0.35, p = .724$). Furthermore, the NJT-H ($\beta = .27, t = 2.33, p = .022$) showed a significant and positive beta weight in predicting supervisory ratings of task performance, explaining 5.7% additional variance above and beyond HEXACO H-H (ΔF [1, 89] = 5.42, p = .022). Additionally, the NJT-H ($\beta = .25, t = 2.13, p = .036$) also explained 5.6% unique variance in task performance above and beyond the variance explained by the six HEXACO traits (ΔF [1, 84] = 5.31, p = .024; see Supplementary Material, Table S7).

Altogether, these exploratory research analyses showed that the NJT-H is positively related to task performance, and explains unique variance in task performance, above and beyond the variance explained by HEXACO H-H and the complete HEXACO.

Table 6. Hierarchical Regression Analyses with Predictors of Task Performance in Study 2

	Mc	odel 1	M	odel 2	M	odel 3
	β	95% CI	β	95% CI	β	95% CI
TP	R^2	= .001	$R^2 = .002$	$(\Delta R^2 = .001)$	$R^2 = .059$	$(\Delta R^2 = .057)$
IF	03	[23; .18]	02	[23; .19]	03	[24; .17]
Н			.04	[17; .25]	08	[31; .15]
NJT-H					.27*	[.04; .49]

Note. IF = Interaction Frequency, H = Honesty-Humility, NJT-H = Normative Judgment Test of Honesty-Humility, TP = Task performance.

N = 93.

* ρ < .05; ** ρ < .01; *** ρ < .001 (two-tailed).

DISCUSSION STUDY 2

Overall, Study 2 provides additional support for the construct- and criterion-related validity of the NJT-H (for an overview of the results of Study 1 and Study 2, see Table 7).

In line with our expectations, the NJT-H is modestly and positively associated with HEXACO H-H. Furthermore, of the other five HEXACO traits, the NJT-H only shows a weak positive correlation with Conscientiousness. The NJT-H is negatively associated with self-ratings and supervisory ratings of CWB and is positively associated with supervisory ratings of OCB, and explains unique variance in the supervisory ratings of CWB and OCB, above and beyond the variance explained by HEXACO H-H. Additional analyses showed that the NJT-H also explains unique variance in supervisory ratings of CWB and OCB, above and beyond the variance explained by the six HEXACO traits. Lastly, the exploratory research analyses showed that the NJT-H is positively associated with task performance, and explains unique variance in task performance, above and beyond the variance explained by HEXACO traits.

 Table 7. An Overview of the Hypotheses, Research Questions, and their Empirical Support

#	Hypothesis or research question	Support
H1	The NJT-H is modestly and positively correlated with HEXACO H-H.	Yes
H2	The NJT-H is not significantly correlated with any of the five other HEXACO traits.	Partial support (weak correlation with Conscientiousness)
H3	The NJT-H is negatively correlated with CWB.	Yes
H4	The NJT-H is negatively correlated with OCB.	Yes
H5	The NJT-H explains unique variance in CWB above the variance explained by HEXACO $\operatorname{H-H}$	Yes
H6	The NJT-H explains unique variance in OCB above the variance explained by HEXACO H-H.	Yes
RQ1	To what extent is the NJT-H correlated with task performance?	r = .25
RQ2	To what extent does the NJT-H explain unique variance task performance, above and beyond the variance explained by HEXACO H-H?	5.7% unique variance explained

GENERAL DISCUSSION

The present research addresses recent calls from scholars to examine and advance alternative personality assessment methods, such as implicit instruments, for organizational contexts (Funder, 2002; Sackett et al., 2017). Here, we have developed the NJT-H, an implicit instrument of H-H that is based on the PSAM paradigm by Vargas et al. (2004). The NJT-H was subject to validation in this research. In two studies, we assessed the NJT-H, the HEXACO traits, and self-reported and supervisory ratings of the employees' CWB, OCB, and task performance. The findings provide initial support for the construct- and criterion-related validity and the incremental validity of the NJT-H.

THEORETICAL IMPLICATIONS

The present study contributes to the literature on implicit instruments of personality (Back et al., 2009; Bosson et al., 2000; Hofmann et al., 2005; James et al., 2005) by validating an implicit instrument of H-H based on the PSAM paradigm. Whilst Vargas et al. (2004) provided a proof of concept of the PSAM, our research with the NJT-H demonstrates that this paradigm is useful for assessing trait H-H in the organizational context. Specifically, we find support for the construct-related validity of the NJT-H in both studies. The current research reveals a modest and positive relationship between the NJT-H and HEXACO H-H, and the NJT-H showed no significant correlations with the other HEXACO traits (apart from a weak significant correlation with Conscientiousness in Study 2). In line with previous studies on implicit instruments (e.g., Back et al., 2009; James et al., 2005), we treated a modest positive correlation between scores on the implicit and explicit instrument of the same trait as evidence for the convergent validity of the instrument. In line with this interpretation, we argue that it is important to develop test validation guidelines for implicit instruments that acknowledge modest correlations with explicit instruments of the same construct as evidence for the convergent validity.

Moreover, the current research contributes to the literature on employee work behavior by revealing that the NJT-H predicts CWB, OCB, and task performance, and also explains unique variance in these employee behaviors above and beyond the variance explained by HEXACO H-H and the six HEXACO traits. Thus, the NJT-H assesses unique variance in one's personality to a personality self-report measure, and adding the NJT-H to a personality selfreport measure makes it possible to more accurately predict employees' CWBs, OCBs, and task performance. This is an important and a promising finding, as previous research has shown that only trait Conscientiousness is a consistent predictor of these three employee behaviors (Connelly & Ones, 2010; Lee et al., 2019). This also raises the anticipation that an NJT of Conscientiousness might likewise provide additional predictive power up and above self-rated Conscientiousness.

Finally, although it was not the primary goal of our research, the NJT-H may address the problem that more valid selection instruments show larger (in particular ethnic) subgroup score differences and increase the potential for adverse impact (Ployhart & Holtz, 2008). Furthermore, reducing gender and age subgroup score differences is also an important concern to organizations because of their potential influence on workforce diversity (e.g., Sackett et al., 2001). Importantly, Integrity scores, which are highly related to H-H (e.g., Lee et al., 2005; Lee et al., 2019; Marcus et al., 2007), are generally lower among individuals from collectivist cultures (Fine, 2010). Furthermore, in the HEXACO model, H-H is one of the six traits that shows the largest gender differences after Emotionality (Lee & Ashton, 2020), and H-H scores substantially increase with age (Ashton & Lee, 2016). In the present research, we showed that with respect to gender, age, and nationality, the score differences on the NJT-H between demographic groups are small or nonexistent, and we have some evidence that there are smaller score differences on the NJT-H than on the HEXACO H-H scale. Although the robustness of our findings needs to be validated in future research with larger samples, the present findings are nonetheless promising.

PRACTICAL IMPLICATIONS

The NJT-H is an instrument that is easy to administer and cost-efficient compared to other implicit instruments that require specialized test administration expertise, one-on-one administration, and complex scoring procedures (Bing et al., 2007; Lilienfeld et al., 2000). A reliable and valid "low tech" implicit instrument (Vargas et al., 2007) such as the NJT-H could therefore be a feasible alternative or complement to self-report measures in the organizational context. Personnel selection could be one useful application of the NJT-H.

The NJT-H might also be a useful instrument for coaches and employees' development goals. The use of personality assessments in employees' development has increased significantly in the last two decades (McDowall & Redman, 2017; Passmore, 2012). The goal of such development programs is to increase self-awareness (Cseh et al., 2013), which

positively influences employees' well-being (e.g., Harrington & Loffredo, 2011). Importantly, people's self-knowledge is the lowest for evaluative traits (e.g., Vazire & Carlson, 2010), and in particular trait H-H (Thielmann et al., 2021). The present research provides some evidence that the HEXACO H-H scale should be supplemented by the NJT-H to obtain a more detailed picture of someone's level of H-H. Together, these test scores could be used to set goals for self-development.

LIMITATIONS AND FUTURE DIRECTIONS

The first limitation of this research is the investigation of the NJT-H in a low-stakes context. In a high-stakes context, job applicants may be inclined to substantially elevate their scores on personality self-report scales (e.g., Goffin & Christiansen, 2003; Griffith et al., 2007), and they do so mainly for socially desirable traits such as H-H (Anglim et al., 2018). Some studies have empirically investigated the fakability of implicit instruments, and these studies usually show that these instruments are resistant to faking as long as the participant has not been informed about the construct that is being measured (LeBreton et al., 2007 for the CRT; Steffens, 2004 for the IAT). Thus, one important avenue for future research is to investigate the fakability of the NJT-H and compare it to the fakability of personality self-report measures and other implicit instruments.

Future research could also further examine the construct-related validity of the NJT-H. One opportunity is to study the nomological network of the NJT-H. A particularly relevant trait to study in this context is emotional intelligence. In the PSAM paradigm, participants need to judge others, and they do so by comparing the other to themselves (Dunning & Hayes, 1996). This comparison requires self-reflection, which is a crucial aspect of emotional intelligence (e.g., Boyatzis et al., 2000). Another way to study the construct-related validity of the NJT-H is to investigate whether this instrument partly measures response styles. Some research has examined anchoring vignettes (comparable to the NJT-H items) as a tool to assess response styles, aiming to improve the cross-cultural validity of instruments by controlling for the vignette scores (King & Wand, 2007; King et al., 2004). There is some evidence that controlling for anchoring vignette scores style results in improvements in the validity of the test scores, although the effects are usually weak (e.g., He et al., 2017; Primi et al., 2016). Moreover, in Study 2, we showed that the NJT-H is associated with supervisory ratings, which provides support that the instrument assesses personality and not - at least not merely - response styles. Nonetheless, we encourage future research to examine the extent to which the NJT-H assesses H-H versus response styles. One methodology for this is to develop an NJT for a construct that is unrelated to H-H (e.g., Extraversion; Thielmann et al., 2021), and investigate its relationship with the NJT-H.

Another opportunity for future research is to investigate applicant reactions (e.g., perceived fairness, liking; Ryan & Ployhart, 2000) to the NJT-H. Applicant reactions to selection

instruments are essential to study, as these reactions affect applicants' test performance, perceptions of organizational attractiveness, and intentions to accept a job offer (McCarthy et al., 2017). The available empirical work on applicant reactions to implicit instruments in the selection context has raised critical issues. For example, an IAT that was developed to predict training skills and that could be used to hire or promote individuals was perceived by participants to lack procedural justice due to low job-relatedness and too little opportunity to perform (i.e., applicants' feeling that the selection procedure allows them to show their abilities) as judged by the participants (Wright & Meade, 2011). For TAT-like instruments, concerns have been raised about their lack of face validity (Van Rensburg et al., 2019), which could lead to defensive test-takers responses (Ridgeway, 2017). To our knowledge, applicant reactions to CRTs have not been empirically investigated so far (Connelly et al., 2018). Future research needs to examine applicant reactions to the NJT-H, and compare them to the applicant reactions to personality self-reports and other implicit instruments.

Finally, future research could also investigate the cross-cultural validity of the NJT-H. One research suggestion is to test this instrument in collectivistic cultures, where people have a more interdependent (versus independent) self-concept (Markus & Kitayama, 1991). Scholars have argued that an interdependent self-concept might contribute to lower validities of personality self-reports among people from collectivistic cultures, although evidence for this argument is inconsistent (Church & Katigbak, 2017). Arguably, the NJT-H might have a higher criterion-related validity than H-H self-reports in collectivistic cultures, because self-reflection is crucial in personality self-reports, but not likely for the NJT-H. However, this is a prediction that needs to be tested in future research.

CONCLUSION

The present research provides initial evidence for the construct-related validity of the NJT-H, and shows that this instrument is a predictor of CWB, OCB, and task performance. The NJT-H also explains unique variance in these employee behaviors, above the variance explained by HEXACO H-H and also the other five HEXACO traits. While research is necessary to provide more insights into the practical value of the NJT-H (determined by, for instance, its fakability and applicant reactions), the present research by means of the NJT-H suggests that implicit personality measures could form a useful alternative or complement to personality self-reports in the organizational context.

CHAPTER General Discussion

Dan Asfar Vrije Universiteit Amsterdam, the Netherlands NOA B.V., Amsterdam, the Netherlands

GENERAL DISCUSSION

As a consequence of the forced replacement of large groups of refugees due to war and persecution in several MENA (Middle East and North Africa) and neighboring countries, almost 200,000 asylum seekers requested humanitarian protection in the Netherlands between 2014 and 2021 (Eurostat, 2021a). The influx of these recently arrived refugees has underscored the importance of how our society can effectively help refugees successfully integrate into their new receiving country. This dissertation addresses this question using an individual-level psychological approach, by focusing on work-related integration as examined by economic (e.g., employment outcomes and income) and linguistic (local language proficiency) integration dimensions (Harder et al., 2018). Whilst most of the research into individual-level predictors of refugees' linguistic and economic integration outcomes examined sociodemographic and human capital variables such as age, gender, educational attainment, and work experience (Lee et al., 2020), the present dissertation contributes to this literature by concentrating on the influence and incremental validity of refugees' psychological traits. Additionally, several issues related to traditional measurement of psychological traits (which pertain to matters of response styles, self-presentation, and reference groups; see Chapter 1) pose a threat to the validity of cross-cultural personality assessments. This dissertation intended to provide an answer to these issues by advancing the psychological assessment of refugees through the development and validation of a socalled implicit instrument. This implicit instrument assesses the integrity-related personality trait of Honesty-Humility, which has been shown to be an important trait for employment outcomes (e.g., Lee et al., 2019).

SUMMARY OF FINDINGS

Chapter 2 focused on whether it is useful to examine refugees as consisting of separate subgroups based on their psychological profiles. This was done by clustering groups of individual refugees who share similar profiles through means of cluster analysis. The study was conducted among recently arrived refugees in the Netherlands, who mostly came from Syria (n = 2881) and Eritrea (n = 1183), but also from Iran (n = 270), Iraq (n = 188), and Afghanistan (n = 126).¹⁹ Drawing upon the ability-motivation-opportunity (AMO-)framework (Appelbaum et al., 2000), refugees' general mental ability (GMA), achievement motivation, and psychological distress were assessed to identify and investigate psychological clusters. The cluster analysis revealed four clusters: respectively a bright (n = 1377), an incapable (n = 876), a distressed (n = 576), and an undistinctive-ordinary (n = 1357) cluster. In their respective order, these clusters were mainly characterized by a high score on GMA, a low score on GMA, a high score on psychological distress, and a low score on psychological distress, although they also differed in the other psychological traits. These clusters as well as the

refugee demographic subgroups based on nationality, gender, age, and educational level were related to the refugees' level of work search intention and local language proficiency. The results showed that the psychological clusters differed particularly in local language proficiency, and did so more than the demographic subgroups. Specifically, the bright and the incapable clusters, which were the most distinguishable clusters, showed respectively high and low local language proficiency test scores. However, the psychological clusters were substantially less differentiated in their level of work search intention, though the bright and undistinctive-ordinary cluster scored somewhat higher on work search intention than the distressed cluster. The refugee subgroups based on nationality, gender, and educational level showed larger differences in work search intention compared to the psychological clusters, with highest scores among Eritrean and Iranian refugees, male refugees, and highly educated (master's level) refugees.

Chapter 3 focused on the relationship between psychological traits and refugees' local language proficiency, and the potential incremental validity of psychological traits in predicting local language proficiency. Using cross-sectional assessment data of a large group of Syrian (n = 1054) and Eritrean (n = 500) refugees living in the Netherlands, the results showed negative effects of age at the time of arrival and psychological distress, and positive effects of local length of stay and pre-migration educational attainment on Syrian and Eritrean refugees' local language proficiency. Additionally, above and beyond the effects of refugees' demographic variables and psychological distress, GMA and work search intention showed positive relationships with local language proficiency. Moreover, one's work search intention was found to strengthen the effects on local language proficiency for Conscientiousness and Openness (or: Openness to Experience) were observed, although some evidence was found for curvilinear (U-shaped and exponential) relationships between these traits and the outcome.

Chapter 4 introduced and tested a novel framework of individual-difference factors as predictors of refugees' workforce participation, studied as employment (the actual occurrence and speed of finding a job) and longest employment duration (i.e., the longest consecutive employment duration). Additionally, the employment outcome (dependent variable) highest hourly wage was exploratorily examined. This framework organizes individual-difference factors into two hindering and two facilitating variable groups: Impeding demographics and health- and family-related challenges (hindering), and acquired human and social capital and work-relevant traits (facilitating). The framework was examined using time-lagged data with a five-year span among Syrian (n = 1867) and Eritrean (n = 844) refugees living in the Netherlands. The findings revealed that several hindering factors (i.e., older age, being a woman, and physical health problems) and several facilitating factors (i.e., pre-migration educational level, pre-migration work experience, local language proficiency, frequency

¹⁹ The samples of Chapters 2, 3, and 4 are a little overlapping.
of contact with natives, Extraversion, GMA, and work centrality) contributed to predicting one or more of the studied employment outcomes. There were some differences in the predictive validity of the individual-difference factors between refugee groups (i.e., based on age, sex, and nationality). For example, PTSD symptoms predicted employment among younger refugees but not among older refugees, and having left a spouse or children in the country of origin predicted employment among Syrian refugees but not among Eritrean refugees.

Chapter 5 concerned the development and validation of an integrity-related implicit instrument for the organizational context. Whilst Chapter 2, 3, and 4 showed that traditional personality self-report measures have predictive validity for several work-related integration outcomes of refugees, there are also some issues with the cross-cultural validity of traditional personality inventories (cf. Gurven et al., 2013; Laajaj et al., 2019; Smaldino et al., 2019; for a description, see Chapter 1). Implicit instruments offer a viable solution to the cross-cultural validity issues of self-report measures (Uhlmann et al., 2012). Such instruments assess individual attributes that people might not be willing to disclose or are unaware of (Moors et al., 2010), and thus might be less prone to biases of self-report measures (e.g., Vianello et al., 2013). One novel implicit paradigm is the partially structured attitude measure (Vargas et al., 2004), which assesses individuals' attributes through their judgments of the actions of hypothetical persons described in vignettes. Based on this paradigm, the Normative Judgment Test of Honesty-Humility (the NJT-H) was developed. Honesty-Humility is an important trait as it is an important predictor of several employee behaviors and work outcomes (e.g., see Lee et al., 2019) and has been shown to be associated with the cultural adaptation among sojourners (Geeraert et al., 2019). Honesty-Humility has been defined as "the tendency to be fair and genuine in dealing with others, in the sense of cooperating with others even when one might exploit them without suffering retaliation" (Ashton & Lee, 2007, p. 156). In two studies among (N = 230 and N = 124) non-refugee local Dutch employees, the NJT-H's construct- and criterion-related validity were examined. In both studies, the NJT-H was significantly and positively related to Honesty-Humility, and not meaningfully related to the other five HEXACO traits. Furthermore, the NJT-H was negatively related to selfratings and supervisor ratings of counterproductive work behavior and positively related to supervisor ratings of organizational citizenship behavior and task performance. The NJT-H also explained unique variance in these outcomes above and beyond measures of Honesty-Humility and the other five HEXACO traits. The NJT-H is intended to be examined among refugee and migrant samples as described below in the section on future directions.

THEORETICAL CONTRIBUTIONS AND IMPLICATIONS

This dissertation offers several theoretical contributions and corresponding implications to diverse literatures. One contribution pertains to the literature on refugee profiles (Damen et al., 2022). Cluster analysis among refugee samples has been applied to topics such as the identification of rehousing trajectories (De Hoon et al., 2021), socio-cultural starting positions (i.e., a typology of refugees based on their co-ethnic and inter-ethnic social contacts, emotional ties, and cultural value orientation; Damen et al., 2022), and participation profiles (i.e., refugees' course of participation in activities that are economic [paid employment and internships], educational [enrolled in an educational program or a Dutch language course], or societal [voluntary work] in nature; Miltenburg & Dagevos, 2021). This dissertation applied cluster analysis using refugees' psychological individual-differences factors, namely GMA, achievement motivation, and psychological distress. By means of this analysis, the dissertation contributed to this literature by showing that (a) the application of cluster analysis of psychological traits among refugees provides meaningfully distinguishable clusters, (b) the psychological clusters among each other show differences in work search intention and local language proficiency, and (c) the differences in local language proficiency are larger between the psychological clusters than between the demographic subgroups of nationality, gender, age, or educational level. These findings imply that it is useful to study refugee subgroups based on the psychological clusters to which they belong, instead of or in addition to their demographic identities (e.g., gender or age group).

Furthermore, the dissertation contributed to the literature on the predictors of refugees' integration (Lee et al., 2020) by providing evidence for the importance of psychological traits for work-related integration outcomes. So far, this literature has been dominated by the disciplines of sociology and economics, which tend to focus on individual demographics and human and social capital variables. For example, with respect to migrants' local language proficiency, individual-difference factors have been studied widely through the lens of the standard theoretical model of Chiswick and Miller (2007), which contends that there are three general determinants of immigrants' local language acquisition, namely (a) exposure, that is, the extent to which immigrants hear and read the local language, (b) incentives, that is, the advantages one can obtain by mastering the local language weighted against the costs of learning the local language, and (c) efficiency, that is, the innate abilities to learn and acquire a new language. Similarly, migrants' economic integration has mostly been influenced by research from economics that focuses on human and social capital (e.g., educational attainment and ties with native citizens; Chiswick & Miller, 2009). Although such frameworks have been shown to be useful in identifying several predictors of migrants' linguistic and economic integration, they do not encompass psychological individualdifference factors from the personnel psychology literature (e.g., Judge & Zapata, 2015; Roberts et al., 2007; Schmidt & Hunter, 1998). In fact, psychological traits have shown to be important predictors of diverse work outcomes, such as job performance (e.g., Sackett et al., 2022) and actual employment (e.g., Van Hooft et al., 2021). Correspondingly, Chapters 3 and 4 have shown that refugees' individual-difference factors such as GMA, work attitudes (work search intention and work centrality), and personality traits (Extraversion) influence the linguistic and economic integration of refugees, and Chapter 3 additionally revealed

6

evidence for the incremental validity of psychological traits above and beyond frequently studied demographics. These findings imply that the literature on the predictors of refugees' integration outcomes can be advanced by incorporating psychological traits.

Building upon the previous point, another contribution of this dissertation to the literature on refugees' work-related integration was the development and examination of a novel integrative framework containing a wide range of hindering and facilitating individual-difference factors for refugees' path to finding work. Whilst various frameworks exist of the individual-difference predictors of workforce participation (e.g., Kanfer et al., 2001; Van Hooft et al., 2021), these frameworks are mostly applicable to native-born job seekers' and hence lack factors that are relevant for migrants in general (e.g., local language proficiency and frequency of contact with natives), and refugees in particular (e.g., physical health problems). There have been only two prior attempts at developing a theoretical framework of individual-difference factors for understanding refugees' workforce participation (Boss et al., 2021; Lee et al., 2020), but both frameworks (1) have not been quantitatively examined, (2) lack important predictors of workforce participation (e.g., demographics and factors related to acquired human capital in refugees' home country), (3) focus predominantly on hindering factors, (4) do not include refugee-specific family-related challenges, and (5) lack psychological characteristics such as GMA and personality traits. The framework developed in this dissertation addressed the issues described above, and the empirical examination of this framework showed that most of the unique factors in the present framework predict refugees' employment outcomes, including local language proficiency, frequency of contact with natives, physical health problems, GMA, Extraversion, and work centrality. Furthermore, unlike the latter three traits, other established individual-difference predictors of workforce participation among native-born job seekers, such as Agreeableness (Baay et al., 2014; Van Hooft et al., 2021), Emotional Stability (Kanfer et al., 2001), and Conscientiousness (Egan et al., 2017) showed no relation - or a weaker relation in the case of GMA (Vélez-Coto et al., 2021) – with refugees' workforce participation. Together, these findings demonstrate that established individual-difference factors predicting workforce participation do not necessarily generalize to refugees, who face unique barriers compared to native-born job seekers or economic migrants (e.g., Agbényiga et al., 2012), and whose skills are less likely to match the needs of the job market (Lee et al., 2020).

6

Finally, this dissertation contributed to the literature on personality, personnel psychology, and implicit cognition (Back et al., 2009; Sackett et al., 2022; Uhlmann et al., 2012) by providing evidence for the validity of an integrity-related implicit instrument that is based on an understudied paradigm, namely the partially structured attitude measure (Vargas et al., 2004). While Vargas et al. (2004) provided a proof of concept of the implicit paradigm, this dissertation demonstrated that this paradigm is useful for assessing the trait Honesty-Humility in the organizational context. The NJT-H has shown incremental validity over the Honesty-Humility scale and the other personality traits in the prediction of employees' work behaviors.

As such, incorporating the NJT-H (or other NJT's measuring other relevant constructs) in the personnel selection literature could assist in explaining more variance in employees' work behaviors and performance. Although the NJT-H was not examined among refugees in this dissertation, the implicit nature of this instrument has potential benefits with reference to the cross-cultural validity of refugee assessments. For example, traditional self-report measures are problematic in the cross-cultural personality assessments context due to the referencegroup effect (RGE; Heine et al., 2002). This effect refers to "the tendency for people to respond to subjective self-report items by comparing themselves with implicit standards from their culture" (Heine et al., 2008, p. 309). Several studies have shown that using different references based on age, gender, relative, and nationality affects mean-level personality trait test scores (Credé et al., 2010; De Vries et al., 2014; Wood et al., 2012). For example, De Vries et al. (2014) have shown that Turkish-Dutch participants reveal different personality trait scores when they completed the inventory by taking the Turkish-Dutch or Dutch majority group as their reference. Presumably, the RGE is no issue with the NJT-H, as in this instrument, the vignettes are the "reference group", and this reference group is identical for all participants, irrespective of their cultural background.

PRACTICAL IMPLICATIONS

This dissertation offers several practical implications. To support the work-related integration of refugees, courses, interventions, and integration programs must be tailored to refugees' psychological profiles or individual-difference factors (cf. Kosyakova & Laible, 2021). One approach was based on the findings of the cluster analysis of Chapter 2, where the results demonstrate that refugees can be arranged into one of four clusters, and where each cluster shows different levels of work search intention and local language proficiency. Practitioners can identify to which clusters refugees belong to indicate whether they are at "risk" of experiencing a suboptimal integration, and as such, they could offer tailored support in an early stage to prevent negative outcomes.

Another approach to effectively assist refugees in their integration is to focus on specific individual-difference factors. To provide such guidelines, the individual-difference factors are addressed as follows, according to the variable groups to which they are arranged in the framework introduced in Chapter 4. First, the (impeding) demographics consist of the stable (unmalleable) individual-difference factors nationality, sex, and age. Such stable characteristics do not have a direct causal effect on outcomes, but are associated with outcomes through other relevant factors (Lilienfeld et al., 2015). For example, in Chapters 2, 3, and 4, age was negatively associated with economic and linguistic integration outcomes. Among non-refugee samples, some hindering mediating factors that have been identified for older people are their lack of modern job skills (Fossum et al., 1986), low familiarity with modern job-search methods (Gibson et al., 1993; Westaby & Braithwaite, 2003), and employer hiring preferences for younger workers (Ahmed et al., 2012; Derous & Decoster,

2017). Thus, in practice, such mediating factors need to be recognized and dealt with to support refugees in their work-related integration.

With respect to the variable group of health- and family-related challenges, physical health problems, PTSD symptoms, and having a spouse or children in the country of origin can be distinguished. Refugees face higher risks of physical health issues compared to nativeborn citizens and non-refugee migrants (for a review of this issue and explanations for this gap, see Kumar et al., 2021). Additionally, Chapter 4 suggests that physical health is a predictor of workforce participation. Therefore, it is essential to assess a refugee's physical health to determine the feasible pace of work-related integration and the kind of jobs that are suitable based on the refugee's health situation. To do so, practitioners can consider the etiology (e.g., distress, natural autoimmune response), severity (e.g., minor, or major complications), and type (e.g., acute, chronic) of the physical health problem(s), as these factors can influence the ease and opportunities of refugees' work-related integration (Lai et al., 2022). Moreover, with respect to PTSD symptoms, therapeutic interventions have shown to be effective in reducing refugees' trauma-related symptoms, although, like regular client populations, refugees vary considerably in their response to such interventions (Sijbrandij et al., 2016). Hence, it is essential to provide therapeutic support to refugees in need, and closely monitor their mental health improvement to determine a sustainable work-related integration. Altogether, practitioners must tailor their support to refugees' unique physical and mental needs and constraints.

Another variable group, acquired human and social capital, includes integration-facilitating factors (pre-migration educational level, having pre-migration work experience, local language proficiency, and frequency of contact with natives) that refugees can actively influence to a certain degree. Whilst pre-migration educational level and pre-migration work experience are difficult to change, Chapter 4 revealed that refugees who were following an education in the Netherlands had higher odds of being simultaneously employed than refugees who did not follow an education. This finding indicates that, compared to (unmalleable) pre-migration factors, (malleable) post-migration factors can and ideally should be harvested to promote refugees' work-related integration (cf. De Vroome & Van Tubergen, 2010; Van Tubergen, 2010). Furthermore, although refugees' local language learning ability depends on other individual-difference factors (Asfar et al., 2019) and has a substantial heritability component, as was shown by Rimfeld et al. (2015), the willingness to learn another language also influences refugees' local language learning (Dörnyei, 2005). Finally, the frequency of contact with natives can be regarded as a habit that is relatively easy to change (Van Dijk et al., 2022). Altogether, many human and social capital variables are under the individual's voluntary control and can be obtained by time and energy investment, and hence, the harvesting of refugees' human and social capital needs to be stimulated by practitioners.

Finally, the findings from the studies in this dissertation have shown that work-relevant (i.e., psychological) traits are important for refugees' work-related integration (see Chapters 2, 3. and 4). The studied traits were Agreeableness, Extraversion, Emotional Stability, GMA. Conscientiousness, and work centrality. Such traits can be considered malleable to a certain degree. For example, the heritability of most personality traits of adults appears to be between 30 and 60% (Kandler, 2012). However, psychological traits are modifiable by environmental experiences (Merzenich, 2013). Indeed, research indicates that volitional personality trait change is possible, although limited in magnitude (for a systematic review, see Roberts et al., 2017). Nonetheless, instead of focusing on generic personality change, it is easier and more pragmatic to focus on cultivating trait-related micro-level skills and behaviors such as if-then plans (or 'implementation intentions'; Toli et al., 2016) to improve work-related integration outcomes. For example, refugees with lower levels of Extraversion can develop specific implementation intentions and train communicative and social interaction skills to increase their employment odds; Hudson & Fraley, 2015; Liu et al., 2014; Petruzziello et al., 2021; Turban et al., 2009). Such interventions need to be provided and encouraged by practitioners.

The final important practical implication pertains to the administration of the NJT-H. This instrument is easy to administer and cost-efficient compared to other implicit instruments that require specialized test administration expertise, one-on-one administration, and complex scoring procedures (Bing et al., 2007; Lilienfeld et al., 2000; Vargas et al., 2007). Based on the evidence of the incremental validity of the NJT-H among a local Dutch population, the instrument could be a feasible complement to self-report measures. One potential application of the NJT-H is personnel selection. Another potential application of the NJT-H is in employees' development programs to increase self-awareness about one's Honesty-Humility level (Cseh et al., 2013). Self-awareness is particularly challenging for evaluative traits (e.g., Vazire & Carlson, 2010), including Honesty-Humility (Thielmann et al., 2021). As such, a self-reported Honesty-Humility scale can be supplemented by the NJT-H to obtain a more detailed picture of someone's level of Honesty-Humility.

STRENGTHS, LIMITATIONS, AND FUTURE DIRECTIONS

The studies in this dissertation have several strengths. First, the studies described in Chapters 2, 3, and 4 contain the assessment of thousandths of refugees, whilst refugees are a "hard-to-reach" population (e.g., Jones & Newburn, 2001). Second, these three chapters included the assessment of psychological traits (e.g., GMA, personality traits, and work attitudes), which have been rarely studied among refugees (for exceptions, see Hahn et al., 2019; Kosyakova & Laible, 2021; Thum, 2014). Third, these three chapters contain objective integration outcomes of local language proficiency and workforce participation. Finally, with respect to Chapter 5, the criterion-related validity of the NJT-H has been studied using supervisory ratings of employees' performance and behavior. Informant ratings are an antidote to the common source bias, that

is, the possibility that self-report response tendencies explain the relationship between the predictor and the criterion (Podsakoff et al., 2012).

However, despite these strengths, the studies in this dissertation also have some limitations - and corresponding future research opportunities. First, the data used in Chapters 2, 3, and 4 have been collected in an assessment context. Although this context is optimal when one is interested in testing the validity of hypothesized predictors for (integration) outcomes in a real and presumably high-stakes assessment context (such as in personnel selection research; Sackett et al., 2022), this context is suboptimal in terms of finding actual validity estimates for the real-world relationships between individual-difference factors and (integration) outcomes (e.g., personality research; Roberts et al., 2007; Soto, 2019). When the real-world relationship is the prominent question, a potential issue is that refugees might have considered the assessment a high-stakes context as their assessment results influence their education- and work-related opportunities, and hence, this might have provoked socially desirable responses (e.g., Birkeland et al., 2006). Another issue of these data, which only concerns Chapter 4 (on the individual-difference predictors of workforce participation), is that the assessment results might have influenced the counselors' consultation of the refugees. The counselors were trained to use the assessment results to provide customized support for refugees in their integration trajectories, and as such, they were ought to pay special attention to refugees with worrisome profiles (e.g., more PTSD symptoms and low work centrality), compared to refugees with more favorable characteristics. As such, actual effects of individual-difference factors in workforce participation might have been partly masked in this research and thus may have been underestimated. Future research is encouraged to investigate the predictive validity of individual psychological traits of refugees for integration outcomes in research context.

Second, this dissertation focused primarily on refugees from Syria and Eritrea – and to a lesser extent on refugees from Iran, Iraq, and Afghanistan (Chapter 2) – and paid less attention to the theorizing of and explaining differences in outcomes and differential predictive validities among refugee groups themselves. With the available data, it was difficult to examine other refugee groups since the majority of refugees in the data originate from the above-mentioned nationalities, and the data on other refugee groups were too small for robust analyses. Nonetheless, understanding the differences in outcomes and predictive validities of refugee groups is important given that refugees originate from numerous countries. In the last decade, the largest refugee groups that seek protection in the Netherlands were those from Syria, Eritrea, Iraq, Afghanistan, and Iran. Previous publications (e.g., CBS, 2022) and the results reported in Chapters 2, 3, and 4 have shown that these groups differ substantially in integration outcomes. Similarly, three decades ago, most refugees who entered the Netherlands originated from Afghanistan, Iran, Iraq, Somalia, and former Yugoslavia, and were also found to differ in their integration outcomes (e.g., Engbersen et al., 2015; Huijnk et al.,

2015; Maliepaard et al., 2017). The recent influx of primarily Ukrainian and Russian refugees has underscored the importance of identifying the factors that underlie the differences in outcomes and differential predictive validities of refugees from different nationalities. Thus, future research to investigate this issue further should be encouraged.

Third, the theory sections of this dissertation explained why refugees' individual-difference factors may be related to the integration outcomes, but the mediating mechanisms of these relationships were not examined. As an example, Chapter 4 explained why refugees who score high on Extraversion are expected to show a higher workforce participation, but did not examine any mediating effects of this relationship. Some potential explanations for the positive effect of Extraversion on workforce participation include a higher self-confidence and self-efficacy in job search which can result in more job applications and success in job applications (Petruzziello et al., 2021), more positive emotions that offer personal resources relevant to the job search, thereby improving hiring success (Fredrickson, 2001; Turban et al., 2009), and being a more desirable candidate to employers (Huffcutt et al., 2001). The investigation of mediating mechanisms of individual-difference factors is important, as it can provide relevant practical implications.

Fourth, with reference to Chapter 5, the NJT-H could only be validated among local Dutch samples, but not among migrants or refugees. As described above, psychological instruments might not possess equivalent psychometric properties among people from different cultures and nationalities (e.g., Boer et al., 2018). Unpublished findings among a small sample (see Chapter 5), showed that the differences between Dutch respondents without a migration background (n = 177) and Dutch respondents with a migration background (n = 177) 69) were high on the HEXACO Honesty-Humility scale (former group higher average score; d = 1.02, p < .001, but small and in the opposite direction on the NJT-H (latter group higher; d = -0.33, p = .038). Although these findings rest on a small and heterogeneous sample with a migration background and are therefore unstable, they seem to point to the presence of clear cultural differences in test scores in the HEXACO Honesty-Humility scale, but a much smaller cultural difference in NJT-H scores. This finding is promising in the context of the diversity-validity dilemma in personnel selection, that is, using instruments in selection that have high predictive validity and show small ethnic subgroup score differences (De Soete et al., 2012). Following the empirical investigation of cultural differences in other popular implicit instruments such as the thematic apperception test (TAT; Hofer & Chasiotis, 2022), the implicit association test (IAT; Falk et al., 2015), and the conditional reasoning test (CRT; Galić et al., 2014; LeBreton et al., 2022), future research needs to further investigate the cross-cultural validity of the NJT-H.

In addition to these limitations, future research could advance the studies in this dissertation in three additional ways. First, future research could investigate a broader variety of psychological

attributes and measures in the refugee integration context. Some relevant examples of traits include emotional intelligence and social skills (Sackett et al., 2022). There is accumulating evidence that these traits predict important behaviors and performance in the domains of work and education (e.g., Joseph & Newman, 2010; MacCann et al., 2020; Morgeson et al., 2005). In the context of integration success, such traits can compensate for lower levels of GMA (Agnoli et al., 2012). Additionally, future research is encouraged to investigate implicit instruments among refugees, since such instruments could partly address several issues with explicit self-reports in cross-cultural personality assessments (see Chapter 1). One suggestion is the application of the NJT-H among refugee or migrant samples.

Second, this dissertation primarily focused on linguistic (local language proficiency) and economic (workforce participation) integration outcomes, but future research could examine a wider range of integration outcomes. Chapter 4 suggested some additional relevant employment outcomes for refugees, such as underemployment (i.e., overqualification; Campion, 2018; Ortlieb & Weiss, 2020). However, beyond employment-related outcomes and local language proficiency, several integration dimensions have been identified in the literature. Chapter 1 referred to the framework by Harder et al. (2018), which outlines six integration dimensions: The psychological dimension (one's sense of belonging in the host society), the economic dimension (employment outcomes and income), the political dimension (political and civic participation), the social dimension (social ties with natives), the linguistic dimension (the ability to use the local language), and the navigational dimension (managing basic needs in the host country). Likewise, several other frameworks of integration dimensions have recently been proposed in the literature, popular ones being those by Faith and Lessard-Phillips (2022), Ndofor-Tah et al. (2019), OECD/EU (2019), Puma et al. (2018), Lessard-Phillips (2017), and Spencer and Charsley (2016). Despite some differences, these frameworks display many similarities and much overlap in the integration dimensions, which vary from four to seven dimensions (e.g., dimensions related to health and well-being, social identification and identity, and civic and political participation). Future research is encouraged to investigate psychological individual characteristics as predictors of such dimensions.

Third, a final research idea is to study the influence of geographical and local (e.g., city or neighborhood) characteristics in the receiving country and their interaction with individual characteristics in predicting refugees' integration outcomes (cf. Van Riemsdijk & Axelsson, 2021). Several recent research examples have examined geographical and local characteristics in this context, such as the local levels of unemployment, neighborhood deprivation, the proportion of local co-ethnic migrants, and municipality size (Andersson et al., 2019; Azlor et al., 2020; De Lange et al., 2021; Stips & Kis-Katos, 2020; Vogiazides et al., 2021; Wimark et al., 2019). For example, Vogiazides et al. (2021) studied the occupational (mis)match – the match between the refugees' qualifications and their actual employment – and found that in the early years of resettlement, more occupational matches were

observed in urban (versus rural) regions in Sweden. In another recent Swedish study, initial settlement in a neighborhood with a high share of co-ethnic members had a negative effect on workforce participation for refugee women, but not for men (Andersson et al., 2019). Although empirical evidence to explain these effects is lacking, the authors of this Swedish study speculate that traditional patriarchal norms and values might create an environment in more concentrated co-ethnic enclaves that discourage employment of women. However, more empirical work on neighborhood effects is essential to investigate this issue in the Dutch context, as most of the available studies have been conducted in Scandinavian countries. Future research could advance our understanding of the influence of geographical and local characteristics by investigating how they interact with individual-difference factors in predicting refugees' integration outcomes. For example, conscientious refugees might thrive better in geographical areas with more employment opportunities (cf. Boyce et al., 2010). Such research could have important implications for dispersal policies.

CONCLUDING REMARKS

Armed conflict and persecution among other areas in the MENA (Middle East and North Africa) region, coerced millions of people in the last decade to leave their homes and countries to seek protection. The studies in this dissertation emerged from the need to investigate the integration of these refugees in the Netherlands, and to improve the psychological assessment of this group in order to better provide customized and effective integration trajectories for refugees. This research focused on individual characteristics (such as sociodemographic variables) and psychological traits (such as GMA and personality) as predictors of refugees' integration outcomes, and described the development of a novel implicit instrument to assess the important personality trait of Honesty-Humility with the ultimate aim to be utilized for refugee assessments. The findings underscore the importance of the assessment of refugees' individual-difference factors, among which their psychological traits, to provide evidence-based interventions and integration programs for refugees.

APPENDICES



SUMMARY

Through four empirical chapters, this dissertation intended to advance insights into the psychological assessment of refugees in relation to work-related integration. Chapters 2, 3, and 4 concentrated on individual-level predictors of refugees' linguistic and economic integration outcomes. These predictors included sociodemographic and human capital variables such as age, gender, educational attainment, and work experience, but also psychological traits, which have rarely been examined among refugees in the work-related integration context. Furthermore, to advance the psychological assessment of refugees, Chapter 5 focused on the development and validation of a so-called implicit instrument. This implicit instrument assesses the integrity-related personality trait of Honesty-Humility, which has been shown to be an important trait for employment outcomes (e.g., Lee et al., 2019).

Chapter 2 focused on whether it is useful to examine refugees as consisting of separate subgroups based on their psychological profiles. This was done by clustering groups of individual refugees who share similar profiles through means of cluster analysis. The study was conducted among recently arrived refugees in the Netherlands, who mostly came from Syria and Eritrea, but also from Iran, Iraq, and Afghanistan. Drawing upon the ability-motivationopportunity (AMO-)framework (Appelbaum et al., 2000), refugees' general mental ability (GMA), achievement motivation, and psychological distress were assessed to identify and investigate psychological clusters. The cluster analysis revealed four clusters: respectively a bright, an incapable, a distressed, and an undistinctive-ordinary cluster. In their respective order, these clusters were mainly characterized by a high score on GMA, a low score on GMA, a high score on psychological distress, and a low score on psychological distress, although they also differed in the other psychological traits. These clusters as well as the refugee demographic subgroups based on nationality, gender, age, and educational level were related to the refugees' level of work search intention and local language proficiency. The results showed that the psychological clusters differed particularly in local language proficiency, and did so more than the demographic subgroups. Specifically, the bright and the incapable clusters, which were the most distinguishable clusters, showed respectively high and low local language proficiency test scores. However, the psychological clusters were substantially less differentiated in their level of work search intention, though the bright and undistinctive-ordinary cluster scored somewhat higher on work search intention than the distressed cluster. The refugee subgroups based on nationality, gender, and educational level showed larger differences in work search intention compared to the psychological clusters, with highest scores among Eritrean and Iranian refugees, male refugees, and highly educated (master's level) refugees.

Chapter 3 focused on the relationship between psychological traits and refugees' local language proficiency, and the potential incremental validity of psychological traits in predicting local language proficiency. Using cross-sectional assessment data of a large group of Syrian and Eritrean refugees living in the Netherlands, the results showed negative

effects of age at the time of arrival and psychological distress, and positive effects of local length of stay and pre-migration educational attainment on Syrian and Eritrean refugees' local language proficiency. Additionally, above and beyond the effects of refugees' demographic variables and psychological distress, GMA and work search intention showed positive relationships with local language proficiency. Moreover, one's work search intention was found to strengthen the effect of GMA on local language proficiency. Contrary to the predictions, no positive linear effects on local language proficiency for Conscientiousness and Openness (or: Openness to Experience) were observed, although some evidence was found for curvilinear (U-shaped and exponential) relationships between these traits and the outcome.

Chapter 4 introduced and tested a novel framework of individual-difference factors as predictors of refugees' workforce participation, studied as employment (the actual occurrence and speed of finding a job) and longest employment duration (i.e., the longest consecutive employment duration). Additionally, the employment outcome (dependent variable) highest hourly wage was exploratorily examined. This framework organizes individual-difference factors into two hindering and two facilitating variable groups: Impeding demographics and health- and family-related challenges (hindering), and acquired human and social capital and work-relevant traits (facilitating). The framework was examined using time-lagged data with a five-year span among Syrian and Eritrean refugees living in the Netherlands. The findings revealed that several hindering factors (i.e., older age, being a woman, and physical health problems) and several facilitating factors (i.e., pre-migration educational level, pre-migration work experience, local language proficiency, frequency of contact with natives, Extraversion, GMA, and work centrality) contributed to predicting one or more of the studied employment outcomes. There were some differences in the predictive validity of the individual-difference factors between refugee groups (i.e., based on age, sex, and nationality). For example, PTSD symptoms predicted employment among younger refugees but not among older refugees, and having left a spouse or children in the country of origin predicted employment among Syrian refugees but not among Eritrean refugees.

Chapter 5 concerned the development and validation of an integrity-related implicit instrument for the organizational context. Whilst Chapter 2, 3, and 4 showed that traditional personality self-report measures have predictive validity for several work-related integration outcomes of refugees, there are also some issues with the cross-cultural validity of traditional personality inventories (cf. Gurven et al., 2013; Laajaj et al., 2019; Smaldino et al., 2019; for a description, see Chapter 1). Implicit instruments offer a viable solution to the cross-cultural validity issues of self-report measures (Uhlmann et al., 2012). Such instruments assess individual attributes that people might not be willing to disclose or are unaware of (Moors et al., 2010), and thus might be less prone to biases of self-report measures (e.g., Vianello

et al., 2013). One novel implicit paradigm is the partially structured attitude measure (Vargas et al., 2004), which assesses individuals' attributes through their judgments of the actions of hypothetical persons described in vignettes. Based on this paradiam, the Normative Judgment Test of Honesty-Humility (the NJT-H) was developed. Honesty-Humility is an important trait as it is an important predictor of several employee behaviors and work outcomes (e.g., see Lee et al., 2019) and has been shown to be associated with the cultural adaptation among sojourners (Geeraert et al., 2019). Honesty-Humility has been defined as "the tendency to be fair and genuine in dealing with others, in the sense of cooperating with others even when one might exploit them without suffering retaliation" (Ashton & Lee, 2007, p. 156). In two studies among non-refugee local Dutch employees, the NJT-H's constructand criterion-related validity were examined. In both studies, the NJT-H was significantly and positively related to Honesty-Humility, and not meaningfully related to the other five HEXACO traits. Furthermore, the NJT-H was negatively related to self-ratings and supervisor ratings of counterproductive work behavior and positively related to supervisor ratings of organizational citizenship behavior and task performance. The NJT-H also explained unique variance in these outcomes above and beyond measures of Honesty-Humility and the other five HEXACO traits. The NJT-H is intended to be examined among refugee and migrant samples as described below in the section on future directions.

Drie jaar lang waren mijn ouders en ik als peuter vanuit Irak op de vlucht om uiteindelijk op 1 september 1996 in Nederland asiel aan te vragen. Met deze achtergrond is het bijzonder om 26 jaar later mijn proefschrift in te dienen over de psychologische assessment van vluchtelingen in Nederland. Deze prestatie had ik nooit kunnen behalen zonder de hulp van de mensen om mij heen. Ik wil daarom graag een aantal personen in het bijzonder bedanken.

Allereerst dank ik Stichting NOA en haar bestuursleden (Marise Born, Nico Bleichrodt, Pieter Drenth, Henk van der Flier, Janneke Oostrom, Rutger Kappe en Gerlof Sijtsma) voor de financiering van mijn promotieonderzoek. Dankzij het initiatief van deze stichting en de toewijding van haar bestuursleden, kon ik in de voetsporen treden van NOA-promovendi Anita de Vries (2012) en Djurre Holtrop (2016). Ik hoop van harte dat Stichting NOA zich blijft inzetten voor wetenschappelijk onderzoek naar de psychologische assessment van etnische minderheden. Ik wil ook de leescommissie hartelijk bedanken voor het lezen en kritisch evalueren van mijn proefschrift.

Daarnaast gaat mijn lof naar mijn fantastische begeleiders. Graag wil ik Janneke Oostrom, mijn copromotor, bedanken voor haar enorm fijne begeleiding en voortdurende betrokkenheid. Vijf jaar lang heb ik vrijwel wekelijks een beroep gedaan op Jannekes advies en ideeën. Ze was altijd beschikbaar wanneer ik iets nodig had. Zelfs als ze druk was met andere taken of onderwijs, beantwoordde ze mijn mailtjes binnen een dag. Haar begeleiding en omgang is heel prettig: begripvol, aanmoedigend en op Socratische wijze sturend. Ook belangrijk: wát is Janneke creatief! Ze heeft vaak gelijk een ideale oplossing voor een ingewikkeld probleem. Haar creatieve kijk op wetenschappelijke en praktische vraagstukken heeft me erg geholpen in mijn promotieonderzoek.

Ik dank Marise Born, mijn promotor, dat ze ondanks haar overvolle werkweken altijd tijd maakte om mee te denken met mijn onderzoek, advies te geven en me te motiveren. Ik had me echt geen betere promotor kunnen wensen. Marise is niet alleen een deskundige geleerde, maar ook een erg leuk mens door haar oprechte interesse in anderen, haar begrip, openheid en gevoel voor humor – met name dat laatste siert de professor enorm. Terwijl ik me soms bezwaard voelde omdat ik veel van haar tijd vroeg, nam Marise juist graag de tijd om te vragen over mijn leven buiten werk, zoals over mijn familie en relatie. Erg leuk! En laat ik hier dan ook eerlijk over zijn: er zijn weinig mensen als Marise die me zo jaloers hebben gemaakt met hun eloquentie – ja, dit soort moeilijke woorden zitten in het alledaagse vocabulaire van Marise.

Verder dank ik Remko van den Berg, mijn NOA-werkgever, voor zijn flexibiliteit en ondersteuning tijdens mijn promotieonderzoek. Terwijl goede studies commercieel van belang zijn voor NOA, gaf Remko mij de volledige autonomie en vertrouwde hij het onderzoek

helemaal aan mij toe. Hij is pragmatisch, lief, persoonlijk betrokken, ondersteunend en menselijk. Zijn deur stond letterlijk en figuurlijk altijd open. Dat alles heeft ervoor gezorgd dat ik mijn werk als adviseur en onderzoeker bij NOA goed heb kunnen combineren met mijn promotieonderzoek.

Ik wil ook Mark van Vugt, mijn voormalige promotor, bedanken voor zijn belangrijke bijdrage aan mijn promotieonderzoek. Mark kende mij al sinds mijn studie psychologie, en ik geloof dat ik mijn aanstelling als promovendus deels aan zijn toevertrouwen te danken heb. Marks sociaalpsychologische en evolutionaire kijk op het thema van mijn promotieonderzoek was erg interessant en waardevol. Hij heeft mij geleerd om kritisch en met een sterk theoretische invalshoek naar mijn onderzoek te kijken. Daarnaast is Mark van alle mark-ten thuis: hij is een sportfanaat, politiekexpert, geschiedeniskenner, boekenliefhebber en een humorist. Ik heb altijd genoten van onze gesprekken over voetbal en politiek!

Natuurlijk dank ik ook graag mijn collega's van NOA en de VU. Het zijn er eigenlijk te veel om te noemen, dus ik belicht hier liever alleen twee personen. Ik dank Corine Sonke voor haar belangrijke bijdrage aan mijn begeleidingscommissie. Als persoon die zelf gepromoveerd is op een onderwerp dat relevant is aan die van mijn promotieonderzoek wist ze me altijd goed te helpen met haar ervaring en kennis. Daarnaast dank ik Reinout de Vries voor het delen van zijn inzichten op het gebied van de persoonlijkheidspsychologie. Reinout maakte geen deel uit van mijn begeleidingscommissie, maar schroomde nooit om zijn ideeën met me te delen en me van feedback te voorzien. Vaak was deze feedback erg belangrijk voor mijn onderzoek.

Tot slot wil ik de belangrijkste personen in mijn leven bedanken: mijn vader Dawood, mijn moeder Jinan, mijn zusje Carla, mijn vrouw Mariam en onze lieve Emilia. Ik dank mijn ouders voor hun liefdevolle opvoeding, hun grenzeloze inspanningen, geduld en ondersteuning tijdens mijn studie en promotieonderzoek. Mijn studie- en werkprestatie kan ik voor een groot deel aan hen toeschrijven. Ik ben blij dat ik hun zorg heb kunnen verzilveren met de mooie titel die ik met dit proefschrift behaal. Ik dank mijn zusje omdat haar luisterend oor en haar flauwekul mij erg goed konden doen bij moeilijkere momenten in mijn promotieonderzoek. Ik dank ook Mariam hiervoor, eveneens voor al haar liefde en de mooie momenten. Overigens heb ik nooit geweten in hoeverre mijn familie mijn onderzoek echt begreep of gewoon toeknikte, maar ik ben er wel zeker van dat ze iets leuks en leerzaams hebben meegekregen. En Emilia, ons wondertje, jij mag nog eerst even kinderboekjes lezen voordat je aan mijn proefschrift begint.

Oké, genoeg bedankt. Na vijf jaar lang werken aan dit thema is het ook fijn om hier met een sprankje nostalgie een punt achter te kunnen zetten. Tijdens mijn promotieonderzoek heb ik niet alleen veel geleerd over mijn onderwerp, maar ook over het academische werkveld, de psychologische consultancy en het werken met professionals. Ik heb ondervonden dat de wetenschap mijn leergierigheid en nieuwsgierigheid voedt, maar in mijn optiek tegelijkertijd ook wat 'afstandelijk' is: het biedt inzichten aan de hand van patronen in data waarin anonieme individuen worden weergegeven als regels met veel getallen. Achter deze getallen schuilen unieke verledens, persoonlijke verhalen en emoties, die even belangrijk zijn om mensen te kunnen begrijpen en helpen. Daar verdiep ik me nu in als psycholoog in de klinische praktijk.

- Abdou, L. H. (2019). Immigrant integration: the governance of ethno-cultural differences. *Comparative Migration Studies*, 7(1), Article 15. https://doi.org/10.1186/s40878-019-0124-8
- Ackerman, P. L., & Heggestad, E. D. (1997). Intelligence, personality, and interests: Evidence for overlapping traits. *Psychological Bulletin*, 121(2), 219–245. <u>https://doi.org/10.1037/0033-2909.121.2.219</u>
- Acock, A. C. (2014). A Gentle Introduction to Stata (4th ed.). Stata Press.
- Agbényiga, D. L., Barrie, S., Djelaj, V., & Nawyn, S. J. (2012). Expanding our community: Independent and interdependent factors impacting refugee successful community resettlement. *Advances in Social Work*, *13*(2), 306–324. <u>https://doi. org/10.18060/1956</u>
- Ager, A., & Strang, A. (2008). Understanding integration: A conceptual framework. *Journal of Refugee Studies*, *21*(2), 166–191. <u>https://doi.org/10.1093/jrs/ fen016</u>
- Agnoli, S., Mancini, G., Pozzoli, T., Baldaro, B., Russo, P. M., & Surcinelli, P. (2012). The interaction between emotional intelligence and cognitive ability in predicting scholastic performance in school-aged children. *Personality and Individual Differences*, 53(5), 660–665. <u>https://doi.</u> org/10.1016/j.paid.2012.05.020
- Ahmed, A. M., Andersson, L., & Hammarstedt, M. (2012). Does age matter for employability? A field experiment on ageism in the Swedish labour market. *Applied Economics Letters*, *19*(4), 403– 406. <u>http://dx.doi.org/10.1080/13504851.2011.5811</u> 99
- Aiyar, M. S., Barkbu, M. B. B., Batini, N., Berger, M. H., Detragiache, M. E., Dizioli, A., ... & Topalova, P. (2016). The refugee surge in Europe: Economic challenges. International Monetary Fund. <u>https://</u> doi.org/10.5089/9781513552590.006
- Akl, E. A., Briel, M., You, J. J., Sun, X., Johnston, B. C., Busse, J. W., Mulla, S., Lamontagne, F., Bassler, D., Vera, C., Alshurafa, M., Katsios, C. M., Zhou, Q., Cukierman-Yaffe, T., Gangji, A., Mills, E. J., Walter, S. D., Cook, D. J., Schunemann, H. J., ... Guyatt, G. H. (2012). Potential impact on estimated treatment effects of information lost to follow-up in randomised controlled trials (LOST-IT): Systematic review. *BMJ*, 344, e2809. <u>https://doi.org/10.1136/</u> bmj.e2809
- Al Ariss, A., Koall, I., Özbilgin, M., & Suutari, V. (2012). Careers of skilled migrants: towards a theoretical and methodological expansion. *Journal of Management Development, 31*(2), 92–101. <u>https://doi.org/10.1108/02621711211199511</u>
- Aldashev, A., Gernandt, J., & Thomsen, S. L. (2009). Language usage, participation, employment and earnings: Evidence for foreigners in West Germany with multiple sources of

selection. *Labour Economics*, *16*(3), 330–341. https://doi.org/10.1016/j.labeco.2008.11.004

- Allik, J., & Realo, A. (2016). How valid are culture-level mean personality scores? In A. T. Church (Ed.), *Personality across cultures*. Praeger.
- Andersson, R., Musterd, S., & Galster, G. (2019). Port-of-entry neighborhood and its effects on the economic success of refugees in Sweden. International Migration Review, 53(3), 671–705. https://doi.org/10.1177/0197918318781785
- Anglim, J., Lievens, F., Everton, L., Grant, S. L., & Marty, A. (2018). HEXACO personality predicts counterproductive work behavior and organizational citizenship behavior in low-stakes and job applicant contexts. *Journal of Research in Personality*, 77, 11–20. <u>https://doi.org/ doi:10.1016/j.jrp.2018.09.003A</u>
- Apers, C., Lang, J. W., & Derous, E. (2019). Who earns more? Explicit traits, implicit motives and income growth trajectories. *Journal of Vocational Behavior*, *110*, 214–228. <u>https://doi.org/10.1016/j.</u> jvb.2018.12.004
- Appel, M., Weber, S., & Kronberger, N. (2015). The influence of stereotype threat on immigrants: Review and meta-analysis. *Frontiers in Psychology*, 6, Article 900. <u>https://doi.org/10.3389/</u> <u>fpsyg.2015.00900</u>
- Appelbaum, E., Bailey, T., Berg, P., & Kalleberg, A. L. (2000). Manufacturing advantage: Why highperformance work systems pay off. ILR Press.
- Arthur, W. Jr., Day, E. A., McNelly, T. L., & Edens, P. S. (2003). A meta-analysis of the criterionrelated validity of assessment center dimensions. *Personnel Psychology*, 56(1), 125–153. https://doi.org/10.1111/j.1744-6570.2003.tb00146.x
- Asendorpf, J. B., Van De Schoot, R., Denissen, J. J., & Hutteman, R. (2014). Reducing bias due to systematic attrition in longitudinal studies: The benefits of multiple imputation. *International Journal of Behavioral Development*, 38(5), 453– 460. https://doi.org/10.1177/0165025414542713
- Asfar, D., Born, M. P., Oostrom, J. K., & Vugt, M. (2019). Psychological individual differences as predictors of refugees' local language proficiency. *European Journal of Social Psychology*, 49(7), 1385–1400. https://doi.org/10.1002/ejsp.2592
- Ashton, M. C., & Lee, K. (2007). Empirical, theoretical, and practical advantages of the HEXACO model of personality structure. *Personality and Social Psychology Review*, *11*(2), 150–166. <u>https://doi.org/10.1177/1088868306294907</u>
- Ashton, M. C., & Lee, K. (2008). The prediction of Honesty–Humility-related criteria by the HEXACO and Five-Factor Models of personality. *Journal of Research in Personality*, 42(5), 1216–1228. <u>https://</u> doi.org/10.1016/j.jrp.2008.03.006
- Ashton, M. C., & Lee, K. (2009). The HEXACO-60: A short measure of the major dimensions

Δ

 of
 personality.
 Journal
 of
 Personality

 Assessment,
 91(4),
 340–345.
 <u>https://doi.org/10.1080/00223890902935878</u>

- Ashton, M. C., & Lee, K. (2016). Age trends in HEXACO-PI-R self-reports. *Journal of Research in Personality, 64*, 102–111. <u>https://doi.org/10.1016/j.</u> jrp.2016.08.008
- Ashton, M. C., & Lee, K. (2020). Objections to the HEXACO Model of Personality Structure—and why those Objections Fail. *European Journal* of *Personality*, 34(4), 492–510. <u>https://doi.</u> org/10.1002/per.2242
- Ashton, M. C., Lee, K., & De Vries, R. E. (2014). The HEXACO Honesty-Humility, Agreeableness, and Emotionality Factors. *Personality and Social Psychology Review*, 18(2), 139–152. <u>https://doi.org/10.1177/1088868314523838</u>
- Ashton, M. C., Lee, K., Perugini, M., Szarota, P., De Vries, R. E., Di Blas, L., Boies, K., & De Raad, B. (2004). A six-factor structure of personality-descriptive adjectives: solutions from psycholexical studies in seven languages. *Journal of Personality and Social Psychology*, *86*(2), 356–366. <u>https://doi.org/10.1037/0022-3514.86.2.356</u>
- Avrić, B., Odekerken, M., De Vries, S., & Van Rooijen, M. (2019). *De Financiële zelfredzaamheid van statushouders: Belemmeringen en bouwstenen.* Kennisplatform Integratie & Samenleving. <u>https://www.kis.nl/sites/default/files/financiele-</u> zelfredzaamheid-statushouders.pdf
- Azlor, L., Damm, A. P., & Schultz-Nielsen, M. L. (2020). Local labour demand and immigrant employment. *Labour Economics, 63*, Article 101808. <u>https://doi.org/10.1016/j.labeco.2020.101808</u>
- Azuero, A. (2016). A note on the magnitude of hazard ratios. *Cancer, 122*(8), 1298–1299. <u>https://doi. org/10.1002/cncr.29924</u>
- Baay, P. E., Van Aken, M. A. G., Van der Lippe, T., & De Ridder, D. T. D. (2014). Personality moderates the links of social identity with work motivation and job searching. *Frontiers in Psychology*, 5, 1–8. <u>https://doi.org/10.3389/fpsyg.2014.01044</u>
- Back, M. D., Schmukle, S. C., & Egloff, B. (2009). Predicting actual behavior from the explicit and implicit self-concept of personality. *Journal of Personality and Social Psychology*, 97(3), 533– 548. <u>https://doi.org/10.1037/a0016229</u>
- Bakker, L., & Dagevos, J. (2017). Arbeidsmarktparticipatie asielmigranten: stroeve start en blijvende achterstand. In Maliepaard, M., Witkamp, B., & Jennissen, R. (Eds). Een kwestie van tijd? De integratie van asielmigranten: Een cohortonderzoek (pp. 63–76). WODC.
- Bakker, L., Dagevos, J., & Engbersen, G. (2014). The importance of resources and security in the Socio-economic integration of refugees. A study on the impact of length of stay in asylum accommodation and residence status on Socio-

economic integration for the four largest refugee groups in the Netherlands. *Journal of International Migration and Integration*, *15*(3), 431–448. <u>https://</u> <u>doi.org/10.1007/s12134-013-0296-2</u>

- Bakker, L., Dagevos, J., & Engbersen, G. (2017). Explaining the refugee gap: a longitudinal study on labour market participation of refugees in the Netherlands. *Journal of Ethnic and Migration Studies*, 43(11), 1775–1791. <u>https://doi.org/10.1080/1</u> <u>369183X.2016.1251835</u>
- Bambra, C., & Eikemo, T. A. (2009). Welfare state regimes, unemployment and health: a comparative study of the relationship between unemployment and selfreported health in 23 European countries. *Journal* of Epidemiology & Community Health, 63(2), 92– 98. <u>https://doi.org/10.1136/jech.2008.077354</u>
- Baranik, L. E. (2020). Employment and attitudes toward women among Syrian refugees. *Personnel Review*, 50(4), 1233–1252. <u>https://doi.org/10.1108/</u> <u>pr-10-2018-0435</u>
- Bargh, J. A., & Chartrand, T. L. (1999). The unbearable automaticity of being. *American Psychologist*, 54(7), 462–479. <u>https://doi.org/10.1037/0003-066x.54.7.462</u>
- Baron-Epel, O., Kaplan, G., Weinstein, R., & Green, M. S. (2010). Extreme and acquiescence bias in a bi-ethnic population. *European Journal of Public Health*, 20(5), 543–548. <u>https://doi.org/10.1093/ eurpub/ckq052</u>
- Barrick, M. R., & Mount, M. K. (1991). The big five personality dimensions and job performance: a meta-analysis. *Personnel Psychology*, 44(1), 1–26. <u>https://doi.org/10.1111/j.1744-6570.1991.tb00688.x</u>
- Barslund, M., Di Bartolomeo, A., & Ludolph, L. (2017). *Gender Inequality and Integration of Non-EU Migrants in the EU.* CEPS Policy Insight. <u>https://www.ceps.eu/ceps-publications/gender-</u> <u>inequality-and-integration-non-eu-migrants-eu/</u>
- Bear, J. B., & Glick, P. (2017). Breadwinner bonus and caregiver penalty in workplace rewards for men and women. *Social Psychological and Personality Science*, 8(7), 780–788. <u>https://doi.org/10.1177/1948550616683016</u>
- Beauregard, K. S., & Dunning, D. (1998). Turning up the contrast: Self-enhancement motives prompt egocentric contrast effects in social judgments. *Journal of Personality and Social Psychology*, 74(3), 606–621. <u>https://doi.org/10.1037/0022-3514.74.3.606</u>
- Becker, G. (1964). Human capital: A theoretical and empirical analysis with special reference to education. Columbia University Press.
- Beckers, P., & Borghans, L. (2011). Segregation in neighbourhoods and labour market outcomes of immigrants: Evidence from random assignment in the Netherlands. UNUMERIT Working Paper Series, Article 2011–068. <u>https://ideas.repec. org/p/unm/unumer/2011068.html</u>

- Beddoes, K., & Pawley, A. L. (2014). 'Different people have different priorities': work–family balance, gender, and the discourse of choice. *Studies in Higher Education*, 39(9), 1573–1585. <u>https://doi.or</u> g/10.1080/03075079.2013.801432
- Beenstock, M., Chiswick, B. R., & Repetto, G. L. (2001). The effect of linguistic distance and country of origin on immigrant language skills: Application to Israel. *International Migration*, 39(3), 33–60. https://doi.org/10.1111/1468-2435.00155
- Beiser, M., & Hou, F. (2000). Gender differences in language acquisition and employment consequences among southeast Asian refugees in Canada. *Canadian Public Policy*, *26*(3), 311–330. https://doi.org/10.2307/3552403
- Beiser, M., & Hou, F. (2001). Language acquisition, unemployment and depressive disorder among Southeast Asian refugees: a 10-year study. Social Science & Medicine, 53(10), 1321–1334. <u>https://doi.org/10.1016/s0277-9536(00)00412-3</u>
- Bennett, R. J., & Robinson, S. L. (2000). Development of a measure of workplace deviance. *Journal of Applied Psychology*, 85(3), 349–360. <u>https://doi.org/10.1037/0021-9010.85.3.349</u>
- Bergold, S., & Steinmayr, R. (2018). Personality and intelligence interact in the prediction of academic achievement. *Journal of Intelligence*, 6(2), 27–45. https://doi.org/10.3390/jintelligence6020027
- Bernerth, J. B., & Aguinis, H. (2016). A critical review and best-practice recommendations for control variable usage: Bernerth and Aguinis. *Personnel Psychology*, 69(1), 229–283. <u>https://doi.org/10.1111/</u> peps.12103
- Berry, C. M., Carpenter, N. C., & Barratt, C. L. (2012). Do other-reports of counterproductive work behavior provide an incremental contribution over selfreports? A meta-analytic comparison. *Journal of Applied Psychology*, 97(3), 613–636. <u>https://doi. org/10.1037/a0026739</u>
- Berry, C. M., Ones, D. S., & Sackett, P. R. (2007). Interpersonal deviance, organizational deviance, and their common correlates: A review and meta-analysis. *Journal of Applied Psychology*, 92(2), 410–424. <u>https://doi.org/10.1037/0021-9010.92.2.410</u>
- Berry, J. W. (1997). Immigration, acculturation, and adaptation. Applied Psychology, 46(1), 5–34. <u>https://doi.org/10.1111/j.1464-0597.1997.tb01087.x</u>
- Bertone, S. (2004). From factory fodder to multicultural mediators: A typology of NESB immigrant work experiences in Australia [Unpublished doctoral dissertation]. University of Sydney.
- Bijl, R., & Verweij, A. (2012). Measuring and monitoring immigrant integration in Europe. Netherlands Institute for Social Research. <u>https://library.wur.nl/</u> <u>WebQuery/titel/1984030</u>
- Bing, M. N., Stewart, S. M., Davison, H. K., Green, P. D., McIntyre, M. D., & James, L. R. (2007). An

integrative typology of personality assessment for aggression: Implications for predicting counterproductive workplace behavior. *Journal of Applied Psychology*, *92*(3), 722–744. <u>https://doi. org/10.1037/0021-9010.92.3.722</u>

- Birkeland, S. A., Manson, T. M., Kisamore, J. L., Brannick, M. T., & Smith, M. A. (2006). A meta-analytic investigation of job applicant faking on personality measures: Job applicant faking on personality measures. *International Journal of Selection and Assessment*, *14*(4), 317–335. <u>https://doi.org/10.1111/</u> j.1468-2389.2006.00354.x
- Bisello, M., & Mascherini, M. (2017). The gender employment gap: Costs and policy responses. Intereconomics, 52(1), 24–27. <u>https://doi.org/10.1007/s10272-017-0638-y</u>
- Blackmore, R., Boyle, J. A., Fazel, M., Ranasinha, S., Gray, K. M., Fitzgerald, G., Misso, M., & Gibson-Helm, M. (2020). The prevalence of mental illness in refugees and asylum seekers: A systematic review and meta-analysis. *PLOS Medicine*, *17*(9), e1003337.<u>https://doi.org/10.1371/journal.</u> pmed.1003337
- Blau, P. M. (1964). Exchange and power in social life. Routledge. <u>https://doi.org/10.4324/9780203792643</u>
- Blevins, C. A., Weathers, F. W., Davis, M. T., Witte, T. K., & Domino, J. L. (2015). The Posttraumatic Stress Disorder Checklist for DSM-5 (PCL-5): Development and initial psychometric evaluation. *Journal of Traumatic Stress*, 28(6), 489–498. <u>https://doi.org/10.1002/jts.22059</u>
- Bloch, A. (2002). Refugees' opportunities and barriers in employment and training. Corporate Document Services. <u>https://www.researchgate.</u> <u>net/publication/237210603_Refugees'</u> <u>Opportunities_and_Barriers_in_Employment_ and_Training</u>
- Boer, D., Hanke, K., & He, J. (2018). On detecting systematic measurement error in cross-cultural research: A review and critical reflection on equivalence and invariance tests. *Journal of Cross-Cultural Psychology*, 49(5), 713–734. <u>https://</u> doi.org/10.1177/0022022117749042
- Bogic, M., Njoku, A., & Priebe, S. (2015). Long-term mental health of war-refugees: a systematic literature review. BMC International Health and Human Rights, 15(1), Article 29. <u>https://doi.org/10.1186/s12914-015-0064-9</u>
- Bolino, M. C., Harvey, J., & Bachrach, D. G. (2012). A self-regulation approach to understanding citizenship behavior in organizations. Organizational Behavior and Human Decision Processes, 119(1), 126–139. <u>https://doi.org/10.1016/j. obhdp.2012.05.006</u>
- Boneva, B. S., & Frieze, I. H. (2001). Toward a concept of a migrant personality. *Journal of Social Issues*, 57(3), 477–491. <u>https://doi.org/10.1111/0022-</u>

Α

Δ

4537.00224

- Borjas, G. J., & Chiswick, B. R. (2019). Foundations of migration economics. Oxford University Press. <u>https://doi.org/10.1093/</u> <u>oso/9780198788072.001.0001</u>
- Borman, W. C., & Motowidlo, S. J. (1997). Task performance and contextual performance: The meaning for personnel selection research. *Human Performance*, *10*(2), 99–109. <u>https://doi.</u> org/10.1207/s15327043hup1002_3
- Boselie, P. (2010). High performance work practices in the health care sector: a Dutch case study. *International Journal of Manpower*, *31*(1), 42–58. https://doi.org/10.1108/01437721011031685
- Boss, H. C., Lee, C. S., Bourdage, J. S., & Hamilton, L. K. (2021). Developing and testing a framework for understanding refugees' job search processes. *Equality, Diversity and Inclusion: An International Journal, 41*(4), 568–591. <u>https://doi.org/10.1108/</u> EDI-01-2021-0031
- Bosson, J. K., Swann, W. B., & Pennebaker, J. W. (2000). Stalking the perfect measure of implicit selfesteem: The blind men and the elephant revisited? *Journal of Personality and Social Psychology*, 79(4), 631–643. <u>https://doi.org/10.1037/0022-</u> 3514.79.4.631
- Boudreaux, M. J., Ferrell, B. T., Hundley, N. A., & Sherman, R. A. (2022). A personality-based measure of employability. *Journal of Personnel Psychology*, *21*(1), 11–22. <u>http://dx.doi.org/10.1027/1866-5888/</u> <u>a000283</u>
- Bourdage, J. S., Lee, K., Lee, J. H., & Shin, K.-H. (2012). Motives for organizational citizenship behavior: Personality correlates and coworker ratings of OCB. *Human Performance*, *25*(3), 179–200. https://doi.org/10.1080/08959285.2012.683904
- Bowerman, B. L., & O'Connell, R. T. (1990). *Linear* statistical models: An applied approach (2nd ed.). Duxbury.
- Boyatzis, R. E., Goleman, D., & Rhee, K. S. (2000). Clustering competence in emotional intelligence: Insights from the Emotional Competence Inventory. In R. Bar-On & J. D. A. Parker (Eds.) The handbook of emotional intelligence: Theory, development, assessment, and application at home, school, and in the workplace (pp. 343– 362). Jossey-Bass.
- Boyce, C. J., Wood, A. M., & Brown, G. D. A. (2010). The dark side of conscientiousness: Conscientious people experience greater drops in life satisfaction following unemployment. *Journal of Research in Personality*, 44(4), 535–539. <u>https://</u> doi.org/10.1016/j.jrp.2010.05.001
- Brell, C., Dustmann, C., & Preston, I. (2020). The labor market integration of refugee migrants in high-income countries. *Journal of Economic Perspectives*, 34(1), 94–121. <u>https://doi.org/10.1257/</u> jep.34.1.94

Brusco, M. J., Singh, R., Cradit, J. D., & Steinley, D. (2017). Cluster analysis in empirical OM research: survey and recommendations. *International Journal of Operations & Production Management*, 37(3), 300–320. <u>https://doi.org/10.1108/</u> ijopm-08-2015-0493

- Burns, K., Ellegood, W. A., Bernard Bracy, J. M., Duncan, M., & Sweeney, D. C. (2019). Early College Credit Programs Positively Impact Student Success. *Journal of Advanced Academics*, *30*(1), 27–49. https://doi.org/10.1177/1932202x18801274
- Busato, V. V., Prins, F. J., Elshout, J. J., & Hamaker, C. (2000). Intellectual ability, learning style, personality, achievement motivation and academic success of psychology students in higher education. *Personality and Individual Differences*, 29(6), 1057–1068. <u>https://doi.</u> org/10.1016/s0191-8869(99)00253-6
- Cairney, J., Veldhuizen, S., Wade, T. J., Kurdyak, P., & Streiner, D. L. (2007). Evaluation of 2 measures of psychological distress as screeners for depression in the general population. *The Canadian Journal of Psychiatry*, 52(2), 111–120. https://doi.org/10.1177/070674370705200209
- Camara, W. J., & Schneider, D. L. (1994). Integrity tests: Facts and unresolved issues. *American Psychologist*, 49(2), 112–119. <u>https://doi.</u> <u>org/10.1037/0003-066x.49.2.112</u>
- Campion, E. D. (2018). The career adaptive refugee: exploring the structural and personal barriers to refugee resettlement. *Journal of Vocational Behavior, 105,* 6–16. <u>https://doi.org/10.1016/j.</u> jvb.2017.10.008
- Carliner, G. (2000). The language ability of U.S. immigrants: Assimilation and cohort effects. *International Migration Review, 34*(1), 158-182. <u>https://doi.org/10.1177/019791830003400107</u>
- Carroll, J. B. (2005). The Three-Stratum Theory of Cognitive Abilities. In D. P. Flanagan & P. L. Harrison (Eds.), *Contemporary Intellectual Assessment: Theories, Tests, and Issues* (pp. 69– 76). The Guilford Press.
- Caspi, A., & Moffitt, T. E. (1993). When do individual differences matter? A paradoxical theory of personality coherence. *Psychological Inquiry*, 4(4), 247–271. <u>https://doi.org/10.1207/s15327965pli0404_1</u>
- Caspi, A., Wright, B. R. E., Moffitt, T. E., & Silva, P. A. (1998). Early failure in the labor market: Childhood and adolescent predictors of unemployment in the transition to adulthood. *American Sociological Review*, 63(3), 424–451. <u>https://doi. org/10.2307/2657557</u>
- Castles, S., Korac, M., Vasta, E., & Vertovec, S. (2002). Integration: mapping the field. Home Office Online Report 28/03, <u>http://webarchive.nationalarchives.</u> gov.uk/
- Cattell, R. B. (1971). Abilities: Their structure, growth, and

action. Houghton Mifflin.

- CBS. (2021). Asiel en integratie 2021. Centraal Bureau voor de Statistiek. <u>https://www.cbs.nl/-/media/</u> pdf/2021/15/asielenintegratie2021.pdf
- CBS. (2022). Asiel en integratie 2022. Centraal Bureau voor de Statistiek. <u>https://longreads.cbs.nl/</u> asielenintegratie-2022/
- Central Intelligence Agency. (2022). In *The world* factbook. https://www.cia.gov/the-world-factbook/ countries/
- Cerda-Navarro, A., Salva-Mut, F., & Comas Forgas, R. (2019). A typology of students in intermediate vocational education and training programmes based on student engagement factors, sociodemographic characteristics and intentions of dropping out. *European Journal of Education*, 54(4), 635–650. https://doi.org/10.1111/ejed.12361
- Chandler, C. R., & Tsai, Y. (2001). Social factors influencing immigration attitudes: an analysis of data from the General Social Survey. *The Social Science Journal*, 38(2), 177–188. <u>https://doi.org/10.1016/s0362-3319(01)00106-9</u>
- Charlson, F., Van Ommeren, M., Flaxman, A., Cornett, J., Whiteford, H., & Saxena, S. (2019). New WHO prevalence estimates of mental disorders in conflict settings: a systematic review and metaanalysis. *The Lancet*, 394(10194), 240–248. https://doi.org/10.1016/s0140-6736(19)30934-1
- Cheng, Z., Wang, B. Z., & Taksa, L. (2021a). Labour force participation and employment of humanitarian migrants: Evidence from the building a new life in Australia longitudinal data. *Journal of Business Ethics*, *168*(4), 697–720. <u>https://doi.org/10.1007/</u> s10551-019-04179-8
- Cheng, Z., Wang, B. Z., Jiang, Z., Taksa, L., & Tani, M. (2021b). English skills and early labour market integration: Evidence from humanitarian migrants in Australia. *International Migration*, 00, 1–21. https://doi.org/10.1111/imig.12889
- Cheung, G. W., & Rensvold, R. B. (2002). Evaluating goodness-of-fit indexes for testing measurement invariance. *Structural Equation Modeling*, 9(2), 233–255. <u>https://doi.org/10.1207/</u> s15328007sem0902_5
- Chiswick, B. R., & Miller, P. W. (2001). A model of destination-language acquisition: Application to male immigrants in Canada. *Demography*, 38(3), 391–409. https://doi.org/10.1353/dem.2001.0025
- Chiswick, B. R., & Miller, P. W. (2002). Immigrant earnings: Language skills, linguistic concentrations and the business cycle. *Journal of Population Economics*, 15(1), 31–57. <u>https://doi.org/10.1007/p100003838</u>
- Chiswick, B. R., & Miller, P. W. (2007). Modeling Immigrants' Language Skills. In B. R. Chiswick (Ed.), *Immigration* (pp. 75–128). Emerald Group Publishing Limited.
- Chiswick, B. R., & Miller, P. W. (2008). Why is the payoff to schooling smaller for immigrants?

Labour Economics, 15(6), 1317–1340. <u>https://doi.org/10.1016/j.labeco.2008.01.001</u>

- Chiswick, B. R., & Miller, P. W. (2009). The international transferability of immigrants' human capital. *Economics of Education Review*, 28(2), 162–169. https://doi.org/10.1016/j.econedurev.2008.07.002
- Chiswick, B. R., & Repetto, G. (2001). Immigrant adjustment in Israel: literacy and fluency in Hebrew and earnings. In Djajic, S. (Ed.), International migration: Trends, policy and economic impact (pp. 204 – 228). Routledge.
- Chiswick, B. R., & Wenz, M. (2006). The linguistic and economic adjustment of Soviet Jewish immigrants in the United States, 1980–2000. Emerald Group Publishing Limited.
- Chiswick, B. R., Rebhun, U., & Beider, N. (2020). Language acquisition, employment status, and the earnings of Jewish and non-Jewish immigrants in Israel. *International Migration*, *58*(2), 205–232. https://doi.org/10.1111/imig.12634
- Chiswick, B. R., & Wang, Z. (2016). Social contacts, Dutch language proficiency and immigrant economic performance in the Netherlands: a longitudinal study. <u>https://doi.org/10.2139/ssrn.2742548</u>
- Chuah, S. C., Drasgow, F., & Roberts, B. W. (2006). Personality assessment: Does the medium matter? No. Journal of Research in Personality, 40(4), 359–376. <u>https://doi.org/10.1016/j.jrp.2005.01.006</u>
- Church, A. T., & Katigbak, S. K. (2017). Trait consistency and validity across cultures: Examining trait and cultural psychology perspectives. In A. T. Church (Ed.), *The Praeger handbook of personality across cultures* (pp. 279–308). Praeger.
- Ciani, A. C., & Capiluppi, C. (2011). Gene flow by selective emigration as a possible cause for personality differences between small islands and mainland populations. *European Journal of Personality*, 25(1), 53–64. https://doi.org/10.1002/per.774
- CINOP, Citogroep, Bureau ICE, & BVE Raad. (2002). Naar een portfolio NT2. Taalportfolio voor leerders van het Nederlands. CINOP.
- Cohen, F., Kemeny, M. E., Zegans, L. S., Johnson, P., Kearney, K. A., & Stites, D. P. (2007). Immune function declines with unemployment and recovers after stressor termination. *Psychosomatic Medicine*, 69(3), 225–234. <u>https://doi.org/10.1097/ psy.0b013e31803139a6</u>
- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences*. Lawrence Erlbaum Associates.
- Cohen, J. (1992). A power primer. *Psychological Bulletin*, 112(1), 155–159. <u>https://doi.org/10.1037/0033-</u> 2909.112.1.155
- Colic-Peisker, V. (2005). 'At least you're the right colour': Identity and social inclusion of Bosnian refugees in Australia. *Journal of Ethnic and Migration Studies*, *31*(4), 615–638. <u>https://doi. org/10.1080/13691830500109720</u>

Colic-Peisker, V., & Tilbury, F. (2006). Employment niches for recent refugees: Segmented labour market in twenty-first century Australia. *Journal* of Refugee Studies, 19(2), 203–229. <u>https://doi.</u> org/10.1093/jrs/fej016

- Colquitt, J. A., Sabey, T. B., Rodell, J. B., & Hill, E. T. (2019). Content validation guidelines: Evaluation criteria for definitional correspondence and definitional distinctiveness. *The Journal of Applied Psychology*, *104*(10), 1243–1265. <u>https://</u> doi.org/10.1037/ap10000406
- Colquitt, J. A., Scott, B. A., Rodell, J. B., Long, D. M., Zapata, C. P., Conlon, D. E., & Wesson, M. J. (2013). Justice at the millennium, a decade later: a meta-analytic test of social exchange and affect-based perspectives. *The Journal of Applied Psychology*, *98*(2), 199–236. <u>https://doi.org/10.1037/a0031757</u>
- Connelly, B. S., & Ones, D. S. (2010). An other perspective on personality: meta-analytic integration of observers' accuracy and predictive validity. *Psychological Bulletin*, *136*(6), 1092–1122. https://doi.org/10.1037/a0021212
- Connelly, B. S., Ones, D. S., & Hülsheger, U. R. (2018). Personality in industrial, work and organizational psychology: Theory, measurement and application. In D. S. Ones, N. Anderson, C. Viswesvaran, & H. K. Sinangil (Eds.), *The SAGE* handbook of industrial, work and organizational psychology (2nd ed., Vol. 1, pp. 320–365). Sage. https://doi.org/10.4135/9781473914940.n13
- Conner, M., McEachan, R., Jackson, C., McMillan, B., Woolridge, M., & Lawton, R. (2013). Moderating effect of socioeconomic status on the relationship between health cognitions and behaviors. *Annals* of *Behavioral Medicine*, *46*(1), 19–30. <u>https://doi.org/10.1007/s12160-013-9481-y</u>
- Costa, P. T., & McCrae, R. R. (1992). Normal personality assessment in clinical practice: The NEO Personality Inventory. *Psychological Assessment*, 4(1), 5–13. https://doi.org/10.1037/1040-3590.4.1.5
- Costa, P. T., & McCrae, R. R. (1992). Revised NEO Personality Inventory and NEO Five-Factor Inventory: Professional manual. Psychological Assessment Resources.
- Council of Europe. (1997). Measurement and indicators of integration. <u>https://www.coe.int/t/</u> dg3/migration/archives/documentation/Series_ <u>Community_Relations/Measurement_indicators_</u> integration_en.pdf
- Couttenier, M., Petrencu, V., Rohner, D., & Thoenig, M. (2019). The violent legacy of conflict: Evidence on asylum seekers, crime, and public policy in Switzerland. *American Economic Review*, 109(12), 4378–4425. <u>https://doi.org/10.1257/aer.20170263</u>
- Coward, W. M., & Sackett, P. R. (1990). Linearity of ability-performance relationships: A reconfirmation. *Journal of Applied*

Psychology, 75(3), 297–300. <u>https://doi.</u> org/10.1037/0021-9010.75.3.297

- Coyle, T. R. (2015). Relations among general intelligence (g), aptitude tests, and GPA: Linear effects dominate. *Intelligence*, 53, 16–22. <u>https://doi.org/10.1016/j.intell.2015.08.005</u>
- Credé, M., Bashshur, M., & Niehorster, S. (2010). Reference group effects in the measurement of personality and attitudes. *Journal of Personality Assessment*, 92(5), 390–399. <u>https://doi.org/10.10</u> <u>80/00223891.2010.497393</u>
- Cseh, M., Davis, E. B., & Khilji, S. E. (2013). Developing a global mindset: learning of global leaders. *European Journal of Training* and Development, 37(5), 489-499. <u>https://doi.</u> org/10.1108/03090591311327303
- Cucina, J. M., & Vasilopoulos, N. L. (2005). Nonlinear personality–performance relationships and the spurious moderating effects of traitedness. *Journal of Personality*, *73*(1), 227–259. https://doi.org/10.1111/j.1467-6494.2004.00309.x
- Cunningham, W. A., Preacher, K. J., & Banaji, M. R. (2001). Implicit attitude measures: Consistency, stability, and convergent validity. *Psychological Science*, *12*(2), 163–170. <u>https://doi.org/10.1111/1467-9280.00328</u>
- Dagevos, J., Huijnk, W., Maliepaard, M., & Miltenburg, E. (2018). De integratie van Syriërs heeft nog een lange weg te gaan [The integration of Syrians has still a long way to go]. *Sociaal Bestek*, *80*(4), 50–51. <u>https://doi.org/10.1007/s41196-018-0100-0</u>
- Dagevos, J., Jongen, E., & Muns, S. (2020). Promising policies for integration on the labour market: summary. The Netherlands Institute for Social Research. <u>https://english.scp.nl/publications/</u> publications/2020/10/21/promising-policies-forintegration-on-the-labour-market
- Dalal, R. S. (2005). A meta-analysis of the relationship between organizational citizenship behavior and counterproductive work behavior. *Journal of Applied Psychology*, 90(2), 1241–1255. <u>https://doi.org/10.1037/0021-9010.90.6.1241</u>
- Dalal, R. S. (2007). Contextual performance / prosocial behavior / organizational citizenship behavior. In S. G. Rogelberg (Ed.) *Encyclopedia of industrial/* organizational psychology (Vol. 1, pp.103–106). Sage. <u>https://doi.org/10.4135/9781412952651.n44</u>
- Damen, R., Huijnk, W., & Dagevos, J. (2022). Sociocultural starting positions among recently arrived Syrian refugees in the Netherlands: A latent class analysis. International Journal of Intercultural Relations, 87, 72–84. <u>https://doi.org/10.1016/j.</u> jjintrel.2022.01.009
- Danner, D., Aichholzer, J., & Rammstedt, B. (2015). Acquiescence in personality questionnaires: Relevance, domain specificity, and stability. *Journal of Research in Personality*, *57*, 119–130. https://doi.org/10.1016/j.jrp.2015.05.004

- De Hoon, M., Vink, M., & Schmeets, H. (2021). On the move again? Residential trajectories of refugees after obtaining asylum in the Netherlands. *Population, Space and Place, 27*(2), Article e2386. <u>https://doi.org/10.1002/psp.2386</u>
- De Houwer, J., & Moors, A. (2010). Implicit measures: Similarities and differences. In B. Gawronski & B. K. Payne (Eds.), Handbook of implicit social cognition: Measurement, theory, and applications (pp. 176–193). Guilford Press.
- De Lange, T., Berntsen, L., Hanoeman, R., & Haidar, O. (2021). Highly skilled entrepreneurial refugees: Legal and practical barriers and enablers to start up in the Netherlands. *International Migration*, 59(4), 74–87. https://doi.org/10.1111/imig.12745
- De Raad, B., Barelds, D. P. H., Timmerman, M. E., De Roover, K., Mlačić, B., & Church, A. T. (2014). Towards a pan–cultural personality structure: Input from 11 psycholexical studies. *European Journal of Personality*, 28(5), 497–510. <u>https://doi.org/10.1002/per.1953</u>
- De Soete, B., Lievens, F., & Druart, C. (2012). An update on the diversity-validity dilemma in personnel selection: A review. *Psychological Topics*, 21(3), 399–424. <u>https://psycnet.apa.org/ fulltext/2013-03108-002.pdf</u>
- De Vries, A., De Vries, R. E., Born, M. P., & Van den Berg, R. H. (2014). Persoonlijkheid als voorspeller van werkprestatie en contraproductiefwerkgedrag:Het belang van specifieke persoonlijkheidsmetingen [Personality as a predictor of job performance and counterproductive work behavior: the importance of specificity in personality measurement]. Gedrag en Organisatie, 27(4), 407–427. <u>https://doi.</u> org/10.5553/GenO/092150772014027004003
- De Vries, R. E. (2012). Personality predictors of leadership styles and the self-other agreement problem. *The Leadership Quarterly, 23*(5), 809– 821. <u>https://doi.org/10.1016/j.leaqua.2012.03.002</u>
- De Vries, R. E. (2013). The 24-item brief HEXACO inventory (BHI). Journal of Research in Personality, 47(6), 871–880. https://doi.org/10.1016/j.jrp.2013.09.003
- De Vries, R. E., & Van Gelder, J.-L. (2015). Explaining workplace delinquency: The role of Honesty– Humility, ethical culture, and employee surveillance. *Personality and Individual Differences*, 86, 112–116. <u>https://doi.org/10.1016/j.</u> paid.2015.06.008
- De Vries, R. E., Ashton, M. C., & Lee, K. (2009). De zes belangrijkste persoonlijkheidsdimensies en de HEXACO Persoonlijkheidsvragenlijst [The six most important personality dimensions and the HEXACO Personality Inventory]. *Gedrag & Organisatie, 22,* 208–250.
- De Vries, R. E., Pathak, R. D., Van Gelder, J. L., & Singh, G. (2017). Explaining Unethical Business Decisions: The role of personality, environment, and states. *Personality and Individual Differences*, 117,

188–197. https://doi.org/10.1016/j.paid.2017.06.007

- De Vries, R. E., Pronk, J., Olthof, T., & Goossens, F. A. (2020). Getting along and/or getting ahead: Differential HEXACO personality correlates of likeability and popularity among adolescents. *European Journal of Personality*, 34(2), 245–261. https://doi.org/10.1002/per.2243
- De Vroome, T., & Van Tubergen, F. (2010). The employment experience of refugees in the Netherlands. *International Migration Review*, 44(2), 376–403. <u>https://doi.org/10.1111/j.1747-7379.2010.00810.x</u>
- Deary, I. J., Strand, S., Smith, P., & Fernandes, C. (2007). Intelligence and educational achievement. Intelligence, 35(1), 13–21. <u>https://doi.org/10.1016/j.</u> intell.2006.02.001
- Derous, E., & Decoster, J. (2017). Implicit age cues in resumes: Subtle effects on hiring discrimination. *Frontiers in Psychology*, *8*, 1–15. <u>https://doi.org/10.3389/fpsyg.2017.01321</u>
- DeSimone, J. A., & James, L. R. (2015). An item analysis of the Conditional Reasoning Test of Aggression. *Journal of Applied Psychology*, 100(6), 1872–1886. <u>https://doi.org/10.1037/ap10000026</u>
- Dettenborn, L., Tietze, A., Bruckner, F., & Kirschbaum, C. (2010). Higher cortisol content in hair among long-term unemployed individuals compared to controls. *Psychoneuroendocrinology*, 35(9), 1404–1409. <u>https://doi.org/10.1016/j.</u> <u>psyneuen.2010.04.006</u>
- DeYoung, C. G., Quilty, L. C., & Peterson, J. B. (2007). Between facets and domains: 10 aspects of the Big Five. *Journal of Personality and Social Psychology*, 93(5), 880–896. <u>https://doi.org/10.1037/0022-3514.93.5.880</u>
- Di Domenico, S. I., & Fournier, M. A. (2015). Able, ready, and willing: Examining the additive and interactive effects of intelligence, conscientiousness, and autonomous motivation on undergraduate academic performance. *Learning and Individual Differences*, 40, 156–162. <u>https://doi.org/10.1016/j.</u> <u>lindif.2015.03.016</u>
- Diamantopoulos, A., Reynolds, N. L., & Simintiras, A. C. (2006). The impact of response styles on the stability of cross-national comparisons. *Journal of Business Research*, 59(8), 925–935. <u>https://doi. org/10.1016/j.jbusres.2006.03.001</u>
- Diefendorff, J. M., Brown, D. J., Kamin, A. M., & Lord, R. G. (2002). Examining the roles of job involvement and work centrality in predicting organizational citizenship behaviors and job performance. *Journal of Organizational Behavior, 23*(1), 93–108. <u>https://doi.org/10.1002/job.123</u>
- Dietl, E., & Meurs, J. A. (2019). Implicit core selfevaluations and work outcomes: Validating an indirect measure. *Journal of Occupational and Organizational Psychology*, 92(1), 169–190. <u>https://</u> doi.org/10.1111/joop.12244

Α

- Digman, J. M. (1990). Personality structure: Emergence of the Five-Factor Model. Annual Review of Psychology, 41(1), 417–440. <u>https://doi.org/10.1146/</u> annurev.ps.41.020190.002221
- Dijksterhuis, A., & Nordgren, L. F. (2006). A theory of unconscious thought. *Perspectives on Psychological Science*, 1(2), 95–109. <u>https://doi.org/10.1111/j.1745-6916.2006.00007.x</u>
- Donato, K. M., & Ferris, E. (2020). Refugee integration in Canada, Europe, and the United States: Perspectives from research. *The Annals* of the American Academy of Political and Social Science, 690(1), 7–35. <u>https://doi.</u> org/10.1177/0002716220943169
- Donker, T., Comijs, H., Cuijpers, P., Terluin, B., Nolen, W., Zitman, F., & Penninx, B. (2010). The validity of the Dutch K10 and extended K10 screening scales for depressive and anxiety disorders. *Psychiatry Research*, *176*(1), 45–50. <u>https://doi.org/10.1016/j. psychres.2009.01.012</u>
- Dörnyei, Z. (2005). The Psychology of the Language Learner: Individual Differences in Second Language Acquisition. Lawrence Erlbaum Associates.
- Dourleijn, E., & Dagevos, J. (2011). Vluchtelingengroepen in Nederland: Over de integratie van Afghaanse, Iraakse, Iraanse en Somalische migranten [Refugee groups in The Netherlands: About the integration of Afghan, Iraqi, Iranian, and Somalian refugees]. Netherlands Institute for Social Research. <u>https://repository. scp.nl/bitstream/handle/publications/715/ http____www.scp.nl_8080_gvisapi_dsplug.</u> pdf?sequence=1&isAllowed=y
- Drever, A. I., & Hoffmeister, O. (2008). Immigrants and social networks in a job-scarce environment: The case of Germany. *International Migration Review*, 42(2), 425–448. <u>https://doi.org/10.1111/j.1747-7379.2008.00130.x</u>
- Duckworth, A. L., & Seligman, M. E. P. (2005). Selfdiscipline outdoes IQ in predicting academic performance of adolescents. *Psychological Science*, *16*(12), 939–944. <u>https://doi.org/10.1111/</u> j.1467-9280.2005.01641.x
- Duckworth, A. L., Quinn, P. D., Lynam, D. R., Loeber, R., & Stouthamer-Loeber, M. (2011). Role of test motivation in intelligence testing. *Proceedings of* the National Academy of Sciences, 108(19), 7716– 7720. <u>https://doi.org/10.1073/pnas.1018601108</u>
- Dudley, N. M., McFarland, L. A., Goodman, S. A., Hunt, S. T., & Sydell, E. J. (2005). Racial differences in socially desirable responding in selection contexts: magnitude and consequences. *Journal* of *Personality Assessment*, *85*(1), 50–64. <u>https:// doi.org/10.1207/s15327752jpa8501_05</u>
- Dudley, N. M., Orvis, K. A., Lebiecki, J. E., & Cortina, J. M. (2006). A meta-analytic investigation of conscientiousness in the prediction of job

performance: Examining the intercorrelations and the incremental validity of narrow traits. *Journal* of Applied Psychology, 91(1), 40–57. <u>http://dx.doi.</u> org/10.1037/0021-9010.91.1.40

- Due, C., Callaghan, P., Reilly, A., Flavel, J., & Ziersch, A. (2021). Employment for women with refugee and asylum seeker backgrounds in Australia: An overview of workforce participation and available support programmes. *International Migration*. <u>https://doi.org/10.1111/imig.12848</u>
- Dunlop, P. D., & Lee, K. (2004). Workplace deviance, organizational citizenship behavior, and business unit performance: The bad apples do spoil the whole barrel. *Journal of Organizational Behavior*, 25(1), 67–80. https://doi.org/10.1002/job.243
- Dunning, D. (2012). The relation of self to social perception. In M. R. Leary & J. P. Tangney (Eds.), *Handbook of self and identity* (pp. 481–501). Guilford Press.
- Dunning, D., & Cohen, G. L. (1992). Egocentric definitions of traits and abilities in social judgment. *Journal* of *Personality and Social Psychology*, 63(3), 341– 355. <u>https://doi.org/10.1037/0022-3514.63.3.341</u>
- Dunning, D., & Hayes, A. F. (1996). Evidence for egocentric comparison in social judgment. Journal of Personality and Social Psychology, 71(2), 213–229. <u>https://doi.org/10.1037/0022-3514.71.2.213</u>
- Dusseldorp, E., Hofstetter, H., & Sonke, C. (2018). Landelijke doorontwikkeling van de UWV Werkverkenner: eindrapportage [National development of the UWV Werkverkenner:final report]. TNO. <u>http://resolver.tudelft.nl/</u> uuid:330d4615-255b-44be-9c93-29556b7507c6
- Dustmann, C. (1994). Speaking fluency, writing fluency and earnings of migrants. *Journal of Population Economics, 7*(2), 133–156. <u>https://doi.org/10.1007/</u> <u>bf00173616</u>
- Dustmann, C., & Fabbri, F. (2003). Language proficiency and labour market performance of immigrants in the UK. *The Economic Journal*, *113*(489), 695–717. https://doi.org/10.1111/1468-0297.t01-1-00151
- Dustmann, C., Fasani, F., Frattini, T., Minale, L., & Schönberg, U. (2017). On the economics and politics of refugee migration. *Economic Policy*, 32(91), 497–550. <u>https://doi.org/10.1093/epolic/ eix008</u>
- Echterhoff, G., Hellmann, J. H., Back, M. D., Kärtner, J., Morina, N., & Hertel, G. (2020). Psychological antecedents of refugee integration (PARI). *Perspectives on Psychological Science*, *15*(4), 856–879. <u>https://doi.org/10.1177/1745691619898838</u>
- Edele, A., Seuring, J., Kristen, C., & Stanat, P. (2015). Why bother with testing? The validity of immigrants' self-assessed language proficiency. *Social Science Research*, *52*, 99–123. <u>https://doi.org/10.1016/j.ssresearch.2014.12.017</u>

- Egan, M., Daly, M., Delaney, L., Boyce, C. J., & Wood, A. M. (2017). Adolescent conscientiousness predicts lower lifetime unemployment. *Journal of Applied Psychology*, *102*(4), 700–709. <u>https://doi.org/10.1037/apl0000167</u>
- Egloff, B., Schwerdtfeger, A., & Schmukle, S. C. (2005). Temporal stability of the implicit association test-anxiety. *Journal of Personality Assessment*, 84(1), 82–88. <u>https://doi.org/10.1207/ s15327752jpa8401_14</u>
- Eidelman, S., & Biernat, M. (2007). Getting more from success: Standard raising as esteem maintenance. *Journal of Personality and Social Psychology*, 92(5), 759–774. <u>https://doi. org/10.1037/0022-3514.92.5.759</u>
- Eisnecker, P., & Schacht, D. (2016). Half of the refugees in Germany found their first job through social contacts. *DIW Economic Bulletin, 6*(34/35), 414– 421.
- El-Sheikh, A. A., Abonazel, M. R., & Gamil, N. (2017). A review of software packages for structural equation modeling: A comparative study. *Applied Mathematics and Physics*, 5(3), 85–94. <u>https://doi. org/10.12691/amp-5-3-2</u>
- Engbersen, G., Dagevos, J., Jennissen, R., Bakker, L., Leerkens, A., Klaver, J. & Odé, A. (2015). *Geen tijd te verliezen: Van opvang naar integratie van asielmigranten* [No time to lose: From reception to integration of asylum migrants]. Wetenschappelijke Raad voor het Regeringsbeleid. <u>https://repository.</u> wodc.nl/handle/20.50012832/2950
- Epstein, S. (1994). Integration of the cognitive and the psychodynamic unconscious. *American Psychologist*, 49(8), 709–724. <u>https://doi.org/10.1037/0003-066x.49.8.709</u>
- Esser, H. (2004). Does the "new" immigration require a "new" theory of intergenerational integration? International Migration Review, 38(3), 1126–1159. https://doi.org/10.1111/j.1747-7379.2004.tb00231.x
- Esser, H. (2006). *Migration, language and integration.* Social Science Research Center Berlin. <u>https://citeseerx.ist.psu.edu/viewdoc/download?doi=10.11.376.2951&rep=rep1&type=pdf</u>
- Eurostat. (2021a). Migrant integration statistics labour market indicators. <u>https://ec.europa.eu/eurostat/</u> <u>statistics-explained/index.php?title=Migrant_</u> <u>integration_statistics_%E2%80%93_labour_</u> market_indicators#Employment_rates
- Eurostat. (2021b). Population by Age Group, Sex, and Country of Birth. Retrieved August 22, 2022, from https://ec.europa.eu/eurostat/en/web/productsdatasets/-/MIGR_POP3CTB
- Eurostat. (2022). Migrant integration statistics labour market indicators. <u>Retrieved August 22, 2022</u>, from <u>https://ec.europa.eu/eurostat/statisticsexplained/index.php?title=Migrant_integration statistics_%E2%80%93_labour_market_ indicators</u>

- Evers, A., Lucassen, W., Meijer, R., & Sijtsma, K. (2015). *COTAN review system for evaluating test quality.* <u>https://www.psynip.nl/uw-beroep/cotan/</u>
- Faist, T. (2000). Transnationalization in international migration: implications for the study of citizenship and culture. *Ethnic and Racial Studies*, 23(2), 189– 222. https://doi.org/10.1080/014198700329024
- Fajth, V., & Lessard-Phillips, L. (2022). Multidimensionality in the integration of first- and second-generation migrants in Europe: A conceptual and empirical investigation. *International Migration Review*. https://doi.org/10.1177/01979183221089290
- Falk, C. F., Heine, S. J., Takemura, K., Zhang, C. X. J., & Hsu, C.-W. (2015). Are implicit self-esteem measures valid for assessing individual and cultural differences?: Implicit self-esteem. *Journal* of *Personality*, 83(1), 56–68. <u>https://doi.org/10.1111/ jopy.12082</u>
- Farsides, T., & Woodfield, R. (2003). Individual differences and undergraduate academic success: the roles of personality, intelligence, and application. *Personality and Individual Differences*, 34(7), 1225–1243. <u>https://doi.org/10.1016/s0191-8869(02)00111-3</u>
- Fasani, F., Frattini, T., & Minale, L. (2022). (The struggle for) refugee integration into the labour market: evidence from Europe. *Journal of Economic Geography, 00*, 1–43. <u>https://doi.org/10.1093/jeg/ lbab011</u>
- Fassaert, T., De Wit, M. A. S., Tuinebreijer, W. C., Wouters, H., Verhoeff, A. P., Beekman, A. T. F., & Dekker, J. (2009). Psychometric properties of an interviewer-administered version of the Kessler Psychological Distress scale (K10) among Dutch, Moroccan and Turkish respondents. *International Journal of Methods in Psychiatric Research*, 18(3), 159–168. https://doi.org/10.1002/mpr.288
- Fazel, M., Wheeler, J., & Danesh, J. (2005). Prevalence of serious mental disorder in 7000 refugees resettled in western countries: A systematic review. *The Lancet*, 365(9467), 1309–1314. <u>https:// doi.org/10.1016/s0140-6736(05)61027-6</u>
- Fazio, R. H. (1990). Multiple processes by which attitudes guide behavior: The MODE model as an integrative framework. Advances in Experimental Social Psychology, 23, 75–109. <u>https://doi. org/10.1016/S0065-2601(08)60318-4</u>
- Fazio, R. H., & Olson, M. A. (2003). Implicit measures in social cognition research: Their meaning and use. *Annual Review of Psychology*, 54(1), 297–327. <u>https://doi.org/10.1146/annurev.</u> psych.54.101601.145225
- Feeney, A. (2000). Refugee employment. Local Economy, 15(4), 343–349. <u>https://doi.</u> org/10.1080/026909400750068086
- Fennelly, K., & Palasz, N. (2003). English language proficiency of immigrants and refugees in the Twin Cities metropolitan area. *International*

Α

Migration, *41*(5), 93–125. <u>https://doi.org/10.1111/</u> j.0020-7985.2003.00262.x

- Fergusson, D. M., John Horwood, L., & Ridder, E. M. (2005). Show me the child at seven II: childhood intelligence and later outcomes in adolescence and young adulthood. *Journal of Child Psychology* and Psychiatry, 46(8), 850–858. <u>https://doi.org/10.1111/j.1469-7610.2005.01472.x</u>
- Ferris, G. R., Witt, L. A., & Hochwarter, W. A. (2001). Interaction of social skill and general mental ability on job performance and salary. *Journal of Applied Psychology*, 86(6), 1075–1082. <u>https://doi. org/10.1037/0021-9010.86.6.1075</u>
- Fetvadjiev, V. H., Meiring, D., Van de Vijver, F. J., Nel, J. A., & Hill, C. (2015). The South African Personality Inventory (SAPI): A culture-informed instrument for the country's main ethnocultural groups. *Psychological Assessment*, 27, 827–837(3), 827–837. <u>https://doi.org/10.1037/ pss0000078</u>
- Fine, S. (2010). Cross-Cultural Integrity Testing as a Marker of Regional Corruption Rates. International Journal of Selection and Assessment, 18(3), 251–259. <u>https://doi.org/10.1111/j.1468-</u> 2389.2010.00508.x
- Fine, S., & Gottlieb-Litvin, Y. (2013). Justifying counterproductive work behaviors and an integrity-based conditional reasoning test: Back to the drawing board?: Conditional reasoning and integrity testing. *International Journal of Selection and Assessment, 21*(3), 328–333. <u>https://doi. org/10.1111/ijsa.12042</u>
- Fossum, J. A., Arvey, R. D., Paradise, C. A., & Robbins, N. E. (1986). Modeling the skills obsolescence process: A psychological/economic integration. *The Academy of Management Review*, *11*(2), 362– 374. <u>https://doi.org/10.5465/amr.1986.4283130</u>
- Frank, K. (2013). Immigrant employment success in Canada: Examining the rate of obtaining a job match. International Migration Review, 47(1), 76– 105. <u>https://doi.org/10.1111/imre.12014</u>
- Fredrickson, B. L. (2001). The role of positive emotions in positive psychology: The broaden-andbuild theory of positive emotions. *American Psychologist*, 56(3), 218–226. <u>https://doi.org/10.1037/0003-066X.56.3.218</u>
- Friedberg, R. M. (2000). You can't take it with you? Immigrant assimilation and the portability of human capital. *Journal of Labor Economics*, 18(2), 221–251. <u>https://doi.org/10.1086/209957</u>
- Fugate, M., Kinicki, A. J., & Ashforth, B. E. (2004). Employability: A psycho-social construct, its dimensions, and applications. *Journal of Vocational Behavior*, 65(1), 14–38. <u>https://doi.org/10.1016/j.jvb.2003.10.005</u>
- Funder, D. C. (2002). Personality psychology: Current status and some issues for the future. *Journal of Research in Personality*, 36(6), 638–639. https://

doi.org/10.1016/S0092-6566(02)00515-9

- Furukawa, T. A., Kessler, R. C., Slade, T., & Andrews, G. (2003). The performance of the K6 and K10 screening scales for psychological distress in the Australian National Survey of Mental Health and Well-Being. *Psychological Medicine*, *33*(2), 357– 362. https://doi.org/10.1017/s0033291702006700
- Galić, Z., Scherer, K. T., & LeBreton, J. M. (2014). Validity evidence for a Croatian version of the Conditional Reasoning Test for Aggression. *International Journal of Selection and Assessment*, 22(4), 343– 354. <u>https://doi.org/10.1111/ijsa.12082</u>
- Gambaro, L., Kreyenfeld, M., Schacht, D., & Spieß, C. K. (2018). Refugees in Germany with children still living abroad have lowest life satisfaction. *DIW Weekly Report*, 8(42), 415–425. <u>http://hdl.handle. net/10419/183869</u>
- Gargalianou, V., Muehlfeld, K., Urbig, D., & Van Witteloostuijn, A. (2015). The effects of gender and personality on foreign language anxiety among adult multilinguals. Universitätsbibliothek Wuppertal, University Library. <u>https://www. econstor.eu/bitstream/10419/111688/1/826935257.</u> pdf
- Garkisch, M., Heidingsfelder, J., & Beckmann, M. (2017). Third sector organizations and migration: A systematic literature review on the contribution of third sector organizations in view of flight, migration and refugee crises. *Voluntas*, 28(5), 1839–1880. <u>https://doi.org/10.1007/s11266-017-9895-4</u>
- Geeraert, N., Li, R., Ward, C., Gelfand, M., & Demes, K. A. (2019). A tight spot: How personality moderates the impact of social norms on sojourner adaptation. *Psychological Science*, *30*(3), 333– 342. https://doi.org/10.1177/0956797618815488
- Gericke, D., Burmeister, A., Löwe, J., Deller, J., & Pundt, L. (2018). How do refugees use their social capital for successful labor market integration? An exploratory analysis in Germany. *Journal* of Vocational Behavior, 105, 46–61. <u>https://doi. org/10.1016/j.jvb.2017.12.002</u>
- Gibson, K. J., Zerbe, W. J., & Franken, R. E. (1993). Employers' perceptions of the reemployment barriers faced by older job hunters. *Relations Industrielles,* 48(2), 321–335. <u>http://dx.doi.</u> <u>org/10.7202/050857ar</u>
- Gnambs, T. (2017). Human capital and reemployment success: The role of cognitive abilities and personality. *Journal of Intelligence*, 5(1), Article 9. https://doi.org/10.3390/jintelligence5010009
- Goffin, R. D., & Christiansen, N. D. (2003). Correcting personality tests for faking: A review of popular personality tests and an initial survey of researchers. *International Journal of Selection* and Assessment, 11(4), 340–344. <u>https://doi.org/10.1111/j.0965-075x.2003.00256.x</u>
- Goldberg, L. R. (1999). A broad-bandwidth, public-

domain, personality inventory measuring the lower-level facets of several five-factor models. In I. Mervielde, I. J. Deary, F. De Fruyt, & F. Ostendorf (Eds.), *Personality psychology in Europe* (Vol. 7, pp. 7–28). Tilburg University Press.

- Goldberg, L. R., Johnson, J. A., Eber, H. W., Hogan, R., Ashton, M. C., Cloninger, C. R., & Gough, H. C. (2006). The International Personality Item Pool and the future of public-domain personality measures. *Journal of Research in Personality*, 40(1), 84–96. <u>https://doi.org/10.1016/j.jrp.2005.08.007</u>
- Goodman, S. A., & Svyantek, D. J. (1999). Personorganization fit and contextual performance: Do shared values matter. *Journal of Vocational Behavior*, 55(2), 254–275. <u>https://doi.org/10.1006/</u> jvbe.1998.1682
- Gore, P. A. (2000). Cluster analysis. In Tinsley, H. E. & Brown, S. D. (Eds.), Handbook of Applied Multivariate Statistics and Mathematical Modeling (pp. 297–321). Academic Press.
- Greenwald, A. G., & Banaji, M. R. (1995). Implicit social cognition: attitudes, self-esteem, and stereotypes. *Psychological Review*, 102(1), 4–27. https://doi.org/10.1037/0033-295X.102.1.4
- Greenwald, A. G., McGhee, D. E., & Schwartz, J. L. K. (1998). Measuring individual differences in implicit cognition: the implicit association test. *Journal of Personality and Social Psychology*, 74(6), 1464– 1480. <u>https://doi.org/10.1037/0022-3514.74.6.1464</u>
- Griffith, R. L., Chmielowski, T., & Yoshita, Y. (2007). Do applicants fake? An examination of the frequency of applicant faking behavior. *Personnel Review*, 36(3), 341–355. <u>https://doi.org/10.1108/00483480710731310</u>
- Guerrero, L., & Rothstein, M. G. (2012). Antecedents of underemployment: Job search of skilled immigrants in Canada. *Applied Psychology*, *61*(2), 323–346. <u>https://doi.org/10.1111/j.1464-0597.2011.00470.x</u>
- Gurven, M., von Rueden, C., Massenkoff, M., Kaplan, H., & Lero Vie, M. (2013). How universal is the Big Five? Testing the five-factor model of personality variation among forager–farmers in the Bolivian Amazon. *Journal of Personality and Social Psychology*, *104*(2), 354–370. <u>https://doi.org/10.1037/a0030841</u>
- Guven, C., & Islam, A. (2015). Age at migration, language proficiency, and socioeconomic outcomes: evidence from Australia. *Demography*, 52(2), 513– 542. https://doi.org/10.1007/s13524-015-0373-6
- Hahn, E., Richter, D., Schupp, J., & Back, M. D. (2019). Predictors of refugee adjustment: The importance of cognitive skills and personality. *Collabra: Psychology*, 5(1), 23. <u>https://doi.org/10.1525/</u> <u>collabra.212</u>
- Hainmueller, J., Hangartner, D., & Lawrence, D. (2016). When lives are put on hold: Lengthy asylum processes decrease employment among

refugees. Science Advances, 2(8), Article e1600432. https://doi.org/10.1126/sciadv.1600432

- Hair, J. F. Jr., Black, W. C., Babin, B. J., & Anderson, R. E. (2010). *Multivariate Data Analysis*. Pearson.
- Harari, M. B., McCombs, K., & Wiernik, B. M. (2021). Movement Capital, RAW model, or circumstances? A meta-analysis of perceived employability predictors. *Journal of Vocational Behavior*, 131, 103657. https://doi.org/10.1016/j.jvb.2021.103657
- Harari, M. B., & Viswesvaran, C. (2018). Individual job performance. In D. S. Ones, N. Anderson, & C. Viswesvaran (Eds.), *The sage handbook of industrial, work and organizational psychology* (pp. 55–72). SAGE Publications Ltd.
- Harder, N., Figueroa, L., Gillum, R. M., Hangartner, D., Laitin, D. D., & Hainmueller, J. (2018). Multidimensional measure of immigrant integration. *Proceedings of the National Academy* of Sciences, 115(45), 11483–11488. <u>https://doi. org/10.1073/pnas.1808793115</u>
- Harrington, R., & Loffredo, D. A. (2011). Insight, rumination, and self- reflection as predictors of well-being. *The Journal of Psychology*, 145(1), 39–57. <u>https://</u> doi.org/10.1080/00223980.2010.528072
- Harris, S., Dykxhoorn, J., Hollander, A.-C., Dalman, C., & Kirkbride, J. B. (2019). Substance use disorders in refugee and migrant groups in Sweden: A nationwide cohort study of 1.2 million people. *PLOS Medicine*, *16*(11), e1002944. <u>https://doi.org/10.1371/journal.pmed.1002944</u>
- Hartog, J., & Zorlu, A. (2009). How important is homeland education for refugees' economic position in The Netherlands?. *Journal of Population Economics*, *22*(1), 219–246. <u>https://doi.org/10.1007/</u> s00148-007-0142-y
- Hayfron, J. E. (2001). Language training, language proficiency and earnings of immigrants in Norway. *Applied Economics*, 33(15), 1971–1979. <u>https://doi.org/10.1080/00036840010018630</u>
- He, J., Buchholz, J., & Klieme, E. (2017). Effects of anchoring vignettes on comparability and predictive validity of student self-reports in 64 cultures. *Journal of Cross-Cultural Psychology*, 48(3), 319–334. <u>https://</u> doi.org/10.1177/0022022116687395
- Heine, S. J. (2001). Self as cultural product: an examination of East Asian and North American selves. *Journal of Personality*, 69(6), 881–905. https://doi.org/10.1111/1467-6494.696168
- Heine, S. J., & Buchtel, E. E. (2009). Personality: The universal and the culturally specific. Annual Review of Psychology, 60(1), 369–394. <u>https://doi.</u> org/10.1146/annurev.psych.60.110707.163655
- Heine, S. J., Buchtel, E. E., & Norenzayan, A. (2008). What do cross-national comparisons of personality traits tell us? The case of conscientiousness. *Psychological Science*, *19*(4), 309–313. https://doi.org/10.1111/j.1467-9280.2008.02085.x

Α

Heine, S. J., Lehman, D. R., Peng, K., & Greenholtz, J. (2002). What's wrong with cross-cultural comparisons of subjective Likert scales?: The reference-group effect. *Journal of Personality* and Social Psychology, 82(6), 903–918. <u>https://</u> doi.org/10.1037/0022-3514.82.6.903

- Henrich, J., Heine, S. J., & Norenzayan, A. (2010). Most people are not WEIRD. *Nature*, 466(7302), 29–29. https://doi.org/10.1038/466029a
- Herrnstein, R. J., & Murray, C. (1994). The bell curve: Intelligence and class structure in American life. Simon & Schuster
- Hilbig, B. E., & Zettler, I. (2009). Pillars of cooperation: Honesty–Humility, social value orientations, and economic behavior. *Journal of Research in Personality*, 43(3), 516–519. <u>https://doi.</u> org/10.1016/j.jrp.2009.01.003
- Hirschfeld, R. R., Lawson, L., & Mossholder, K. W. (2004). Moderators of the relationship between cognitive ability and performance: General versus context– specific achievement motivation. *Journal of Applied Social Psychology*, 34(11), 2389–2409. https://doi.org/10.1111/j.1559-1816.2004.tb01983.x
- Hodson, G., Book, A., Visser, B. A., Volk, A. A., Ashton, M. C., & Lee, K. (2018). Is the dark triad common factor distinct from low honesty-humility?. *Journal* of Research in Personality, 73, 123–129. <u>https://</u> doi.org/10.1016/j.jrp.2017.11.012
- Hofer, J., & Chasiotis, A. (2022). Implicit motives across cultures. *Online Readings in Psychology* and *Culture*, 4(1), Article 5. <u>https://doi.</u> org/10.9707/2307-09191097
- Hoffmann, S., Mai, R., & Cristescu, A. (2013). Do culture-dependent response styles distort substantial relationships? *International Business Review*, 22(5), 814–827. <u>https://doi.org/10.1016/j.</u> ibusrev.2013.01.008
- Hofmann, W., Gawronski, B., Gschwendner, T., Le, H., & Schmitt, M. (2005). A meta-analysis on the correlation between the Implicit Association Test and explicit self-report measures. *Personality and Social Psychology Bulletin*, 31(10), 1369–1385. https://doi.org/10.1177/0146167205275613
- Hofstede, G. (2001). Culture's consequences: Comparing values, behaviors, institutions and organizations across nations. Sage publications.
- Hofstede, G. (2011). Dimensionalizing Cultures: The Hofstede Model in Context. Online Readings in Psychology and Culture, 2(1). <u>https://doi.org/10.9707/2307-0919.1014</u>
- Hogan, J., & Holland, B. (2003). Using theory to evaluate personality and job-performance relations: A socioanalytic perspective. *Journal* of *Applied Psychology*, *88*(1), 100–112. <u>https://doi. org/10.1037/0021-9010.88.1.100</u>
- Hogan, R., & Chamorro-Premuzic, T. (2015). Personality and career success. In M. L. Cooper & R. J. Larsen (Eds.), *Handbook of personality*

and social psychology (Vol. 4, pp. 619–638). American Psychological Association. <u>https://doi.org/10.1037/14343-028</u>

- Hogan, R., Chamorro-Premuzic, T., & Kaiser, R. B. (2013). Employability and career success: Bridging the gap between theory and reality. *Industrial and Organizational Psychology*, 6(1), 3–16. <u>https://doi.org/10.1111/iops.12001</u>
- Holland, J. L. (1959). A theory of vocational choice. Journal of Counseling Psychology, 6(1), 35–45. https://doi.org/10.1037/h0040767
- Hollifield, M., Warner, T. D., Lian, N., Krakow, B., Jenkins, J. H., Kesler, J., Stevenson, J., & Westermeyer, J. (2002). Measuring trauma and health status in refugees. *JAMA*, *288*(5), 611–621. <u>https://doi. org/10.1001/jama.288.5.611</u>
- Holtrop, D. J. (2016). Improving personality and interest measurement for purposes of selection and assessment [Doctoral dissertation, Vrije Universiteit Amsterdam]. VU Research Portal. <u>https://research.vu.nl/files/42161060/</u> complete%20dissertation.pdf
- Holtrop, D., Born, M. P., De Vries, A., & De Vries, R. E. (2014). A matter of context: A comparison of two types of contextualized personality measures. *Personality and Individual Differences*, *68*(1), 135–165. <u>https://</u> <u>doi.org/10.1016/j.paid.2014.04.029</u>
- Horyniak, D., Melo, J. S., Farrell, R. M., Ojeda, V. D., & Strathdee, S. A. (2016). Epidemiology of substance use among forced migrants: a global systematic review. *PLOS ONE*, *11*(7), e0159134. <u>https://doi. org/10.1371/journal.pone.0159134</u>
- Hou, F., & Beiser, M. (2006). Learning the language of a new country: A ten-year study of English acquisition by South-East Asian refugees in Canada. International Migration, 44(1), 135–165. https://doi.org/10.1111/j.1468-2435.2006.00358.x
- Hough, L. M., & Oswald, F. L. (2000). Personnel selection: Looking toward the future—remembering the past. Annual Review of Psychology, 51(1), 631–664. <u>https://doi.org/10.1146/annurev.psych.51.1.631</u>
- Howes, S. S., Henning, J., Mills, M. J., & Huffman, A. H. (2018). Yes Virginia, there is a gender disparity problem—And it goes beyond STEM. *Industrial* and Organizational Psychology, 11(2), 318–323. <u>https://doi.org/10.1017/iop.2018.22</u>
- Hudson, N. W., & Fraley, R. C. (2015). Volitional personality trait change: Can people choose to change their personality traits? *Journal of Personality and Social Psychology*, *109*(3), 490–507. <u>https://doi.org/10.1037/pspp0000021</u>
- Huffcutt, A. I., Conway, J. M., Roth, P. L., & Stone, N. J. (2001). Identification and meta-analytic assessment of psychological constructs measured in employment interviews. *Journal of Applied Psychology*, 86(5), 897–913. <u>https://doi.org/10.1037/0021-9010.86.5.897</u>
- Huisman, P. (2020). Hoe de toeslagenaffaire kon

gebeuren. Management kinderopvang, 26(2), 36–37. <u>https://doi.org/10.1007/s41190-020-0260-2</u>

- Huijnk, W., Dagevos, J., Gijsberts, M., & Andriessen, I. (2015). Werelden van verschil, over de sociaalculturele afstand en positie van migrantengroepen in Nederland [Worlds of Difference. On the sociocultural distance and position of migrant groups in the Netherlands]. Sociaal en Cultureel Planbureau. <u>https://www.scp.nl/publicaties/ publicaties/2015/12/16/werelden-van-verschil</u>
- Hunkler, C., & Khourshed, M. (2020). The role of trauma for integration. The case of Syrian refugees. *Soziale Welt*, 71(1–2), 90–122. <u>https://doi.</u> org/10.2139/ssrn.3496726
- Hunkler, C., Edele, A., & Schipolowski, S. (2021). The role of educational resources in the labor market integration of refugees. The case of Syrian asylum seekers in Germany. *Journal for Educational Research Online*, 13(1), 157–181.<u>https://doi. org/10.25656/01:22069</u>
- Hutcheson, G., & Sofroniou, N. (1999). The multivariate social scientist. Sage. <u>https://dx.doi.</u> org/10.4135/9780857028075
- Hwang, S.-S., & Xi, J. (2008). Structural and individual covariates of English language proficiency. *Social Forces*, *86*(3), 1079–1104. <u>https://doi.org/10.1353/</u> <u>sof.0.0017</u>
- Ibrahim, H., Ertl, V., Catani, C., Ismail, A. A., & Neuner, F. (2018). The validity of Posttraumatic Stress Disorder Checklist for DSM-5 (PCL-5) as screening instrument with Kurdish and Arab displaced populations living in the Kurdistan region of Iraq. *BMC Psychiatry*, 18(1). Article 259. <u>https://doi.org/10.1186/s12888-018-1839-z</u>
- Ion, A., Iliescu, D., Aldhafri, S., Rana, N., Ratanadilok, K., Widyanti, A., & Nedelcea, C. (2017). A crosscultural analysis of personality structure through the lens of the HEXACO model. *Journal of Personality Assessment*, 99(1), 25–34. <u>https://doi.org/10.1080/00223891.2016.1187155</u>
- Jackson, D. N. (1974). *Personality Research Form manual* (2nd ed.). Research Psychologists Press.
- James, L. R. (1998). Measurement of personality via conditional reasoning. *Organizational Research Methods*, 1(2), 131–163. <u>https://doi.org/10.1177/109442819812001</u>
- James, L. R., & LeBreton, J. M. (2012). Assessing the implicit personality through conditional reasoning. American Psychological Association. https://doi.org/10.1037/13095-000
- James, L. R., McIntyre, M. D., Glisson, C. A., Green, P. D., Patton, T. W., LeBreton, J. M., Frost, B. C., Russell, S. M., Sablynski, C. J., Mitchell, T. R., & Williams, L. J. (2005). A conditional reasoning measure for aggression. *Organizational Research Methods*, 8(1), 69–99. <u>https://doi. org/10.1177/1094428104272182</u>

moderator in the curvilinear relationships between job demands, and job performance and job satisfaction. Academy of Management Journal, 44(5), 1039–1050. <u>https://doi. org/10.5465/3069447</u>

- Jaramillo, F., Carrillat, F. A., & Locander, W. B. (2005). A meta-analytic comparison of managerial ratings and self-evaluations. *Journal of Personal Selling & Sales Management*, 25(4), 315–328.
- Jennissen, R. P. W. (2011). De Nederlandse migratiekaart: achtergronden en ontwikkelingen van verschillende internationale migratietypen. Wetenschappelijk Onderzoek- en Documentatiecentrum. <u>https://repository.wodc.nl/</u> handle/20.500.12832/1249
- Jensen, A. R. (1998). *The g factor: The science of mental ability* (Vol. 648). Praeger.
- Jokela, M. (2009). Personality predicts migration within and between U.S. states. *Journal of Research in Personality,* 43(1), 79–83. <u>https://doi.org/10.1016/j.</u> jrp.2008.09.005
- Joly, D. (1996). Haven or Hell? Asylum policies and refugees in Europe. Macmillan.
- Jones, T., & Newburn, T. (2001). Widening access: Improving police relations with hard to reach groups. Home Office, Policing and Reducing Crime Unit. <u>https://www.ojp.gov/ncjrs/virtuallibrary/abstracts/widening-access-improvingpolice-relations-hard-reach-groups</u>
- Jongen, E., Muns, S., Thijs, C., De Boer, H. W., Dagevos, J., Dillingh, R., Ebregt, J., Huijnk, W., Karpinska, K., Klaver, J., Schaper, J., De Wilde, T., & Zulkarnain, A. (2020). Kansrijk integratiebeleid op de arbeidsmarkt [Promising integration policy in the labor market]. <u>https://www.scp.nl/publicaties/ publicaties/2020/10/15/kansrijk-integratiebeleidop-de-arbeidsmarkt</u>
- Joseph, D. L., & Newman, D. A. (2010). Emotional intelligence: An integrative meta-analysis and cascading model. *Journal of Applied Psychology*, 95(1), 54–78. <u>https://doi.org/10.1037/a0017286</u>
- Judge, T. A., & Bono, J. E. (2001). Relationship of core self-evaluations traits—self-esteem, generalized self-efficacy, locus of control, and emotional stability—with job satisfaction and job performance: A meta-analysis. Journal of Applied Psychology, 86(1), 80–92. <u>https://doi.org/10.1037/0021-9010.861.80</u>
- Judge, T. A., & Ilies, R. (2002). Relationship of personality to performance motivation: A meta-analytic review. *Journal of Applied Psychology*, 87(4), 797–807. https://doi.org/10.1037/0021-9010.87.4.797
- Judge, T. A., & Zapata, C. P. (2015). The personsituation debate revisited: Effect of situation strength and trait activation on the validity of the big five personality traits in predicting job performance. Academy of Management Journal, 58(4), 1149–1179. https://doi.org/10.5465/

Janssen, O. (2001). Fairness perceptions as a

amj.2010.0837

- Judge, T. A., Cable, D. M., Boudreau, J. W., & Bretz, R. D. Jr. (1995). An empirical investigation of the predictors of executive career success. *Personnel Psychology*, 48(3), 485–519. <u>https://doi.org/10.1111/j.1744-6570.1995.tb01767.x</u>
- Judge, T. A., Higgins, C. A., Thoresen, C. J., & Barrick, M. R. (1999). The big five personality traits, general mental ability, and career success across the life span. *Personnel Psychology*, 52(3), 621–652. https://doi.org/10.1111/j.1744-6570.1999.tb00174.x
- Judge, T. A., Rodell, J. B., Klinger, R. L., Simon, L. S., & Crawford, E. R. (2013). Hierarchical representations of the five-factor model of personality in predicting job performance: Integrating three organizing frameworks with two theoretical perspectives. *Journal of Applied Psychology*, *98*(6), 875–925. https://doi.org/10.1037/a0033901
- Kacmar, K. M., Witt, L. A., Zivnuska, S., & Gully, S. M. (2003). The interactive effect of leadermember exchange and communication frequency on performance ratings. *Journal of Applied Psychology*, *88*(4), 764–772. <u>https://doi.org/10.1037/0021-9010.88.4.764</u>
- Kaiser, H. F. (1970). A second-generation little jiffy. *Psychometrika*, 35(4), 401–415. <u>https://doi.org/10.1007/bf02291817</u>
- Kalleberg, A. L., & Mastekaasa, A. (2001). Satisfied movers, committed stayers: The impact of job mobility on work attitudes in Norway. *Work* and Occupations, 28(2), 183–209. <u>https://doi.</u> org/10.1177/0730888401028002004
- Kanas, A., & Van Tubergen, F. (2009). The impact of origin and host country schooling on the economic performance of immigrants. Social Forces, 88(2), 893–915. https://doi.org/10.1353/sof.0.0269
- Kanas, A., Chiswick, B. R., Van der Lippe, T., & Van Tubergen, F. (2012). Social contacts and the economic performance of immigrants: A panel study of immigrants in Germany. *International Migration Review*, 46(3), 680–709. <u>https://doi. org/10.1111/j.1747-7379.2012.00901.x</u>
- Kanas, A., Van Tubergen, F., & Van der Lippe, T. (2011). The role of social contacts in the employment status of immigrants: A panel study of immigrants in Germany. *International Sociology*, 26(1), 95– 122. <u>https://doi.org/10.1177/0268580910380977</u>
- Kandler, C. (2012). Nature and nurture in personality development: The case of neuroticism and extraversion. *Current Directions in Psychological Science*, *21*(5), 290–296. <u>https://doi. org/10.1177/0963721412452557</u>
- Kanfer, R., & Heggestad, E. D. (1997). Motivational traits and skills: A person-centered approach to work motivation. *Research in Organizational Behavior*, 19, 1–56.
- Kanfer, R., Wanberg, C. R., & Kantrowitz, T. M. (2001). Job search and employment: A personality–

motivational analysis and meta-analytic review. Journal of Applied Psychology, 86(5), 837–855. https://doi.org/10.1037/0021-9010.86.5.837

- Kassambara, A., Kosinski, M., Biecek, P., & Fabian, S. (2017). Package 'survminer'. <u>https://CRAN.R-project.org/package=survminer</u>.
- Keita, S., & Schewe, P. (2021). Out of sight, out of mind? Terror in the home country, family reunification options, and the well-being of refugees. World Development, 146, Article 105562. <u>https://doi. org/10.1016/j.worlddev.2021.105562</u>
- Kelloway, E. K., Loughlin, C., Barling, J., & Nault, A. (2002). Self-reported counterproductive behaviors and organizational citizenship behaviors: Separate but related constructs. *International Journal* of Selection and Assessment, 10(1-2), 143–151. <u>https://doi.org/10.1111/1468-2389.00201</u>
- Kessler, R. C. (2000). Posttraumatic stress disorder: the burden to the individual and to society. *Journal of Clinical Psychiatry*, 61, 4–12.
- Kessler, R. C., Andrews, G., Colpe, L. J., Hiripi, E., Mroczek, D. K., Normand, S. L. T., Walters, E. E., & Zaslavsky, A. M. (2002). Short screening scales to monitor population prevalences and trends in non-specific psychological distress. *Psychological Medicine*, 32(6), 959–976. <u>https://doi.org/10.1017/ s0033291702006074</u>
- Kessler, R. C., Barker, P. R., Colpe, L. J., Epstein, J. F., Gfroerer, J. C., Hiripi, E., Howes, M. J., Normand, S.-L. T., Manderscheid, R. W., Walters, E. E., & Zaslavsky, A. M. (2003). Screening for serious mental illness in the general population. *Archives* of *General Psychiatry*, 60(2), 184–189. <u>https://doi. org/10.1001/archpsyc.60.2.184</u>
- Khawaja, N. G., & Hebbani, A. (2018). Does employment status vary by demographics? An exploratory study of former refugees resettled in Australia. Australian Social Work, 71(1), 71–85. https://doi.org/10.1080/0312407X.2017.1376103
- King, G., & Wand, J. (2007). Comparing incomparable survey responses: Evaluating and selecting anchoring vignettes. *Political Analysis*, 15(1), 46– 66. <u>https://doi.org/10.1093/pan/mpl011</u>
- King, G., Murray, C. J. L., Salomon, J. A., & Tandon, A. (2004). Enhancing the validity and cross-cultural comparability of measurement in survey research. *American Political Science Review*, 98(1), 191–207. https://doi.org/10.1017/s000305540400108x
- Kirkman, B. L., & Shapiro, D. L. (2001). The impact of team members' cultural values on productivity, cooperation, and empowerment in selfmanaging work teams. *Journal of Cross-Cultural Psychology*, 32(5), 597–617. <u>https://doi.org/10.1177/0022022101032005005</u>
- Klehe, U.-C., & Anderson, N. (2007). Working hard and working smart: Motivation and ability during typical and maximum performance. *Journal of Applied Psychology*, 92(4), 978–992. <u>https://doi.</u>

org/10.1037/0021-9010.92.4.978

- Klehe, U.-C, Koen, J., & De Pater, I. E. (2012). Ending on the scrap heap: The experience of job loss and job search among older workers. In J. W. Hedge & W. C. Borman (Eds.), *The Oxford handbook of work and aging* (pp. 313–340). Oxford University Press.
- Kofman, E. (2014). Towards a gendered evaluation of (highly) skilled immigration policies in Europe. International Migration, 52(3), 116–128. <u>https://doi.org/10.1111/imig.12121</u>
- Koopmans, R. (2010). Trade-offs between equality and difference: immigrant integration, multiculturalism and the welfare state in cross-national perspective. *Journal of Ethnic and Migration Studies*, 36(1), 1–26. <u>https://doi.org/10.1080/13691830903250881</u>
- Koser, K. (2007). Refugees, transnationalism and the state. Journal of Ethnic and Migration Studies, 33(2), 233–254. <u>https://doi.</u> org/10.1080/13691830601154195
- Kosny, A., Yanar, B., Begum, M., Al-khooly, D., Premji, S., Lay, M. A., & Smith, P. M. (2020). Safe employment integration of recent immigrants and refugees. *Journal of International Migration and Integration*, 21(3), 807–827. <u>https://doi.org/10.1007/s12134-019-00685-w</u>
- Kostek, J. A. (2012). Work centrality: A meta-analysis of the nomological network [Master's thesis, Bowling Green State University]. <u>https://etd.ohiolink.</u> <u>edu/apexprod/rws_etd/send_file/send?accession=bgsu1339525599&disposition=inline</u>
- Kosyakova, Y., & Laible, M. C. (2021). Socioemotional Skills and Refugees' Language Acquisition. Centre for Research and Analysis of Migration, Department of Economics, University College London. <u>https://ideas.repec.org/p/crm/ wpaper/2130.html</u>
- Krahn, H., Derwing, T., Mulder, M., & Wilkinson, L. (2000). Educated and underemployed: Refugee integration into the Canadian labour market. *Journal of International Migration and Integration*, 1(1), 59–84. <u>https://doi.org/10.1007/s12134-000-1008-2</u>
- Kristen, C., Mühlau, P., & Schacht, D. (2016). Language acquisition of recently arrived immigrants in England, Germany, Ireland, and the Netherlands. *Ethnicities*, 16(2), 180–212. <u>https://doi.org/10.1177/1468796815616157</u>
- Kumar, G. S., Beeler, J. A., Seagle, E. E., & Jentes, E. S. (2021). Long-term physical health outcomes of resettled refugee populations in the United States: A scoping review. *Journal of Immigrant* and *Minority Health*, 23(4), 813–823. <u>https://doi. org/10.1007/s10903-021-01146-2</u>
- Kuncel, N. R., & Beatty, A. (2013). Thinking at work: Intelligence, critical thinking, job knowledge, and reasoning. In K. Geisinger, B. A. Bracken, J. F Carlson, J.-I. Hansen, N. R. Kuncel, S. P. Reise, &

M. C. Rodriguez (Eds.), APA handbook of testing and assessment in psychology (pp. 417–435). American Psychological Association. <u>http://dx.doi.</u> org/10.1037/14047-024

- Laajaj, R., Macours, K., Pinzon Hernandez, D. A., Arias, O., Gosling, S. D., Potter, J., Rubio-Codina, M., & Vakis, R. (2019). Challenges to capture the big five personality traits in non-WEIRD populations. *Science Advances*, 5(7), Article eaaw5226. <u>https://doi.org/10.1126/sciadv.</u> <u>aaw5226</u>
- Lacroix, M., Baffoe, M., & Liguori, M. (2015). Refugee community organizations in Canada: From the margins to the mainstream? A challenge and opportunity for social workers. *International Journal of Social Welfare*, *24*(1), 62–72. <u>https://doi. org/10.1111/ijsw.12110</u>
- LaHuis, D. M., Martin, N. R., & Avis, J. M. (2005). Investigating nonlinear conscientiousness–job performance relations for clerical employees. *Human Performance*, *18*(3), 199–212. <u>https://doi.org/10.1207/s15327043hup1803_1</u>
- Lai, H., Due, C., & Ziersch, A. (2022). The relationship between employment and health for people from refugee and asylum-seeking backgrounds: A systematic review of quantitative studies. SSM - Population Health, 18, 101075. <u>https://doi.org/10.1016/j.ssmph.2022.101075</u>
- Lancee, B. (2012). Immigrant performance in the labour market: Bonding and bridging social capital. Amsterdam University Press. <u>https://doi.</u> org/10.26530/oapen_418150
- Lane, S., Raymond, M. R., Haladyna, T. M., & Downing, S. M. (2016). Test development process. In S. Lane, M. R. Raymond, & T. M. Haladyna (Eds.), *Handbook of test development* (2nd ed., pp. 3–18). Routledge. https://doi.org/10.4324/9780203102961
- Lang, J. W., Zettler, I., Ewen, C., & Hülsheger, U. R. (2012). Implicit motives, explicit traits, and task and contextual performance at work. *Journal of Applied Psychology*, 97(6), 1201–1217. <u>https://doi. org/10.1037/a0029556</u>
- Le, H., Oh, I.-S., Robbins, S. B., Ilies, R., Holland, E., & Westrick, P. (2011). Too much of a good thing: Curvilinear relationships between personality traits and job performance. *Journal of Applied Psychology*, 96(1), 113–133. <u>https://doi.org/10.1037/</u> <u>a0021016</u>
- Leavitt, K., Fong, C. T., & Greenwald, A. G. (2011). Asking about well-being gets you half an answer: Intra-individual processes of implicit and explicit job attitudes. *Journal of Organizational Behavior*, 32(4), 672–687. <u>https://doi.org/10.1002/job.746</u>
- LeBel, E. P., & Paunonen, S. V. (2011). Sexy but often unreliable: The impact of unreliability on the replicability of experimental findings with implicit measures. *Personality and Social*

Α

Psychology Bulletin, 37(4), 570–583. <u>https://doi.org/10.1177/0146167211400619</u>

- LeBreton, J. M., Barksdale, C. D., Robin, J., & James, L. R. (2007). Measurement issues associated with conditional reasoning tests: Indirect measurement and test faking. *Journal of Applied Psychology*, 92(1), 1–16. <u>https://doi.org/10.1037/0021-9010.92.1.1</u>
- LeBreton, J. M., Reichin, S. L., Nijenhuis, J., Cremers, M., & Heijden-Lek, K. (2022). Validity evidence and measurement equivalence for the Dutch translation of the conditional reasoning test for aggression. *Applied Psychology*, *71*(2), 710–739. https://doi.org/10.1111/apps.12309
- Lee, E. S., Szkudlarek, B., Nguyen, D. C., & Nardon, L. (2020). Unveiling the canvas ceiling: A multidisciplinary literature review of refugee employment and workforce integration. International Journal of Management Reviews, 22(2), 193–216. <u>https://doi.org/10.1111/</u> ijmr.12222
- Lee, K., & Allen, N. J. (2002). Organizational citizenship behavior and workplace deviance: The role of affect and cognitions. *Journal of Applied Psychology*, *87*(1), 131–142. <u>https://doi.org/10.1037/0021-9010.87.1.131</u>
- Lee, K., & Ashton, M. C. (2004). Psychometric properties of the HEXACO Personality Inventory. *Multivariate Behavioral Research*, *39*(2), 329–358. <u>https://doi. org/10.1207/s15327906mbr3902_8</u>
- Lee, K., & Ashton, M. C. (2008). The HEXACO personality factors in the indigenous personality lexicons of English and 11 other languages. *Journal* of *Personality*, 76(5), 1001–1054. <u>https://doi. org/10.1111/j.1467-6494.2008.00512.x</u>
- Lee, K., & Ashton, M. C. (2020). Sex differences in HEXACO personality characteristics across countries and ethnicities. *Journal of Personality*, *88*(6), 1075–1090. <u>https://doi.org/10.1111/jopy.12551</u>
- Lee, K., Ashton, M. C., & De Vries, R. E. (2005). Predicting workplace delinquency and integrity with the HEXACO and five-factor models of personality structure. *Human Performance*, *18*(2), 179–197. https://doi.org/10.1207/s15327043hup1802_4
- Lee, Y., Berry, C. M., & Gonzalez-Mulé, E. (2019). The importance of being humble: A meta-analysis and incremental validity analysis of the relationship between honesty-humility and job performance. *Journal of Applied Psychology*, 104(12), 1535– 1546. https://doi.org/10.1037/apI0000421
- LePine, J. A., & Van Dyne, L. (2001). Voice and cooperative behavior as contrasting forms of contextual performance: Evidence of differential relationships with big five personality characteristics and cognitive ability. *Journal of Applied Psychology*, *86*(2), 326–336. <u>https://doi. org/10.1037/0021-9010.86.2.326</u>

Lessard-Phillips, L. (2017). Exploring the dimensionality

180 | APPENDICES

Α

of ethnic minority adaptation in Britain: An analysis across ethnic and generational lines. *Sociology*, *51*(3), 626–645. <u>https://doi.org/10.1177/0038038515609030</u>

- Levy, F., & Cannon, C. (2016). The Bloomberg job skills report 2016: What recruiters want. <u>https://www. bloomberg.com/graphics/2016-job-skills-report/</u>
- Liebau, E., & Salikutluk, Z. (2016). Viele Geflüchtete brachten Berufserfahrung mit, aber nur ein Teil einen Berufsabschluss [Many refugees have job experiences but only a fraction has a vocational qualification]. *DIW Wochenbericht, 35,* 732–740. <u>https://www.diw.de/documents/publikationen/73/ diw_01.c.541799.de/16-35-3.pdf</u>
- Lilienfeld, S. O., Sauvigné, K. C., Lynn, S. J., Cautin, R. L., Latzman, R. D., & Waldman, I. D. (2015). Fifty psychological and psychiatric terms to avoid: a list of inaccurate, misleading, misused, ambiguous, and logically confused words and phrases. *Frontiers in Psychology*, 6, Article 1100. https://doi.org/10.3389/fpsyg.2015.01100
- Lilienfeld, S. O., Wood, J. M., & Garb, H. N. (2000). The scientific status of projective techniques. *Psychological Science in the Public Interest,* 1(2), 27–66. <u>https://doi.org/10.1111/1529-1006.002</u>
- Lind, N. (2014). Legatum Prosperity Index. Encyclopedia of Quality of Life and Well-Being Research, 3529–3530. <u>https://doi.org/10.1007/978-94-007-0753-5_1619</u>
- Lindqvist, E., & Vestman, R. (2011). The Labor Market Returns to Cognitive and Noncognitive Ability: Evidence from the Swedish Enlistment. *American Economic Journal: Applied Economics, 3*(1), 101– 128. <u>https://doi.org/10.1257/app.3.1.101</u>
- Lippens, L., Vermeiren, S., & Baert, S. (2022). The State of Hiring Discrimination: A Meta-Analysis of (Almost) All Recent Correspondence Experiments. *SSRN Electronic Journal*. Article 14966. <u>https://doi. org/10.2139/ssrn.4114491</u>
- Liu, S., Huang, J. L., & Wang, M. (2014). Effectiveness of job search interventions: a meta-analytic review. *Psychological Bulletin*, *140*(4), 1009–1041. <u>https://doi.org/10.1037/a0035923</u>
- Loch, D. (2014). Integration as a sociological concept and national model for immigrants: scope and limits. *Identities*, *21*(6), 623–632. <u>https://doi.org/10. 1080/1070289x.2014.908776</u>
- Lubinski, D. (2009). Exceptional cognitive ability: The phenotype. *Behavior Genetics*, *39*(4), 350–358. https://doi.org/10.1007/s10519-009-9273-0
- Lubinski, D., & Dawis, R. V. (1992). Aptitudes, skills, and proficiencies. In M. D. Dunnette & L. M. Hough (Eds.), *Handbook of industrial and organizational psychology* (2nd ed., Vol. 3, pp. 1–59). Consulting Psychologists Press.
- Lundborg, P. (2013). Refugees' Employment integration in Sweden: cultural distance and labor

market performance. *Review of International Economics*, *21*(2), 219–232. <u>https://doi.org/10.1111/roie.12032</u>

- Lundborg, P., & Skedinger, P. (2016). Employer attitudes towards refugee immigrants: Findings from a Swedish survey. *International Labour Review*, *155*(2), 315–337. <u>https://doi.org/10.1111/</u> ilr.12026
- MacCann, C., Jiang, Y., Brown, L. E. R., Double, K. S., Bucich, M., & Minbashian, A. (2020). Emotional intelligence predicts academic performance: A meta-analysis. *Psychological Bulletin*, 146(2), 150– 186. https://doi.org/10.1037/bul0000219
- Mackey, J. D., McAllister, C. P., Maher, L. P., & Wang, G. (2019). Leaders and followers behaving badly: A meta-analytic examination of curvilinear relationships between destructive leadership and followers' workplace behaviors. *Personnel Psychology*, 72(1), 3–47. <u>https://doi.org/10.1111/ peps.12286</u>
- Maliepaard, M., Witkamp, B., & Jennissen, R. (2017). *Een kwestie van tijd? De integratie van asielmigranten: een cohortonderzoek.* WODC. <u>https://repository.wodc.nl/handle/20.500.12832/181</u>
- Marcus, B., Lee, K., & Ashton, M. C. (2007). Personality dimensions explaining relationships between integrity tests and counterproductive behavior: big five, or one in addition? *Personnel Psychology*, *60*(1), 1–34. <u>https://doi.org/10.1111/j.1744-6570.2007.00063.x</u>
- Markus, H. R., & Kitayama, S. (1991). Culture and the self: Implications for cognition, emotion, and motivation. *Psychological Review*, 98(2), 224– 253. https://doi.org/10.1037/0033-295x.98.2.224
- Markus, H. R., & Kitayama, S. (1998). The cultural psychology of personality. *Journal of Cross-Cultural Psychology*, 29(1), 63–87. <u>https://doi.</u> org/10.1177/0022022198291004
- Maxwell, B. A., Pryor, F. L., & Smith, C. (2002). Cluster analysis in cross-cultural research. *World Cultures*, *13*(1), 22–38. <u>https://works.swarthmore.</u> <u>edu/fac-economics/228</u>
- McAllister, D. J. (1995). Affect-and cognition-based trust as foundations for interpersonal cooperation in organizations. Academy of Management Journal, 38(1), 24–59. https://www.jstor.org/stable/256727
- McArdle, S., Waters, L., Briscoe, J. P., & Hall, D. T. (2007). Employability during unemployment: Adaptability, career identity and human and social capital. *Journal of Vocational Behavior*, 71(2), 247–264. https://doi.org/10.1016/j.jvb.2007.06.003
- McCarthy, J. M., Bauer, T. N., Truxillo, D. M., Anderson, N. R., Costa, A. C., & Ahmed, S. M. (2017). Applicant perspectives during selection: A review addressing "So what?," "What's new?," and "Where to next?". *Journal of Management*, 43(6), 1693– 1725. https://doi.org/10.1177/0149206316681846

McCrae, R. R., & Costa, P. T., Jr. (2008). The five-factor

theory of personality. In O. P. John, R. W. Robins, & L. A. Pervin (Eds.), *Handbook of personality: Theory and research* (pp. 159–181). Guilford Press.

- McDonald, J. D. (2008). Measuring personality constructs: The advantages and disadvantages of self-reports, informant reports and behavioural assessments. *Enguire*, 1(1), 75–94.
- McDowall, A., & Redman, A. (2017, April 5). Psychological assessment – an overview of theoretical, practical and industry trends [Video]. YouTube. <u>https://youtu.be/Sa-kU5qwilE</u>
- McIntosh, A., & Cockburn-Wootten, C. (2019). Refugeefocused service providers: improving the welcome in New Zealand. *The Service Industries Journal*, 39(9–10), 701–716. <u>https://doi.org/10.1080</u> /02642069.2018.1472243
- McKee-Ryan, F., Song, Z., Wanberg, C. R., & Kinicki, A. J. (2005). Psychological and physical well-being during unemployment: A meta-analytic study. *Journal of Applied Psychology*, 90(1), 53–76. <u>https://doi.org/10.1037/0021-9010.90.1.53</u>
- Meier, K. J., & O'Toole, L. J. Jr. (2013). I think (I am doing well), therefore I am: Assessing the validity of administrators' self-assessments of performance. International Public Management Journal, 16(1), 1–27. <u>https://doi.org/10.1080/109674</u> 94.2013.796253
- Merzenich, M. (2013). Soft-wired: How the new science of brain plasticity can change your life. Parnassus.
- Mesch, G. S. (2003). Language proficiency among new immigrants: The role of human capital and societal conditions: The case of immigrants from the Fsu in Israel. *Sociological Perspectives*, *46*(1), 41–58. https://doi.org/10.1525/sop.2003.46.1.41
- Miltenburg, E., & Dagevos, J. (2021). Looking beyond employment: participation profiles of Syrian refugees in the Netherlands. *International Migration*, 59(4), 59–73. <u>http://doi.org/10.1111/</u> imig.12785
- Mitchell, M. S., & Ambrose, M. L. (2007). Abusive supervision and workplace deviance and the moderating effects of negative reciprocity beliefs. *Journal of Applied Psychology*, *92*(4), 1159–1168. <u>https://doi.org/10.1037/0021-</u> 9010.92.4.1159
- Mitchell, T. R., & Nebeker, D. M. (1973). Expectancy theory predictions of academic effort and performance. *Journal of Applied Psychology*, 57(1), 61–67. <u>https://doi.org/10.1037/h0034181</u>
- Moon, H., Hollenbeck, J. R., Marinova, S., & Humphrey, S. E. (2008). Beneath the surface: Uncovering the relationship between extraversion and organizational citizenship behavior through a facet approach. *International Journal of Selection and Assessment*, *16*(2), 143–154. <u>https://doi. org/10.1111/j.1468-2389.2008.00419.x</u>
- Moors, A., Spruyt, A., & de Houwer, J. (2010). In search of a measure that qualifies as implicit:

Δ

Recommendations based on a decompositional view of automaticity. In B. Gawronski & K. B. Payne (Eds.), *Handbook of implicit social cognition: Measurement, theory, and applications* (pp. 19–37). Guilford Press.

- Morgan, C. D., & Murray, H. A. (1935). A method for investigating fantasies: The Thematic Apperception Test. *Archives of Neurology & Psychiatry*, 34(2), 289–306. <u>https://doi.org/10.1001/</u> archneurpsyc.1935.02250200049005
- Morgeson, F. P., Reider, M. H., & Campion, M. A. (2005). Selecting individuals in team settings: The importance of social skills, personality characteristics, and teamwork knowledge. *Personnel Psychology*, 58(3), 583-611. https://doi.org/10.1111/j.1744-6570.2005.655.x
- Morren, M., Gelissen, J. P. T. M., & Vermunt, J. K. (2012). Response strategies and response styles in cross-cultural surveys. Cross-Cultural Research, 46(3), 255–279. <u>https://doi.</u> org/10.1177/1069397112440939
- Mossholder, K. W., Settoon, R. P., & Henagan, S. C. (2005). A relational perspective on rurnover: Examining structural, attitudinal, and behavioral predictors. *Academy of Management Journal*, 48(4), 607– 618. https://doi.org/10.5465/amj.2005.17843941
- Mount, M. K., Barrick, M. R., & Strauss, J. P. (1999). The joint relationship of conscientiousness and ability with performance: Test of the interaction hypothesis. *Journal of Management*, 25(5), 707– 721. <u>https://doi.org/10.1177/014920639902500505</u>
- MOW. (1987). The Meaning of Working. Academic Press.
- Muck, P. M., Hell, B., & Gosling, S. D. (2007). Construct Validation of a Short Five-Factor Model Instrument. *European Journal of Psychological Assessment*, 23(3), 166–175. <u>https://doi.org/10.1027/1015-</u> 5759.23.3.166
- Ndofor-Tah, C., Strang, A. Phillimore, J. Morrice, L. Michael, L. Wood, P. & Simmons. J. (2019). *Home Office Indicators of Integration Framework* 2019. UK Home Office. <u>http://sro.sussex.ac.uk/</u> id/eprint/84107/1/__smbhome.uscs.susx.ac.uk_ dm50_Desktop_Morrice%20Home%20Office.pdf
- Neisser, U., Boodoo, G., Bouchard, T. J., Jr., Boykin, A. W., Brody, N., Ceci, S. J., Halpern, D. F., Loehlin, J. C., Perloff, R., Sternberg, R. J., & Urbina, S. (1996). Intelligence: Knowns and unknowns. *American Psychologist*, *51*(2), 77–101. <u>https://doi. org/10.1037/0003-066X.51.2.77</u>
- Newman, A., Dunwoodie, K., Jiang, Z., & Nielsen, I. (2022). Openness to experience and the career adaptability of refugees: How do career optimism and family social support matter? *Journal of Career Assessment*, 30(2), 309–328. <u>https://doi. org/10.1177/10690727211041532</u>
- Ng, T. W. H., & Feldman, D. C. (2008). The relationship of age to ten dimensions of job performance. *Journal of Applied Psychology*, 93(2), 392–423.

https://doi.org/10.1037/0021-9010.93.2.392

- Ng, T. W. H., Eby, L. T., Sorensen, K. L., & Feldman, D. C. (2005). Predictors of objective and subjective career success: A meta-analysis. *Personnel Psychology*, 58(2), 367–408. <u>https://doi.org/10.1111/</u> j.1744-6570.2005.00515.x
- Ng, T. W. H., Lam, S. S., & Feldman, D. C. (2016). Organizational citizenship behavior and counterproductive work behavior: Do males and females differ?. *Journal of Vocational Behavior*, 93, 11–32. <u>https://doi.org/10.1016/j.</u> jvb.2015.12.005
- Nielsen, T. M., Hrivnak, G. A., & Shaw, M. (2009). Organizational citizenship behavior and performance: A meta-analysis of group-level research. *Small Group Research*, *40*(5), 555–577. https://doi.org/10.1177/1046496409339630
- Niven, K., Holman, D., & Totterdell, P. (2012). How to win friendship and trust by influencing people's feelings: An investigation of interpersonal affect regulation and the quality of relationships. *Human Relations*, 65(6), 777–805. <u>https://doi. org/10.1177/0018726712439909</u>
- NOA. (2005). Arbeids Motivatie en Werkzoek Vragenlijst [Work Motivation and Work Search Questionnaire]. NOA.
- NOA. (2006). Handleiding & Normen NOA Taaltest Nederlands [Manual & Norms NOA Dutch Language Test]. NOA.
- NOA. (2009). Handleiding Multiculturele Persoonlijkheidstest Big Six [Manual Multicultural Personality Test Big Six]. NOA.
- NOA. (2016). Gebruikersinformatie Competentietest (CT) [User Information Competency Test (CT)]. NOA.
- NOA. (2021). Aanbesteding SZW-MVO Leerbaarheidstoets [Unpublished manuscript]. NOA.
- Noftle, E. E., & Robins, R. W. (2007). Personality predictors of academic outcomes: Big five correlates of GPA and SAT scores. *Journal of Personality and Social Psychology*, 93(1), 116–130. <u>https://doi. org/10.1037/0022-3514.93.1.116</u>
- Nuffic. (2022). Education systems. Retrieved August 22, 2022, from <u>https://www.nuffic.nl/en/subjects/</u> <u>diploma/education-systems</u>
- Nye, C. D., Ma, J., & Wee, S. (2022). Cognitive ability and job performance: Meta-analytic evidence for the validity of narrow cognitive abilities. *Journal of Business and Psychology*, 37(6), 1119–1139. <u>https:// doi.org/10.1007/s10869-022-09796-1</u>
- O'Boyle, E. H., Forsyth, D. R., Banks, G. C., & McDaniel, M. A. (2012). A meta-analysis of the Dark Triad and work behavior: A social exchange perspective. *Journal of Applied Psychology*, 97(3), 557–579. https://doi.org/10.1037/a0025679
- O'Connor, M. C., & Paunonen, S. V. (2007). Big Five

personality predictors of post-secondary academic performance. *Personality and Individual Differences*, 43(5), 971–990. <u>https://doi.</u> org/10.1016/j.paid.2007.03.017

- Odé, A., & Dagevos, J. (2017). Statushouders aan het werk: De moeizame positieverwerving op de arbeidsmarkt en de betekenis van beleidsfactoren. *Mens & Maatschappij, 92*(4), 447–468. <u>https://doi.org/10.5117/mem2017.4.ode</u>
- OECD. (2016). How are refugees faring on the labour market in Europe?. <u>https://data.europa.eu/</u> doi/10.2767/350756
- OECD/EU. (2019). Settling In 2018: Indicators of Immigrant Integration. OECD Publishing. <u>https://</u> doi.org/10.1787/9789264307216-en
- OECD/EU. (2015). Indicators of Immigrant Integration 2015: Settling In. OECD Publishing. <u>https://doi.org/10.1787/9789264234024-en</u>
- Oh, I.-S., Le, H., Whitman, D. S., Kim, K., Yoo, T.-Y., Hwang, J.-O., & Kim, C.-S. (2014). The incremental validity of honesty–humility over cognitive ability and the big five personality traits. *Human Performance*, 27(3), 206–224. <u>https://doi.org/10.10</u> 80/08959285.2014.913594
- Oh, I.-S., Lee, K., Ashton, M. C., & De Vries, R. E. (2011). Are dishonest extraverts more harmful than dishonest introverts? The interaction effects of honesty-humility and extraversion in predicting workplace deviance. Applied Psychology: An International Review, 60, 496–516. <u>http://dx.doi. org/10.1111/j.1464-0597.2011.00445.x</u>
- Ones, D. S., Viswesvaran, C., & Schmidt, F. L. (1993). Comprehensive meta-analysis of integrity test validities: Findings and implications for personnel selection and theories of job performance. *Journal of Applied Psychology*, 78(4), 679–703. https://doi.org/10.1037/0021-9010.78.4.679
- Organ, D. (1988). Organizational citizenship behavior: The good soldier syndrome. Lexington Books.
- Organ, D. W., Podsakoff, P. M., & MacKenzie, S. B. (2006). Organizational citizenship behavior: Its nature, antecedents, and consequences. Sage Publications. <u>https://doi.org/10.4135/9781452231082</u>
- Ortlieb, R., & Weiss, S. (2020). Job quality of refugees in Austria: Trade-offs between multiple workplace characteristics. *German Journal of Human Resource Management*, 34(4), 418– 442. <u>https://</u> doi.org/10.1177/2397002220914224
- Pace, V. L., & Brannick, M. T. (2010). How similar are personality scales of the "same" construct? A meta-analytic investigation. *Personality and Individual Differences*, 49(7), 669–676. <u>https://doi.org/10.1016/j.paid.2010.06.014</u>
- Pajic, S., Ulceluse, M., Kismihók, G., Mol, S. T., & Den Hartog, D. N. (2018). Antecedents of job search self-efficacy of Syrian refugees in Greece and the Netherlands. *Journal of Vocational Behavior*, 105,

159-172. https://doi.org/10.1016/j.jvb.2017.11.001

- Park, G., Lubinski, D., & Benbow, C. P. (2008). Ability differences among people who have commensurate degrees matter for scientific creativity. *Psychological Science*, *19*(10), 957–961. <u>https://doi.org/10.1111/j.1467-9280.2008.02182.x</u>
- Passmore, J. (2012). Psychometrics in coaching: Using psychological and psychometric tools for development (2nd ed). Kogan Page.
- Paul, K. I., & Moser, K. (2009). Unemployment impairs mental health: Meta-analyses. *Journal of Vocational Behavior*, 74(3), 264–282. <u>https://doi.org/10.1016/j.jvb.2009.01.001</u>
- Paulhus, D. L. (1991). Measurement and control of response Bias. In J. P. Robinson, P. R. Shaver, & L. S. Wright (Eds.), *Measures of personality* and social psychological attitudes (pp. 17–59). Academic.
- Paulhus, D. L., & Williams, K. M. (2002). The dark triad of personality: Narcissism, Machiavellianism, and psychopathy. *Journal of Research in Personality*, 36(6), 556–563. <u>https://doi.org/10.1016/s0092-6566(02)00505-6</u>
- Payne, B. K., Burkley, M. A., & Stokes, M. B. (2008). Why do implicit and explicit attitude tests diverge? The role of structural fit. *Journal of Personality* and Social Psychology, 94(1), 16–31. <u>https://doi. org/10.1037/0022-3514.94.1.16</u>
- Penninx, R. (2019). Problems of and solutions for the study of immigrant integration. *Comparative Migration Studies*, 7(1), Article 13. <u>https://doi.</u> org/10.1186/s40878-019-0122-x
- Perry, S. J., Hunter, E. M., Witt, L. A., & Harris, K. J. (2010). P = f (conscientiousness × ability): Examining the facets of conscientiousness. *Human Performance,* 23(4), 343–360. <u>https://doi.org/10.1080/08959285</u> .2010.501045
- Petruzziello, G., Mariani, M. G., Chiesa, R., & Guglielmi, D. (2021). Self-efficacy and job search success for new graduates. *Personnel Review*, 50(1), 225– 243. https://doi.org/10.1108/pr-01-2019-0009
- Pishghadam, R., & Khajavy, G. H. (2013). Intelligence and metacognition as predictors of foreign language achievement: A structural equation modeling approach. *Learning and Individual Differences, 24*, 176–181. <u>https://doi.org/10.1016/j.</u> lindif.2012.12.004
- Pletzer, J. L., Bentvelzen, M., Oostrom, J. K., & De Vries, R. E. (2019). A meta-analysis of the relations between personality and workplace deviance: Big Five versus HEXACO. *Journal of Vocational Behavior*, *112*, 369–383. <u>https://doi.org/10.1016/j. jvb.2019.04.004</u>
- Pletzer, J. L., Oostrom, J. K., & De Vries, R. E. (2021). HEXACO personality and organizational citizenship behavior: A domain-and facet-level meta-analysis. *Human Performance*, 1–22. <u>https://</u> doi.org/10.1080/08959285.2021.1891072

- Pletzer, J. L., Oostrom, J. K., Bentvelzen, M., & De Vries, R. E. (2020). Comparing domain-and facet-level relations of the HEXACO personality model with workplace deviance: A meta-analysis. *Personality* and Individual Differences, 152, Article 109539. http://dx.doi.org/10.1016/j.paid.2019.109539
- Ployhart, R. E., & Holtz, B. C. (2008). The diversity–validity dilemma: Strategies for reducing racioethnic and sex subgroup differences and adverse impact in selection. *Personnel Psychology*, 61(1), 153–172. https://doi.org/10.1111/j.1744-6570.2008.00109.x
- Podsakoff, N. P., Whiting, S. W., Podsakoff, P. M., & Blume, B. D. (2009). Individual- and organizational-level consequences of organizational citizenship behaviors: A meta-analysis. *Journal of Applied Psychology, 94*(1), 122–141. <u>https://doi.org/10.1037/</u> a0013079
- Podsakoff, P. M., MacKenzie, S. B., & Podsakoff, N. P. (2012). Sources of method bias in social science research and recommendations on how to control it. *Annual Review of Psychology*, 63(1), 539–569. <u>https://doi.org/10.1146/annurevpsych-120710-100452</u>
- Podsakoff, P. M., MacKenzie, S. B., Lee, J.-Y., & Podsakoff, N. P. (2003). Common method biases in behavioral research: A critical review of the literature and recommended remedies. *Journal of Applied Psychology*, *88*(5), 879–903. <u>https://doi. org/10.1037/0021-9010.88.5.879</u>
- Podsakoff, P. M., MacKenzie, S. B., Paine, J. B., & Bachrach, D. G. (2000). Organizational citizenship behaviors: A critical review of the theoretical and empirical literature and suggestions for future research. *Journal of Management*, *26*(3), 513–563. https://doi.org/10.1177/014920630002600307
- Poropat, A. E. (2009). A meta-analysis of the five-factor model of personality and academic performance. *Psychological Bulletin, 135*(2), 322-338. <u>https:// doi.org/10.1037/a0014996</u>
- Porter, M., & Haslam, N. (2005). Predisplacement and postdisplacement factors associated with mental health of refugees and internally displaced persons. *JAMA*, 294(5), 602–612. <u>https://doi.org/10.1001/jama.294.5.602</u>
- Portes, A. (2000). The two meanings of social capital. Sociological Forum, 15(1), 1–11. <u>https://doi.org/10.1023/A:1007537902813</u>
- Portes, A., & Rumbaut, R. G. (2006). *Immigrant America:* A portrait. University of California Press. <u>https://doi.org/10.1525/9780520940482</u>
- Potocky-Tripodi, M. (2004). The role of social capital in immigrant and refugee economic adaptation. *Journal of Social Service Research*, *31*(1), 59–91. <u>https://doi.org/10.1300/j079v31n01_04</u>
- Primi, R., Zanon, C., Santos, D., De Fruyt, F., & John, O. P. (2016). Anchoring vignettes: Can they make adolescent self-reports of social-emotional skills more reliable, discriminant, and criterion-

valid? European Journal of Psychological Assessment, 32(1), 39–51. <u>https://doi.</u> org/10.1027/1015-5759/a000336

- Protzko, J., & Schooler, J. W. (2019). Kids these days: Why the youth of today seem lacking. *Science Advances*, 5(10), Article eaav5916. <u>https://doi.</u> org/10.1126/sciadv.aav5916
- Puma, J. E., Lichtenstein, G., & Stein, P. (2018). The RISE survey: Developing and implementing a valid and reliable quantitative measure of refugee integration in the United States. *Journal* of Refugee Studies, 31(4), 605–625. <u>https://doi.org/10.1093/jrs/fex047</u>
- Rammstedt, B., & John, O. P. (2007). Measuring personality in one minute or less: A 10-item short version of the Big Five Inventory in English and German. Journal of Research in Personality, 41(1), 203–212. https://doi.org/10.1016/j.jrp.2006.02.001
- Razenberg, I., Kahmann, M., & De Gruijter, M. (2018). *Mind the gap. Barrières en mogelijkheden voor de arbeidsparticipatie van vluchtelingenvrouwen.* <u>https://www.kis.nl/sites/default/files/bestanden/</u> <u>Publicaties/barrieres-mogelijkheden-</u> <u>arbeidsparticipatie-vluchtelingenvrouwen.pdf</u>
- Ree, M. J., Earles, J. A., & Teachout, M. S. (1994). Predicting job performance: Not much more than g. *Journal of Applied Psychology*, 79, 518–524. <u>http://dx.doi.org/10.1037/0021-9010.79.4.518</u>
- Renner, W., & Senft, B. (2013). Predictors of Unemployment in Refugees. Social Behavior and Personality: An International Journal, 41(2), 263– 270. https://doi.org/10.2224/sbp.2013.41.2.263
- Ridgeway, C. (2017). Projective measures and occupational assessment. In B. Cripps (Ed.), *Psychometric testing: Critical perspectives* (pp. 213–220). John Wiley & Sons. <u>https://doi.org/10.1002/9781119183020</u>
- Rimfeld, K., Dale, P. S., & Plomin, R. (2015). How specific is second language-learning ability? A twin study exploring the contributions of first language achievement and intelligence to second language achievement. *Translational Psychiatry*, 5(9), Article e638. <u>https://doi.org/10.1038/tp.2015.128</u>
- Robbins, S. B., Lauver, K., Le, H., Davis, D., Langley, R., & Carlstrom, A. (2004). Do psychosocial and study skill factors predict college outcomes? A metaanalysis. *Psychological Bulletin*, 130(2), 261–288. https://doi.org/10.1037/0033-2909.130.2.261
- Roberts, B. W., Kuncel, N. R., Shiner, R., Caspi, A., & Goldberg, L. R. (2007). The power of personality: The comparative validity of personality traits, socioeconomic status, and cognitive ability for predicting important life outcomes. *Perspectives* on *Psychological Science*, 2(4), 313–345. <u>https://</u> doi.org/10.111/j.1745-6916.2007.00047.x
- Roberts, B. W., Luo, J., Briley, D. A., Chow, P. I., Su, R., & Hill, P. L. (2017). A systematic review of personality trait change through intervention. *Psychological*

Bulletin, 143(2), 117–141. <u>https://doi.org/10.1037/</u> bul0000088

- Robertson, K. F., Smeets, S., Lubinski, D., & Benbow, C. P. (2010). Beyond the threshold hypothesis: Even among the gifted and top math/science graduate students, cognitive abilities, vocational interests, and lifestyle preferences matter for career choice, performance, and persistence. *Current Directions* in *Psychological Science*, *19*(6), 346–351. <u>https:// doi.org/10.1177/0963721410391442</u>
- Robie, C., & Ryan, A. M. (1999). Effects of nonlinearity and heteroscedasticity on the validity of conscientiousness in predicting overall job performance. *International Journal of Selection and Assessment*, 7(3), 157–169. <u>https://doi.</u> org/10.1111/1468-2389.00115
- Robinson, S. L., & Bennett, R. J. (1995). A typology of deviant workplace behaviors: A multidimensional scaling study. Academy of Management Journal, 38(2), 555–572. https://doi.org/10.5465/256693
- Rosseel, Y. (2012). Lavaan: An R package for structural equation modeling. *Journal of Statistical Software*, 48(2), 1–36. <u>https://doi.org/10.18637/jss.v048.i02</u>
- Rothstein, M. G., & Goffin, R. D. (2006). The use of personality measures in personnel selection: What does current research support? *Human Resource Management Review*, *16*(2), 155–180. https://doi.org/10.1016/j.hrmr.2006.03.004
- Rotundo, M., & Sackett, P. R. (2002). The relative importance of task, citizenship, and counterproductive performance to global ratings of job performance: A policy-capturing approach. *Journal of Applied Psychology, 87*(1), 66–80. <u>http://dx.doi.org/10.1037/0021-9010.871.66</u>
- Rubenstein, A. L., Eberly, M. B., Lee, T. W., & Mitchell, T. R. (2018). Surveying the forest: A meta-analysis, moderator investigation, and future-oriented discussion of the antecedents of voluntary employee turnover. *Personnel Psychology*, *71*(1), 23–65. https://doi.org/10.1111/peps.12226
- Rudmin, F. W. (2003). Critical history of the acculturation psychology of assimilation, separation, integration, and marginalization. *Review of General Psychology*, 7(1), 3–37. <u>https://doi.org/10.1037/1089-2680.7.1.3</u>
- Ruiz, I., & Vargas-Silva, C. (2018). Differences in labour market outcomes between natives, refugees and other migrants in the UK, *Journal of Economic Geography*, 18 (4), 855–885, <u>https://doi. org/10.1093/jeg/lby027</u>
- Ryan, A. M., & Ployhart, R. E. (2000). Applicants' perceptions of selection procedures and decisions: A critical review and agenda for the future. *Journal of Management*, *26*(3), 565–606. https://doi.org/10.1177/014920630002600308
- Sackett, P. R., & DeVore, C. J. (2001). Counterproductive behaviors at work. In N. Anderson, D. S. Ones, H. K. Sinangil, & C. Viswesvaran (Eds.), Handbook of

industrial, work, and organizational psychology (Vol. 1, pp. 145–151). Sage. <u>http://dx.doi.</u> org/10.4135/9781848608320.n9

- Sackett, P. R., & Walmsley, P. T. (2014). Which personality attributes are most important in the workplace? *Perspectives on Psychological Science*, 9(5), 538– 551. https://doi.org/10.1177/1745691614543972
- Sackett, P. R., Gruys, M. L., & Ellingson, J. E. (1998). Ability–personality interactions when predicting job performance. *Journal of Applied Psychology*, *83*(4), 545–556. <u>https://doi.org/10.1037/0021-</u> 9010.83.4.545
- Sackett, P. R., Lievens, F., Van Iddekinge, C. H., & Kuncel, N. R. (2017). Individual differences and their measurement: A review of 100 years of research. *Journal of Applied Psychology*, 102(3), 254–273. <u>https://doi.org/10.1037/ap10000151</u>
- Sackett, P. R., Schmitt, N., Ellingson, J. E., & Kabin, M. B. (2001). High-stakes testing in employment, credentialing, and higher education: Prospects in a post-affirmative-action world. *American Psychologist*, 56(4), 302–318. <u>https://doi.org/10.1037/0003-066x.56.4.302</u>
- Sackett, P. R., Zhang, C., Berry, C. M., & Lievens, F. (2022). Revisiting meta-analytic estimates of validity in personnel selection: Addressing systematic overcorrection for restriction of range. *Journal of Applied Psychology*, 107(11), 2040–2068.

https://doi.org/10.1037/apl0000994

- Saharso, S. (2019). Who needs integration? Debating a central, yet increasingly contested concept in migration studies. *Comparative Migration Studies*, 7(1), Article 16. <u>https://doi.org/10.1186/s40878-019-0123-9</u>
- Saks, A. M. (2005). Job search success: A review and integration of the predictors, behaviors and outcomes. In S. D. Brown & R. W. Lent (Eds.), *Career development and counseling: Putting theory and research to work* (pp. 155–179). John Wiley & Sons, Inc.
- Saks, A. M., & Ashforth, B. E. (1999). Effects of individual differences and job search behaviors on the employment status of recent university graduates. *Journal of Vocational Behavior*, *54*(2), 335–349. https://doi.org/10.1006/jvbe.1998.1665
- Salgado, J. F., Anderson, N., Moscoso, S., Bertua, C., De Fruyt, F., & Rolland, J. P. (2003). A meta-analytic study of general mental ability validity for different occupations in the European community. *Journal* of Applied Psychology, 88(6), 1068–1081. <u>https:// doi.org/10.1037/0021-9010.88.6.1068</u>
- Salikutluk, Z., Giesecke, J., & Kroh, M. (2016). Refugees entered the labor market later than other migrants. *DIW Economic Bulletin*, 6(34/35), 407–413. <u>https://www.diw.de/documents/</u> <u>publikationen/73/diw_01.c.542012.de/diw_econ_</u> bull_2016-34-5.pdf

Saucier, G. (2009). Recurrent personality dimensions in

inclusive lexical studies: Indications for a big six structure. Journal of Personality, 77(5), 1577–1614. https://doi.org/10.1111/j.1467-6494.2009.00593.x

- Schellekens, P. (2001). English language as a barrier to employment, education and training. DfES. <u>https://</u> www.researchgate.net/publication/268412244_ English_Language_As_A_Barrier_To_ Employment_Education_And_Training
- Schinkel, W. (2018). Against 'immigrant integration': for an end to neocolonial knowledge production. *Comparative Migration Studies*, 6(1), Article 31. <u>https://doi.org/10.1186/s40878-018-0095-1</u>
- Schmidt, F. L., & Hunter, J. E. (1998). The validity and utility of selection methods in personnel psychology: Practical and theoretical implications of 85 years of research findings. *Psychological Bulletin*, 124(2), 262–274. <u>https://doi.org/10.1037/0033-</u> 2909.124.2.262
- Schmidt, F. L., & Hunter, J. E. (2004). General mental ability in the world of work: Occupational attainment and job performance. *Journal of Personality and Social Psychology*, 86(1), 162–173. https://doi.org/10.1037/0022-3514.86.1.162
- Schmitt, D. P., Allik, J., McCrae, R. R., & Benet-Martínez, V. (2007). The geographic distribution of big five personality traits: Patterns and profiles of human self-description across 56 nations. *Journal of Cross-Cultural Psychology*, 38(2), 173–212. <u>https:// doi.org/10.1177/0022022106297299</u>
- Schneider, J., & Crul, M. (2010). New insights into assimilation and integration theory: Introduction to the special issue. *Ethnic and Racial Studies*, 33(7), 1143–1148. <u>https://doi.org/10.1080/01419871003777809</u>
- Schock, K., Böttche, M., Rosner, R., Wenk-Ansohn, M., & Knaevelsrud, C. (2016). Impact of new traumatic or stressful life events on pre-existing PTSD in traumatized refugees: results of a longitudinal study. *European Journal of Psychotraumatology*, 7(1), Article 32106. <u>https://doi.org/10.3402/ejpt. v7.32106</u>
- Schultheiss, O. C., Liening, S. H., & Schad, D. (2008). The reliability of a Picture Story Exercise measure of implicit motives: Estimates of internal consistency, retest reliability, and ipsative stability. *Journal of Research in Personality*, 42(6), 1560–1571. <u>https://</u> doi.org/10.1016/j.jrp.2008.07.008
- Schwartz, S. H. (1992). Universals in the content and structure of values: Theoretical advances and empirical tests in 20 countries. Advances in Experimental Social Psychology, 1–65. https://doi. org/10.1016/s0065-2601(08)60281-6
- Sedikides, C., Gaertner, L., & Toguchi, Y. (2003). Pancultural self-enhancement. *Journal of Personality and Social Psychology, 84*(1), 60–79. https://doi.org/10.1037/0022-3514.84.1.60
- Seidelsohn, K., Flick, U., & Hirseland, A. (2020).

Refugees' labor market integration in the context of a polarized public discourse. *Qualitative Inquiry, 26*(2), 216–226. <u>https://doi.org/10.1177/1077800419857097</u>

- Seligman, M. E. P., & Csikszentmihalyi, M. (2000). Positive psychology: An introduction. American Psychologist, 55(1), 5–14. <u>https://doi.org/10.1037/0003-066x.55.1.5</u>
- Shaffer, J. A., & Postlethwaite, B. E. (2012). A matter of context: A meta-analytic investigation of the relative validity of contextualized and noncontextualized personality measures. *Personnel Psychology*, 65(3), 445–494. https://doi.org/10.1111/j.1744-6570.2012.01250.x
- Sharabi, M., & Harpaz, I. (2007). Changes in work centrality and other life areas in Israel: A longitudinal study. *Journal of Human Values*, *13*(2), 95–106. <u>https://</u> <u>doi.org/10.1177/097168580701300203</u>
- Sharabi, M., & Harpaz, I. (2010). Improving employees' work centrality improves organizational performance: Work events and work centrality relationships. Human Resource Development International, 13(4), 379–392. <u>https://doi.org/10.10</u> 80/13678868.2010.501960
- Sheeran, P., & Webb, T. L. (2016). The intention– behavior gap. Social and Personality Psychology Compass, 10(9), 503–518. <u>https://doi.org/10.1111/</u> <u>spc3.12265</u>
- Sherif, M., & Hovland, C. I. (1961). Social judgment: Assimilation and contrast effects in communication and attitude change. Yale University Press.
- Shields, M. A., & Price, S. W. (2002). The English language fluency and occupational success of ethnic minority immigrant men living in English metropolitan areas. *Journal of Population Economics*, 15(1), 137–160. <u>https://doi.org/10.1007/ pl00003836</u>
- Shiferaw, D. & Hagos, H. (2002). Refugees and Progression Routes to Employment. Refugee Council. https://lemosandcrane.co.uk/resources/ Refugee%20Council%20-%20Refugees%20 and%20progression%20routes%20to%20 employment.pdf
- Shiner, R. L., Masten, A. S., & Roberts, J. M. (2003). Childhood personality foreshadows adult personality and life outcomes two decades later. *Journal of Personality*, *71*(6), 1145–1170. <u>https://doi.org/10.1111/1467-6494.7106010</u>
- Shweder, R. A. (1991). *Thinking through cultures: Expeditions in cultural psychology.* Harvard University Press.
- Sienkiewicz, M. E., Amalathas, A., Iverson, K. M., Smith, B. N., & Mitchell, K. S. (2020). Examining the association between trauma exposure and work-related outcomes in women veterans. *International Journal of Environmental Research* and Public Health, 17(12), Article 4585. <u>https://doi.</u> org/10.3390/ijerph17124585

- Sijbrandij, M., Kunovski, I., & Cuijpers, P. (2016). Effectiveness of internet-delivered cognitive behavioral therapy for posttraumatic stress disorder: A systematic review and metaanalysis. *Depression and Anxiety*, *33*(9), 783–791. https://doi.org/10.1002/da.22533
- Skehan, P. (1991). Individual differences in second language learning. Studies in Second Language Acquisition, 13(2), 275–298. <u>https://doi.org/10.1017/ s0272263100009979</u>
- Smaldino, P. E., Lukaszewski, A., von Rueden, C., & Gurven, M. (2019). Niche diversity can explain cross-cultural differences in personality structure. *Nature Human Behaviour*, 3(12), 1276–1283. <u>https://doi.org/10.1038/s41562-019-0730-3</u>
- Smith, C. A., Organ, D. W., & Near, J. P. (1983). Organizational citizenship behavior: Its nature and antecedents. *Journal of Applied Psychology*, 68(4), 653–663. <u>https://doi.org/10.1037/0021-9010.68.4.653</u>
- Smith, M. (1994). A theory of the validity of predictors in selection. Journal of Occupational and Organizational Psychology, 67(1), 13–31. <u>https://</u> doi.org/10.1111/j.2044-8325.1994.tb00546.x

Snir, R., & Harpaz, I. (2002). Work-leisure relations: Leisure orientation and the meaning of work. *Journal of Leisure Research*, 34(2), 178–203. https://doi.org/10.1080/00222216.2002.11949968

- Soto, C. J. (2019). How replicable are links between personality traits and consequential Life Outcomes? The Life Outcomes of Personality Replication Project. *Psychological Science*, 30(5), 711–727. <u>https://doi.org/10.1177/0956797619831612</u>
- Spector, P. E., Bauer, J. A., & Fox, S. (2010). Measurement artifacts in the assessment of counterproductive work behavior and organizational citizenship behavior: Do we know what we think we know?. Journal of Applied Psychology, 95(4), 781–790. <u>https://doi.org/10.1037/a0019477</u>
- Spence, J. T., Pred, R. S., & Helmreich, R. L. (1989). Achievement strivings, scholastic aptitude, and academic performance: A follow-up to "Impatience versus achievement strivings in the Type A pattern." *Journal of Applied Psychology*, 74(1), 176–178. <u>https://doi.org/10.1037/0021-</u> 9010.74.1176
- Spencer, S., & Charsley, K. (2016). Conceptualising integration: a framework for empirical research, taking marriage migration as a case study. Comparative Migration Studies, 4(1), Article 18. <u>https://doi.org/10.1186/s40878-016-0035-x</u>
- Steele, C. M., & Aronson, J. (1995). Stereotype threat and the intellectual test performance of African Americans. *Journal of Personality and Social Psychology*, 69(5), 797–811. <u>https://doi. org/10.1037/0022-3514.69.5.797</u>
- Steele, C. M., Spencer, S. J., & Aronson, J. (2002). Contending with group image: The psychology of

stereotype and social identity threat. In *Advances in experimental social psychology* (Vol. 34, pp. 379–440). Academic Press.

- Steffens, M. C. (2004). Is the implicit association test immune to faking?. Experimental Psychology, 51(3), 165–179. <u>https://doi.org/10.1027/1618-3169.51.3.165</u>
- Stenberg, A., & Westerlund, O. (2008). Does comprehensive education work for the long-term unemployed? *Labour Economics*, 15(1), 54–67. https://doi.org/10.1016/j.labeco.2006.11.005
- Sterckx, L., Fessehazion, M., & Teklemariam, B. E. (2018). Eritrese statushouders in Nederland. <u>https://www.scp.nl/publicaties/publicaties/2018/11/16/eritrese-statushouders-in-nederland</u>
- Sternberg, R. J., Grigorenko, E., & Bundy, D. A. (2001). The Predictive Value of IQ. Merrill-Palmer Quarterly, 47(1), 1–41. <u>https://doi.org/10.1353/mpg.2001.0005</u>
- Stips, F., & Kis-Katos, K. (2020). The impact of co-national networks on asylum seekers' employment: Quasiexperimental evidence from Germany. *PLoS One*, *15*(8), Article e0236996. <u>https://doi.</u> org/10.1371/journal.pone.0236996
- Szkudlarek, B. (2019). Engaging business in refugee employment: The employer's perspective. University of Sydney. https://hdl.handle. net/2123/21513
- Tabor, A. S., Milfont, T. L., & Ward, C. (2015). The migrant personality revisited: Individual differences and international mobility intentions. *New Zealand Journal of Psychology*, 44(2), 89–95.
- Tan, P. N., Steinbach, M., & Kumar, V. (2016). *Introduction to data mining*. Pearson Education.
- Taylor, M. (2006). Employers fail to find enough graduates with social skills. The Guardian. <u>https://</u> theguardian.com/money/2006/feb/07/graduates. highereducation
- Taylor, S. E., & Brown, J. D. (1988). Illusion and wellbeing: a social psychological perspective on mental health. *Psychological Bulletin*, 103(2), 193– 210. https://doi.org/10.1037/0033-2909.103.2.193
- Te Nijenhuis, J., & Hartmann, P. (2006). Spearman's "Law of Diminishing Returns" in samples of Dutch and immigrant children and adults. *Intelligence*, *34*(5), 437–447. <u>https://doi.org/10.1016/j.</u> intell.2006.02.002
- Tett, R. P., Jackson, D. N., & Rothstein, M. (1991). Personality measures as predictors of job performance: A meta-analytic review. *Personnel Psychology*, 44(4), 703–742. <u>https://doi.org/10.1111/j.1744-6570.1991.tb00696.x</u>
- Tharmaseelan, N., Inkson, K., & Carr, S. C. (2010). Migration and career success: testing a timesequenced model. *Career Development International*, *15*(3), 218–238. <u>https://doi.org/10.1108/13620431011053712</u>
- Therneau, T. (2021). A Package for Survival Analysis in R. R package version 3.2-13, <u>https://CRAN.R-</u>

project.org/package=survival

- Thielmann, I., & De Vries, R. E. (2021). Who wants to change and how? On the trait-specificity of personality change goals. *Journal of Personality and Social Psychology*, *121*(5), 1112–1139. <u>https:// doi.org/10.1037/pspp0000304</u>
- Thielmann, I., Moshagen, M., Hilbig, B., & Zettler, I. (2021). On the comparability of basic personality models: Meta-analytic correspondence, scope, and orthogonality of the Big Five and HEXACO dimensions. *European Journal of Personality*. https://doi.org/10.1177/08902070211026793
- Thijssen, L., Van Tubergen, F., Coenders, M., Hellpap, R., & Jak, S. (2022). Discrimination of black and Muslim minority groups in Western societies: Evidence from a meta-analysis of field experiments. *The International Migration Review*, 56(3), 843–880. https://doi.org/10.1177/01979183211045044
- Thum, A. (2014). Labor market integration of German immigrants and their children: Does personality matter? SSRN Electronic Journal, Article 723. https://doi.org/10.2139/ssrn.2547396
- Tierens, H., Dries, N., Smet, M., & Sels, L. (2021). Multiple-membership survival analysis and its applications in organizational behavior and management research. *Organizational Research Methods*, 24(2), 412–442. <u>https://doi.</u> org/10.1177/1094428119877452
- Titova, L., Werner, K. M., & Sheldon, K. M. (2018). Translating positive psychology. *Translational Issues in Psychological Science*, 4(3), 211–214. <u>https://doi.org/10.1037/tps0000170</u>
- Toli, A., Webb, T. L., & Hardy, G. E. (2016). Does forming implementation intentions help people with mental health problems to achieve goals? A meta-analysis of experimental studies with clinical and analogue samples. *British Journal of Clinical Psychology*, 55(1), 69–90. <u>https://doi.org/10.1111/ bjc.12086</u>
- Trapnell, P. D., & Wiggins, J. S. (1990). Extension of the Interpersonal Adjective Scales to include the Big Five dimensions of personality. *Journal of Personality and Social Psychology*, 59(4), 781– 790. https://doi.org/10.1037/0022-3514.59.4.781
- Turban, D. B., Stevens, C. K., & Lee, F. K. (2009). Effects of conscientiousness and extraversion on new labor market entrants' job search: The mediating role of metacognitive activities and positive emotions. *Personnel Psychology*, 62(3), 553–573. <u>https://doi.org/10.1111/j.1744-6570.2009.01148.x</u>
- Uhlmann, E. L., Leavitt, K., Menges, J. I., Koopman, J., Howe, M., & Johnson, R. E. (2012). Getting explicit about the implicit: A taxonomy of implicit measures and guide for their use in organizational research. *Organizational Research Methods*, *15*(4), 553– 601. https://doi.org/10.1177/1094428112442750
- UNHCR. (2013). Global Trends 2013. <u>https://www.unhcr.</u> org/ph/wp-content/uploads/sites/28/2017/03/

GlobalTrends2013.pdf

- UNHCR. (2017). Global trends: Forced displacement in 2016. http://www.unhcr.org/5943e8a34.pdf.
- UNHCR. (2021). *Mid-year trends*. Retrieved August 3, 2022 from <u>https://www.unhcr.org/statistics/</u> <u>unhcrstats/618ae4694/mid-year-trends-2021.html</u>
- UNHCR. (2022a). *Refugee statistics*. Retrieved August 3, 2022 from <u>https://www.unrefugees.org/</u> <u>refugee-facts/statistics/</u>
- UNHCR. (2022b). Global trends report. Retrieved August 3, 2022 from <u>https://www.unhcr.org/62a9d1494/</u> global-trends-report-2021
- Valtonen, K. (1994). The Adaptation of Vietnamese Refugees in Finland. *Journal of Refugee Studies*, 7(1), 63–78. <u>https://doi.org/10.1093/jrs/71.63</u>
- Van Breemen, J., Van den Berg, R. H., & Beerepoot, S. (2018). Selecteren op cognitieve capaciteiten: verschillen in capaciteitentestscores tussen jongvolwassenen met en zondermigratieachtergrond in Nederland. Gedrag & Organisatie, 37(4), 314–343. <u>https://doi. org/10.5117/2018.031.004.002</u>
- Van de Vijver, F. J., & Tanzer, N. K. (1997). Bias and equivalence in cross-cultural assessment: An overview. *European Review of Applied Psychology*, 54(2), 119–135. <u>https://doi.org/10.1016/j. erap.2003.12.004</u>
- Van den Berg, R. H. (2001). *Psychologisch onderzoek in een multiculturele samenleving* [Psychological research in a multicultural society; Doctoral dissertation, Vrije Universiteit Amsterdam]. VU Research Portal.
- Van der Slik, F. W., Van Hout, R. W., & Schepens, J. J. (2015). The gender gap in second language acquisition: Gender differences in the acquisition of Dutch among immigrants from 88 countries with 49 mother tongues. *PloS One*, *10*(11), Article e0142056. <u>https://doi.org/10.1371/journal. pone.0142056</u>
- Van Dijk, H., Knappert, L., Muis, Q., & Alkhaled, S. (2022). Roomies for life? An assessment of how staying with a local facilitates refugee integration. *Journal* of Immigrant & Refugee Studies, 20(3), 321–335. https://doi.org/10.1080/15562948.2021.1923879
- Van Hooft, E. A. J., & Noordzij, G. (2009). The effects of goal orientation on job search and reemployment: A field experiment among unemployed job seekers. *Journal of Applied Psychology*, 94(6), 1581–1590. <u>http://dx.doi.org/10.1037/a0017592</u>
- Van Hooft, E. A. J., Born, M. P., Taris, T. W., & Van der Flier, H. (2005). Predictors and outcomes of job search behavior: The moderating effects of gender and family situation. *Journal of Vocational Behavior*, 67(2), 133–152. <u>https://doi.org/10.1016/j. jvb.2004.11.005</u>
- Van Hooft, E. A. J., Born, M. P., Taris, T. W., Van der Flier, H., & Blonk, R. W. B. (2004). Predictors of job search behavior among employed and

unemployed people. *Personnel Psychology*, 57(1), 25–59. <u>https://doi.org/10.1111/j.1744-6570.2004.</u> tb02483.x

- Van Hooft, E. A. J., Kammeyer-Mueller, J. D., Wanberg, C. R., Kanfer, R., & Basbug, G. (2021). Job search and employment success: A quantitative review and future research agenda. *Journal of Applied Psychology*, 106(5), 674–713. <u>https://doi. org/10.1037/apl0000675</u>
- Van Iddekinge, C. H., Aguinis, H., Mackey, J. D., & DeOrtentiis, P. S. (2018). A meta-analysis of the interactive, additive, and relative effects of cognitive ability and motivation on performance. *Journal of Management*, 44(1), 249–279. <u>https://</u> doi.org/10.1177/0149206317702220
- Van Niejenhuis, C., Otten, S., & Flache, A. (2018). Sojourners' second language learning and integration. The moderating effect of multicultural personality traits. *International Journal of Intercultural Relations*, 63, 68–79. <u>https://doi.org/10.1016/j.ijintrel.2018.01.001</u>
- Van Niejenhuis, C., Van der Werf, M. P., & Otten, S. (2015). Predictors of immigrants' second-language proficiency: a Dutch study of immigrants with a low level of societal participation and secondlanguage proficiency. *International Journal of the Sociology of Language*, 2015(236), 75–100. https://doi.org/10.1515/ijsl-2015-0022
- Van Rensburg, Y. E. J., De Kock, F. S., & Derous, E. (2019). 'Going implicit': Using implicit measures in organizations. *Gedrag en Organisatie*, 32(3), 131–161.
- Van Rensburg, Y.-E. J., De Kock, F. S., De Vries, R. E., & Derous, E. (2022). Measuring honesty-humility with an implicit association test (IAT): Construct and criterion validity. *Journal of Research in Personality*, 99, Article 104234. <u>https://doi. org/10.1016/j.jrp.2022.104234</u>
- Van Riemsdijk, M., & Axelsson, L. (2021). Introduction "Labour market integration of highly skilled refugees in Sweden, Germany and the Netherlands." International Migration, 59(4), 3–12. https://doi.org/10.1111/imig.12883
- Van Tubergen, F. (2010). Determinants of second language proficiency among refugees in the Netherlands. *Social Forces*, *89*(2), 515–534. https://doi.org/10.1353/sof.2010.0092
- Van Tubergen, F., & Kalmijn, M. (2005). Destinationlanguage proficiency in cross-national perspective: A study of immigrant groups in nine western countries. *American Journal of Sociology*, 110(5), 1412–1457. <u>https://doi.org/10.1086/428931</u>
- Van Tubergen, F., & Wierenga, M. (2011). The language acquisition of male immigrants in a multilingual destination: Turks and Moroccans in Belgium. *Journal of Ethnic and Migration Studies*, 37(7), 1039–1057. <u>https://doi.org/10.1080/</u> 1369183x.2011.572476

- Van Vianen, A. E. M. (2018). Person–Environment Fit: A Review of Its Basic Tenets. Annual Review of Organizational Psychology and Organizational Behavior, 5(1), 75–101. <u>https://doi.org/10.1146/</u> annurev-orgpsych-032117-104702
- Vardi, Y., & Weitz, E. (2004). *Misbehavior in organizations: Theory, research, and management.* Lawrence Erlbaum Associates Publishers.
- Vargas, P. T., Von Hippel, W., & Petty, R. E. (2004). Using partially structured attitude measures to enhance the attitude-behavior relationship. *Personality and Social Psychology Bulletin*, 30(2), 197–211. <u>https://</u> doi.org/10.1177/0146167203259931
- Vargas, P., Sekaquaptewa, D., & Von Hippel, W. (2007). Armed only with paper and pencil: "Low-tech" measures of implicit attitudes. In B. Wittenbrink & N. Schwarz (Eds.), *Implicit measures of attitudes* (pp. 103–124). Guilford Press.
- Vasilopoulos, N. L., Cucina, J. M., & Hunter, A. E. (2007). Personality and training proficiency: Issues of bandwidth-fidelity and curvilinearity. *Journal* of Occupational and Organizational Psychology, 80(1), 109–131. <u>https://doi.</u> org/10.1348/096317906x102114
- Vazire, S., & Carlson, E. N. (2010). Self-knowledge of personality: Do people know themselves?. Social and Personality Psychology Compass, 4(8), 605–620. <u>https://doi.org/10.1111/j.1751-</u> 9004.2010.00280.x
- Vecchione, M., Dentale, F., Alessandri, G., & Barbaranelli, C. (2014). Fakability of implicit and explicit measures of the Big Five: Research findings from organizational settings. *International Journal of Selection and Assessment*, 22(2), 211–218. <u>https:// doi.org/10.1111/ijsa.12070</u>
- Vélez-Coto, M., Rute-Pérez, S., Pérez-García, M., & Caracuel, A. (2021). Unemployment and general cognitive ability: A review and meta-analysis. *Journal of Economic Psychology*, 87, 102430. https://doi.org/10.1016/j.joep.2021.102430
- Verhoeven, L., & Vermeer, A. (2002). Communicative competence and personality dimensions in first and second language learners. *Applied Psycholinguistics*, 23(3), 361–374. <u>https://doi. org/10.1017/s014271640200303x</u>
- Vianello, M., Schnabel, K., Sriram, N., & Nosek, B. (2013). Gender differences in implicit and explicit personality traits. *Personality and Individual Differences*, 55(8), 994–999. <u>https://doi.org/10.1016/j.paid.2013.08.008</u>
- Vinokur, A. D., & Schul, Y. (2002). The web of coping resources and pathways to reemployment following a job loss. *Journal of Occupational Health Psychology*, 7(1), 68–83. <u>https://doi.org/10.1037/1076-8998.71.68</u>
- Vogiazides, L., Bengtsson, H. & Axelsson, L. (2021) Geographies of occupational (mis) match: The case of highly educated refugees and family

migrants in Sweden. Retrieved August 22, 2022, from <u>https://www.humangeo.su.se/english/</u> about-us/news/new-report-geographies-ofoccupational-mis-match-1.543697

Von Stumm, S., Hell, B., & Chamorro-Premuzic, T. (2011). The hungry mind: Intellectual curiosity is the third pillar of academic performance: Intellectual curiosity is the third pillar of academic performance. Perspectives on *Psychological Science*, 6(6), 574–588. <u>https://doi.org/10.1177/1745691611421204</u>

Vroom, V. H. (1964). Work and motivation. Wiley.

- Wanberg, C. R., Hough, L. M., & Song, Z. (2002). Predictive validity of a multidisciplinary model of reemployment success. *Journal of Applied Psychology*, 87(6), 1100–1120. <u>https://doi. org/10.1037/0021-9010.87.6.1100</u>
- Wanberg, C. R., Kanfer, R., Hamann, D. J., & Zhang, Z. (2016). Age and reemployment success after job loss: An integrative model and meta-analysis. *Psychological Bulletin*, 142(4), 400–426. <u>https:// doi.org/10.1037/bul0000019</u>
- Ward, C., Fischer, R., Zaid Lam, F. S., & Hall, L. (2009). The convergent, discriminant, and incremental validity of scores on a self-report measure of cultural intelligence. *Educational and Psychological Measurement*, 69(1), 85–105. https://doi.org/10.1177/0013164408322001
- Waxman, P. (2001). The economic adjustment of recently arrived Bosnian, Afghan and Iraqi refugees in Sydney, Australia. International Migration Review, 35(2), 472–505. <u>https://doi.org/10.1111/j.1747-7379.2001.tb00026.x</u>
- Westaby, J. D., & Braithwaite, K. N. (2003). Specific factors underlying reemployment self-efficacy: Comparing control belief and motivational reason methods for the recently unemployed. *The Journal* of Applied Behavioral Science, 39(4), 415–437. http://dx.doi.org/10.1177/0021886303261234
- Whetzel, D. L., McDaniel, M. A., Yost, A. P., & Kim, N. (2010). Linearity of Personality– Performance Relationships: A large-scale examination. International Journal of Selection and Assessment, 18, 310–320. doi:10.1111/j.1468-2389.2010.00514.x
- Wihler, A., Meurs, J. A., Momm, T. D., John, J., & Blickle, G. (2017). Conscientiousness, extraversion, and field sales performance: Combining narrow personality, social skill, emotional stability, and nonlinearity. *Personality and Individual Differences*, 104, 291–296. <u>https://doi.org/10.1016/j. paid.2016.07.045</u>
- Wiita, N. E., Meyer, R. D., Kelly, E. D., & Collins, B. J. (2020). Not aggressive or just faking it? Examining faking and faking detection on the Conditional Reasoning Test of Aggression. Organizational Research Methods, 23(1), 96–123. <u>https://doi.</u> org/10.1177/1094428117703685

- Wilmot, M. P., Wanberg, C. R., Kammeyer-Mueller, J. D., & Ones, D. S. (2019). Extraversion advantages at work: A quantitative review and synthesis of the meta-analytic evidence. *Journal of Applied Psychology*, 104(12), 1447–1470. <u>https://doi.org/10.1037/ap10000415</u>
- Wimark, T., Haandrikman, K., & Nielsen, M. M. (2019). Migrant labour market integration: The association between initial settlement and subsequent employment and income among migrants. *Geografiska Annaler: Series B, Human Geography*, 101(2), 118–137. <u>https://doi.org/10.1080/</u> 04353684.2019.1581987
- Witt, L. A., Burke, L. A., Barrick, M. R., & Mount, M. K. (2002). The interactive effects of conscientiousness and agreeableness on job performance. *Journal of Applied Psychology*, *87*(1), 164–169. <u>https://doi.org/10.1037/0021-9010.871.164</u>
- Wood, A. M., Brown, G. D. A., Maltby, J., & Watkinson, P. (2012). How are personality judgments made? A cognitive model of reference group effects, personality scale responses, and behavioral reactions. *Journal of Personality*, 80(5), 1275–1311. https://doi.org/10.1111/j.1467-6494.2012.00763.x
- Wortman, J., & Wood, D. (2011). The personality traits of liked people. Journal of Research in Personality, 45(6), 519–528. <u>https://doi.org/10.1016/j.jrp.2011.06.006</u>
- Wortmann, J. H., Jordan, A. H., Weathers, F. W., Resick, P. A., Dondanville, K. A., Hall-Clark, B., Foa, E. B., Young-McCaughan, S., Yarvis, J. S., Hembree, E. A., Mintz, J., Peterson, A. L., & Litz, B. T. (2016). Psychometric analysis of the PTSD Checklist-5 (PCL-5) among treatment-seeking military service members. *Psychological Assessment, 28*(11), 1392–1403. https://doi.org/10.1037/pas0000260
- Wright, A. M., Dhalimi, A., Lumley, M. A., Jamil, H., Pole, N., Arnetz, J. E., & Arnetz, B. B. (2016). Unemployment in Iraqi refugees: The interaction of pre and postdisplacement trauma. *Scandinavian Journal of Psychology*, 57(6), 564–570. <u>https://doi.org/10.1111/ sjop.12320</u>
- Wright, N. A., & Meade, A. W. (2011, April 13–16). Predictive validity and procedural justice of the implicit association test. 26th Annual Conference Program of the APA Division 14 Society for Industrial and Organizational Psychology (SIOP), Chicago, IL, United States.
- Zeinoun, P., Daouk-Öyry, L., Choueiri, L., & Van de Vijver, F. J. (2017). A mixed-methods study of personality conceptions in the Levant: Jordan, Lebanon, Syria, and the West Bank. *Journal of Personality and Social Psychology*, *113*(3), 453–465. <u>https:// doi.org/10.1037/pspp0000148</u>
- Zhang, J., & Ziegler, M. (2015). Interaction effects between openness and fluid intelligence predicting scholastic performance. *Journal of Intelligence*, 3(3), 91–110. <u>https://doi.org/10.3390/</u> jintelligence3030091

- Ziegler, M., & Peikert, A. (2018). How specific abilities might throw "g" a curve: An idea on how to capitalize on the predictive validity of specific cognitive abilities. *Journal of Intelligence*, 6(3), Article 41. https://doi.org/10.3390/jintelligence6030041
- Ziegler, M., Danay, E., Heene, M., Asendorpf, J., & Bühner, M. (2012). Openness, fluid intelligence, and crystallized intelligence: Toward an integrative model. *Journal of Research in Personality*, 46(2), 173–183. https://doi.org/10.1016/j.jrp.2012.01.002
- Ziegler, M., Knogler, M., & Bühner, M. (2009). Conscientiousness, achievement striving, and intelligence as performance predictors in a sample of German psychology students: Always a linear relationship?. *Learning and Individual Differences*, *19*(2), 288–292. <u>https://doi. org/10.1016/j.lindif.2009.02.001</u>
- Zikic, J., & Klehe, U.-C. (2021). Going against the grain: The role of skilled migrants' self-regulation in finding quality employment. *Journal of Organizational Behavior*, 42(8), 1023–1041. <u>https://</u> <u>doi.org/10.1002/job.2550</u>
- Zorlu, A., & Hartog, J. (2018). The Impact of Language on Socioeconomic Integration of Immigrants. https://doi.org/10.2139/ssrn.3170274
- Zschirnt, E., & Ruedin, D. (2016). Ethnic discrimination in hiring decisions: a meta-analysis of correspondence tests 1990–2015. *Journal of Ethnic and Migration Studies*, 42(7), 1115–1134. https://doi.org/10.2139/ssrn.2597554

Α

CHAPTER 3

Table S1. Model Fit Indices	of Measured Constructs	s for the Refugee Gro	ups Separately

Construct	Refugee group	X ²	Df	CFI	TLI	RMSEA	SRMR
Psychological distress	Syrians	205.752	35	.959	.947	.071	.033
	Eritreans	113.394	35	.951	.937	.069	.042
GMA	Syrians	2528.260	1650	.920	.914	.020	.037
	Eritreans	2082.800	1650	.894	.886	.023	.057
Work search intention	Syrians	1360.831	35	.790	.730	.190	.094
	Eritreans	565.982	35	.770	.704	.175	.090
Conscientiousness	Syrians	305.073	35	.848	.804	.086	.059
	Eritreans	71.481	35	.929	.909	.046	.043
Openness	Syrians	173.365	35	.952	.938	.061	.037
	Eritreans	111.656	35	.909	.882	.067	.048

Note. The refugee group differences in CFI values in this table differ somewhat from the delta CFI values that are reported in the manuscript. This is because the values in the manuscript represent metric measurement invariance exclusively, whereas the values in this table represent the overall CFA findings.

Variable	D	W	CIS.	W	US.	M	US.	σ	Gender	AGP	1105	FA	ΡD	GMA	IS/M	1 I P
	3	0	0	S	S	Э	5	,	00000	281	1	ì	1		5	1
Emotional stability	.64	3.39	0.50	3.33	0.51	3.53	0.44	0.43**	21**	.02	*90.	*	36**	01	.19**	.06*
Conscientiousness	69.	3.86	0.45	3.97	0.42	3.65	0.45	0.74**	05	.14**	06*	.05	07*	.21**	*90.	03
Extraversion	.57	3.46	0.45	3.51	0.45	3.34	0.43	0.39**	.11**	.07**	04	*90.	09**	.16**	**60.	.03
Agreeableness	77.	3.93	0.47	4.00	0.44	3.77	0.50	0.49**	09**	.19**	09**	.07*	02	.23**	.11**	.04
Openness	.83	3.90	0.53	4.00	0.50	3.69	0.53	0.60**	14**	.14**	05*	*90.	03	.26**	.15**	.02
Integrity	.55	2.84	0.47	2.87	0.47	2.77	0.48	0.21**	**60.	.18**	.02	00	07**	.04	10**	04
Note, α = alpha coefficient.	The st	ubscript	- and -	respective	IV represe	ant the ov	erall san	= (N)	1547), the S	Svrian refu	dee aroup	$(n_{-} = 1053)$	3). and the	Eritrean r	efuaee aroi	= <i>- u</i>) an

ability; WSI = work search intention; medium, 0.50 = large (Cohen, 1988). mental á 0.30 = m general I nall. psychological distress; GMA = 1 Eritrean refugees, where 0.10 personality dimension. nent; PD = p_{1}^{2} Syrian and 0 for each p_{2} educational attainment ifferences between Syri of scores is 1.00-5.00 fc of scores EA = (the diff ange of $\delta_{0,S}$ and ϵ respectively S = local length of stay; E hen's d effect size for th sample. The possible ran a = 1. LLoS =y. d = Cohethe total same iguage proficiency. d = re computed for the to <.01 (two-tailed). Female = 0, le pro Male Gender, Ma : local langu ons are ** p < .0o 5; 494). LLP = Corre

CHAPTER 4

The Application of the Exclusion Criteria

In October 2019, all the available assessment data were retrieved. This resulted in a dataset of 9459 refugees. The assessment data and the register data of CBS were linked by zip code (at the time of the assessment), date of birth, and sex. In total, data of 8332 refugees were successfully matched (i.e., 88.1% of the refugees in the original dataset). However, we only used the data of refugees from Syria and Eritrea (n = 6667), as these groups compromise most subjects in the dataset (respectively 62.6% and 17.4%). Additionally, we excluded participants from the analyses who (a) passed away (minus seven participants), (b) left the Netherlands (minus 1376 participants), (c) did not receive a residence permit (minus 169 participants), (d) completed a different version of the assessment (minus 1921 participants), or (e) who received their residence permit after July 2017, since their employment data are not available far enough into the future for conducting the analyses (minus 483 participants). After excluding these refugees, the final dataset consisted of 2711 refugees ($n_{Syria} = 1867$, $n_{Entrea} = 844$).

MPT-BS-QS Basic Pilot Study Results

The pilot study with the English version of the MPT-BS-QS Basic was conducted among an international sample in Prolific (N = 175). The nationalities of the five largest groups in the present dataset were South Africa (n = 72; 41.1%), Portugal (n = 23; 13.1%), Poland (n =19; 10.9%), Italy (n = 15; 8.6%), and Spain (n = 11; 6.3%). Most participants were female (111; 63.4%), and the sample was on average 28.82 years old (SD = 8.27). Participants completed the MPT-BS-QS Basic and the IPIP-100 Big Five measure (Goldberg et al., 2006). Table S1 presents the item-level correlations of the MPT-BS-QS Basic with the IPIP dimensions. Table S2 presents the correlations between the MPT-BS-QS Basic scales and the IPIP dimensions.

Table S1. MPT-BS-QS Basic Item-Leve	Correlations with the IPIP Dimensions
-------------------------------------	---------------------------------------

	0	С	E	А	Ν
A001	.267**	.214**	.329**	.342**	.057
A002	.152*	.318**	.664**	.228**	374**
A003	.252**	.129	.594**	.303**	187*
A004	.243**	.111	.117	.290**	069
A005	.299**	.131	.298**	.293**	.106
A006	.255**	.143	.365**	.437**	085
A007	.393**	.074	.271**	.442**	027
A008	.157*	.117	.515**	.273**	170*
A009	.098	.192*	.266**	.394**	199**
A010	.205**	.189*	.266**	.410**	.024
X001	.316**	.245**	.507**	058	129
X002	.312**	.314**	.316**	.004	164*
X003	.198**	.155*	.646**	.141	230**
X004	.296**	.246**	.456**	.053	158*
X005	.150*	.323**	.425**	.049	252**
X006	.290**	.347**	.631**	.071	211**
X007	.165*	.276**	.473**	.263**	395**
X008	.114	.248**	.322**	127	078
X009	.101	.251**	.237**	130	166*
X010	.142	.264**	.249**	.014	206**
ES001	008	.289**	.299**	.122	554**
ES002	.133	.241**	.195**	.364**	438**
ES003	.002	.027	013	.057	151*
ES004	.151*	.394**	.419**	.223**	548**
ES005	.110	.392**	.382**	.107	588**
ES006	.017	.154*	.237**	.092	443**
ES007	.107	.234**	.132	.342**	414**
ES008	.036	.132	.269**	047	484**
ES009	.026	.281**	.140	.271**	403**
ES010	.062	.343**	.128	.302**	577**
C001	.070	.694**	.168*	.204**	249**
C002	.036	.430**	.075	.065	199**
C003	.362**	.289**	.226**	.061	034
C004	.011	.326**	.060	.194*	261**
C005	.180*	.487**	.326**	.070	288**
C006	.101	.542**	.239**	.247**	344**
C007	.125	.440**	.092	.158*	334**
C008	155*	076	336**	087	.196**
C009	006	.453**	.033	.120	174*
C010	.054	.358**	.000	.036	083

Note. O = Openness to Experience, C = Conscientiousness, E/X = Extraversion, A = Agreeableness, N = Neuroticism, ES = Emotional Stability.

Table S2. Correlations Between the Study's MPT-BS-QS Basic Scales and the IPIP Dimensions

	0	С	E	А	Ν
Agreeableness	.356**	.230**	.403**	.551**	056
Extraversion	.291**	.401**	.712**	.136	321**
Emotional Stability	.123	.462**	.396**	.332**	807**
Conscientiousness	.168*	.726**	.197**	.185*	324**

Α

Scale Item Exclusion

Below, we describe the items that have been excluded from the MPT-BS-QS Basic scales and the work centrality scale of the AWV in the analyses based on the results of the Prolific pilot study and the reliability analyses in the refugee dataset.

Agreeableness

The pilot showed that item A002 ("Other people say that I make contact easily"), item A003 ("I like to be with other people"), and item A008 ("I like to have people around me") correlate highly with Extraversion and therefore need to be excluded from the scale. The alpha coefficient of the remaining 7-item scale equaled .69 for Syrian refugees and .73 for Eritrean refugees.

Extraversion

The alpha coefficient of the 10-item scale is equal to .62 for Syrian refugees and .43 for Eritrean refugees. Excluding items from the scale does not significantly improve the internal consistency of the scale for both refugee groups. However, adding items A002, A003, and A008 (Agreeableness, see above) improves the alpha coefficient to .70 for Syrian refugees and .53 for Eritrean refugees.

Emotional Stability

Item ES003 ("I can hide my feelings well") and item ES006 ("I'm not easily scared") have negative or low item-rest correlations among both Syrian refugees (r = -.04 and r = .21 respectively) and Eritrean refugees (r = -.27 and r = .20 respectively). The alpha coefficient of the 10-item scale is .70 for Syrian refugees and .49 for Eritrean refugees. The alpha coefficient of the scale without item ES003 and item ES006 is .75 for Syrian refugees and .62 for Eritrean refugees.

Conscientiousness

Item C006 ("I'm sloppy") and item C008 ("I like it when things in my life go in the same way") have low item-rest correlations among both Syrian refugees (r = .29 and r = .03, respectively) and Eritrean refugees (r = .06 and r = .16, respectively). The alpha coefficient of the 10-item scale is .70 for Syrian refugees and .71 for Eritrean refugees. The alpha coefficient for the 8-item scale without item C006 and item C008 is .75 for Syrian refugees and .76 for Eritrean refugees.

Work Centrality

Item 3 ("The most I want to travel (one way) for paid work is") has a low item-rest correlation among both Syrian (r = .28) and Eritrean (r = .06) refugees. The alpha coefficient of the 7-item scale is .71 for Syrian refugees and .66 for Eritrean refugees. The alpha coefficient for the 6-item scale without item 3 is .72 for Syrian refugees and .72 for Eritrean refugees.

CHAPTER 5

NJT-H item revision

Old item

Jane works at a lunchroom in the city centre. On Thursday, there are always two individuals at the city square who work for a marketing company, promoting products like chocolate and drinks. Jane and the two marketing employees help each other: Jane gives them free lunch, and they give her many of the products that they are supposed to distribute to the centre visitors.

How dishonest do you consider Jane to be?

New item

Jane works in a lunchroom. In front of the lunchroom, there are usually two product promotors, distributing products like chocolate and drinks. Jane gives them a free lunch, and in return, they give her many of the products that they are supposed to distribute to the people outside. Jane takes these products home with her. How dishonest do you consider Jane to be?

Item statistics

Table S1. NJT-	H Item-Level F	actor Loadings and	Correlations with	the HEXACO Dimensions

	F1	Н	E	Х	А	С	0
1	0.236	.204*	.104	.013	.107	.054	007
2	0.404	.299**	060	.042	.043	.061	.147
3	0.458	.312**	.020	109	.100	.142	.083
4	0.363	.189*	.057	.115	.043	.075	038
5	0.440	.217*	.040	.173	.014	.170	039
6	0.338	.139	.011	.042	.134	.058	.121
7	0.439	.168	030	.112	.059	.049	033
8	0.513	.240**	131	.121	.117	.047	008
9	0.515	.175	047	.047	.220*	028	.130
10	0.583	.301**	010	.169	.047	.216*	128
11	0.612	.227*	097	.305**	.036	.187*	.043
12	0.319	.274**	096	.012	.128	042	015
13	0.295	.038	069	.123	060	.071	.047
14	0.244	.206*	.054	.075	.123	024	115
15	0.391	.053	.050	067	036	.099	131
16	0.388	.239**	.207*	036	009	.224*	181*
17	0.397	.182*	.079	.284**	.055	.148	050

* p < .05; ** p < .01 (two-tailed).

		L.		
- 1	1		ι.	
		٦	L.	

STUDY 1

CWB and OCB in Study 1

		С	WB			00	В	
	1	Nodel 1	M	odel 2	Mo	odel 1	Mc	odel 2
	β	95% CI	β	95% CI	β	95% CI	β	95% CI
Н	34**	[47;22]	28**	[41;16]	10	[22; .03]	15*	[28;03
E	.04	[08; .16]	.07	[04; .19]	08	[20; .04]	11	[22; .01]
Х	.03	[10; .16]	.03	[10; .15]	.25***	[.12; .38]	.26***	[.13; .38]
A	10	[22; .02]	08	[20;04]	.05	[07; .17]	.03	[08; .15
С	21**	[34;08]	20**	[33;08]	.34***	[.21; .47]	.33***	[.20; .46]
0	.01	[11; .13]	01	[13; .11]	03	[15; .09]	01	[12; .11
NJT-H			23**	[35;11]			.22***	[.10; .34]
\mathbb{R}^2		.218		.266		223	.:	266
F	1	0.34***	11	1.48***	10	.57***	11	.39***
ΔR^2				.048			.(043
ΔF			14	1.55***			12	87***

Table S2. Hierarchical Regression Analyses with the HEXACO traits and the NJT-H as Predictors of

Note. H = Honesty-Humility, E = Emotionality, X = Extraversion, A = Agreeableness, C = Conscientiousness, O = Openness to Experience, OCB = Organizational citizenship behavior, NJT-H = Normative Judgment Test of Honesty-Humility, CWB = Counterproductive work behavior.

N = 230.

* p < .05; ** p < .01; *** p < .001 (two-tailed).

STUDY 2

Results of the analyses with 23 NJT-H items

Alpha coefficient

The alpha coefficient of the NJT-H with 23 items is .79.

Confirmatory factor analysis

The single-factor model: χ^2 (230) = 429.07, p < .001, CFI = .575, TLI = .533, RMSEA = .084 (90% CI = .071; .096), SRMR = .093.

Results

In the current, there were no significant differences between men and women on the NJT-H (respectively M = 3.45, SD = 0.42 and M = 3.46, SD = 0.42; t[119] = -0.15, p = .882, d = 0.02). Furthermore, employees' age was positively correlated with HEXACO H-H (r = .36, p < .001) and the NJT-H (r = .28, p = .002), and these correlations did not differ significantly from each other (z = 0.80, p = .212).

The results revealed a significant modest positive correlation between the NJT-H and

HEXACO H-H (r = .38, p < .001) and no significant correlation between the NJT-H and the other five HEXACO traits. Furthermore, we predicted that the NJT-H is negatively related to self-ratings and supervisor ratings of CWB. In line with these hypotheses, we found a correlation of r = -.24 (p = .008) between the NJT-H and self-ratings of CWB, and a correlation of r = -.34 (p < .001) between the NJT-H and supervisory ratings of CWB. Furthermore, the NJT-H was positively correlated with supervisory ratings of OCB (r = .47, p < .001) and task performance (r = .25, p = .016).

We predicted that the NJT-H explains unique variance in CWB and OCB, above and beyond the variance explained by HEXACO H-H. To test these hypotheses, hierarchical regression analyses were conducted with self-reported CWB and supervisory ratings of CWB and OCB as the dependent variables. For self-ratings of CWB, HEXACO H-H was included in the first step (Model 1), and the NJT-H was added in the second step (Model 2). For supervisory ratings of CWB, interaction frequency was included as a control variable in the first step (Model 1), HEXACO H-H was included in the second step (Model 2), and the NJT-H was added in the second step (Model 2), and the NJT-H was added in the third step (Model 3). In the hierarchical regression analysis with self-ratings of CWB as the dependent variable, H-H showed a negative beta-weight in Model 1 (β = -.31, *t* = -3.32, *p* = .001), and Model 2 showed that the NJT-H does not explain unique variance in self-ratings of CWB above and beyond the HEXACO traits (β = -.07, *t* = -0.74, *p* = .164).

In the hierarchical regression analysis with supervisory ratings of CWB as the dependent variable, interaction frequency ($\beta = .25$, t = 2.47, p = .015) explained 6.3% variance (Model 1; F[1, 92] = 6.11, p = .015), and the model with H-H ($\beta = .40, t = .4.01, p < .001$) explained 34.4% of the variance in supervisory ratings of CWB (Model 2; F[7, 92] = 6.35, p < .001). Model 3 showed that the NJT-H is significantly negatively related to supervisory ratings of CWB (B = -.20, t = -2.12, p = .037), and explained 3.3% additional variance in supervisory ratings of CWB above and beyond the variance explained in Model 2 (ΔF [1, 84] = 4.49, p = .037). The analyses of OCB showed that, after controlling for interaction frequency ($\beta = .11$, t = 1.06, p =.290), H-H (β = .35, t = 3.14, p = .002) is significant positive relationship with OCB. Model 2 explained 21.1% of the variance in OCB (F[7, 92] = 3.24, p = .004). In Model 3, the NJT-H was significantly positively related to OCB ($\beta = .34$, t = 3.42, p = .001). Model 3 explained 9.6% additional variance in OCB above and beyond the variance explained in Model 2 (ΔF [1, 84] = 11.67, p = .001). Finally, the analyses of task performance showed that, after controlling for interaction frequency ($\beta = -.03$, t = -0.25, p = .805), HEXACO H-H showed no significant relationship with task performance (F[7, 92] = 0.75, p = .634), but Model 3 showed that the NJT-H remained significantly positively related to task performance ($\beta = .26, t = 2.30, p =$.024), and explained 5.6% unique variance in task performance above and beyond Model $2 (\Delta F[1, 84] = 5.31, p = .024).$

Table S3. Correlation Between the 23 NJT-H and Variables in Study 2

	-
Variable	Г
1. Gender (male = 0, female = 1)	.01
2. Age	.28**
3. Working hours	.16
4. Interaction frequency	.04
5. 17-item NJT-H	.95**
6. H	.38**
7. E	.06
8. X	.16
9. A	.12
10. C	.18
11. O	08
12. OCB	.47**
13. CWB-E	24**
14. CWB-S	34**
15. Task performance	.25*

* *p* < .05; ** *p* < .01 (two-tailed).

 Table S4. Hierarchical Regression Analyses of CWB and OCB with the 23-item version of the NJT-H in Study 2

		Model 1			Model 2	
	β		95% CI	β	ç	95% CI
CWB-E		$R^2 = .168$	3	$R^2 =$	= .176 (<i>∆R</i> ² = .00)8)
Н	41***	*	[57;24]	37***	[]	55;20]
NJT-H				10	[28; .08]
	Мо	del 1	М	odel 2	Mc	del 3
	β	95% CI	β	95% CI	β	95% CI
CWB-S	$R^{2} =$.063	R ² = .241	1 (<i>∆R</i> ² = .178)	R ² = .281	(<i>∆R</i> ² = .039)
IF	.25*	[.05; .45]	.21*	[.02; .39]	.22*	[.04; .40]
Н			43***	[61;24]	34**	[54;15]
NJT-H					21*	[41;02]
	Мо	del 1	М	odel 2	Mc	del 3
	β	95% CI	β	95% CI	β	95% CI
OCB	$R^{2} =$.012	$R^2 = .160$) (<i>∆R</i> ² = .148)	R ² = .276	$(\Delta R^2 = .115)$
IF	.11	[10; .32]	.15	[04; .35]	.12	[06; .30]
Н			.39***	[.19; .58]	.25*	[.05; .44]
NJT-H					.37***	[.19; .58]
	Мо	del 1	М	odel 2	Model 3	
	β	95% CI	β	95% CI	β	95% CI
TP	$R^{2} =$.001	$R^2 = .102$	$2(\Delta R^2 = .001)$	$R^2 = .068$	$(\Delta R^2 = .066)$
IF	03	[23; .18]	02	[23; .19]	04	[25; .16]
Н			.04	[17; .25]	07	[29; .15]
NJT-H					.28*	[.06; .50]

Note. IF = Interaction Frequency, H = Honesty-Humility, NJT-H = Normative Judgment Test of Honesty-Humility, CWB-E = Counterproductive work behavior employee self-reports, CWB-S = Counterproductive work behavior supervisory rating, OCB = Organizational citizenship behavior, TP = Task performance. N = 123 and N = 93.

* p < .05; ** p < .01; *** p < .001 (two-tailed).

Α

nen.	
req	
n F	
acti	
nter	
le li	
rriat	
il Va	
ntrc	
S	
t th€	
noų	
Wit	
Y 2,	
stud	
in S	
nce	
rma	
erfo	
ik P	
Tas	
and	
CB,	
O m	
CWE	
s of (
tors	
edic	
Pre	
witi	
'SeS	
naly	
Чu	
ssic	
egre	
al R€	
hico	
rarc	
Hie	
S5.	
ble	
Та	

5

		CWB-S)CB		<u>г</u>		
	Mo	odel 1	W	odel 2	Model 1	Model 2	Moo	tel 1	Mod	el 2
	β	95% CI	β	95% CI	β 95% CI	β 95% CI	β	95% CI	β	95% CI
Т	45***	[63;26]	31**	[50;11]	.37*** [.17; .56]	.20* [00; .43]	.04	[16; .25]	07	[30; .15]
NJT-H			32**	[52; .13]		.39*** [.19; .59]			.26*	04; 49]
\mathbb{R}^2		.200		.285	.137	.263		.002		.058
F	2.	:2.71***		17.95***	14.50***	16.04***		0.15		2.78
ΔR^2				.085		.125				.057
ΔF				10.75**		15.31***				5.41*
	Innoct, Unimi	ilih / E = Emotionalih / V	- Eviteor corori				A - H TI N COCC	lormotive leader	to toot too	Lococh,

e. H = Honesty-Humility, E = Emotionality, X = Extraversion, A = Agreeableness, C = Conscientiousness, O = Openness to Experience, NJT ality, CWB-S = Counterproductive work behavior supervisory rating, OCB = Organizational citizenship behavior, TP = Task performance. = 93. = 93. < .05; ** *p* < .01; *** *p* < .001 (two-tailed). īΖ

Q.

Table S6. Hierarchical Regression Analyses with the HEXACO Traits and the NJT-H as Predictors of CWB and OCB in Study 2

		Model	1		Model 2		
	β		95% CI	β		95% CI	
CWB-E		$R^2 = .22$	23	R	$R^2 = .226 (\Delta R^2 = .004)$		
Н	31**		[50;13]	29**	[·	48;09]	
E	.02		[15; .18]	.02	[15; .19]	
×	11		[28; .05]	10	[27; .06]	
А	11		[29; .07]	11	[29; .07]	
С	15		[33; .02]	15	[33; .03]	
0	.13		[03; .30]	.13	[04; .30]	
NJT-H				07	[25; .11]	
	Mod	lel 1	M	odel 2	Mo	odel 3	
	β	95% CI	β	95% CI	β	95% CI	
CWB-S	$R^{2} =$.063	$R^2 = .344$	4 (<i>ΔR</i> ² = .281)	$R^2 = .418$	β (<i>ΔR</i> ² = .074)	
IF	.25*	[.05; .45]	.18	[01; .37]	.18*	[.00; .36]	
Н			40**	[60;20]	28**	[48;07]	
E			.03	[15; .20]	.03	[14; .20]	
×			19*	[37; .01]	15	[32; .03]	
A			.07	[13; .26]	.07	[11; .25]	
С			12	[31; .08]	11	[29; .08]	
0			22*	[40;03]	22*	[40;05]	
NJT-H					31**	[49;12]	
	Mod	lel 1	M	odel 2	Mo	odel 3	
	β	95% CI	β	95% CI	β	95% CI	
OCB	$R^{2} =$.012	$R^2 = .21^{\circ}$	1 (<i>△R</i> ² = .198)	$R^2 = .312$	$(\Delta R^2 = .101)$	
IF	.11	[10; .32]	.12	[09; .33]	.11	[08; .31]	
Н			.35**	[.13; .56]	.20	[02; .42]	
E			.10	[09; .30]	.10	[09; .28]	
X			.20*	[.00; .40]	.15	[04; .34]	
A			.03	[18; .24]	.02	[17; .22]	
С			.05	[17; .26]	.04	[16; .24]	
0			.01	[19; .21]	.02	[17; .21]	
NJT-H					.36**	[.16; .56]	

Note. IF = Interaction Frequency, H = Honesty-Humility, E = Emotionality, X = Extraversion, A = Agreeableness, C = Conscientiousness, O = Openness to Experience, NJT-H = Normative Judgment Test of Honesty-Humility, CWB-E = Counterproductive work behavior employee self-reports, CWB-S = Counterproductive work behavior supervisory rating, OCB = Organizational citizenship behavior.

N = 123 and N = 93. * p < .05; ** p < .01; *** p < .001 (two-tailed).

	Model 1		Model 2		Model 3	
	β	95% CI	β	95% CI	β	95% CI
TP	R^2	= .00	$R^2 = .058 (\Delta R^2 = .057)$		R ² = .106 (∆R ² = .048)	
IF	03	[23; .18]	.00	[22; .23]	00	[22; .22]
Н			04	[27; .20]	14	[39; .11]
E			.10	[11; .32]	.10	[11; .31]
Х			.09	[13; .31]	.06	[16; .27]
A			.04	[19; .27]	.04	[18; .26]
С			.17	[06; .40]	.16	[06; .39]
0			.09	[13; .31]	.09	[12; .31]
NJT-H					.25*	[.02; .48]

 Table S7. Hierarchical Regression Analyses the HEXACO Traits and the NJT-H as Predictors of Task

 Performance in Study 2

Note. IF = Interaction Frequency, H = Honesty-Humility, E = Emotionality, X = Extraversion, A = Agreeableness, C = Conscientiousness, O = Openness to Experience, NJT-H = Normative Judgment Test of Honesty-Humility, TP = Task performance.

N = 93.

* p < .05; ** p < .01; *** p < .001 (two-tailed).

